

Wood, Stacy (2012) Regulation of adrenal corticosteroidogenesis: the role of microRNAs in the control of aldosterone synthase and 11β -hydroxylase expression. PhD thesis.

http://theses.gla.ac.uk/3252/

Copyright and moral rights for this thesis are retained by the author

A copy can be downloaded for personal non-commercial research or study, without prior permission or charge

This thesis cannot be reproduced or quoted extensively from without first obtaining permission in writing from the Author

The content must not be changed in any way or sold commercially in any format or medium without the formal permission of the Author

When referring to this work, full bibliographic details including the author, title, awarding institution and date of the thesis must be given.

Glasgow Theses Service http://theses.gla.ac.uk/ theses@gla.ac.uk

Regulation of Adrenal Corticosteroidogenesis: the role of microRNAs in the control of aldosterone synthase and 11β-hydroxylase expression.

Stacy Wood BSc. (Hons)

Thesis submitted for the degree of Doctor of Philosophy to the University of Glasgow

> Institute of Cardiovascular and Medical Sciences College of Medical, Veterinary and Life Sciences University of Glasgow

> > June 2011

© Stacy Wood 2011

Summary

Hypertension is a major risk factor for all cardiovascular disease, which is the largest known cause of global mortality. Essential hypertension-that is hypertension of unknown cause-is thought to have genetic and environmental risk factors. The best studied genetic system is that concerning corticosteroid biosynthesis. In humans, the principal glucocorticoid is cortisol, the main function of which is the control of intermediary metabolism; the major mineralocorticoid is aldosterone, which affects electrolyte and acid-base homeostasis. These steroid hormones are produced in the adrenal cortex through a series of biosynthetic reactions and under the influence of multiple regulatory factors. The final step in cortisol and aldosterone production involves, respectively, the cytochrome p450 enzymes, 11B-hydroxylase and aldosterone synthase. These are encoded by the *CYP11B1* and *CYP11B2* genes which have a similar sequence and are highly polymorphic and lie, in tandem, on human chromosome 8.

Regulation of CYP11B1 and CYP11B2 mRNA abundance and of aldosterone and cortisol production have been extensively investigated. These studies have identified that there are several polymorphisms located across the locus which are associated with an increased aldosterone to renin ratio (ARR; used as an indicator of aldosterone regulation), inefficient 11B-hydroxylation and essential hypertension. However, to date, no underlying mechanism for these associations has been established. Regulation of expression by transcription factors has been widely studied but, in this thesis, it is the role of a novel regulator, microRNA (miRNA) that is central.

miRNAs are short, non-coding RNAs which negatively regulate mRNA abundance They are transcribed from endogenous loci, then undergo a series of enzymatic maturation reactions that result in the production of a single-stranded molecule of approximately 20 nucleotides. They function by associating with a group of proteins known as the RNA-induced silencing complex (RISC) and targeting the 3' untranslated region (3'UTR) of specific target mRNAs which they bind with imperfect complementarity. There are approximately 1100 human miRNAs, which have been implicated in the regulation of a range of target mRNAs and in several pathologies including cancer and cardiovascular disease. The aim of this project was to investigate what role, if any, miRNAs have in the regulation of *CYP11B1* and *CYP11B2* expression and in corticosteroid production.

The studies in <u>Chapter 3</u> investigated miRNA regulation of corticosteroidogenesis in the adrenal cell line H295R. miRNA levels were universally reduced by targeting Dicer mRNA, a key component of the miRNA synthetic pathway, with short interfering RNA (siRNA). This study identified all of the CYP450 enzymes of the corticosteroidogenic pathway (CYP11A1, CYP17A1, CYP21A1, CYP11B1 and CYP11B2) as likely candidates for miR-mediated regulation based on mRNA and steroid analysis. The study also suggested that StAR, 3BHSDII and 11BHSDII are not modulated by miRNAs. To determine whether apparent miRNA regulation of *CYP11B1* and *CYP11B2* expression occurs by direct action at their 3'UTRs, reporter constructs were generated and tested. Under both basal and stimulated (AngII) conditions, these studies support a regulatory mechanism involving the 3'UTR of *CYP11B1* and *CYP11B2*. This chapter therefore provides evidence for miRNA-mediated regulation of corticosteroidogenesis.

In <u>Chapter 4</u>, putative miRNA target sites in the *CYP11B1* and *CYP11B2* 3'UTR were identified using bioinformatic prediction algorithms and the miRNA expression profile of the normal human adrenal, as determined by microarray analysis. Based on miRNA target site prediction and analyses of the 3'UTR sequences (including such parameters as relative length, predicted sequence conservation and RNA secondary structure), *in silico* methods indicated the possibility that miRNAs can target *CYP11B1* and *CYP11B2* mRNA. Furthermore, the expression of 107 miRNAs in the normal adrenal gland was confirmed. Cross-referencing of microarray expression and bioinformatic data identified 16 adrenal miRNAs predicted to bind putative sites in *CYP11B1* and 16 predicted to bind *CYP11B2*; 12 of these miRNAs were common to both genes.

These formed the basis of the miRNA target validation studies in <u>Chapter 5</u>. Sixteen adrenal miRNAs identified by bioinformatic analysis were tested individually *in vitro*. This was achieved by measuring mRNA expression, steroid production and 3'UTR reporter construct activity following artificially induced increases or reductions in the levels of specific miRNAs. These studies identified some miRNAs as being false positive predictions, while certain others were validated. The miRNA that gave the most striking and consistent results, for targeting both *CYP11B1* and *CYP11B2*, was miR-24, which significantly decreased mRNA levels and steroid production. Analysis of adrenal miRNAs predicted only to target the *CYP11B2* 3'UTR confirmed miR-125a-5p and miR-125b as novel regulators, although effects on steroid secretion remain to be assessed. The studies in this chapter are the first to report of miRNA-mediated regulation of *CYP11B1* and *CYP11B2* expression.

Finally, in <u>Chapter 6</u>, the miRNA expression profiles of four aldosteroneproducing adenoma (APA) samples were generated and compared to those of normal adrenal gland. Analysis identified 67 miRNAs expressed within the APAs; 54 were also present in the normal tissue. The levels of several miRNAs, including miR-24 and miR-125a-5p, were shown to be differentially expressed between the tissue types. This chapter also describes polymorphisms within the 3'UTR of the *CYP11B1* gene, generated from 26 normotensive patients. No novel SNPs were identified, but three are located in putative miRNA-binding sites. Previously, sequence analysis of the *CYP11B2* 3'UTR had been used to map miRNA binding sites, this identified two miRNA-binding sites which mapped to a known SNP. Taken together, the studies in this chapter provide a foundation for exploring altered miRNA function and/or expression within the adrenal gland.

In summary, the results presented in this thesis support a role for miRNAmediated regulation of corticosteroidogenesis through actions on *CYP11B1* and *CYP11B2* expression. It demonstrates that miRNA are present in the adrenal gland, that miRNA-binding sites are present on the 3'UTR of relevant mRNAs, and that miRNAs are capable of post-transcriptional regulation that significantly alters mRNA abundance and steroid production. My findings describe a novel regulatory mechanism of corticosteroidogenesis. Whether this mechanism is altered in diseases such as essential hypertension remains to be elucidated. If so, miRNAs could, in the longer term, be used as targets for novel therapies or as biomarkers to classify more precisely specific pathologies.

Table of Contents

Title		1
Summary		2
Table of Conte	nts	5
List of Figures		
List of Tables		12
Publications		14
Publications		14
Acknowledgem	ents	15
Declaration		16
Definitions/Abb	previations	17
4		22
1 Introductio	Drassura and Hupartansian	۲۲
	Pressure and Hypertension	۲4۲4 ۲۷
	voortonsion	20 20
1.1.2 Hy	sontial Hyportonsion	20 20
1.1.J L3	Lifestyle Pisk Factors	29 20
1 1 3 7	Genetic Risk Factors	2 zn
1.1.3.2 1.2 Cortice	osteroids	34 عد
12 COLLEC	pe Adrenal Gland	+34 عد
1.2.1 Fn	zymes of the Corticosteroid Pathway	
122	Cytochrome P450 Enzymes	
1.2.2.2	Hydroxysteroid Dehydrogenases	
1.2.3 Ch	olesterol Transfer into the Mitochondria	
1.2.4 Cc	orticosteroidogenesis	
1.2.5 Cc	ontrol of Secretion of Corticosteroids	
1.2.5.1	Angiotensin II (AngII)	45
1.2.5.2	Potassium lons (K^{+})	46
1.2.5.3	Adrenocorticotropic Hormone (ACTH)	47
1.2.6 Tr	anscriptional Regulation of CYP11B Genes	50
1.2.6.1	Transcription Regulation of CYP11B2	50
1.2.6.2	Transcriptional Regulation of CYP11B1	52
1.2.7 Po	st-Translational Regulation	53
1.3 Physiol	logical Actions of Corticosteroids	54
1.3.1 Mi	neralocorticoids	54
1.3.1.1	Aldosterone-Induced Proteins	
1.3.1.2	Non-Genomic Actions of Aldosterone	60
1.3.1.3	Aldosterone Actions in Non-Epithelial Cells	
1.3.2 Gl		
1.4 The Ro	ole of Corticosteroids in Hypertension	
1.4.1 Se	Condary Hypertension	
1.4.1.1	The Aldesterone to Denin Datia (ADD)	
1.4.1.Z	Disorders of Corticol Production	00 00 //
1.4.1.3	Managanic Causes of Hypertension	00 67
1.4.1.4 1.4.7 Co	munugenic causes of hypercension	07 ∩7
1/171	The CYD11R1 and CYD11R2 Gener	70 ∩7
1 4 7 7	Genetic Variation in the CYP11R Locus	70 7∩
1.5 RNA In	terference (RNAi)	

1.5.1	Short Interfering RNA	75
1.5.2	microRNAs	75
1.5.2	.1 Nomenclature	76
1.5.2	.2 Discovery	76
1.5.2	.3 Synthesis	79
1.5.2	.4 Turnover	
1.6 mil	RNA-Mediated Gene Silencing	86
1.6.1	miRNA Target Recognition	86
1.6.2	miRNA-Mediated mRNA Repression	
1.6.2	.1 mRNA Destabilisation	87
1.6.2	.2 Translation Repression	
1.7 Exa	amples of Gene Regulation by miRNAs	
1.7.1	miRNAs and Cancer	
1.7.2	miRNAs and Cardiovascular Disease	
1.8 The	erapeutic Potential of miRNAs	
1.8.1	Current Clinical Trials	
1.8.2	miRNAs as Biomarkers	
1.9 mil	RNAs and Endocrine Disease	100
1.10 Ain	ns	102
2 Materia	al and Methods	103
2.1 Ce	ll Culture	104
2.1.1	Cryopreservation	104
2.1.2	Revival of Cell Stocks from Liquid Nitrogen	104
2.1.3	Sub-Culturing Cells	105
2.1.4	Counting Cells	
2.1.5	Stimulation of Cells with Angiotensin II	
2.2 10	tal RNA Isolation	
2.2.1	Total RNA Isolation from Cells and Frozen Tissue	
2.2.2	Total RNA Isolation from Formalin-Fixed Paraffin-Embe	dded lissue.
2 2 2		107
Z.Z.3	DNase Treatment of RNA	108
2.2.4 2.2.5	Agilant Bioppalyzer	109
2.2.3	Agriefit Diodilalyser	109
2.3 IIII 2.4 Do	verse Transcription	
2.4 Ke	Improm IIM Powerse Transcription System	
2.4.1	TagMan [®] microPNA Povorso Transcription	
2.4.2	miScript Poverse Transcription Kit	113
2. 4 .5 2.5 Ou	antitative Real-Time Polymerase Chain Reaction	
2.5 Qu	Universal Probel ibrary	114
2.5.1	TagMan [®] gRT-PCR	115
2.5.2	miScrint aRT-PCR System	116
2.5.5	Analysis of aRT-PCR Results	117
2.3.4 2.6 Pol	vmerase Chain Reaction	118
2.0 10	PCR of CYP11B1 3'UTR	118
2.6.2	Agarose Gel Electrophoresis	
2.7 DN	A Sequencing Reaction	120
2.7.1	PCR Purification	
2.7.2	Sequencing Reaction	
2.7.3		
	Sequencing Reaction Purification	
2./.4	Sequencing Reaction Purification	
2.7.4 2.8 pE	Sequencing Reaction Purification Automated Cycle Sequencing ZX - 3'UTR Reporter Construct	

	2.8.1	Transformation of Competent Cells	122
	2.8.2	Small Scale DNA Purification	124
	2.8.3	Large Scale DNA Purification	124
	2.8.4	Restriction Endonuclease Digestion	125
	2.9 Trar	nsient Cell Line Transfection	126
	2.9.1	Preparation of Small Molecules	126
	2.9.2	Small Molecule Transient Transfection	126
	2.9.3	Reporter Construct Transient Transfection	
	294	Co-transfection	128
	2 10 Dual	Reporter Luciferase Assav	128
	2.10 Dua 2.11 Ster	oid Quantification	129
	2.11 Dete	ermination of Protein Concentration	129
	2.12 Dec	istical Analysis	130
	2.15 Stat		
3	Investiga	ating a Role for miRNA Regulation in Adrenal Cells	. 131
	3.1 Intro	oduction	132
	3.2 Aim	S	133
	3.3 Met	hods	134
	3 3 1	Transfection of H295R Cells with siRNA molecules	134
	3 3 2	3'IITR Reporter Construct Plasmids	134
	3.3.2	Investigation of 3'IITR Activity In Vitro	134
	334	Statistical Analysis	135
	3.4 Resi	llte	136
	3/1	Confirmation of siPNA Transfection	136
	3.4.7	siPNA-mediated Dicer-knockdown in H295P cells	137
	3 4 3	Verification of the pE7X 3'LITP Construct	1/1
	3 1 1	Response of 3'IITR Reporter Construct Activity to Angli Stimula	tion
	J. 7 .7	Response of 5 of R Reporter construct Activity to Angli Stimute	1/18
	3.5 Disc		151
	J.J DISC		
4	Normal	Adrenal Gland microRNA Profiling and Target-Site Prediction	. 155
-	4.1 Intro	oduction	
	4.2 Aim	5	160
	4.3 Mate	erials and Methods	
	4.3.1	Bioinformatic Analysis	. 161
	431	1 Identification of the Genomic Location of miRNAs	161
	431	2 Investigation of the Evolutionary Sequence Conservation	of the
	CYP11	B1 and CYP11B2 3'IITR	161
	431	3 Structural Prediction of the 3'IITR of <i>CYP11B1</i> and <i>CYP11B</i>	32 161
	431	4 Bioinformatic miRNA Target Site Prediction	161
	437	miRNA Expression Profiling of Normal Adrenal Glands	167
	<u>1.3.2</u>	miRNA Microarray Analysis of miRNA Expression	162
		aPT-DCR Validation of miRNA Expression	162
	A 2 2	Statistical Analysis	162
		שנת אותנאשוש היידי אותנאשוש היידי שלא אותנאשוש היידי שלא אותנאשוש היידי שלא אותנאשוש היידי שלא שלא אותנאשוש היי אול ב	161
		Conomic Prodiction of pro miDNA Coding Degions	164
	4.4.1 / / ว	Applysis of the 2'LITE of the CVD11P1 and CVD11P2 Cares	104 125
	4.4.Z	Analysis of the 3 OTK of the CIPITOL did CIPITOZ Genes	100
	4.4.5	michal Supression Drefiling of the Advance Cland	001
	4.4.4	Adversel miDNAs with CVD11P1 and CVD11P2 Binding City	170
	4.4.5	Aurenat-mikinas with CTPTIBT and CTPTIBZ Binding Sites	1/9
		LISS I/MD	1 × 6

5	miRNA	A Target-Site Validation in the 3'UTR of CYP11B1 and CYP11B.	2. 194
	5.1 In	troduction	195
	5.2 Ai	ims	196
	5.3 Ma	aterials and Methods	197
	5.3.1	Investigation of miRNA Binding to 3'UTR Reporter Construct .	197
	5.3.2	Pre-miR™ or Anti-miR™ Transfection of H295R Cells	197
	5.3.3	Statistical Analysis	197
	5.4 Re	esults	199
	5.4.1	Assessing Potential Repression of the pEZX Reporter Construct	t199
	5.4.2	miRNA Targeting to the 3'UTR of CYP11B1 mRNA	201
	5.4.3	miRNA Targeting to the 3'UTR of CYP11B2 mRNA	204
	5.4.4	miRNAs with Unique Target Sites on the CYP11B2 3'UTR	207
	5.4.5	Assessment of H295R Transfection with Pre-miR [™] and Ant	i-miR™
	Moleci	ules	209
	5.4.6	Assessment of mRNA effect on CYP11B1 and CYP11B2	mRNA
	Abund	ance and Steroid Production.	214
	5.4./	Effect of CYP11B2-Specific miRNAs on CYP11B2 mRNA Abunda	nce
	F (0		
	5.4.8	Investigating miRNA Clustered with miR-24.	
	5.4.9	Investigation of the miRNA Binding Sites of miR-24, miR-1.	25a-5p
	and m	1K-1ZDD	231
	5.5 DI		233
6	Altere	ed Adrenal-miRNA Expression and Function	240
Ŭ	6.1 In	troduction	
	6.2 Ai	ims	
	6.3 M	ethods	243
	6.3.1	miRNA Expression Profiling of Aldosterone-Producing Adenom	as.243
	6.3.	1.1 Statistical Analysis of Expression Data	243
	6.3.2	Genotyping of the 3'UTR of CYP11B1	243
	6.3.	2.1 Subjects	243
	6.3.		
		2.2 Sequence Analysis	244
	6.3.	2.2 Sequence Analysis2.3 Statistical Analysis of Genetic Sequences	244
	6.3. 6.4 Re	 2.2 Sequence Analysis 2.3 Statistical Analysis of Genetic Sequences esults 	244 244 245
	6.3. 6.4 Re 6.4.1	 2.2 Sequence Analysis 2.3 Statistical Analysis of Genetic Sequences esults miRNA Expression Profiling of Aldosterone-Producing Ad 	244 244 245 enoma
	6.3. 6.4 Re 6.4.1 Tissue	 2.2 Sequence Analysis	244 244 245 enoma 245
	6.3. 6.4 Re 6.4.1 Tissue 6.4.2	 2.2 Sequence Analysis 2.3 Statistical Analysis of Genetic Sequences esults miRNA Expression Profiling of Aldosterone-Producing Ad APA miRNAs Targeting the CYP11B1 and CYP11B2 Genes 	244 244 245 enoma 245 250
	6.3. 6.4 Re 6.4.1 Tissue 6.4.2 6.4.3	 2.2 Sequence Analysis	244 244 245 enoma 245 250 es .253
	6.3. 6.4 Re 6.4.1 Tissue 6.4.2 6.4.3 6.4.4	 2.2 Sequence Analysis 2.3 Statistical Analysis of Genetic Sequences esults miRNA Expression Profiling of Aldosterone-Producing Ad APA miRNAs Targeting the CYP11B1 and CYP11B2 Genes. Genetic Variation at the 3'UTR of CYP11B1 and CYP11B2 Gene Impact of Polymorphisms on miRNA Binding Sites 	244 244 245 enoma 245 250 es .253 260
	6.3. 6.4 Re 6.4.1 Tissue 6.4.2 6.4.3 6.4.4 6.5 Di	 2.2 Sequence Analysis	243 244 245 enoma 245 250 es .253 260 263
	6.3. 6.4 Re 6.4.1 Tissue 6.4.2 6.4.3 6.4.3 6.4.4 6.5 Di	 2.2 Sequence Analysis	244 244 245 enoma 245 250 es .253 260 263
7	6.3. 6.4 Re 6.4.1 Tissue 6.4.2 6.4.3 6.4.4 6.5 Di Gener	 2.2 Sequence Analysis	243 244 245 enoma 245 250 es .253 260 263 263
7	6.3. 6.4 Re 6.4.1 Tissue 6.4.2 6.4.3 6.4.4 6.5 Di Gener Appen	 2.2 Sequence Analysis	243 enoma 245 enoma 245 250 es .253 260 263 263 268 268

List of Figures

Figure 1-1. Mechanisms of Arterial Pressure Regulation	25
Figure 1-2. The Renin-Angiotensin System	27
Figure 1-3. Cross-Sectional Illustration of the Layers of the Adrenal Gland	35
Figure 1-4. Structure of Corticosteroids and Intermediate Compounds	39
Figure 1-5. Corticosteroid Biosynthetic Pathway.	41
Figure 1-6. Hypothalamic Pituitary Axis Regulation of Cortisol Secretion	49
Figure 1-7. Regions of Transcriptional Importance in the CYP11B Genes.	50
Figure 1-8. Transcription Factor Binding Sites on the CYP11B2 Promoter	52
Figure 1-9. Transcription Factor Binding Sites on the CYP11B1 Promoter	53
Figure 1-10. Classical Mechanism of Aldosterone Action in Epithelial Cells	59
Figure 1-11. The Exon and Intron Structure of the CYP11B1 and CYP11B2 Gene	es.
-	70
Figure 1-12. The Location of the -344 SNP and the Intronic Conversion	ion
Polymorphism.	71
Figure 1-13. The Location of the CYP11B1 Promoter SNPs: -1859 and -1889	74
Figure 1-14. Overview of miRNA Maturation and Degradation	78
Figure 1-15. Transcription of microRNA from an Intergenic Region	81
Figure 1-16. Nucleotide Sequence of a pre-miR	81
Figure 1-17. miRNAs Generated from miRtrons.	82
Figure 1-18. miRNA-Mediated mRNA Destabilisation.	89
Figure 1-19. miRNA-Mediated Translation Repression: Post-initiation Mechanism	ns.
-	93
Figure 1-20. Methods of Inhibiting miRNAs.	98
Figure 1-21. RNA Structural Modifications.	98
Figure 2-1. pEZX Reporter Constructs1	123
Figure 3-1. Assessment of the GAPDH-targeted siRNA on GAPDH mRNA in H29)5R
Cells	136
Figure 3-2. Assessment of the Effect of Dicer1 siRNAs on Dicer1 mRNA in H29	€75
Cells	139
Figure 3-3. Assessment of the Effect of Dicer1 siRNAs on Steroidogenic mRNAs	s in
H295R Cells1	140
Figure 3-4. Assessment of the Effect of Dicer1 siRNAs on CYP11B1 and CYP11	1B2
mRNA in H295R Cells1	141
Figure 3-5. Assessment of the Effect of Dicer1 siRNAs on Steroid Production	in
H295R Cells (aldosterone pathway)1	142
Figure 3-6. Assessment of the Effect of Dicer1 siRNAs on Steroid Production	in
H295R Cells (cortisol pathway)1	143
Figure 3-7. Restriction Endonuclease Digestion of pEZX-Control Plasmid1	145
Figure 3-8. Restriction Endonuclease Digestion of $pEZX-B1$ and $pEZX-B2$ Plasm	11d.
	146
Figure 3-9. Sequence Analysis of PELX Plasmids	14/
Figure 3-10. Assessment of Relative Levels of Reporter Construct Activities1	149
Figure 3-11. Response of pEZX 3'UIR Reporter Constructs to Angiotensin	
Stimulation	150

Figure 4-2. Predicted RNA Secondary Structure of the 3'UTR of the CYP11B1 and Figure 4-3. Mammalian Evolutionary Conservation of the 3'UTR of the CYP11B1 Figure 4-4. Agilent Bioanalyser Electropherogram of Adrenal RNA Samples.....173 Figure 4-5. Normal Adrenal Gland miRNA Microarray Output......175 Figure 4-6. Average Microarray Signals of miRNAs Expressed in the Adrenal Figure 4-8. Venn Diagram of Adrenal miRNAs with Putative miRNA-Target Sites in the 3'UTR of the CYP11B1 and CYP11B2 Genes181 Figure 5-1. siRNA Verification of the pEZX-B1 and pEZX-B2 Reporter Plasmids. Figure 5-2. Effect of miRNAs on the Luciferase Activity of the pEZX-B1 Reporter Figure 5-3. Effect of miRNAs on the Luciferase Activity of the pEZX-B1 Reporter Figure 5-4. Effect of miRNAs on the Luciferase Activity of the pEZX-B2 Reporter Figure 5-5. Effect of miRNAs on the Luciferase Activity of the pEZX-B2 Reporter Figure 5-6. Effect of CYP11B2-Specific Targeting miRNAs on the Luciferase Figure 5-7. Mature miR-24 Levels Following Transfection with Various Concentrations of miR-24 Pre-miR[™] or Anti-miR[™]......211 Figure 5-8. Mature miRNAs Levels Post-Transfection in H295R Cells.212 Figure 5-9. Verification of Pre-miR[™] and Anti-miR[™] Action Using Positive Control Figure 5-10. Assessment of miR-10b Effects on CYP11B1 and CYP11B2 mRNA Figure 5-11. Assessment of miR-140-3p Effect on CYP11B1 and CYP11B2 mRNA Figure 5-12. Assessment of miR-143 Effect on CYP11B1 and CYP11B2 mRNA Levels Figure 5-13. Assessment of miR-185 Effect on CYP11B1 and CYP11B2 mRNA Levels Figure 5-14. Assessment of miR-24 Effect on CYP11B1 and CYP11B2 mRNA Levels Figure 5-15. Assessment of miR-382 Effect on CYP11B1 and CYP11B2 mRNA Levels Figure 5-16. Assessment of miR-638 Effect on CYP11B1 and CYP11B2 mRNA Levels Figure 5-17. Assessment of miR-940 Effect on CYP11B1 and CYP11B2 mRNA Levels Figure 5-18. Assessment of miR-125a-5p and miR-125b Effect on CYP11B2 mRNA Figure 5-19. CYP11B2 and CYP11B1 3'UTR Binding of miR-24 Cluster miRNAs. .230 Figure 5-20. Predicted Secondary Structure of miR-24 Target Sites in the CYP11B1 and CYP11B2 3'UTRs......232

Figure 5-21. Predicted Secondary Structure of miR-125a-5p and miR-125b Target Sites in the <i>CYP11B2</i> 3'UTR232
Figure 6-1. Aldosterone-Producing Adenoma miRNA Microarray Output246 Figure 6-2. Average Microarray Signals of miRNAs Expressed in Aldosterone-
Producing Adenoma
Figure 6-3. Number of miRNAs Expressed in Normal and APA Samples248
Figure 6-4. Correlation of miRNA Expression in Normal and APA Adrenal Tissue. 248
Figure 6-5. Venn Diagram of APA Adrenal miRNAs Predicted to Bind to the <i>CYP11B1</i> and <i>CYP11B2</i> Genes
Figure 6-6. Putative miRNA Binding Sites in CYP11B1 and CYP11B2 and miRNA
Expression in Adrenal Tissue251
Figure 6-7. Differentially Expressed, Predicted CYP11B1 or CYP11B2 Targeting miRNAs
Figure 6-8. CYP11B1 3'UTR PCR Amplification in MONICA Patients255
Figure 6-9. Genotype Analysis of CYP11B1 3'UTR255
Figure 6-10. LD Plot and Haplotypes of Polymorphisms in the CYP11B1 3'UTR. 257
Figure 6-11. Genotype Analysis of CYP11B2 3'UTR
Figure 6-12. LD Plot and Haplotypes of Polymorphisms in the CYP11B2 3'UTR. 259
Figure 6-13. Schematic Diagram of SNPs and miRNA Binding Sites in the CYP11B1
3°UIK
3'UTR

List of Tables

Table 1-1. Classification of Blood Pressure (Chobanian <i>et al.</i> , 2003).28Table 1-2. Name, Function and Location of the Layers of the Adrenal Cortex 35Table 1-3. Identities and Abbreviations of Key Adrenal Steroids.40Table 1-4. Reactions Catalysed by Aldosterone Synthase.44Table 1-5. Cardiovascular Effects of Glucocorticoids (Walker, 2007).63Table 1-6. Subtypes of Primary Aldosteronism (taken from Young (2007).65Table 1-7. P-bodies Degradation Components and Function.88
Table 2-1. Primers Used to Amplify CYP11B1 3'UTR.119Table 2-2. Endonuclease Enzymes Used for Plasmid Digestion.125Table 2-3. Constituents of Transfection Reaction.127Table 2-4. Constituents of Transfection Reaction for pEZX-Reporter Construct.127
Table 4-1. Bioinformatic Databases Used for Screening miRNA Target Sites.163Table 4-2. Genetic Location of Human Corticosteroid Genes and the PredictedmiRNAs Sequences.164Table 4-3. Bioinformatic miRNA Target Site Predictions for Corticosteroidogenic
Genes
Table 4-6. Identities of miRNAs with Putative Binding Sites in the CYP11B2 Gene, but not CYP11B1.171Table 4-7. qRT-PCR Expression Validation of Selected miRNAs.177Table 4-8. Genomic and Sequence Information of Adrenal-miRNAs Predicted to Bind to CYP11B1 and CYP11B2.182Table 4-9. Bioinformatic and Microarray Information of Adrenal-miRNA with Putative Binding Sites in the CYP11B1 3'UTR.183Table 4-10. Bioinformatic and Microarray Information of Adrenal-miRNA with Putative Binding Sites in the CYP11B2 3'UTR.184
Table 5-1. Pre-miR ^{M} and Anti-miR ^{M} molecules.198Table 5-2. siRNA Molecules Targeted to the CYP11B1 and CYP11B2 3'UTR.199Table 5-3. Summary of miRNA Target-Site Validation Experiments for the CYP11B1 Gene.227Table 5-4. Summary of miRNA Target-Site Validation Experiments for the CYP11B2 Gene.228
Table 6-1. Primers Used to Sequence the CYP11B1 3'UTR.244Table 6-2. Normalised Adrenal and APA Microarray Signals.249Table 6-3. Characteristics of Polymorphisms Genotyped in the CYP11B1 3'UTR.
Table 6-4. Polymorphisms Identified in the CYP11B1 3'UTR (grouped according to -344 and IC genotype).256Table 6-5. Statistical Analysis of CYP11B1 Polymorphisms.257Table 6-6. Characteristics of the Polymorphisms Genotyped in the CYP11B2 3'UTR.259

Table 6-7. Polymorphisms Identified in the CYP11B2 3'UTR (grouped a	ccording to
-344 and IC genotype).	
Table 6-8. Statistical Analysis of CYP11B2 Polymorphisms.	259
Table 8-7-1. Universal Probe Library qRT-PCR Assays	
Table 8-7-2. TaqMan [®] miRNA Assays	275
Table 8-7-3. miScript Primer Assays	275

Publications

Abstracts and Awards

Wood S, Forbes GH, MacKenzie SM, Connell JMC, Davies E (2009) "MicroRNA: a novel post transcriptional regulator of 11B-hydroxylase (*CYP11B1*) and aldosterone synthase (*CYP11B2*)". Endocrine Abstracts (19) P321

Wood S, MacKenzie SM, Stewart PM, Connell JMC and Davies E (2010) "MicroRNA expression profiling of non-tumorous adrenal tissue and modulation of 11B-hydroxylase (*CYP11B1*) & aldosterone synthase (*CYP11B2*) genes by miR-24". Endocrine Abstracts (21) P362

Wood S, MacKenzie SM, Stewart PM, Connell JMC and Davies E (2010) "Human adrenal miRNAs: expression profiling of non-tumorous tissue and modulation of adrenocortical genes by miR-24". Keystone Symposia, RNA Silencing: Mechanism, Biology and Application; Poster Communication.

Wood S, MacKenzie SM, Fraser R, Ingram M, Stewart PM, Connell JMC, Davies E (2010) "Differential Expression of MicroRNAs in Non-Tumorous Adrenal Tissue and Aldosterone Producing Adenoma and the Regulation of CYP11B1 and CYP11B2 Genes by miRNA-24".

36th International Aldosterone Conference; Young Investigator Award Winner.

Wood S, MacKenzie SM, Ingram M, Fraser R, Stewart PM, Connell JMC, Davies E (2010) "Regulation of *CYP11B1/B2* Genes by miRNA-24 & MicroRNA Expression Profiling of Non-Tumourous Adrenal Tissue & Aldosterone-Producing Adenoma". International Congress on Hormonal Steroids, Hormones and Cancer; <u>Poster Prize Winner.</u>

Wood S, Ejaz A, Livie C, MacKenzie SM, Connell JMC, Davies E (2011). "Altered miR-125 and miR-134 expression in aldosterone-producing adenoma and post-transcriptional regulation of the CYP11B2 gene". Endocrine Abstracts (25) OC2.4

Wood S, MacKenzie SM, Fraser R, Connell JMC, Davies E (2011) "Dicer-dependent miRNAs regulate adrenal steroidogenesis".

Endocrine Society's 92nd Annual Meeting; Presidential Poster Prize Nominee

Acknowledgements

Firstly, I would like to thank my supervisors Prof. Eleanor Davies and Dr Scott MacKenzie for their expertise, guidance, trust and enthusiasm throughout the project. For their continued mentorship, I would like to thank Prof. John Connell and Prof. Bob Fraser.

I would like to thank all the members past and present of the Blood Pressure Group; they taught me many skills, both lab-based and otherwise. I'd like to acknowledge Mary Ingram for performing LC/MS:MS steroid measurements. Thanks to Samantha who taught me that if something is worth doing, it is worth doing it correctly and to Frances, we experienced all the PhD problems and successes together. It looks like we aren't sharing a PhD after all.

A big thank you to all the friends I have made in during my PhD, which began by meeting Helen and Jen on day one. Since then we have shared all the ups and downs, and many, many cups of tea. Thanks to Louise, Alette and Ayesha for keeping me on my toes and reminding me of what first year optimism feels like. And finally, a huge thanks to all the girls: Carolyn, Monica, Jenny, Lynda, Ruth, Wendy, Yani and Nic for their friendship, especially Monica who continues to be an amazing friend.

A special thank you to my school friends who know me better than anyone and always manage to put a smile on my face; and to Lindsay, Debra and Katie, who I met as an undergraduate at Aberdeen University and who continue to be great friends.

I would like to express my gratitude to my family: Dad, Liz, Kevin and Leanne and to my partner Craig, by dedicating this thesis to them. Their support, encouragement and most importantly love has made this all possible.

Declaration

I declare that the work presented in this thesis is, to the best of my knowledge and belief, original and my own work, unless otherwise stated in the text. This work has not been submitted previously for a higher degree. It was carried out under the supervision of Professor Eleanor Davies and Dr Scott MacKenzie in the institute for Cardiovascular and Medical Sciences.

Stacy Wood June 2011

Definitions/Abbreviations

* strand	Passenger strand/non functional
[K+]e	Extracellular potassium ion concentration
11BHSDII	11B-hydroxysteroid dehydrogenase type 2
12-HETE	12-Hydroxyeicosatetraenoic acid
12-LO	12-lipoxygenase
17-OH-P	17-Hydroxyprogesterone
17-OH-PREG	17-Hydroxypregenolone
18-OHB	18-hydroxycorticosterone
18-OH-B	18-Hydroxycorticosterone
18-OHDOC	18-hydroxy-11-deoxycorticosterone
19-OHDOC	19-hydroxy-11-deoxycorticosterone
3' UTR	3 prime untranslated region
3B-HSD	3B-hydroxysteroid dehydrogenase
5' UTR	5 prime untranslated region
Α	Adenosine
A'dione	Androstenedione
ACE	Angiotension Converting Enzyme
AD1- AD6	Adrenal transcription factor binding site 1 - 6
ADARs	Adenosine deaminase action on RNA
AGO	Argonaut Protein
AGT	Gene encoding angiontensinogen
AIP	Aldosterone-induced Proteins
AKR1C3	Aldo-keto-reductase family
Aldo	Aldosterone
AME	Apparent Mineralocorticoid Excess
Angli	Angiotensin II
ANOVA	Analysis of variance
AP-1	Activation protein-1
APA	Aldosterone-producing adenoma
AP-O	Aminopeptidase-O
AR	Androgen receptor
ARE	Adenylate Uridylate-rich element
ARR	Aldosterone to renin ratio
AT1R	Angiotensin II receptor type I
AT2R	Angiotensin II receptor type II
ATCC	American Type Culture Collection
ACTH	Adrenocorticotropic hormone
ATF	Activating transcription factors
ATP	Adenosine triphosphate
AU	Arbitrary units
В	Corticosterone
BCA	Bicinchoninic acid
bp	Base pairs
BRIGHT	British Genetics of Hypertension
BSA	Bovine serum albumin

С	Cholesterol
Ca ²⁺	Calcium ions
САН	Congenital adrenal hyperplasia
Calmodulin or CaM	Calcium-modulating protein
CaMK	CaM-dependent protein kinases
cAMP	Cyclic adenosine monophosphate
CAT-1	Cationic amino acid transporter
CCD	Cortical collecting duct
cDNA	Complementary DNA
c-fos	Protein product of FOS gene
CHARGE	Cohorts for Heart and Ageing Research in Genome
	Epidemiology BP consortium
CHIF	Channel-inducing factor
CLL	Chronic lymphocytic leukaemia
CMV	Cytomegalovirus
CNS	central nervous system
CO	Cardiac output
CRE	cAMP response element
CREB	CRE-binding protein
CRH	Corticotrophin-releasing hormone
Ct	Cycle threshold
C-terminus	Carboxyl terminal domain
CVD	Cardiovascular diseases
CYP11B1	11B-Hydroxylase
CYP11B2	Aldosterone Synthase
CYP17A1	17g-Hydroxylase
CYP450s	Cytochrome P450 enzymes
DAG	Diacylglycerol
Dcp2	Decapping enzyme 2
DGCR8	Drosha and Di George critical region 8
DHEA	Dehvdroxyprogesterone
DHFR	Dihydrofolate reducatase
DLRA	Dual Luciferase Reporter Assav
DMSO	Dimethyl sulphoxide
DNA	Deoxyribonucleic acid
dNTPs	Deoxynucleoside triphosphates
DOC	11-Deoxycorticosterone
dsRBD	Double stranded RNA binding domain
dsRNA	Double-stranded RNA
E	Cortisone
- FDTA	Ethylenediamine tetra-acetic acid
elF	Eukarvotic initiation factor
ENaC	Epithelial sodium channel
ER	Estrogen receptors
FRK	Extracellular signal-regulated kinase
ET-1	Endothelin-1
- · ·	

F	Cortisol
FAM	Fluorescein
FCS	Fetal calf serum
FFPE	Formalin-fixed Paraffin-embedded
FH	Familial hyperaldosteronism
FOS	Gene encoding c-fos
g	G-force (relative centrifugal force)
G	Guanine
GAPDH	Glyceraldehyde 3-phosphate dehydrogenase
GILZ	Glucocorticoid-induced leucine-zipper
GPCR	G-protein coupled receptors
GR	Glucocorticoid receptor
GRA	Glucocorticoid Remediable Aldosteronism
GRE	Glucocorticoid-responsive elements
GTP	Guanosine-5'-triphosphate (GTP)
GW182	Glycine(G)-tryptophan(W) protein of 182 kD
GWAS	Genome-wide association study
HDL	High-density lipoproteins
HEPES	4-(2-hydroxyethyl)-1-piperazineethanesulfonic acid
HITS-CLIP	High throughput sequencing of cross-linking immunoprecipitation
HMGA2	Gene encoding High-mobility group AT-hook 2
HPA	Hypothalamic-pituitary-adrenal
HPLC	High-performance liquid chromatography
HRE	Hormone-response elements
hsp	Heat shock protein
HWE	Hardy-Weinberg equilibrium
IC	Intron conversion
IHA	Bilateral idiopathic hyperplasia
INTERSALT	International Study of Salt and Blood Pressure
IP3	1,4,5-trisphosphate
IRES	Internal ribosome entry site
ITS	Insulin-transferrin-selenium
IU	International Unit
K^+	Potassium ions
kb	Kilobases
kcal/mol	Kilocalorie per mole
Kd	Dissociation Constant
kD	KiloDalton
KiRasA	GTP-dependent signalling protein K-ras2
KIT	Gene encoding tyrosine-protein kinase Kit
LAR II	Luciferase assay buffer II
LB	Luria Broth
LC/MS:MS	Liquid chromatography tandem mass spectrometry
LD	Linkage disequilibrium
LDL	Low-density lipoproteins
LNA	Lock Nucleic Acid
Lod	Logarithmic odds ratio

LPP	Gene encoding lipoma-preferred partner is a protein
MAF	Minor allele frequency
MFE	Minimum free-energy
miRISC	miRNA-Induced silencing complex
miRNA	Micro ribonucleic acid (microRNA)
mΜ	Millimolar
mm	Millimetre
MONICA	North Glasgow Monitoring of Trends and Determinants in CVD
MR	Mineralocorticoid receptor
mRNA	Messenger ribonucleic acid
MTHFR	Gene encoding Methylenetetrahydrofolate reductase
NaOAc	Sodium acetate
NAPDH	Nicotinamide adenine dinucleotide phosphate
NBRE-1	Neuronal growth factor-induced clone B response element
Nedd4-2	Neural precursor cell expressed, developmentally down-regulated 4?2
nGRE	Negative glucocorticoid-responsive elements
NGRI-B	Nneuronal growth factor-induced clone B
NHE	Na+/H+ exchanger
nm	Nanometer
nM	Nanomolar (nanomoles per litre)
NR3C2	Gene encoding mineralocorticoid receptor
N-terminus	Amino acid terminal domain
Obs. Het	Observed Heterozygosity
Р	Progesterone
p16	Cyclin-dependent kinase inhibitor 2A
PA	Primary Aldosteronism
PABP	Poly A binding protein
PACT	Protein activator of PKR
PAZ	Piwi-Argonaute-Zwille
P-bodies	Processing-bodies
PBS	Phosphate buffered saline
PCR	Polymerase Chain Reaction
pezx	Reporter construct
pg	Picogram
piwiRNA	P-element-induced wimpy testis RNA
PKA	Protein kinase A
PKC	Protein kinase C
PKR	Interferon-induced protein kinase
PLB	Passive Lysis Buffer
PUMC	Pro-opiometanocortin
	Progesterone receptor
PREG	Pregnenolone
pre-mikina/pre-mik	
pri-mikNA/pri-mik	Primary Mikina
	Phosphatase and tensin nomologue
	Gene encoding
GKI-PCK	Quantilative Real-time Polymerase chain reaction

QTL	Quantitative trait loci		
RALES	Randomized Aldactone Evaluation Study		
Ran-GTP	RAs-related Nuclear protein-Guanosine-5'-triphosphate		
RAS	Renin-Angiotensin System		
REST	Repressor element 1-silencing transcription factor		
RIN	RNA integrity number		
RISC	RNA induced silencing complex		
RLB	Reporter Lysis Buffer		
RLU	Relative light units		
RNA	Ribonucleic acid		
RNAi	RNA interference		
rpm	Revolutions per minute		
RT	Reverse transcriptase		
S	11-Deoxycortisol		
SEM	Standard error of the mean		
SF-1	Steroidogenic factor-1		
SGK1	Serum- and glucocorticoid-regulated kinase 1		
SILAC	Stable isotope labelling with amino acids in cell culture		
siRNA	Short interfering ribonucleic acid		
SLIT2	Gene encoding Slit homolog 2 protein		
SNP	Single nucleotide polymorphism		
SPRI	Solid phase reversible immobilization		
ssRNA	Single stranded RNA		
StAR	Steroidogenic Acute Regulatory		
SV40	Simian virus 40		
Т	I hymine		
T TASK channel	TWIK-related acid-sensitive potassium channel		
T TASK channel TBE	Thymne TWIK-related acid-sensitive potassium channel Tris/Borate/EDTA		
T TASK channel TBE TFS	Thymine TWIK-related acid-sensitive potassium channel Tris/Borate/EDTA Transcription factors		
T TASK channel TBE TFS THDOC	Thymne TWIK-related acid-sensitive potassium channel Tris/Borate/EDTA Transcription factors Tetrahydrodeoxycorticosterone		
T TASK channel TBE TFS THDOC THS	TWIK-related acid-sensitive potassium channel Tris/Borate/EDTA Transcription factors Tetrahydrodeoxycorticosterone Tetrahydrodeoxycortisol		
T TASK channel TBE TFS THDOC THS TIMP3	TWIK-related acid-sensitive potassium channel Tris/Borate/EDTA Transcription factors Tetrahydrodeoxycorticosterone Tetrahydrodeoxycortisol Tissue inhibitor of metalloproteinase 3		
T TASK channel TBE TFS THDOC THS TIMP3 TPR	TWIK-related acid-sensitive potassium channel Tris/Borate/EDTA Transcription factors Tetrahydrodeoxycorticosterone Tetrahydrodeoxycortisol Tissue inhibitor of metalloproteinase 3 Total peripheral resistance		
T TASK channel TBE TFS THDOC THS TIMP3 TPR TRBP	TWIK-related acid-sensitive potassium channel Tris/Borate/EDTA Transcription factors Tetrahydrodeoxycorticosterone Tetrahydrodeoxycortisol Tissue inhibitor of metalloproteinase 3 Total peripheral resistance Tar RNA binding protein		
T TASK channel TBE TFS THDOC THS TIMP3 TPR TRBP tRNA	TWIK-related acid-sensitive potassium channel Tris/Borate/EDTA Transcription factors Tetrahydrodeoxycorticosterone Tetrahydrodeoxycortisol Tissue inhibitor of metalloproteinase 3 Total peripheral resistance Tar RNA binding protein Transfer RNA		
T TASK channel TBE TFS THDOC THS TIMP3 TPR TRBP tRNA U	Thymne TWIK-related acid-sensitive potassium channel Tris/Borate/EDTA Transcription factors Tetrahydrodeoxycorticosterone Tetrahydrodeoxycortisol Tissue inhibitor of metalloproteinase 3 Total peripheral resistance Tar RNA binding protein Transfer RNA Unit		
T TASK channel TBE TFS THDOC THS TIMP3 TPR TRBP tRNA U UPL	Twik-related acid-sensitive potassium channel Tris/Borate/EDTA Transcription factors Tetrahydrodeoxycorticosterone Tetrahydrodeoxycortisol Tissue inhibitor of metalloproteinase 3 Total peripheral resistance Tar RNA binding protein Transfer RNA Unit Universal ProbeLibrary		
T TASK channel TBE TFS THDOC THS TIMP3 TPR TRBP tRNA U UPL UPL UTR	Thymne TWIK-related acid-sensitive potassium channel Tris/Borate/EDTA Transcription factors Tetrahydrodeoxycorticosterone Tetrahydrodeoxycortisol Tissue inhibitor of metalloproteinase 3 Total peripheral resistance Tar RNA binding protein Transfer RNA Unit Universal ProbeLibrary Untranslated Regeion		
T TASK channel TBE TFS THDOC THS TIMP3 TPR TRBP tRNA U UPL UPL UTR UV	Thymne TWIK-related acid-sensitive potassium channel Tris/Borate/EDTA Transcription factors Tetrahydrodeoxycorticosterone Tetrahydrodeoxycortisol Tissue inhibitor of metalloproteinase 3 Total peripheral resistance Tar RNA binding protein Transfer RNA Unit Universal ProbeLibrary Untranslated Regeion Ultraviolet		
T TASK channel TBE TFS THDOC THS TIMP3 TPR TRBP tRNA U UPL UPL UTR UV V	Thymne TWIK-related acid-sensitive potassium channel Tris/Borate/EDTA Transcription factors Tetrahydrodeoxycorticosterone Tetrahydrodeoxycortisol Tissue inhibitor of metalloproteinase 3 Total peripheral resistance Tar RNA binding protein Transfer RNA Unit Universal ProbeLibrary Untranslated Regeion Ultraviolet Voltage		
T TASK channel TBE TFS THDOC THS TIMP3 TPR TRBP tRNA U UPL UPL UTR UV V V	Thymne TWIK-related acid-sensitive potassium channel Tris/Borate/EDTA Transcription factors Tetrahydrodeoxycorticosterone Tetrahydrodeoxycortisol Tissue inhibitor of metalloproteinase 3 Total peripheral resistance Tar RNA binding protein Transfer RNA Unit Universal ProbeLibrary Untranslated Regeion Ultraviolet Voltage Volume/volume		
T TASK channel TBE TFS THDOC THS TIMP3 TPR TRBP tRNA U UPL UPL UTR UV V V V V	Thymne TWIK-related acid-sensitive potassium channel Tris/Borate/EDTA Transcription factors Tetrahydrodeoxycorticosterone Tetrahydrodeoxycortisol Tissue inhibitor of metalloproteinase 3 Total peripheral resistance Tar RNA binding protein Transfer RNA Unit Universal ProbeLibrary Untranslated Regeion Ultraviolet Voltage Volume/volume Working reagent		
T TASK channel TBE TFS THDOC THS TIMP3 TPR TRBP tRNA U UPL UTR UV V V V V V V V	Thymne TWIK-related acid-sensitive potassium channel Tris/Borate/EDTA Transcription factors Tetrahydrodeoxycorticosterone Tetrahydrodeoxycortisol Tissue inhibitor of metalloproteinase 3 Total peripheral resistance Tar RNA binding protein Transfer RNA Unit Universal ProbeLibrary Untranslated Regeion Ultraviolet Voltage Volume/volume Working reagent Wild-type		
T TASK channel TBE TFS THDOC THS TIMP3 TPR TRBP tRNA U UPL UPL UPL UTR UV V V V V V V V V V V V V V V V V V V	Thymine TWIK-related acid-sensitive potassium channel Tris/Borate/EDTA Transcription factors Tetrahydrodeoxycorticosterone Tetrahydrodeoxycortisol Tissue inhibitor of metalloproteinase 3 Total peripheral resistance Tar RNA binding protein Transfer RNA Unit Universal ProbeLibrary Untranslated Regeion Ultraviolet Voltage Volume/volume Working reagent Wild-type 5'-3' exoribonuclease 2		
T TASK channel TBE TFS THDOC THS TIMP3 TPR TRBP tRNA U UPL UTR UV V V V V V V V V V V V V V V V V V V	Thymne TWIK-related acid-sensitive potassium channel Tris/Borate/EDTA Transcription factors Tetrahydrodeoxycorticosterone Tetrahydrodeoxycortisol Tissue inhibitor of metalloproteinase 3 Total peripheral resistance Tar RNA binding protein Transfer RNA Unit Universal ProbeLibrary Untranslated Regeion Ultraviolet Voltage Volume/volume Working reagent Wild-type 5'-3' exoribonuclease 2 Zona fasciculata		
T TASK channel TBE TFS THDOC THS TIMP3 TPR TRBP tRNA U UPL UPL UPL UTR UV V V V V V V V V V V X R S Z G	Thymne TWIK-related acid-sensitive potassium channel Tris/Borate/EDTA Transcription factors Tetrahydrodeoxycorticosterone Tetrahydrodeoxycortisol Tissue inhibitor of metalloproteinase 3 Total peripheral resistance Tar RNA binding protein Transfer RNA Unit Universal ProbeLibrary Untranslated Regeion Ultraviolet Voltage Volume/volume Working reagent Wild-type 5'-3' exoribonuclease 2 Zona fasciculata Zona glomerulosa		
T TASK channel TBE TFS THDOC THS TIMP3 TPR TRBP tRNA U UPL UTR UV V V V V V V V V V V V V V V V V V V	Thymne TWIK-related acid-sensitive potassium channel Tris/Borate/EDTA Transcription factors Tetrahydrodeoxycorticosterone Tetrahydrodeoxycortisol Tissue inhibitor of metalloproteinase 3 Total peripheral resistance Tar RNA binding protein Transfer RNA Unit Universal ProbeLibrary Untranslated Regeion Ultraviolet Voltage Volume/volume Working reagent Wild-type 5'-3' exoribonuclease 2 Zona fasciculata Zona glomerulosa Zona reticularis		

1 Introduction

High blood pressure and associated cardiovascular and cerebrovascular disease are massive contributors to morbidity and early mortality in developed societies. While lifestyle factors such as smoking, high saturated fat and high salt diets and reduced rates of physical exercise are important factors, variations in the efficiency of several physiological systems, often genetically-based, contribute significantly.

The adrenal cortex is among the most important of these. In the following review, the structure of the adrenal cortex and the nature of its principle products, the corticosteroids, are described. The synthesis and control of secretion of the two major categories of corticosteroids, those which affect electrolyte and acid-base homeostasis, the mineralocorticoids, and those whose main function is in the control of intermediary metabolism, the glucocorticoids, are briefly outlined. Their mechanism of action in target tissue is then described, addressing briefly classical receptor-linked processes and presenting information on the more recently discovered non-classical modes of action.

The importance of the adrenal cortex in hypertension will be discussed by describing the effects of corticosteroids in experimental animal models and in a group of relatively uncommon disorders of corticosteroidogenesis. However, the most frequent form of hypertension is primary or essential hypertension, that is, hypertension of unknown aetiology. Studies over recent decades strongly suggest that small, genetically-related differences in the control of secretion and rate of catabolism of corticosteroids may predispose to hypertension and cardiovascular diseases. These studies are outlined and evaluated.

In the control of metabolism, mRNA transcription is initiated in response to specific agonist-induced transcription factors. The resulting mRNA may be modified prior to the process of translation into new protein and the longevity and eventually destruction of mRNA is determined by other biochemical processes. The efficiency of each of these components may contribute to biological variation of clinical significance. In the final sections of this review, the structure and control of expression of two key genes in corticosteroid synthesis are described in detail. Of particular relevance to this thesis, the identification and action of novel RNA species, microRNAs, powerful modulators

of mRNA is analysed. It is with these microRNAs that the experimental work described in sections 3, 4, 5 and 6 is concerned.

1.1 Blood Pressure and Hypertension

The heart contracts periodically, with each beat it ejects a bolus of blood into the circulation. That creates a cyclic change in blood pressure throughout the vasculature. Blood pressure is highest immediately after the contraction, known as systolic blood pressure and lowest prior to the next contraction, known as diastolic blood pressure. Blood pressure can be defined as the cardiac output (CO, the volume of blood ejected from the heart) multiplied by the total peripheral resistance (TPR, the sum of resistance in vasculature), these are under independent control (Figure 1-1).

Blood pressure is maintained by a variety of homeostatic controls (Figure 1-1). These must tolerate everyday stresses such as changes in posture and exercise to ensure continuous perfusion of vital organs. Under healthy circumstances, normal adult blood pressure is defined as systolic pressure under 120 mmHg and diastolic pressure under 80 mmHg (Chobanian *et al.*, 2003) but this is arbitrary. This is further addressed in Section 1.1.2. Blood pressure is under the control of several complex regulatory systems including the sympathetic nervous system, endocrine stimuli, local tissue control (for example, flow-autoregulation) and finally regulation of sodium and water balance by the kidneys (Figure 1-1). The renal component is under the control of the renin-angiotensin system, which through the production of angiotensin II (AngII) regulates the synthesis of aldosterone and will be described in Section 1.1.1.



Figure 1-1. Mechanisms of Arterial Pressure Regulation.

Adapted from Cowley (2006).

1.1.1 The Renin-Angiotensin System (RAS)

The renin-angiotensin system (RAS) (Figure 1-2) is composed of a series of tightly-regulated enzymatic reactions, the ultimate product of which is Angiotensin II (AngII), a pressor octapeptide. Activation of the RAS cascade occurs when low blood pressure is detected by the renal afferent arteriole or when decreased NaCl concentration is sensed by specialised epithelial cells of the distal tubule called the macula densa, stimulating the secretion of renin from renal juxtaglomerular cells. Renin is an aspartyl protease which is originally synthesised from preprorenin, subsequently cleaved to prorenin and finally to renin by the prorenin processing enzyme. Renin catalyses the cleavage of angiotensin I, an inactive decapeptide, are cleaved by angiotensin converting enzyme (ACE) to produce the vasoactive, octapeptide AngII (Figure 1-2). ACE was first discovered in the 1950s and is a membrane-bound zinc metalloprotease located on vasculature endothelial surfaces in the lungs, the kidneys, the intestine and the placenta.

AngII has many actions in several target tissues including, the vasculature, the adrenal cortex, the kidney, the brain and adipose tissue. It binds to specific transmembrane G-protein coupled receptors (GPCR) of two types: type 1 (AT1R) or type 2 (AT2R) (Figure 1-2) (Timmermans *et al.*, 1993). Once activated, the different subtypes of receptors mediate opposing effects on resistant vessels, the AT1R promoting vasoconstriction and the AT2R receptor vasodilation. Both receptors have similar affinities for AngII, are widely distributed and yet have heterogeneous expression patterns. In human adults AT1R serves as the principle mediator of blood pressure regulation (Timmermans *et al.*, 1993;Shanmugam *et al.*, 1996). By acting on AT1R, AngII contributes to blood pressure regulation by three main actions: it promotes the release of aldosterone from the adrenal cortex, it increases the re-absorption of sodium in the distal convoluted tubules of the kidney and finally, it causes vasoconstriction.



Figure 1-2. The Renin-Angiotensin System.

ACE: angiotensin converting enzyme; AT1R: angiotensin II receptor type 1; AT2R: angiotensin II receptor type 2.

1.1.2 Hypertension

Hypertension; that is, blood pressure above the upper limit of the normal range, is the major risk factor for all cardiovascular diseases (CVD); 54% of death from strokes and 47% of death in patients of ischemic heart disease can be attributed to hypertension (Lawes *et al.*, 2008). CVD is the largest known cause of mortality globally (Chobanian *et al.*, 2003).

Blood pressure is subject to short term variability and obtaining a 'representative' clinical value in clinic is difficult. Therefore, defining hypertension is problematic (Carretero and Oparil, 2000). As a necessary, though arbitrary, guide hypertension is defined as a systolic blood pressure greater than 140 mmHg or a diastolic blood pressure over 90 mmHg. However, risk varies incrementally with blood pressure (Chobanian *et al.*, 2003), even at levels below 140/90 mmHg ('pre-hypertensive') (Table 1-1) (Kshirsagar *et al.*, 2006). This emphasises that small changes in the factors governing blood pressure can have significant detrimental effects.

In only a small number of patients (approximately 5%) can an underlying medical cause of their hypertension be identified (Section 1.4.1.1). The remaining 95% are classified as having essential hypertension (also known as primary hypertension or idiopathic hypertension) (Section 1.1.3).

Classification	Systolic BP (mm Hg)	Diastolic BP (mm Hg)
Normal	<120	and <80
Pre-hypertension	120-139	or 80-89
Stage 1 Hypertension	140-159	or 90-99
Stage 2 Hypertension	>160	or >100

Table 1-1. Classification of Blood Pressure (Chobanian et al., 2003).

1.1.3 Essential Hypertension

The aetiology of essential hypertension remains elusive, despite large numbers of investigative studies. However, there are several known risk factors which contribute to the development of hypertension which can be split into lifestyle or genetic risk factors. The next section will briefly discuss the effects of these on blood pressure.

1.1.3.1 Lifestyle Risk Factors

Environmental risk factors include high sodium intake and low potassium intake, obesity, stress, lack of exercise, smoking and alcohol intake. Of these, sodium and potassium intake are the most relevant to the studies of this thesis because they alter the production and action of corticosteroids (Section 1.2.5 and 1.3).

The International Study of Salt and Blood Pressure (INTERSALT) project was a large, prospective epidemiological study which involved 52 centres from 32 countries. Results from 10,079 individuals identified a strong link between sodium excretion and systolic blood pressure that was independent of any other risk factors (Stamler *et al.*, 1989). Other intervention studies reduced salt intake in either normotensive or hypertensive people and showed that blood pressure was reduced on low sodium diets. This is true for both acute (less than 4 weeks) or prolonged salt restricted diets, which reduced salt intake from 10 g/day to 5 g/day (Macgregor *et al.*, 1982;Meneton *et al.*, 2005). Moreover, a recent meta-analysis of 19 separate cohorts demonstrated that high sodium intake (average increase in salt intake of 5 g/day) increased the risk of stroke and total incidence of CVD (Strazzullo *et al.*, 2009).

INTERSALT identified a relationship between potassium intake and blood pressure; urinary excretion of potassium had an independent, inverse association with systolic blood pressure. Also, the ratio of sodium: potassium excretion was positively and significantly related to both systolic and diastolic blood pressure (Stamler *et al.*, 1989). A meta-analysis of 33 randomised controlled trials (totalling 2609 subjects), which each investigated the use of potassium supplements, found that exogenous potassium significantly reduced blood pressure (Whelton *et al.*, 1997). Further, hypertensive animals fed high

potassium and high sodium diets had lower blood pressures than those on high sodium, standard potassium diets (Adrogue and Madias, 2007). Corticosteroids both respond to, and participate in, the control of electrolyte status (Sections 1.2.5 and 1.3).

1.1.3.2 Genetic Risk Factors

The studies in this thesis relate to the control of gene transcription. It is therefore relevant briefly to consider how genetic variation and disease in the population is studied. Evidence implicating a genetic link to blood pressure regulation and pathology is strong and arises from several sources. Several key family studies suggest a genetic link; individual members of a family unit have a greater similarity in their blood pressures when compared to individuals from other families (Longini *et al.*, 1984). Findings from twin studies found a greater concordance of blood pressure in monozygotic twins than in dizygotic twins (Feinleib *et al.*, 1977). Furthermore, the blood pressures of biological siblings is more highly correlated than those of adopted siblings (Biron *et al.*, 1976;Rice *et al.*, 1989). Other evidence of a genetic link arises from several defined monogenic diseases which affect blood pressure (Section 1.4.1.4).

Heritability is a parameter measuring the likelihood that a trait is attributed to genetic rather than environment factors. Blood pressure has a relatively high percentage of heritability, reported 15-40% for systolic blood pressure and 15-30% for diastolic. A systematic review of family studies estimated the joint heritability rate to be approximately 35% (Ward, 1990). Thus, evidence supports a genetic risk to blood pressure.

Hypertension is a polygenic disorder, where the combination of several detrimental genetic traits is manifested as a small increase in blood pressure. However, to date, the identity of the genes responsible for this association remains elusive, although information obtained from the study of rare monogenic disorders has made a useful contribution to our understanding of the genetics of hypertension (Section 1.4.1.4). Also, several other scientific approaches such as linkage and association studies have identified causative genetic defects that may allow for more focused preventative measures and treatment.

Association Studies

Genome-wide association studies (GWAS) as the name suggest, span the entire genome and tend to involve the highest number of subjects as well as being the most expensive method to perform. GWAS assesses over a million single nucleotide polymorphisms (SNPs) located throughout the genome to identify variations important to a disease trait. GWAS studies are carried out in large populations, in cases and controls or in individuals. The aim is to identify novel associations of SNPs with diseases, and been useful in the study of many diseases (Burton *et al.*, 2007).

However, the complex nature of blood pressure, the problem of choosing appropriate controls and the inadequate coverage of SNPs in commercial gene arrays have prevented GWAS from being as successful for the investigation of hypertension. Of the twelve independent blood pressure GWAS studies (all of European origin), only two have successfully identified regions with positive associations to blood pressure: the Global BPGen consortium and the CHARGE (Cohorts for Heart and Ageing Research in Genome Epidemiology BP consortium) (Newton-Cheh et al., 2009;Levy et al., 2009). In total, 14 independent genetic loci have been identified for blood pressure traits, several of which lie near genes which encode enzymes (kinases and cyp450 (see Section 1.2.2.1)), solute channels, transcription factors, a cell signalling protein and a structural protein (Ehret, 2010). Two of the candidates, the CYP17A1 gene and the MTHFR gene are plausible candidates (their significance is explained in Section 1.4.1.4), the role and implication of the others remains to be understood. The analysis of this in GWAS data predicts that each associated allele would represent a change of about 1 mm Hg in systolic pressure (Ehret, 2010). Thus, genetic variations may each have a small impact on blood pressure but the coexistence of several detrimental loci could have a cumulative effect.

Linkage Analysis

Linkage analysis studies aim to identify genomic loci, known as quantitative trait loci (QTL) which are genetically linked, heritable and may be attributable to the disease being investigated. The largest study of this type has been the British Genetics of Hypertension (BRIGHT) study that identified two QTL (Caulfield *et al.*, 2003). However, the results have been criticised due to the location of the QTL being close to the end of chromosomes, this is often problematic as linkage programmes can generate errors at the extremes of chromosome, and the use of multiple testing which requires a higher Lod (logarithmic odds ratio, a score used to indicate the likelihood of linkage) of 3.6 to attain significance. In the case of BRIGHT the was Lod score was only 2.5 (Binder, 2007;Delles *et al.*, 2010). Individual linkage studies have identified QTLs on nearly all human chromosomes but these are rarely replicated by other studies (Binder, 2007). Thus, linkage analysis is not an accurate or reproducible method for studying the importance of genes and hypertension.

Candidate Gene Analysis

An alternative approach, candidate gene studies, dissects the variability of genes with a known physiological link to blood pressure, such as those involved in salt and fluid homeostasis (e.g. RAS) or the adrenergic system. The results of these studies have been inconsistent, frequently due to poor study design (selection of heterogeneous cases and controls) or SNP coverage. One of the most extensively studied genes using this approach is the AGT gene which encodes angiotensinogen, a pivotal enzyme in the RAS (Figure 1-2). Many SNPs have been identified in this gene in coding and non-coding DNA regions. Two SNPs, T174M and M235T, are associated with hypertension in some studies but not others (Dickson and Sigmund, 2006). A meta-analysis of several studies failed to confirm these associations but the T allele at position 235 was associated with an increased relative risk of hypertension (Sethi *et al.*, 2003). Another candidate is the ACE gene, which again has a high frequency of polymorphisms, the one of most interest being a 287 base-pair insertion/deletion within intron 16. This polymorphism is associated with increased ACE plasma levels, coronary artery disease and endothelial dysfunction, as measured by monitoring cardiac blood flow following infusion of acetylcholine and sodium nitroprusside (Prasad et al., 2000). However it does not play a role in hypertension (Jeunemaitre, 2008). The genes that provided the focus in this thesis, *CYP11B2* and *CYP11B1*, are also well-studied candidate genes and will be discussed in greater detail in sections 1.2.2, 1.2.6 and 1.4.2.

To summarise, there are significant genetic and lifestyle components to blood pressure regulation and the development of hypertension. Essential hypertension is a complex disease and the genetic aetiology is difficult to study. One candidate system which studies have shown to be important for blood pressure regulation and the development of essential hypertension is the adrenal cortex. The main mineralocorticoid, aldosterone and, main glucocorticoid, cortisol are synthesised here and altered synthesis plays a role in the aetiology of essential hypertension. The next three sections of this review will focus on these compounds, their actions and the evidence of their dysregulation in several pathologies.

1.2 Corticosteroids

Steroid hormones are classified by their structural characteristics and biological function. Examples include estrogens, androgens, progestins, corticosteroids and vitamin D. All steroids are derived from cholesterol. Corticosteroids are products of the adrenal cortex and, based on their function, can be subdivided into mineralocorticoids and glucocorticoids, the main human examples of which are aldosterone and cortisol, respectively. The adrenal cortex also synthesises androgens and small quantities of estrogens.

1.2.1 The Adrenal Gland

The adrenal glands are paired endocrine organs that lie anatomically superior to each kidney. Each adrenal gland has independent blood, lymph and nervous supplies and can therefore function independently of one another. The glands are surrounded by an outer fibrous capsule. The functional tissue of the adrenal gland is divided into two separate anatomical regions, which have specialised functions. The inner medulla is responsible for producing and secreting catecholamines. The primary function of the outer region, the adrenal cortex, is to produce and secrete steroid hormones, for example aldosterone and cortisol.

The adrenal cortex is divided into three zones, which are functionally and histologically distinguishable. The names, position and functions of each zone are outlined in Table 1-2 and illustrated in Figure 1-3.

Zone	Position	Steroids Produced
Zona glomerulosa (ZG)	Outer	Mineralocorticoids
Zona fasciculata (ZF)	Middle	Glucocorticoids
Zona reticularis (ZR)	Inner	Sex steroids and glucocorticoids

Table 1-2. Name, Function and Location of the Layers of the Adrenal Cortex.



Figure 1-3. Cross-Sectional Illustration of the Layers of the Adrenal Gland.

Adrenal gland showing inner medulla, outer cortex and capsule. Magnified section showing schematic and haematoxylin and eosin staining of individual layers (taken from the McGraw-Hill online learning centre; chapter 10 Endocrine System, http://www.mhhe.com/biosci/ap/seeleyessentials/student/olc/chap10out.html).
1.2.2 Enzymes of the Corticosteroid Pathway

Two classes of enzymes are required for corticosteroidogenesis: five cytochrome P450 enzymes and one hydroxysteroid dehydrogenase. Characteristics of each of these enzymes are briefly discussed below; the succeeding sections will describe the features and functions of these enzymes in relation to corticosteroid synthesis. Steroid synthesis involves a complex pathway of intermediates; the names and common abbreviations of components in the pathway are in Table 1-3 and their structures illustrated in Figure 1-4.

1.2.2.1 Cytochrome P450 Enzymes

To convert cholesterol to the C21 corticosteroids (Figure 1-4), most of its side chain must be removed and the pregnane structure modified by a series of oxidative reactions. Androgen synthesis requires the complete removal of the side chain. The majority of enzymes involved are mixed function oxidases or mono-oxygenases belonging to the cytochrome P450 group of enzymes (CYP450s). These are a large class of haemoproteins which catalyse oxidative reactions in a wide range of cells. They are so named because, when in a carbon-monoxide-bound form, they produce an absorption band at 450 nm. CYP450s are a very diverse family which is sub-classified into 69 families based on structural and phylogenetic similarities (Nelson *et al.*, 1996). CYP450s are involved in a wide range of cellular functions including xenobiotic metabolism, steroid synthesis and lipid synthesis. They catalyse the addition of an oxygen atom from a molecule of oxygen to the substrate in a locus and stereo-specific manner, utilising an electron donor, usually NAPDH. The remaining oxygen atom is reduced to water. The stoichiometry of the reaction is;

 $R-H + NADPH + H^+ + O_2 \longrightarrow R-OH + NADP^+ + H_2O$

1.2.2.2 Hydroxysteroid Dehydrogenases

Hydroxysteroid dehydrogenases are bidirectional, non-metalloenzymes that are expressed in several tissue types. There are several different isoforms of each dehydrogenase, each with a specific expression pattern, cellular localisation, catalytic activity and each is encoded by a different gene. All dehydrogenase enzymes utilise co-factors, which act as hydrogen donors or acceptors, for example NAD⁺/NADP⁺. Within the adrenal cortex, 3B-hydroxysteroid dehydrogenase (3B-HSD) is essential for the synthesis of mineralocorticoids, glucocorticoids and androgens. In contrast, hydroxysteroid dehydrogenases expressed in peripheral tissue, for example in the kidney, are responsible for metabolism of steroids (Penning, 1997;Payne and Hales, 2004).

1.2.3 Cholesterol Transfer into the Mitochondria

Cholesterol is a sterol and is the precursor of all steroids Figure 1-4. It is comprised of four hydrocarbon rings, a hydrocarbon tail (carbons 20-27) and a 3B-hydroxyl group. Cholesterol can be synthesised *de novo* in the adrenal cortex but most is derived from extra-adrenal sources, predominantly in the liver or the diet. Cholesterol is amphipathic and must be incorporated in low-density lipoproteins (LDL) or high-density lipoproteins (HDL) to circulate in the blood.

In humans, the adrenal gland preferentially uses cholesterol derived from LDL in circulating plasma, which is internalised into the adrenal cells by receptormediated endocytosis whereas, in rats the main source is from HDL (Gwynne and Strauss, 1982). Once inside the cell, cholesterol must be transported across the outer mitochondrial membrane. Several transporters have been identified which facilitate this transfer but it is now widely accepted that the principle transporter is the steroidogenic acute regulatory (StAR) protein. Some cholesterol can diffuse across the membrane but its hydrophobic nature means that the rate of diffusion is extremely slow and contributes little to steroidogenesis (Phillips *et al.*, 1987). Cholesterol transport via StAR is the rate-limiting step of corticosteroidogenesis.

Steroidogenic Acute Regulatory (StAR) Protein

StAR is expressed exclusively in steroid-producing tissues. The StAR protein is generated in the cytoplasm and directed to the mitochondrion by an NH_2 (N)-terminal targeting motif. Once localised to the mitochondrion, the targeting motif is cleaved from a 37 kD mature protein to produce a 30 kD transporter. The carboxyl (C)-terminus region of the protein is also crucial to StAR function; truncation of the C-terminus severely decreased the activity of StAR (Arakane *et al.*, 1996). The C-terminus is believed to be crucial for the association of StAR with the outer mitochondrial membrane and conformational changes important to its activity (Yaworsky *et al.*, 2005).

Cholesterol transport by StAR was first observed in rat adrenal cells stimulated with either adrenocorticotropic hormone (ACTH) or dibutyryl cyclic adenosine monophosphate (cAMP) (Krueger and Ormejohnson, 1983). Cloning and expression of the protein confirmed its role in steroidogenesis (Stocco and Clark, 1996) and StAR-knockout mice have dramatically reduced levels of steroid production and increased levels of stimulatory trophins (owing to a negative feedback mechanism, see section 1.2.5). Moreover, these knockout mice have impaired growth and die prematurely, highlighting the essential functions of corticosteroids (Caron *et al.*, 1997). Patients with mutations in the StAR gene can show an almost complete lack of corticosteroids, and can be lethal in newborn infants. This condition is known as lipoid congenital adrenal hyperplasia (Lin *et al.*, 1995). Although StAR's key role is now established, the mechanism by which it transports cholesterol is not fully understood. However, four modes of action have been proposed (for review see Miller (2007)).





The carbon atoms are numbered in the cholesterol molecule.

|--|

Common Name	Abbreviation	Systematic Name
Cholesterol	С	Cholest-5-en-3B-ol
Pregnenolone	PREG	3B-hydroxypregn-5-ene-20-one
Progesterone	Р	pregn-4-ene-3,20-dione
17-Hydroxypregenolone	17-OH-PREG	3B,17α-dihydroxypregn-5-ene-20-one
17-Hydroxyprogesterone	17-OH-P	17α-hydroxypregn-4-ene-3,20-dione
Dehydroxyprogesterone	DHEA	3B-hydroxyandrost-5-ene-17-one
Androstenedione	A'dione	androst-4-ene-3,17-dione
11-Deoxycortisol	S	17α,21-dihydroxypregn-4-ene-3,20-dione
Cortisol	F	11B,17α,21-trihydroxyprepn-4-ene-3,20-dione
Cortisone	E	17α,21-dihydroxypregn-4-ene-3,11,20-trione
11-Deoxycorticosterone	DOC	21-dihydroxypregn-4-ene-3,20-dione
Corticosterone	В	11B,21-dihydroxypregn-4-ene-3,20-dione
18-Hydroxycorticosterone	18-OH-B	11B,18,21-trihydroxypregn-4-ene-3,20-dione
Aldosterone	Aldo	11B-21-dihydroxypregn-4-ene-3,20-dione-18-al





Pathway of corticosteroid synthesis in the adrenal cortex. Dashed lines represent the zone of the adrenal cortex in which each reaction occurs.

1.2.4 Corticosteroidogenesis

Cholesterol Side-Chain Cleavage (P450scc)

The first step in steroidogenesis involves the cleavage of cholesterol to pregnenolone (Figure 1-5), which is mediated by the cholesterol side-chain cleavage enzyme (CYP11A1 or P450ssc). CYP11A1 is encoded by the *CYP11A1* gene, which is located on human chromosome 15. It is expressed in all three zones of the adrenal cortex. The enzyme is located on the inner mitochondrial membrane and catalyses three separate oxidative reactions (Figure 1-4); the first is hydroxylation at C22 and the second the hydroxylation of C20, producing the intermediate compound, 20,22R,hydroxyl-cholesterol. The final reaction catalysed by CYP11A1 is cleavage between C22 and C20, which shortens the side chain of cholesterol by six amino acids, creating the C21 steroid pregnenolone (Figure 1-5). Each reaction requires one molecule of O_2 and one of NADPH, acting via ferredoxin and ferredoxin reductase, to donate electrons. Pregnenolone is the precursor for all steroids of the adrenal cortex (Lieberman and Lin, 2001) and is released into the cytoplasm for further modifications.

3B-Hydroxysteroid Dehydrogenase (3B-HSD)

Pregnenolone is converted to progesterone by 3B-hydroxysteroid dehydrogenase (3B-HSD). This enzyme is located in the membrane of the smooth endoplasmic catalyses three alterative reactions, reticulum and pregnenolone to 17α-pregnenolone 17α-progesterone progesterone, to and dehydroepiandrosterone (DHEA) to androstenedione (Figure 1-5). Two human isoforms of 3B-HSD have been identified: type I (3B-HSDI) and type 2 (3B-HSDI). Each is encoded by a separate gene, both located on human chromosome 1 (Berube et al., 1989; Lachance et al., 1991). The predominant isoform in the human adrenal cortex is 3B-HSDII. Mutations of this isoform have been identified and associated with congenital adrenal hyperplasia (CAH) (Section 1.4.1.4) (Rheaume et al., 1992).

The corticosteroidogenesis pathway is different in the zona fasciculata and glomerulosa (and reticularis), due to the local enzyme expression profile which is cell type-dependent (Table 1-2). There is little or no evidence of exchange of substrate between zones.

Zona Fasciculata Pathway

The major glucocorticoid, cortisol can be synthesised from pregnenolone or progesterone, but it is probably the pregnenolone pathway that is more important (Figure 1-5). Both compounds result from 17α -hydroxylase catalysis to produce 17-OH-pregnenolone and 17-OH-progesterone, respectively. As mentioned above 17-OH-progesterone can also be synthesised through the reduction of 17-OH-pregnenolone by 3B-HSDII.

 17α -Hydroxylase is encoded by the *CYP17A1* gene, located on chromosome 10 and expressed in the zona fasciculata and zona reticularis. It is a dualfunctioning enzyme with both 17α -hydroxylase activity and 17, 20-lyase activity (Figure 1-5). While 17α -hydroxylation produces 17-OH-pregnenolone and 17-OHprogesterone, 17, 20-Lyase activity (predominant in ZR) removes the residual side chain and is necessary for the synthesis of the C19 androgens; it also converts 17-OH-pregnenolone to DHEA and 17-OH-progesterone to androstenedione.

The next zona fasciculata step involves 21-hydroxylation, catalysed by 21α -hydroxylase (CYP21A1), which converts progesterone and 17-OH-progesterone to 11-deoxycorticosterone (DOC) and 11-deoxycortisol (S), respectively. This enzyme is encoded by the *CYP21A1* gene on human chromosome 6 (White *et al.*, 1986) and expressed in all cell types of the adrenal cortex (Shinzawa *et al.*, 1988).

The final step in cortisol synthesis is the 11-hydroxylation of 11-deoxycortisol, which is catalysed by the 11B-hydroxylase and occurs in the mitochondria requiring that the 11-deoxy-precursors are again transported to the inner membrane. This enzyme is encoded by the *CYP11B1* gene, which is located on human chromosome 8 (Mornet *et al.*, 1989) and expressed in the zonae fasciculata and reticularis (Ogishima *et al.*, 1992). 11B-Hydroxylase also converts DOC to corticosterone (B), and also catalyses 18- and 19- hydroxylation, converting DOC to 18-hydroxy-11-deoxycorticosterone (18-OHDOC) and 19-hydroxy-11-deoxycorticosterone (18-OHDOC). In addition, it converts corticosterone to 18-hydroxycorticosterone (18-OHB). Corticosterone is the principal glucocorticoid in the rat.

Zona Glomerulosa Pathway

The zona glomerulosa is the site of aldosterone synthesis (Figure 1-5). This zone lacks 17α -hydroxylase and therefore the intermediary substrate synthesised is DOC. To complete aldosterone synthesis, DOC is oxidised in three separate and sequential reactions, each of which is catalysed by a zona glomerulosa-specific enzyme, aldosterone synthase. Aldosterone synthase is encoded by the *CYP11B2* gene, which lies in tandem with the *CYP11B1* gene on human chromosome 8 (Section 1.4.2.1) (Curnow *et al.*, 1991;Kawamoto *et al.*, 1992). Firstly, aldosterone synthase catalyses 11B-hydroxylation of DOC to produce corticosterone (B), C18 of B is then hydroxylated to produce 18-hydroxylsed, this create an unstable structure that is resolved by a spontaneous dehydration, producing an aldehyde group at C18 and resulting in the formation of aldosterone (Denner *et al.*, 1995). The reactions catalysed by aldosterone synthase are summarised in Table 1-4 and the structural modification depicted in Figure 1-4.

Substrate	Reaction	Product
DOC	11B-hydroxylation	Corticosterone (B)
В	18-hydroxylation	18-Hydroxycorticosterone (18-OHB)
Aldosterone	'18-oxidation'	Aldosterone

Table 1-4. Reactions Catalysed by Aldosterone Synthase.

1.2.5 Control of Secretion of Corticosteroids

Expression of the *CYP11B1* and *CYP11B2* genes is primarily regulated by endocrine trophins, including AngII and ACTH. The extracellular concentration of potassium also contributes to their regulation. Circulating AngII and potassium concentrations are the predominant regulators of aldosterone synthesis and share a common intracellular second messenger pathway that increases transcription of the *CYP11B2* gene and production of aldosterone, whereas ACTH is the key regulator of *CYP11B1* gene transcription and cortisol production. These will be described below. The second level of regulation at the molecular level includes regulation of gene transcription and translational and protein formation. This will be discussed in sections 1.2.6 and 1.2.7.

1.2.5.1 Angiotensin II (Angll)

As previously described, Angll synthesis is under the control of the RAS and its release is stimulated by detection of low blood pressure, sodium depletion or hyperkalaemia. It acts on AT1R, exclusively expressed in the ZG of the adrenal cortex (Section 1.2.1). Antagonising AT1Rs inhibits the Angll dependent increase in CYP11B2 mRNA in rabbit and rat adrenal glands (Dudley et al., 1990; Hajnoczky et al., 1992). Angli-mediated effects on adrenal function are achieved by activating multiple intracellular second messenger systems, including the scr family of kinases and the 12-lipoxygenase (12-LO) pathway (Bassett et al., 2004). Furthermore, AnglI stimulation activates membrane-bound phospholipase C which hydrolyses phosphatidylinositol 4,5-bisphosphate (PIP₂) to produce inositol 1,4,5-trisphosphate (IP₃) and diacylglycerol (DAG). Both affect corticosteroidogenesis, IP₃ is soluble and diffuses through the cytoplasm to mobilize bound calcium from endoplasmic reticulum and sacroplasmic reticulum stores, ultimately increasing intracellular free calcium concentrations. By binding calcium-modulating protein (calmodulin or CaM), calcium activates other intracellular pathways, including CaM-dependent protein kinases (CaMK).

Several isoforms of CaMKs are important for corticosteroidogenesis. CaMKI is localized to the ZG and studies using the CAMK inhibitor KN93 and expression vectors in the human adrenocarcinoma cell line, H295R, demonstrated that CaMKI is important for basal *CYP11B2* mRNA expression and also for increased aldosterone production upon stimulation with AngII or K⁺ (Condon *et al.*, 2002). CaMKI is believed to phosphorylate transcription factors that influence *CYP11B2* mRNA transcription (Section 1.2.6).

CaM also activates the CaM-dependent protein phosphatase, calcineurin. A recent study showed that calcineurin mRNA is increased from basal in H295R cells stimulated with AngII, and manipulating calcineurin levels by pharmacological agents, short interfering RNAs (siRNAs) or adenovirus transfer suggest it is a new calcium-dependent regulator of *CYP11B2*, at least *in vitro* (Yamashiro *et al.*, 2010).

In contrast to IP_3 , DAG remains at the cell membrane, initiating translocation of protein kinase C (PKC) to the plasma membrane, then its activation. Several

members of the PKC family have been implicated in adrenal steroidogenesis (Lehoux *et al.*, 2001) and many intermediates, including arachidonate and 12-HETE, may be involved in the stimulation of aldosterone secretion. PKC does not appear to have direct effects on *CYP11B2* transcription but may act by inhibiting the *CYP17A1* gene, thus making aldosterone synthesis the preferred route of steroidogenesis (Bassett *et al.*, 2004).

While AngII is a potent stimulator of *CYP11B2* expression, it also influences other genes that may contribute indirectly to increased production of aldosterone. A recent gene microarray study of stimulated H295R cells identified several genes which has altered expression compared to non-stimulated cells. For example, members of the nerve growth factor IB family successfully increased aldosterone production when exogenously expressed in cells (Romero *et al.*, 2007).

1.2.5.2 Potassium lons (K⁺)

Potassium ions (K⁺) stimulate aldosterone production by both RAS-dependent (alterations in electrolyte balance can modulate the production of AngII synthesis) and RAS-independent direct effects on the adrenal cortex. Increased extracellular K⁺ concentration ([K⁺]_e) increases aldosterone synthesis and secretion.

Increased $[K^+]_e$ causes the ZG cell membrane to depolarise, opening cell membrane voltage-gated calcium channels. This occurs in a concentration dependent manner; changes within physiological ranges activate T-type calcium channels, whereas higher concentrations activate L-Type calcium channels (Lotshaw, 2001). Calcium flux into the cells rapidly increases the intracellular free calcium concentration which, in turn, activates calmodulin and CaM kinases to stimulate aldosterone production in a similar manner to AngII (Section 1.2.5.1) (Pezzi *et al.*, 1997).

Adrenal glomerulosa cells are highly sensitive potassium-sensor cells and exhibit a rapid response to very small changes in $[K^{+}]_{e}$; for example, a small potassium chloride infusion in normal volunteers, which did not significantly change plasma $[K^{+}]$, caused an increase in plasma aldosterone within 30 minutes (Himathongkam *et al.*, 1975). Moreover, in an *in vitro* system that tends to be less sensitive to stimulation, an increase of only 1 mM of K^{\dagger} was sufficient to double the production of aldosterone in primary rat glomerulosa cells (Pralong et al., 1992). This sensitivity arises due to a very large negative resting membrane potential of the ZG cells, consequentially the membrane is highly permeable to potassium. The identity of the channel responsible for maintaining the membrane at this voltage was identified in 2000. The expression of TWIK-related acid-sensitive K^+ (TASK) channel was verified in ZG cells using single-cell PCR. Xenopus Laevis oocytes were used as a model to test the function of these channels and their susceptibility to inhibitors (Czirjak et al., 2000). The voltageinsensitive nature of this class of channels allows them to be constitutively open in ZG cells. Reduction of the levels of TASK mRNA by short interfering RNA (Section 1.5.1) directed against the TASK channel transcript attenuated the potassium current. Further studies identified TASK-3 (KCNK9) as the predominant sub-type in ZG cells (Czirjak and Enyedi, 2002). Homozygous knockout mice have confirmed a role for TASK channels in the regulation of aldosterone production and interestingly show that TASK knockout mice have impaired adrenal zonation, adrenal cortex function and developed a phenotype similar to primary aldosteronism (Section 1.4.1) (Heitzmann et al., 2008). Taken together, these studies demonstrate outward rectifying K⁺ channels are required for corticosteroid synthesis and for the correct development of the adrenal gland.

1.2.5.3 Adrenocorticotropic Hormone (ACTH)

Adrenocorticotropic hormone (ACTH) is the principal secretagogue of cortisol production but also exerts a modest stimulatory effect on aldosterone secretion which is dependent on sodium status. ACTH is the main active component of the hypothalamic-pituitary-adrenal (HPA) axis (Figure 1-6). Typically, central neural pathways are activated by stimuli such as stress and initiate the release of corticotrophin-releasing hormone (CRH) by the hypothalamus. CRH is transferred along the hypophyseal portal circulatory system to stimulate the release of proopiomelanocortin (POMC), from which ACTH is excised by prohormone convertase in the anterior pituitary gland. Cleavage releases ACTH and several other biological peptides, including, melanocyte stimulating hormone and β-endorphin. ACTH secretion exhibits a diurnal rhythm, highest in early morning and lowest late at night.

ACTH is released from the anterior pituitary gland into the systemic circulation and binds with high affinity to the ACTH-receptor. The ACTH-receptor is expressed in adrenal cortical cell membranes and is a member of the G-proteincoupled receptor family. ACTH binding activates adenylate cyclase increasing the intracellular level of cyclic adenosine monophosphate (cAMP) which activates protein kinase A (PKA). These second messengers stimulate *CYP11B1* gene transcription via transcription factors (Section 1.2.6).

ACTH secretion is essential to correct adrenal formation, growth and corticosteroid production whereas, hyper-secretion of cortisol leads to hypertension and Cushing's Syndrome (Section 1.4). To maintain cortisol homeostasis ACTH secretion is subject to negative feedback regulation (Figure 1-6). Cortisol prevents further ACTH secretion by binding to glucocorticoid receptors (GR) in the hypothalamus.



Figure 1-6. Hypothalamic Pituitary Axis Regulation of Cortisol Secretion. POMC: pro-opiomelanocrtin; ACTH: adrenocorticotropic hormone.

1.2.6 Transcriptional Regulation of CYP11B Genes

Transcription factors (TF) are the largest class of DNA-binding proteins. TFs bind to a specific recognition site (or sites) in the 5'-UTR/ promoter region of a gene and function to regulate gene transcription. There are numerous TFs which are cell- and pathway-dependent. Their expression is tightly regulated and they often work in combination with co-factor proteins. They form one of the most important regulatory control systems of gene transcription and understanding how they function with regard to a particular gene can give an important insight in to normo- and patho-physiology. Variation in the 5' promoter structures of the *CYP11B1* and *CYP11B2* genes affect blood pressure regulation and hypertension; this may be because altered sequences influence TF binding this will be discussed further in section 1.4.2.2.

Prior to characterisation of the human *CYP11B1* and *CYP11B2* promoters or of the equivalent regions in rats or mice, the single *CYP11B* bovine gene had been investigated. Several conserved *cis*-elements, known as Ad1 - Ad6 (Adrenal 1-6) were identified (Figure 1-7) (Kirita *et al.*, 1990). Each DNA-binding motif was predicted to bind a nuclear receptor (Bassett *et al.*, 2004). These have since been investigated in the human genes and the findings of these, together with other 5' regulatory mechanisms, will be discussed below.



Figure 1-7. Regions of Transcriptional Importance in the *CYP11B* **Genes.** Ad1-6: adrenal 1-6. Not to scale. (Kirita *et al.,* 1990;Bassett *et al.,* 2004)

1.2.6.1 Transcription Regulation of CYP11B2

In addition to the *cis*-elements detailed above, studies using reporter constructs containing the *CYP11B2* promoter, from which had series of sequence deletions had been made, identified two additional regions of transcriptional importance: a cAMP response element (CRE) and a NBRE-1 site (neuronal growth factor-induced clone B (NGRI-B) response element (Figure 1-8). This study also

confirmed the importance of the Ad5 site in human *CYP11B2* transcriptional regulation (Clyne *et al.*, 1997).

The CRE site binds several TFs, including activating transcription factors 1 and 2 (ATF1 and ATF2) and CRE-binding protein (CREB) (Bassett *et al.*, 2004). All of these TFs can interact with the activation protein-1 (AP-1) complex, specifically the *JUN* and *FOS* genes. Moreover, AngII and ACTH both increase levels of c-fos (the protein product of *FOS*), which can dimerise with Jun to form the AP-1 transcription complex (Romero *et al.*, 2007).

The NBRE-1 site binds members of the NGFI-B nuclear orphan receptor superfamily. Members of the NGFI-B family include NGFI-B, NR4A2 (NURR1), and NR4A3 (NOR1). The family is a member of the *Nur* nuclear receptor family of TFs. There are highly expressed in the adrenal cortex, in a zone-specific manner. Two members of the family, NR4A1 and NR4A2 (nuclear receptor subfamily 4 group A member 1 and 2), are targets for AngII stimulation; NR4A1 also responds to ACTH and NR4A2 to changes in K⁺ concentration. Both have been shown to regulate expression of three steroidogenic-related genes: *3B-HSDII, CYP11A1* and *CYP11B2* (Bassett *et al.*, 2004;Nogueira *et al.*, 2009). A recent study conducted in H295R cells investigated the NBRE-1 site and identified a stimulatory transcriptional role for the other NGFI-B family members, NURR1 and ATF/CREB, a synergistic relationship between them was observed (Nogueira and Rainey, 2010).

The Ad4 site at -344 to -366 of the *CYP11B2* promoter is a consensus match for the steroidogenic factor-1 (SF-1) TF binding site (Figure 1-8). SF-1 is a monomeric orphan nuclear receptor required for correct development and function of the adrenal gland. The site has been the focus of a great detail of research because of the common polymorphism at this site observed in certain hypertensive population; this will be discussed later, in section 1.4.2.2. This research has revealed a positive regulatory function for other genes in the corticosteroid pathway including all steroid hydroxylase genes (Bassett *et al.*, 2002). However, deletion of this sequence (Ad-4) (Section 1.2.6) was without effect on *CYP11B2* expression and further analysis suggested an inhibitory role for SF-1 on *CYP11B2* (Bassett *et al.*, 2002).



Figure 1-8. Transcription Factor Binding Sites on the *CYP11B2* **Promoter.** Not to scale. Adapted from Bassett *et al.*, (2004).

1.2.6.2 Transcriptional Regulation of CYP11B1

Reporter construct analysis of the human *CYP11B1* promoter sequence suggested that, under basal and ACTH stimulation, there are two important regulatory regions:, Ad-1, a cAMP response element (CRE), and Ad-4, a SF-1 consensus sequence (Figure 1-9) (Wang *et al.*, 2000). The CRE site is nearly identical to that observed in the *CYP11B2* gene and mutational studies have proved that it is essential for *CYP11B1* expression during basal conditions (Bassett *et al.*, 2000).

The *CYP11B1* Ad-4 site differs from the *CYP11B2* site by only one nucleotide, but this is sufficient to make SF-1 an effective regulator of *CYP11B1* transcription but not of *CYP11B2* (Bassett *et al.*, 2002). Moreover, *CYP11B1* mRNA is more abundant in H295R cells that over-express SF-1 compared to unmodified cells, and this increase is greater when cells are stimulated with Angll. Targeting SF-1 with an siRNA prevents these increases in *CYP11B1* mRNA (Ye *et al.*, 2009), giving more evidence of the differential effect of SF-1 on the *CYP11B1* and *CYP11B2* genes, which helps to determine the pattern from steroid output of the adrenal cortex. Studies from the same group also identified a regulatory role for SF-1 on several other corticosteroidogenic enzymes, including StAR, CYP11A1 and 17α -hydroxylase (Bassett *et al.*, 2002).

Investigations of the transcriptional regulation of the *CYP11B1* and *CYP11B2* genes have identified numerous transcriptional active proteins, many of which are regulated by AngII, K⁺ and ACTH. However, this area of regulation requires further research and, to date, no good evidence exists to support a role for changes in transcriptional regulation (via TFs) and the development of essential hypertension.



Figure 1-9. Transcription Factor Binding Sites on the *CYP11B1* **Promoter.** Not to scale. Adapted from Wang *et al.*, (2000).

1.2.7 Post-Translational Regulation

Another level at which gene expression can be regulated is by post-translational modification. This includes acetylation, phosphorylation and ubiquitination. There is no reported evidence of post-transcriptional regulation of the *CYP11B1* and *CYP11B2* genes and it is rarely observed in other CYP450 enzymes. However, there is one example: CYP17A1 can be phosphorylated at serine and threonine residues and this can alter the function of the enzyme, with phosphorylation favouring 17,20-lyase activity (Section 1.2.4) (Zhang *et al.*, 1995). Another post-translational modification that is relevant to corticosteroids production is the phosphorylation of bovine adrendoxin (Bureik *et al.*, 2005) and modification of TFs involved in regulating the *CYP11B1* and *CYP11B2* genes (Cammarota *et al.*, 2001;Nogueira and Rainey, 2010).

To summarise, the understanding of transcriptional regulation of the *CYP11B1* and *CYP11B2* genes is relatively well studied. Until recently, it was believed that transcriptional regulation was the dominant factor regulating protein synthesis but novel regulatory mechanisms and regulatory molecules for example microRNAs, has opened new avenues of investigation. microRNAs will be introduced in section 1.5.2.

1.3 Physiological Actions of Corticosteroids

Both aldosterone and cortisol exert their effects by binding to specific nuclear receptors. Nuclear receptors belong to a large superfamily with 49 human members that can be sub-divided on the basis of their receptor dimerization and DNA-binding characteristics. Class I comprises the estrogen receptors (ERa and ERB), the progesterone receptor (PR), the androgen receptor (AR), the glucocorticoid receptor (GR) and the mineralocorticoid receptor (MR). These receptors all share similar structural characteristics; with GR and MR being the most closely related phylogenetically (Mangelsdorf *et al.*, 1995). All have an N-terminal domain, a DNA-binding domain and a C-terminal ligand-binding domain (Arriza *et al.*, 1987).

Unlike other receptors which span or are associated with the plasma membrane, steroid receptors are located in the cellular cytosol (Figure 1-10). In their inactive form, steroid receptors are associated with a complex of chaperone proteins, including heat shock proteins (Hsp90 and Hsp50) (Funder, 1997). Upon binding an agonist, the receptor undergoes a conformational change, dissociates from the chaperone proteins, dimerises and is translocated to the nucleus. The steroid receptor dimer functions as a TF in common with all ligand-bound steroid-binding receptors. The steroid receptor dimer binds consensus hormone-response elements (HRE), which are approximately 15 nucleotides in length and are located in 5' enhancer or promoter regions of target genes (Figure 1-10). Two zinc fingers present in the DNA-binding domain of the MR are responsible for tethering it to the HRE. Gene transcription is initiated, leading to the synthesis of new protein.

1.3.1 Mineralocorticoids

The MR is encoded by the NR3C2 gene, located on human chromosome 4 (Morrison *et al.*, 1990) and is mainly expressed in polarised epithelial cells in such organs as the kidney, colon and salivary glands. It play a key part in regulating electrolyte metabolism and acid base balance and more recently, it has been identified in non-epithelial cells in the brain, heart and vascular tissue (Funder, 2005).

The MR has poor specificity with aldosterone and cortisol demonstrating approximately equal binding affinity. The circulating plasma levels of aldosterone are relatively low, (<1nmol/L) and half is bound to albumin, and the rest readily available for receptor binding. In contrast, the circulating levels of cortisol are 100- to 1000-fold higher and only 5% is bound. Thus, it would be expected that cortisol would be the dominant ligand of MR. However, this is not the case because MR is protected from cortisol by 11B-hydroxysteroid dehydrogenase type 2 (11BHSDII) which is co-localised with MR in epithelial cells (Figure 1-10) (Funder et al., 1988;Edwards et al., 1988). 11BHSDII metabolises cortisol to cortisone, which does not bind the MR. As previously mentioned, DOC has mineralocorticoid properties and may have a higher affinity for MR than its physiological potency implies. A recent study demonstrated that MR, in epithelial tissue, also co-localises with a member of the aldo-keto-reductase family, AKR1C3. The investigators found that AKR1C3 can protect MR from inappropriate DOC activation by converting it to 20α -hydroxy-DOC, an inactive steroid (Sharma et al., 2006).

In epithelial tissues, the principal function of aldosterone-bound MR is to increase sodium reabsorption, linked to fluid retention, and increase excretion of potassium and hydrogen ions. Within the kidney, MR is expressed in the distal convoluted tubule and the cortical collecting duct (CCD) (Todd-Turla *et al.*, 1993). The main cell type in this region, responsible for salt homeostasis are principal cells, which lie adjacent to intercalated cells.

The primary function of intercalated cells is in the control of acid-base balance. They express H^+ -ATPase channels and MRs which are both regulated by aldosterone. In addition to MR, several channels and transporters (discussed below) are expressed within principal cells to facilitate the movement of Na⁺ and K⁺ across apical and basolateral membranes, and these are all subject to aldosterone regulation.

The amiloride-sensitive epithelial sodium channel (ENaC) is located on the apical membrane and allows Na^+ to move from the lumen of the nephron down an electrochemical gradient into the principal cell. This is the rate-limiting step of Na^+ reabsorption (Horisberger, 1998). Also expressed on the apical membrane is the renal outer medullary potassium channel (ROMK), which facilitates passive

movement of K⁺ across the membrane. The basolateral membrane contains three transporters, the Na⁺/K⁺ ATPase, the Na⁺/H⁺ exchanger (NHE) and the ATP sensitive K⁺ channel (Figure 1-10). These all function to complete Na⁺ reabsorption into the blood while maintaining the electrochemical gradient for transport across the apical membrane.

There are 8 human isoforms of NHE and each has different expression patterns and regulatory mechanisms. NHE3 has been identified as the isoform most sensitive to aldosterone and its effect is blocked by spironolactone, an MR antagonist (Drumm *et al.*, 2006). It is believed that aldosterone works in conjunction with the epidermal growth factor receptor to activate NHE3 expression and increase trafficking of the transporter to the membrane surface (Thomas and Harvey, 2010). Aldosterone also has slower effects which may be mediated by protein kinase C (PKC) or by the aldosterone-inducible proteins serum- and glucocorticoid-regulated kinase 1 (SGK1) and channel-inducing factor (CHIF) (Section 1.3.1.1) (Odermatt and Atanasov, 2009;Thomas and Harvey, 2010).

Finally, aldosterone regulation of the ENaC is essential for salt and fluid homeostasis and the steroid can alter the rate of reabsorption in two ways: by altering the number and density of ENaC channels on the cells' surface or by changing the opening probability of ENaC. The former is likely to be a chronic (3-24 hours) effect requiring synthesis of new protein, whilst the latter is more likely to be an acute effect (<3 hours). ENaC is composed of three subunits (α , β and γ) and is continually recycled from the membrane surface and intracellular pools (Butterworth et al., 2005). In rats treated with either aldosterone or dexamethasone, a synthetic corticosteroid, ENaC subunit mRNA was increased in a cell-type specific manner; in the cortex of the kidney, there was an increase in the α -ENaC mRNA, whereas in the distal colon both B-ENaC and γ -ENaC subunits were increased (Asher et al., 1996). A proposed mechanism for direct aldosterone action is that ligand-bound MR binds to the 5' regulatory regions on the ENaC gene increases transcription (Mick et al., 2001). Additionally, aldosterone can have indirect effects on ENaC, which are mainly mediated through Aldosterone-induced Proteins (AIP) (Section 1.3.1.1).

1.3.1.1 Aldosterone-Induced Proteins

The most widely studied AIP is SGK1, a serine-threonine kinase upregulated in response to mineralocorticoids (Figure 1-10). Experiments conducted in epithelial cell lines, adrenalectomised mice and using ectopic expression of SGK1 in Xenopus Laevis oocytes identified a regulatory role for SGK1 through increased ENaC activity (Chen et al., 1999). Moreover, SGK1 mRNA was shown to increase in response to dexamethasone stimulation (Chen et al., 1999). Further evidence of SGK1 involvement in salt homeostasis came from homozygous sgk1 knockout mice, mice on a low NaCl diet had impaired salt reabsorption and low blood pressure, despite high circulating aldosterone levels (Wulff et al., 2002). SGK1 action can affect sodium transport; for example, it phosphorylates the ubiguitin protein ligase, neural precursor cell expressed, developmentally downregulated 4-2 (Nedd4-2) (Figure 1-10). In its unphosphorylated state Nedd4-2 ubiguitinates ENaC, which directs the channel for degradation but phosphorylation inactivates Nedd4-2 and prevents this, leading to increased ENaC at the apical membrane (Debonneville *et al.*, 2001).

Another AIP is the corticosteroid hormone induced factor or channel-inducing factor (CHIF), a small membrane protein (FXYD family) associated with the Na⁺/K⁺ ATPase transporter in renal tissue (Figure 1-10). CHIF mRNA expression is increased in rats administered aldosterone (Brennan and Fuller, 1999). CHIP is homologous to the γ -subunit of Na⁺/K⁺ ATPase and it is thought that it increases Na⁺/K⁺ ATPase affinity for Na⁺, thus enhancing the reabsorption rate of Na⁺. Increased transport of Na⁺ increases the electrochemical gradient of Na⁺, thus also increasing transport across the apical membrane (Beguin *et al.*, 2001).

The mRNA of endothelin-1, another AIP (Figure 1-10), is rapidly increased in response to aldosterone in a rat smooth muscle cell lines (Wolf *et al.*, 2006). This increase was also observed in adrenalectomized rat kidneys and colon following aldosterone treatment (Wong *et al.*, 2007). More recently, aldosterone-induced endothelin-1 was shown to be specifically expressed in the renal collecting duct and its induction to be mediated by two putative HRE present in the endothelin-1 promoter (Stow *et al.*, 2009). The mechanisms of endothelin-1 actions are unclear but *in vitro* and *in vivo* experiments have

demonstrated that it can increase *SGK1* mRNA and may exert its actions via this AIP (Wolf *et al.*, 2006).

The importance of other AIPs is less clear. GTP-dependent signalling protein Kras2 (KiRasA) increases ENaC opening time *in vitro* but experiments in rats detected increased expression only in the colon and not in the kidney (Odermatt and Atanasov, 2009). Another AIP with an unclear mechanism or relevance is glucocorticoid-induced leucine-zipper (GILZ) protein. This antagonises extracellular signal-regulated kinase (ERK), a negative regulator of ENaC; thus GILZ increases Na+ reabsorption (Bhalla *et al.*, 2006).

Taken together, there is substantial evidence that aldosterone affects several transporter systems in the nephron, both directly and indirectly, via AIPs, and that this facilitates the homeostatic regulation of sodium and potassium transport and water balance, which are key factors in blood pressure control.





Aldosterone binds to the mineralocorticoid receptor (MR). This complex dimerizes and translocates to the nucleus. The dimer binds to a hormone-response element (HRE) on the gene promoter and influences transcription. Cortisol is inactivated by 11β-dehydrogenase II (11βHSDII) to form cortisone. ENaC: epithelial sodium channel; GILZ: glucocorticoid-induced leucine zipper protein: ERK; extracellular signal-regulated kinase; ROMK; renal outer medullary potassium channel; Sgk1: serum- and glucocorticoid-regulated kinase 1; CHIF: corticosteroid hormone induced factor; Nedd4-2 neural precursor cell expressed, developmentally down-regulated 4-2.

1.3.1.2 Non-Genomic Actions of Aldosterone

In addition to acting as a ligand-dependent TF in combination with MR, aldosterone has rapid cellular effects which cannot be explained by changes in gene regulation. Classical aldosterone actions mediated via the MR are subject to a lag phase (1-2 hours), can be blocked by MR inhibitors and are susceptible to transcription or translational blockade, for example with puromycin or actinomycin D (Edelman *et al.*, 1963;Horisberger and Diezi, 1984). However, physiological responses to aldosterone have been observed over shorter time frames (1-2 minutes) and are not affected by blockade of the nuclear MR (Funder, 2005).

Aldosterone's capacity to increase sodium exchange in canine erythrocytes was observed in the 1960s, a response which obviously does not involve the nucleus (Spach and Streeten, 1964). This was followed by elegant studies in human leucocytes and the idea of non-genomic aldosterone action was founded (Wehling et al., 1989). A striking finding of these studies was that membrane binding sites had very low K_d , indicating high affinity. These mechanisms have since been demonstrated in renal epithelial cells (Gekle et al., 1996), in several primary renal cell cultures (Koppel et al., 2003), in vascular smooth muscle cells (Christ et al., 1995) and in primary human colon cells (Doolan et al., 1998). In vivo non-genomic effects of aldosterone have been supported by MR knockout mice; ex vivo skin cells from $MR^{-/-}$ mice respond rapidly (<2 minutes) to aldosterone stimulation (Haseroth et al., 1999). Several reports show rapid vasoconstriction in response to aldosterone however, recent evidence demonstrates vasodilation, that may be mediated via nitric oxide; the physiological significant of this mechanism of action in light of the classical mechanism remains unclear (Grossmann and Gekle, 2009).

It has been postulated that the rapid response may be attributed to a novel receptor, the MR receptor working in a novel, non-genomic way or that classical gene transcription may facilitate a signalling cascade more rapidly than is currently estimated (Vinson and Coghlan, 2010). Evidence suggests that this rapid response mechanism involves changes in intracellular free Ca²⁺ concentration, as well as several second messenger systems (Grossmann and Gekle, 2009). A list of proteins induced by aldosterone has been derived from

2D-electrophoresis experiments of yeast extracts, which do not express nuclear receptors (Bohmer *et al.*, 2006). This is promising work but functional experiments will be required to understand fully the mechanism involved.

1.3.1.3 Aldosterone Actions in Non-Epithelial Cells

MR expression in non-epithelial tissues has revealed a novel mechanism of aldosterone action in tissues other than epithelia. MR expression has been identified in many cell types including vascular cells (smooth vascular and endothelial), cardiac cells (fibroblast and cardiomyocytes), neuronal cells (particularly in the CNS) and circulatory cells (monocytes, leucocytes and lymphocytes) (Viengchareun *et al.*, 2007;Odermatt and Atanasov, 2009). With the exception of the vascular cells, the MR is not co-expressed with 11BHSDII. Thus, activation of MR will be dominated by glucocorticoids in these tissues (Alzamora *et al.*, 2000).

The clinical significance of aldosterone's blood pressure controlling properties will be discussed later in section 1.4, but it is worth noting here that aldosterone's actions in the cardiovascular system extend beyond blood pressure control and are often detrimental. An early study showed that rats on a high salt diet had greater cardiac collagen accumulation, fibrosis and left ventricular hypertrophy when given aldosterone in comparison to control animals (Brilla and Weber, 1992). This phenomenon is called cardiac remodelling. A subsequent study showed that these effects were dependent on high salt status and independent of blood pressure changes (Young *et al.*, 1995). Furthermore, a recent study conditionally over-expressing MR in endothelial cells of mice found the animals to have increased vascular resistance, vasoconstriction and hypertension (Cat *et al.*, 2010).

The mechanism of aldosterone-induced cardiovascular damage is unclear; a number genes which appear to be regulated by MR have been identified, including those involved in collagen synthesis, inflammation and cell proliferation (Viengchareun *et al.*, 2007). Prior to these observations, cardiac remodelling had been attributed to indirect effects of aldosterone and DOC mineralocorticoid activity principally raised blood pressure. However, an important clinical heart failure trial (RALES) in the late 1990s showed that the

addition of spironolactone (an MR antagonist) to standard treatment reduced the risk of morbidity and mortality when compared to standard treatment alone (Pitt *et al.*, 1999).

1.3.2 Glucocorticoids

The majority of cortisol's effects are mediated through binding to the glucocorticoid receptor (GR). GR activation is identical to that of MR; the active GR dimer complex translocates to the nucleus and binds to specific glucocorticoid-responsive elements (GRE) located in promoter regions of target genes stimulating their expression. This process is known as transactivation. An alternative method of GR function is transrepression, involving negative GRE (nGRE) sites. Once bound, these act to repress gene transcription. An example of transrepression is the negative feedback mediated by nGRE located in the proopipmalencortin (*POMC*) gene, which is an ACTH precursor gene (Section 1.2.5.3) (Drouin *et al.*, 1993).

The GR is ubiquitously expressed (Ballard *et al.*, 1974) and thus cortisol has a wide range of target tissues and can facilitate a range of effects (Sapolsky *et al.*, 2000). The primary role of cortisol is to mediate the stress response and metabolism of fat and carbohydrate, leading to increased glycogenolysis and blood glucose levels in competition with insulin. Other known roles of cortisol include suppression of the inflammatory and immune responses and, most relevant to this review, regulation of blood pressure. The mechanisms by which cortisol can regulate blood pressure are complex and several examples are listed in Table 1-5.

Target Cell Type	Via GR	Via MR
Vascular Smooth Muscle Cell	Increased contractility Decreased proliferation Decreased migration	Increase perivascular inflammation Changes in vasotone
Endothelial Cell	Decreased vasodilation Decreased angiogenesis	Changes in vasotone
Myocardium		Increased fibrosis
Macrophage	Changes in cytokines Increase apoptosis	
Non-Cardiovascular Organs	Obesity Hypertension Dyslipidaemia Insulin resistance Glucose intolerance Prethombotic	Hypertension Prothrombotic

Table 1-5. Cardiovascular Effects of Glucocorticoids (Walker, 2007).

1.4 The Role of Corticosteroids in Hypertension

The previous sections have discussed the synthesis, regulation and function of corticosteroids and their role in blood pressure homeostasis. The detrimental affects of hypertension and its role as a risk factor for cardiovascular disease was described in section 1.1.2. In approximately 5% of hypertensive patients, there is a definable cause of hypertension, known as secondary hypertension. The following sections describe these rare but important causes. They show clearly that corticosteroids can cause hypertension. More recent studies indicate that corticosteroid metabolism may also be altered in the remaining 95%, so-called essential hypertension. This will be reviewed with particular attention to the *CYP11B1* and *CYP11B2* genes.

1.4.1 Secondary Hypertension

For the purpose of this thesis, and to highlight the importance of the adrenal cortex and hypertension two types of secondary hypertension, first primary aldosteronism which will also introduce the aldosterone to renin ratio and its use as a diagnostic tool; secondly monogenic forms of hypertension will be introduced, paying particular attention to those in which mineralocorticoid production is altered.

1.4.1.1 Primary Aldosteronism

Primary aldosteronism (PA) (or primary hyperaldosteronism) is an important cause of secondary hypertension characterised by increased aldosterone concentration and associated with normal renin levels (Section 1.4.1.2). PA was first recognised by Dr Jerome W. Conn in 1954 in a 34-year old woman with hypertension and muscle spasms and weakness due to hypokalaemia. This led to a diagnosis of excess production of the salt-retaining corticoid from the adrenal cortex. Surgical intervention revealed a right-sided adrenal tumour which, when removed, left the patient free of all symptoms (Conn, 1955). In the following ten years, Conn identified 145 similar cases and proposed that a proportion of essential hypertension cases might have primary aldosteronism.

Since then, the number of essential hypertensive patients with primary aldosteronism is estimated to be 10% of patients (Funder *et al.*, 2008) although in accepting this figure, it is important to stress that the definition of the disorder has been modified since Conn's original description.

Currently, there are several sub-classifications of primary aldosteronism which differ in terms of pathology and treatment. The subtypes of primary aldosteronism are listed in Table 1-6. One of the main causes is a small, solitary aldosterone-producing adenoma (APA), similar to the first case described by Conn; currently the aetiology of APA is unknown. Adenomas are autonomous sources of aldosterone synthesis that is not under the control of AngII stimulation. In cases of the main type of PA, bilateral idiopathic hyperplasia (IHA) the ZG is hyper-responsive to AngII, which leads to the expansion of the ZG. Patients with APA have higher blood pressures than those with IHA (Blumenfeld *et al.*, 1994). Diagnosis and treatment of PA is important as they have a higher rate of cardiac events (stroke, myocardial infarction *etc*) than age-, gender- and blood pressure-matched essential hypertensive patients (Milliez *et al.*, 2005).

Pathology	Prevalence
Aldosterone-producing adenoma (APA)	35%
Bilateral idiopathic hyperplasia (IHA)	60%
Unilateral (primary) adrenal hyperplasia	2%
Pure aldosterone-producing adrenocortical carcinoma	<1%
Familial hyperaldosteronism (FH) FH-I (Glucocorticoid-remediable aldosteronism) FH-II (APA or IHA)	< 1%

Table 1-6. Subtypes of Primary Aldosteronism (taken from Young (2007).

1.4.1.2 The Aldosterone to Renin Ratio (ARR)

The diagnosis of PA is now largely dependent on the aldosterone to renin ratio (ARR). This index was introduced in 1981 and has since been accepted as a useful diagnostic tool for hypertensive diseases as it is relatively simple, inexpensive and can compensate for salt intake, diurnal variation and posture (Hiramatsu *et al.*, 1981;Montori and Young, 2002). A high ARR demonstrates that aldosterone production is inappropriately high for the level of renin, suggesting an error in the efficiency of corticosteroid synthesis. The test has been shown to be useful for diagnosis of PA, hypertension, hypotension, resistant hypertension (Funder *et al.*, 2008). Furthermore, the ARR is highly heritable (38.1%) thus is likely to be at least partly determined by genetic variation (Alvarez-Madrazo *et al.*, 2009). To summarise, aldosterone concentration and electrolyte balance are frequently normal in PA patients, therefore the ARR can detect alterations in the corticosteroid pathway that signify an underlying pathology.

1.4.1.3 Disorders of Cortisol Production

Cortisol deficiency, for example in Addison's Disease leads to hypotension, hypoglycaemia and weight loss (Ten *et al.*, 2001). Conversely, in Cushing's syndrome, where cortisol levels are high, blood pressure is severely elevated. Other clinical examples of cortisol excess associated with hypertension are: unilateral benign or malignant adrenal adenomas or bilateral adrenal hyperplasia or dysplasia, overproduction of ACTH caused by anterior pituitary adenomas (Cushing's disease) or, rarely, by ectopic ACTH-producing tumours. It is also a risk for patient's dependant on long-term glucocorticoid therapy such as autoimmune disease and after heterogenic transplantation. The syndrome is characterised by central adiposity, glucose intolerance and hyperglycaemia, hyperlipidaemia, hypertension and severely increased risk of cardiovascular disease. In addition to its actions mediated through the GR pathologically high levels of cortisol may overwhelm the capacity of 11BHSDII to protect the MR allowing cortisol to exert additional MR-like actions (Section 1.3.1).

1.4.1.4 Monogenic Causes of Hypertension

Another source of secondary hypertension is the mutations arising in genes encoding enzymes of the corticosteroid biosynthetic pathway. This type of Mendelian disease offers an interesting perspective on the contribution of genetics to the development of blood pressure pathology. However, these are rare, only accounting for less than 1% of hypertensive cases. To date, mutations in 8 human genes have been identified as causing hypertension and 9 as causing hypotension (Lifton *et al.*, 2001). All of these mutations result in changes in renal salt reabsorption which cause early onset, severe changes in blood pressure.

Congenital Adrenal Hyperplasia (CAH)

Congenital adrenal hyperplasia (CAH) is a group of autosomal recessive loss-offunction mutations in genes of the corticosteroid biosynthetic pathway. Defects reduce the production of cortisol therefore, stimulating massive continued ACTH production which causes hyperplasia of the cortex as well as abnormal increases in precursor levels (Section 1.2.5). Clinical symptoms result from two abnormalities, increased levels of DOC and altered levels of adrenal androgens.

The symptoms present in CAH patients depends on which enzyme the causal mutation lies. The most common mutations are in the *CYP21A2* gene coding for 21-hydroxylase which accounts for approximately 90-95% of cases of CAH; approximately 75% of patients cannot synthesise appropriate levels of aldosterone and have severe salt-wasting and hypotension (White and Speiser, 2000). As can be inferred from Figure 1-5, 21 α -hydroxylase deficiency will result in insufficient cortisol synthesis, but the resulting increase in ACTH drive will result in high levels of 21-deoxy-precursors (17-OH-pregnenolone and 17-OH-progesterone) and raised androgen synthesis, creating masculinisation in women (Krone *et al.*, 2007).

 17α -Hydroxylase deficiency comprises approximately 5-8% of CAH cases and was identified in the 1960s (Biglieri *et al.*, 1966). Inactivating mutations can be located at many positions in the *CYP17A1* gene and lead to inefficient conversion of 17-OH-pregnenolone to dehydroepiandrosterone (DHEA) (Figure 1-5).This deficiency also affects sex steroid production in the gonads and leads to

feminisation of males and failure of normal development of secondary sex characteristics in females. Massively raised DOC levels cause mineralocorticoid hypertension; high corticosterone level as a substitute glucocorticoid for cortisol.

Mutations in the *CYP11B1* gene (which codes for 11B-hydroxylase) lead to deficient 11B-hydroxylation, creating inefficient conversion of DOC to corticosterone and of 11-deoxycortisol to cortisol (Figure 1-5). The ratio of 11-deoxycortisol to cortisol to cortisol concentration (S:F ratio) is used as an index of 11B-hydroxylase deficiency. This form of CAH presents with mineralocorticoid hypertension and glucocorticoid (cortisol) deficiency. High androgen production leads to virilisation in girls and precocious puberty in boys.

Both 17α-hydroxylase and 11β-hydroxylase deficiencies cause symptoms that are associated with hypermineralocorticoidism, for example hypertension, but in this instance the causative steroid is DOC and not aldosterone. Decreased circulating cortisol concentrations associated with both deficiencies lead to an increased ACTH drive, which further increase the rate of DOC synthesis. Diagnosis is confirmed by genetic screening of the genes involved and analysing androgen concentrations. Treatment is by glucocorticoid replacement therapy which normalises ACTH drive, correcting overproduction of corticosteroid precursors and in turn, correcting the hypertension.

Glucocorticoid Remediable Aldosteronism (GRA)

GRA (familial hyperaldosteronism type I, FH-I) is an autosomal dominant disease which is characterised by high aldosterone, low renin and early onset hypertension. GRA is caused by a chimeric gene generated by unequal crossing over during meiosis at the promoter regions of the *CYP11B1* and *CYP11B2* genes. The hybrid gene contains the promoter sequence of the *CYP11B1* gene and distal regions of the *CYP11B2* gene. Thus, the novel gene is expressed in the ZF and allows ectopic aldosterone production abnormally under the control of ACTH, producing massive excess (Lifton *et al.*, 1992). Aldosterone production does not therefore respond to normal signals such as high sodium levels or expanded extracellular volume (Section 1.2.5). The unique feature of GRA is that, upon treatment with glucocorticoids, the phenotype is completely suppressed, by removing ACTH drive. A second class of familial hyperaldosteronism also exists

(FH-II) and is thought to be more prevalent than type I. Patients do not have a chimeric gene and do not respond to glucocorticoids. The molecular basis of this disease is not known but has been link to genetic variation, identified on human chromosome 7, investigations continue to identify the gene(s) involved (Jeska *et al.*, 2008).

Apparent Mineralocorticoid Excess (AME)

The syndrome of apparent mineralocorticoid excess (AME) is a rare, autosomal recessive hypertensive disorder (Ulick *et al.*, 1979). It is characterised by hypokalemia and metabolic acidosis resulting from low renin and aldosterone levels and is caused by loss of function mutations in the *11BHSDII* gene, which prevents the 11B-dehydrogenase 2 enzyme (11BHSDII) from oxidising cortisol to cortisone in the kidney (Section 1.3.1). This allows cortisol to saturate the renal MR and induce a severe mineralocorticoid form of hypertension (Mune *et al.*, 1995;Stewart *et al.*, 1996). Treatment includes the use of MR antagonists, such as spironolactone. The importance of 11BHSDII is illustrated by knockout mice which exhibit severe hypertension (Kotelevtsev *et al.*, 1999).

Aldosterone Synthase Deficiency

Rare loss of function mutations in the *CYP11B2* genes impairs the final three steps of aldosterone synthesis. This rare condition causes salt-wasting and hyperkalemia, increased plasma renin activity and hypotension, with normal cortisol and sex steroids levels (White, 2004). Two forms of aldosterone synthase deficiency have been described; corticosteroid methyloxidase (CMO) type I and type II (Ulick, 1976).

In summary, PA and monogenic mutations of corticosteroid enzymes are rare in the general population and result in extreme phenotypes, but they provide irrefutable evidence that altered corticosteroid synthesis resulting from a variety of genetic changes can affect blood pressure. Recent evidence suggests that, in essential hypertension (a much more common disease) subtle genetic changes in the genes coding for corticosteroid enzymes may significantly contribute to the heritable component. This is discussed below.

1.4.2 Corticosteroids and Essential Hypertension

The following section will first outline the characteristics of the *CYP11B1* and *CYP11B2* genes and then discuss the role of these genes in essential hypertension, including the association of several polymorphisms present within theses genes with raised blood pressure.

1.4.2.1 The CYP11B1 and CYP11B2 Genes

To recap, the CYP11B1 and CYP11B2 genes encode 11B-hydroxylase and aldosterone synthase, respectively. As previously described, these gene products are crucial to the synthesis of corticosteroids and the maintenance of blood pressure. To date, only one mRNA transcript has been identified for each of the human genes; each has 9 exons and 8 introns (Figure 1-11). The genes are located in tandem on human chromosome 8 and lie only 40 kilobases (kb) apart. Both enzymes are comprised of 503 amino acids and share 95% sequence identity in coding regions and 90% in intronic regions. The 5'- and 3'- regulatory regions are the area of most heterogeneity (Mornet *et al.*, 1989). At present the 3-dimensional structure has not been solved for either protein.



Figure 1-11. The Exon and Intron Structure of the *CYP11B1* **and** *CYP11B2* **Genes.** Located on Human Chromosome 8. Diagram not to scale.

1.4.2.2 Genetic Variation in the CYP11B Locus

Several variants have been identified across the *CYP11B* locus and include SNPs, conversions and insertion/deletions. These variants are in relatively high linkage disequilibrium (LD) (a marker of the likelihood of alleles being inherited together), meaning there are few common haplotypes. One of the most commonly studied SNPs is in the *CYP11B2* promoter located at position -344 (rs1799998). This SNP lies in a binding site for the SF-1 transcription factor and can either be a C (cytosine) or T (thymine) nucleotide, in approximately equal

frequencies (Figure 1-12). A second common mutation is a gene conversion, where part of intron 2 of the *CYP11B1* gene is duplicated in the corresponding intron 2 region of *CYP11B2* (Figure 1-12) (White and Slutsker, 1995). Together, these variants are in tight LD; the frequencies of their haplotypes are : T/Conc (38%), C/Wt (45%) and T/Wt (16%) (Davies *et al.*, 1999).



Figure 1-12. The Location of the -344 SNP and the Intronic Conversion Polymorphism. Diagram not to scale.

Several studies in the late 1990s associated the T allele at -344 (-344T) and the presence of the intron conversion (IC) with raised blood pressure. A case-control study found that the -344T allele was more frequently identified in hypertensive patients than in normotensive controls (Brand et al., 1998). However, this study failed to detect any association of the IC with hypertension possibly due to an equal frequency of this allele in both the cases and control populations (Brand et al., 1998). A small study found that male subjects with the -344T allele had higher urinary aldosterone levels (Hautanen et al., 1998). Moreover, Davies et al., (1999) found that both the -344T allele and the IC were more frequently observed in hypertensive patients than age- and sex- matched normotensive controls. They also showed in the North Glasgow Monitoring of Trends and Determinants in Cardiovascular Disease (MONICA) study, which contained 486 individuals, there was a significant increase in urinary excretion of tetrahydroaldosterone (the main aldosterone metabolite) associated with the -344T allele of heterozygotic subjects (-344 C/T) when compared to homozygous -344C subjects (Davies et al., 1999). This finding was corroborated in a separate study carried out in normotensive subjects by Paillard et al., (1999) who reported significantly higher levels of plasma aldosterone concentration in -344T subjects compared to -344C subjects. A study of a small Japanese cohort identified an association between the -344T allele and low-renin hypertension, but failed to identify any difference plasma in activity or plasma aldosterone concentration (Komiya et al., 2000).
However, not all reports of this type have been consistent; in one study, the -344C allele was associated with increased plasma aldosterone concentration in hypertensive patients (Pojoga et al., 1998) and a large case-control study in Japanese patients failed to detect any association of the -344 SNP and the IC with hypertension (Tsujita et al., 2001). A recent meta-analysis set out to address this disagreement between the studies; it confirmed the association between the -344T allele and increased risk of hypertension (Sookoian et al., 2007). However, the findings were limited. It could not positively associate the -344 SNP with systolic or diastolic blood pressure or define a link between the polymorphisms and aldosterone excretion or plasma concentration (Sookoian et al., 2007). These polymorphisms may have a role in PA; in 27 well-characterised Conn's patients there was a higher frequency of the -344T and IC allele (Inglis et al., 2001) and the -344T allele was associated with a higher ARR in non-selected hypertensive subjects (Lim *et al.*, 2002). Despite the abundance of studies which suggest a link between the CYP11B2 polymorphism and essential hypertension, the causative mechanism has not yet been identified. It was suggested that the -344 SNP altered the binding of SF-1 TF to the promoter to affect transcription but in vitro, the -344T allele has 4-fold less affinity of SF-1 and yet has no effect on gene transcription (Bassett et al., 2002). There are several possible explanations; it may be the -344 and IC polymorphisms are merely markers of a causal variant in near-by genes in this region of high LD. Alternatively, the functional mutation may lie in the relatively under-studied IC region, or that other polymorphisms located across the locus may have yet undiscovered functions.

The first of these proposals that -344T/C is a marker for linked polymorphisms elsewhere has been investigated in the *CYP11B1* gene. Similar to the *CYP11B2* gene, *CYP11B1* is highly polymorphic and these polymorphisms are in tight LD with those in *CYP11B2*. It has been proposed that polymorphisms in *CYP11B1* result in inefficient 11B-hydroxylation, identifiable by altered plasma S:F and DOC:B ratios or of urinary metabolites, tetrahydrodeoxycortisol (THS) to tetrahydrodeoxycorticosterone (THDOC) (Section 1.4.1.4). A mild 11B-hydroxylase inefficiency, while it would not affect cortisol levels, would result in increased ACTH drive over a long time, contributing to the development of essential hypertension and mildly raised aldosterone secretion (Davies *et al.*, 2009). An argument supported by the associated increase in ARR (Lim *et al.*, 2002).

In the presence of impaired efficiency, exogenous ACTH should accentuate abnormal S:F ratio. Essential hypertensive patients given an ACTH infusion had increased S:F ratios compared to controls, supporting reduced 11B-hydroxylation efficiency as a disease marker (Honda *et al.*, 1977;de Simone *et al.*, 1985;Connell *et al.*, 1996). This observation was then linked to polymorphisms in the *CYP11B2* gene in a study undertaken in 92 male subjects; those with the -344T/ intron 2 conversion had higher urinary aldosterone excretion rates than -344C/intron 2 wild-type subjects. Further, they had higher 11-deoxycortisol levels in plasma when stimulated with ACTH (Hautanen *et al.*, 1998). Davies *et al.*, (2001) observed a similar relationship with the polymorphisms and plasma DOC concentration in normotensive subjects and in another normotensive cohort study the excretion rate of the major 11-deoxycortisol metabolite (tetrahydro-11-deoxycortisol) being higher in urine from -344T subjects; supporting the hypothesis of inefficient 11B-hydroxylation (Kennon *et al.*, 2004).

Other genetic variants within CYP11B2 and CYP11B1 genes have also been implicated in altered 11B-hydroxylation efficiency. A large study evaluated the roles of 6 CYP11B2 SNPs and 3 CYP11B1 SNPs, finding a significant association between the presence of a SNP in exon 7 (T4986C) of CYP11B2, plasma cortisol concentrations and the S:F ratio . Additionally, a family-based studied found a significant association between THaldo excretion rate, and a SNP in intron 3 (rs6387) of the CYP11B1 gene (Imrie et al., 2006). However, a complete genetic screen of the CYP11B1 gene in normo- and hyper-tensive subjects could not relate variation in the coding region of this gene to hypertension or to an altered ARR (Barr et al., 2006). Interestingly, a screen of the non-coding regions identified two novel polymorphisms in the promoter region of CYP11B1, at -1889 (G/T) and -1859 (A/G) (Figure 1-13), which associated with the ratio of urinary THS:total cortisol excretion. Moreover, these polymorphisms were found to be transcriptionally functional, responding differently to trophins in vitro. Crucially, these polymorphisms are in strong LD with the -344T/C, IC CYP11B2 polymorphisms (Barr et al., 2007).



Figure 1-13. The Location of the CYP11B1 Promoter SNPs: -1859 and -1889. Diagram not to scale.

To summarise, genetic variation in the *CYP11B1* and *CYP11B2* genes is associated altered corticosteroid metabolism, the ARR and hypertension. The precise causal molecular modifications are uncertain but it appears likely that variation in one *CYP11B* gene can affect the other and this is supported by the strong LD identified across the locus. The foregoing discussion strongly supports the conclusion that small changes: the efficiency of corticosteroid-related enzymes, genetically-based and expressed over a lifetime, can result in clinically important changes. It seems reasonable to suggest that any other mechanism which influences the expression of the *CYP11B* genes may have a similar effect. microRNAs may be just such a mechanism, and are discussed in the following sections.

1.5 RNA Interference (RNAi)

RNA Interference (RNAi) is an evolutionarily conserved, naturally-occurring regulation mechanism in eukaryotes. Utilising small, non-coding, doublestranded (ds) RNA that is complementary to, and binds mRNA, RNAi results in post-transcriptional gene silencing. The mechanism of RNAi was first observed in plants (Napoli *et al.*, 1990) but the molecular mechanism was only fully described several years later in the nematode *Caenorhabditis elegans* (*C. elegans*) by the Melo Group (Fire *et al.*, 1998). RNAi has since been identified in many organisms. High-throughput sequencing has identified several classes of small RNA that facilitate RNAi: short interfering RNA (siRNA), micro RNA (miRNA) and p-element-induced wimpy testis RNA (piRNA or piwiRNA) (Farazi *et al.*, 2008;Kim *et al.*, 2009).

Since their discovery small RNAs have become an effective scientific tool for investigating control of gene expression which has shown them to be important regulators of cellular function with great relevance to understanding disease pathologies.

1.5.1 Short Interfering RNA

siRNAs are small double-stranded RNA molecules that bind to and silence specific mRNA targets. siRNAs can be used experimentally and introduced into cells exogenously but also can be derived endogenously by viral infection or gene transcription. siRNAs have perfect base-pairing complementarity to their target genes and result in mRNA degradation by cleavage thus, siRNAs are important regulators of various cellular functions and have become a useful research tool and offering the possibility of novel therapies (Ghildiyal and Zamore, 2009).

1.5.2 microRNAs

miRNAs are a tightly defined class of endogenous, non-coding RNA molecules that are highly conserved across evolution. The regulatory power of these molecules is a relatively new discovery and was awarded the title of scientific breakthrough of the year by 'Science' in 2002 (Couzin, 2002). miRNAs have been identified in a wide range of eukaryotes but for the purpose of this thesis, the main focus will be on human miRNAs, unless otherwise stated.

1.5.2.1 Nomenclature

The identities and sequences of experimentally-validated miRNAs are deposited in the miRNA repository, miRBase (Griffiths-Jones, 2004; Griffiths-Jones et al., 2006; Griffiths-Jones et al., 2008). The database maintains accurate information on each miRNA identified thus preventing duplication. Leading scientists in the field have devised a strict naming convention for miRNAs (Ambros et al., 2003). Each miRNA is preceded with a species identifier, for example, hsa (human) or mmu (mouse). miRNAs are numbered chronologically from the time of submission to miRBase and, where applicable, miRNAs with identical sequences in different species are given matching numbers. The miRNA number may be followed by a lowercase letter that identifies miRNAs with similar sequences (e.g. hsa-miR-125a and hsa-miR-125b); miRNAs with similar sequences belong to a 'miRNA family'. Two miRNAs may possibly be generated from each hair-pin pre-miRNA (Section 1.6.2.3); these are given the same number but are distinguished by -3p (3') or -5p (5') according to the arm of the hair pin from which the miRNA originated (e.g. hsa-miR-125a-3p and hsa-miR-125a-5p). Alternatively, the strands can be distinguished by an asterisk (*), representing the non-functional miRNA strand (e.g. hsa-miR-149 and hsa-miR-149*). A single miRNA type may arise from two different genetic loci and in this instance, miRNAs are given a numerical suffix to identify their origin (e.g. hsa-miR-24-1, hsa-24-2) (Ambros et al., 2003).

1.5.2.2 Discovery

miRNAs were discovered in 1993, the inaugural miRNA, lin-4, was isolated from *C. elegans* (Lee *et al.*, 1993). By cloning the *lin-4* gene, Lee *et al.* (1993) established that the product was not translated into a protein but instead generated a non-coding RNA. At the same time, Ruvkun and colleagues demonstrated that lin-4 could bind to the 3'UTR (untranslated region) of the *lin-14* gene and reduce protein production (Wightman *et al.*, 1993). Further experiments showed that repression of *lin-14* by lin-4 was an important

mechanism in controlling developmental timing in *C. elegans* (Olsen and Ambros, 1999).

At the time of publication, the full significance of miRNA post-transcriptional regulation was not appreciated. However, with the discovery of the second miRNA in 2000, the importance of miRNA regulation started to become clearer. The candidate miRNA was let-7c; it was found to be highly conserved across species (Reinhart *et al.*, 2000) and to be a member of the large let-7 family, which has several subtypes. Let-7c is capable of regulating a number of developmental genes by binding to their 3'UTR and proved to be important in temporal changes in the late development of *C. elegans*. It is not expressed during early development then increases 10-fold at early pupal stage that is sustained to adulthood (Pasquinelli *et al.*, 2000).

Over the last 10 years, interest in miRNA-mediated gene regulation has accelerated and the latest update of miRBase (v. 16, September 2010) includes sequences of 1048 human miRNAs (Griffiths-Jones, 2004;Griffiths-Jones *et al.*, 2006;Griffiths-Jones *et al.*, 2008). Moreover, novel regulatory pathways have been unveiled since the discovery of miRNA (Section 1.7).



Figure 1-14. Overview of miRNA Maturation and Degradation.

Primary miRNA (Pri-miR) transcripts are transcribed in the nucleus from miRNA genes. A hair-pin structure is cleaved from the pri-miR by Drosha endonuclease. The Pre-miR is exported to the cytoplasm via Exportin-5, further cleaved by Dicer, and then unwound to form a mature miRNA (yellow). This then associates with several proteins known as the miRNA-induced silencing complex (miRISC). This then targets mRNA by binding to the 3'UTR. The passenger strand (black) is degraded by XRN-2.

1.5.2.3 Synthesis

miRNA transcription and maturation is a multistep process that starts with the nuclear transcription of a stem-loop structure from genomic DNA (Figure 1-14).

Nuclear Processing

Primary miRNA transcripts (pri-miR) are transcribed from miRNA-genes, located in intergenic chromosomal regions, by RNA polymerase II or RNA polymerase III (Lee *et al.*, 2004a;Borchert *et al.*, 2006) (Figure 1-15). Pri-miRNAs are several kilobases in length, are polyadenylated and have 5' 7-methyl-guanylate caps. Several different miRNA molecules can be excised from a single polycistronic primiR. miRNAs generated from the same genetic location (*i.e.* from one pri-miR) are termed a miRNA cluster. This is important as their production is subject to the same regulation (Mourelatos *et al.*, 2002) and can be important in pathophysiology for example, in lung cancer samples the expression of all six members of the miR-17-92 cluster is higher than in controls (Hayashita *et al.*, 2005).

Pri-miR transcription may be regulated by a mechanism involving ADARs (adenosine deaminase acting on RNA). ADARs catalyse the substitution of adenosine nucleotides for inosine in a process known as A-to-I editing. This was first described for both human and mouse pri-miR-22 by (Luciano *et al.*, 2004) but has been subsequently identified for other pri-miRs (Blow *et al.*, 2006;Yang *et al.*, 2006). By altering the secondary structure of a pri-miR molecule, A-to-I editing can affect miRNA processing (Section 1.5.2.3 and Figure 1-14) both enhancing synthesis (e.g. pri-miR-142;(Kawahara *et al.*, 2008) or preventing further maturation (Scadden, 2005). Thus, pri-miRs are subject to post-transcriptional regulation prior to the cleavage steps described below.

Hair-pin structures are then cropped from the pri-miR transcript. This is catalysed by a nuclear Microprocessor Complex, of which the primary components are Drosha and Di George critical region 8 (DGCR8, also known as Pasha) proteins (Figure 1-14). Drosha, an RNAse III enzyme, is highly conserved between species and contains two RNAse III domains and a double stranded RNA binding domain (dsRBD). DGCR8 contains two dsRBDs and is responsible for binding to the pri-miRNA and determining the site of cleavage that creates the pre-miRNA. Other proteins involved in the microprocessor are DEAD box RNA helicases, p68 (DDX5) and p72, heterozygous nuclear ribonucleoprotein and SMAD proteins (Gregory *et al.*, 2004;Davis *et al.*, 2008). These co-factors can affect miRNA synthesis; p68 or p72 knockout mice have reduced levels of a specific subset of miRNAs (Fukuda *et al.*, 2007).

The endonucleolytic action of Drosha cleaves the pri-miRNA to produce a hair-pin structure(s), approximately 70 nucleotides in length, called pre-miRNA (pre-miR) (Figure 1-15). The pre-miR is a double-stranded hair-pin RNA molecule with regions of imperfect base-pairing along the hair-pin (Figure 1-16). Other characteristics of a pre-miRNA include a 5' phosphate group, 3' hydroxyl group and a 2-nucleotide overhang at the 3' end. The overhang serves as a recognition motif for further cleavage steps (Lund *et al.*, 2004;Zeng and Cullen, 2004). Single nucleotide polymorphisms in pri-miRNAs can prevent correct processing, for example a SNP in the mature sequence of miR-125a prevents Drosha cleavage leading to an accumulation of pri-miR-125a (Duan *et al.*, 2007). This illustrates the high degree of specificity in miRNA processing.



Figure 1-15. Transcription of microRNA from an Intergenic Region.

miRNA genes are located in the intergenic region (white) of chromosomes and are transcribed by RNA polymerase II to produce a primary-miRNA (pri-miRNA). Splicing produces a pri-miR. This then undergoes maturation to produce a mature miRNA.



Figure 1-16. Nucleotide Sequence of a pre-miR.

An example of the nucleotide sequence for the pre-miR hair-pin of hsa-miR-25. Underlined based show the sequences of the mature miRNA. Taken from Lagos-Quintana *et al.*, (2001).

miRTrons

A Drosha-independent pathway for pre-miR production also exists in mammals, *C. elegans* and plants (Lund *et al.*, 2004;Ruby *et al.*, 2007;Okamura *et al.*, 2007;Berezikov *et al.*, 2007). miRNAs produced in this pathway are generated from miRtrons, where the hair-pin pre-miR structure is located in an intronic sequence of a protein coding gene. They are spliced out during transcription and then processed identically to those produced from the canonical pathway (Figure 1-17).



Figure 1-17. miRNAs Generated from miRtrons.

Gene transcription produces a pre-messenger RNA (pre-mRNA) strand with exons (solid red), introns (solid black) and regulatory regions (red and white stripes). Splicing removes the intron and allows mature mRNA to continue to by translated to produce protein (green). Introns a degraded but may contain a pre-miRNA (yellow), which is spliced out and can produce mature miRNAs.

Nuclear-Cytoplasm Transport

The second stage of miRNA maturation occurs in the cytoplasm (Figure 1-14). The pre-miRNA molecule is transported from the nucleus by Exportin-5 which requires Ran-GTP as a co-factor (Yi *et al.*, 2003). Exportin-5 is postulated to have quality control properties; it will only transport double stranded pre-miRs of a defined length, that have a 3' 2-nucleotide overhang (Lund *et al.*, 2004; Zeng and Cullen, 2004).

Cytoplasmic Processing

Once localised in the cytoplasm, the pre-miRNA is incorporated into the RISC (RNA Induced Silencing Complex) Loading Complex (RLC). The key components of the RLC are Dicer, an RNAse III enzyme, Tar RNA binding protein (TRBP), protein activator of PKR (PACT) and Argonaut 2 (AGO2) (Gregory *et al.*, 2005;Haase *et al.*, 2005;Lee *et al.*, 2006). The pre-miR is tethered by the dsRBD of the TRBP. TRBP activates Dicer by conformational rearrangement (Ma *et al.*, 2008). Dicer cleaves the pre-miRNA at the hairpin loop of the pre-miR to produce a double-stranded miRNA, each strand of which is approximately 22 nucleotides in length. This intermediate structure contains a miRNA guide strand and a miRNA passenger strand (also known as the * strand) (Section 1.5.2.1).

The processing enzymes are essential to miRNA maturation and, as such, loss-offunction mutations in *Dicer1*, but not those in *Dicer2* prevent miRNA production (Lee *et al.*, 2004b). In contrast, depletion of PACT or TRBP only reduces the efficiency of miRNA maturation (Hutvagner *et al.*, 2001;Grishok *et al.*, 2001). Further, Dicer double knockout mice die early in development (Bernstein *et al.*, 2003) emphasising the importance of Dicer in miRNA production and of miRNAregulation in early development. Thus, it is not surprising that Dicer levels must be closely regulated. Moreover, a product of Dicer, let-7 can repress Dicer mRNA, thus providing a negative feedback loop (Forman *et al.*, 2008). Further layers of regulation will undoubtedly be identified in coming years.

Following cleavage, Dicer and the helper-proteins dissociate from the miRNA. The final step in miRNA maturation is the unwinding of the dsRNA releasing the functional guide strand and the non-functional passenger strand. The enzyme that mediates the unwinding of miRNA has not been ascertained but there are several candidates for this roles e.g. p68 which unwinds the let-7c duplex in

mice (Salzman *et al.*, 2007); the DEAD box helicase, RCK/p54, is an alternative candidates in humans (Chu and Rana, 2006). Establishing the mechanism of miRNA unwinding and degradation (Section 1.5.2.4) will be important to understanding miRNA strand selection and expression level.

miRISC

The guide miRNA strand directs the miRNA-induced silencing complex (miRISC) to its target mRNA (Figure 1-14). The passenger strand is degraded. Whilst the full mechanism of strand selection is not known, studies with siRNA and miRNA molecules have shown that the strand with the less stable base-pairing at the 5' end of the duplex is incorporated into the (mi)RISC (Khvorova *et al.*, 2003;Schwarz *et al.*, 2003). The passenger strand is sometimes functional and, in one example, both the guide and passenger strand of a miR species are believed to be active; co-regulation of the REST and Co-REST transcription factors, which are important in Huntington's disease, by hsa-miR-9 and hsa-miR-9*, has recently been described (Packer *et al.*, 2008).

The core component of miRISC is an Argonaut protein. Argonaut proteins are a highly-conserved, ubiquitously-expressed protein family consisting of four members in mammals: AGO1, AGO2, AGO3, AGO4. They comprise a Piwi-Argonaute-Zwille (PAZ) domain, a MID (middle) domain and a PIWI domain. Structural studies in bacteria have shown that the PAZ domain forms a pocket to associate with the 3' nucleotide overhang of the miRNA (Jinek and Doudna, 2009). The PIWI domain resembles RNAse H, an endoribonuclease and therefore is believed to be important for silencing (Jinek and Doudna, 2009). Argonaut proteins also have strand selection properties but this may be more relevant to siRNA silencing (Matranga *et al.*, 2005). In mammalian systems, only AGO2 has cleavage properties and is the predominant mediator of RNAi (Liu *et al.*, 2004). *In vitro* experiments have shown that expression levels of Argonaut proteins correlate with the levels of mature miRNA, whereas their levels of are independent of those of Drosha, DGCR8 and Dicer (Diederichs and Haber, 2007).

1.5.2.4 Turnover

Given the powerful regulatory role of miRNAs and their ability to modify mRNA directly, the levels of active miRNAs are likely to be closely regulated. Similar to

most cellular molecules, the abundance of miRNAs in the cytoplasm is a balance of transcription, processing and decay. Variations in miRNA levels are often associated with disease phenotypes (Section 1.7).

The regulation of miRNA degradation has a dual purpose: first to remove the passenger strand after separation of the pre-miRNA duplex and thus, preventing aberrant miRNA action, and secondly, to clear redundant guide miRNAs. Relatively little is understood about the process of miRNA degradation and the half life of miRNAs. However, recently XRN-2 exonuclease was shown to degrade single-stranded animal miRNA, both guide and passengers *in vitro* (Figure 1-14) (Chatterjee and Grosshans, 2009). Several strategies that protect against degradation have been identified, including 3' adenylation in mammals (Katoh *et al.*, 2009) and 3' methylation in plants (Yu *et al.*, 2005;Li *et al.*, 2005). Moreover, miRNA abundance is correlated with the number and levels of target mRNA (Chi *et al.*, 2009) which may suggest that miRNAs incorporated into miRISC and bound to Argonaut are resistant to degradation.

In summary, miRNA synthesis, maturation and degradation are regulated, multistep, multi-protein processes. Errors occurring during maturation, or polymorphisms in the miRNA sequence can alter the levels of miRNA and therefore have down-stream effects on target protein expression and function.

1.6 miRNA-Mediated Gene Silencing

Canonical miRNA action involves binding to the 3'UTR and consequent repression of mRNA levels, this can occur by transcriptional repression or mRNA degradation and will be discussed in section 1.6.2. Variants of this pathway have been described but it is believed that the majority of miRNAs act via this repressive mechanism.

1.6.1 miRNA Target Recognition

The sequence of miRNAs serves as a target recognition motif for guiding miRISC to target mRNA and thereby tethering the two together. miRNAs predominantly bind to the 3'UTR of mRNA, although there are examples of mammalian miRNA binding to coding regions and 5'UTR regulatory regions (Orom *et al.*, 2008;Duursma *et al.*, 2008;Lee *et al.*, 2009). Most studies, however, focus on binding to the 3'UTR and subsequent changes in target expression.

The degree of miRNA-to-mRNA base-pairing complementarity determines the fate of the targeted mRNA. Perfect base-pairing complementarity leads to mRNA endonucleolytic cleavage by AGO2, a similar mechanism to siRNA-mediated silencing. Perfect base-pairing and cleavage is the prevalent mechanism in plants but is less common in animals. In contrast, imperfect miRNA base-pairing can cause mRNA destabilisation or translational repression. The ability of miRNA to regulate, despite imperfect complementarity, is a special trait of miRNA action in animals. There are a large number of bioinformatic databases which provide researchers with *in silico* tools to predict miRNA:mRNA interaction. However, the ability of miRNAs to act even though complementarity is imperfect increases the difficulty of predicting *bone fide* targets for miRNAs accurately. This will be discussed further in Chapter 5.

1.6.2 miRNA-Mediated mRNA Repression

The original miRNA experiments, that identified *lin-14* as a target for lin-4, only observed repression at the protein level, changes in mRNA levels were negligible (Lee *et al.*, 1993;Reinhart *et al.*, 2000). From this, it was concluded that miRNA

regulation of mRNA expression was only apparent at the level of protein synthesis. However, repetition of these experiments (Bagga *et al.*, 2005) and subsequent studies have shown that miRNAs can reduce both mRNA and protein levels (Lim *et al.*, 2005;Guo *et al.*, 2010).

The exact mechanisms and rules governing miRNA-mediated repression are not fully understood. Current evidence suggests to two models of silencing following miRNA:mRNA target interaction: mRNA destabilisation followed by degradation or mRNA translation repression. There is strong experimental evidence in support of each model and this will be described below.

1.6.2.1 mRNA Destabilisation

miRNAs may prevent translation by destabilising mRNA which can, in turn, lead to degradation. The first step of degradation involves shortening the poly(A) tail (Wu *et al.*, 2006). Removal of the 3' poly (A) tail leaves the mRNA vulnerable to degradation and also prevents binding of PABPs (Poly (A)-binding protein) and translation initiation. mRNA degradation can then proceed in a 3'-5' manner facilitated by exosomes. Alternatively, the 5' m⁷GpppN cap is removed, typically by the decapping enzyme Dcp2, and then the mRNA is degraded in a 5'-3' direction, mediated by XRN1 (Figure 1-18) (Behm-Ansmant *et al.*, 2006).

miRNA-targeted mRNAs destined for degradation are directed to, and accumulate in, cytoplasmic foci called processing-bodies (P-bodies) (Figure 1-18)(Liu *et al.*, 2004). These were first discovered in 1997 and were implicated in miRNA silencing in 2005 (Bashkirov *et al.*, 1997;Pillai *et al.*, 2005). mRNA with shortened poly(A) tails are inducted into P-bodies (Kulkarni *et al.*, 2010). Once internalised, many degradation factors contained in P-bodies degrade the targeted mRNA. Examples of these factors were recently reviewed by Kulkarni *et al.*, (2010) and key factors are summarised in Table 1-7.

miRISC components GW182 (Protein with multiple glycine(G)-tryptophan(W) repeats and a molecular mass of 182 kD) and AGO proteins are required for mRNA de-adenylation, decapping and decay (Behm-Ansmant *et al.*, 2006). Experiments that artificially tethered GW182 to the 3'UTR of a reporter construct identified it as the key protein for decay. Thus, the current model of

miRNA-mediated decay is: miRNA targets the 3'UTR of mRNA, which is bound by the AGO of miRISC and targeted to P-bodies. AGO protein then recruits GW182, which is responsible for mRNA decay (Behm-Ansmant *et al.*, 2006;Li *et al.*, 2008).

P-bodies are motile entities that can move through the cytoplasm and also fluctuate in number and size. Decreasing XRN-1 in human cells, thus, preventing 5'-3' degradation, leads to an increase in the number and size of P-bodies (Sheth and Parker, 2003).

miRNA repression is not always irreversible. An elegant study demonstrated that the cationic amino acid transporter (CAT-1) mRNA is targeted by miR-122 (a liver specific miRNA) and directed to P-bodies (Bhattacharyya *et al.*, 2006). However, under cellular stress conditions, such as oxidative stress (induced with sodium arsenite) or endoplasmic reticulum stress (induced with thapsigargin), the mRNA can be released back in to the cytoplasm, where it can engage again in active translation. The authors demonstrated that this reversal was due to an AU-rich-element binding protein, HuR, which also bound the 3'UTR and provoked CAT-1-induced mRNA release (Bhattacharyya *et al.*, 2006). This raises the possibility that P-bodies may be temporary stores for sequestered mRNA levels.

Component	Function
XNR1	5'-3' Exonuclease
GW182	Degradation
Dcp2:Dcp1	Decapping Complex
Rck	DEAD/H box RNA Helicase
Ccr4, Caf1	3'-5' exoribonuclease (Deadenylase)
AGO (1-4)	RISC Complex

 Table 1-7. P-bodies Degradation Components and Function.



Figure 1-18. miRNA-Mediated mRNA Destabilisation.

mRNA (blue) in its pre-circularized position. miRNA (red) binds the miRNA-induced silencing complex (miRISC) (green) to the 3' untranslated region of mRNA. The top diagram shows CCR4 de-adenylating the 3' poly (A) tail. The bottom figure illustrates the Dcp2:Dcp1 complex dislodging and decapping the mRNA of its 7-methyl guanine (M7G) 5' cap. Each method leads to mRNA being targeted to cytoplasmic foci, called Processing bodies (P-bodies). Within P-bodies, mRNAs are degraded in a bi-directional action, mediated by XRN1 or other exonucleases.

1.6.2.2 Translation Repression

miRNA-mediated silencing may occur at several stages of translation. Translation is a multi-step process, transfer RNA (tRNA) is produce using the sequence of mRNA as a template. tRNA then forms the template for amino acids assembly and protein synthesis. Translation can be divided into three stages: initiation, elongation and termination. Each step must be completed for the production of a correctly aligned and folded protein. Current research seeks to establish whether miRNAs act by preventing initiation or by inhibiting other post-initiation translational steps.

The initiation step of translation begins when a ribosome recognises and binds to the 5' mRNA m⁷GpppN (where N is any nucleotide) cap. This cap is present on all transcribed eukaryotic mRNAs. Several eukaryotic initiation factors (eIFs) are required for cap recognition; there are at least 13 eIFs subtypes. They recruit the small (40S) ribosomal subunit to the 5'UTR of mRNA and interact with poly(A)-binding proteins (PABPs), present at the 3'UTR (Pelletier and Sonenberg, 1988). This interaction brings the ends of the transcript together in what is known as circularisation, which is thought to increase eIFs affinity for PABPs and increase the rate of initiation (Figure 1-19A) (Kahvejian *et al.*, 2005). The initiation process is the rate-limiting step and is under a high degree of regulatory control. In certain circumstances, mRNA translation initiation can be cap-independent. In these cases, internal ribosome entry sites (IRESs) located in the 5'UTR allow interaction with the 40S subunit (Pelletier and Sonenberg, 1988).

Initiation Inhibition

miRNA-mediated repression of translation initiation was first proposed by Humphreys *et al.*, (2005) and Pillai *et al.*, (2005) from experiments with HeLa cells transfected with reporter constructs containing known miRNA targets. They proposed that miRNAs interfere with cap recognition, therefore preventing translation initiation, and the latter study implicated cap protein, eIF4E as a target for miRNA (Figure 1-19B) (Pillai *et al.*, 2005). Several studies have demonstrated that cap-independent translation is not subjected to miRNA repression, thus modulation of cap-recognition in cap-dependent translation appears to be one method for miRNA-mediated silencing.

AGO2, a core component of the mammalian miRISC binds directly to the cap structure (Kiriakidou *et al.*, 2007). The MID region of AGO2 has a highly similar sequence to the cap-binding eIF and may competitively antagonise binding to mRNA. Further evidence of this is illustrated by mutations in the MID region which can prevent AGO2 binding (Kiriakidou *et al.*, 2007). However, the results of more recent studies disagree. One study suggested that the mutations employed by Kiriakidou *et al.*, (2007) affected interaction with GW182 proteins, thus preventing miRISC silencing (Eulalio *et al.*, 2008). Another study cast doubt on the sequence and structural similarities between the eIF and AGO2 (Kinch and Grishin, 2009).

Post-Initiation

The same group who reported the first miRNA, lin-4 first described miRNA repression that occurs post-initiation (Olsen and Ambros, 1999). Much later, in 2006, polysomal sucrose gradients used to assess the presence of active ribosomal translation began to uncover a mechanism for repression. These studies favoured post-initiation repression that prevents successful mRNA translation during peptide elongation or at termination. However, the results of several studies seemed to indicate marginally different modes of repression. In HeLa cells, miRNA-targeted mRNA was located in active polysomes, indicating that active translation was already in progress (Maroney et al., 2006). This was corroborated in experiments using the C. elegans lin-41 3'UTR targeted with let-7a in HeLa cells, and further verified using the translation inhibitor puromycin, that blocks protein synthesis. When used in this study it released let-7a-bound miRISC from polysomes therefore, confirming active translation (Nottrott et al., 2006). Further, co-immunoprecipitation of an artificial N-terminal tag from miRNA-targeted mRNA in this study failed to identify nascent polypeptides, the authors concluded that miRNAs act by recruiting proteases to degrade newly synthesised peptide, termed co-translational protein degradation (Figure 1-19D) (Nottrott *et al.*, 2006).

Finally, Petersen *et al.*, (2006) demonstrated that translation already in progress could be prematurely terminated by the dissociation of the ribosome from mRNA, and that this dissociation could be caused by miRNA. Pre-mature termination of translation, termed 'ribosome drop-off', has been recently verified in a large-scale study of human and mouse cells (Figure 1-19C) (Guo *et al.*, 2010). Taken together, these studies support the concept that miRNA can repress mRNA translation post-initiation.

In summary, miRNA-mediated silencing occurs either by preventing translation or by degrading mRNA prior to translation. Whether these two methods are mutually exclusive is not yet known. Understanding how miRNAs negatively regulate mRNA is necessary for identifying miRNA targets and increases the potential for therapeutic manipulation.



Figure 1-19. miRNA-Mediated Translation Repression: Post-initiation Mechanisms.

A – Classical transcription; mRNA (blue) in circularized position, stabilised by poly (A) binding protein C1 (PABPC1) and eukaryote initiation factors (eIF) 4e and 4G. miRNA (red) binds the miRISC (green) to the 3' UTR of the mRNA. Active ribosomes (orange) are translating mRNA to form nascent polypeptide.

B - Elongation Prevention; Indicating miRISC repression by blocking eIF4E binding to the 7-melthy guanine ($M^{7}G$) cap, thereby preventing mRNA circularisation.

C – 'Ribosomal Drop-off'; miRISC causing ribosome to prematurely 'drop-off' mRNA.

D – Co-translational protein degradation; miRISC causing nascent polypeptide to be degraded by protease (red).

1.7 Examples of Gene Regulation by miRNAs

It has been estimated that miRNAs bind and regulate approximately 30% of all genes (Lewis *et al.*, 2005) and commonly function to 'fine-tune' mRNA levels and protein expression rather than as 'on-off' switches. However, decreased expression of a miRNA or expression of *de novo* miRNAs is common in embryogenesis and development and also in several disease phenotypes. This section will focus on several examples miRNA regulating genes, pathways and diseases.

1.7.1 miRNAs and Cancer

Alterations in miRNA expression have been widely implicated in tumour development and growth. Understanding the role of miRNAs in cancer will, it is to be hoped, increase understanding of pathogenesis, allow earlier diagnosis and hopefully lead to the development of new therapies.

Altered miRNA expression as a participating factor in neoplastic disease was first observed by the Croce group investigating a chromosomal locus which is absent in chronic lymphocytic leukaemia (CLL). They showed that this region coded for two miRNAs, miR-15a and miR-16 and that the production of these was disrupted and down-regulated in 70% of CLL cases (Calin *et al.*, 2002). Sequence analysis of New Zealand Black mice, a model of CLL, identified polymorphisms in the region coding for these miRNAs thus supporting their role in the development of CLL (Calin *et al.*, 2005;Raveche *et al.*, 2007).

To date miRNAs have been implicated in many cancers including those of the lung, colon, prostate and breast; these have been reviewed by Farazi *et al.*, (2011). miRNAs implicated in cancer broadly fall into two groups based on their function: tumour suppressors or oncogenes. However, classing miRNAs in this manner has to be done carefully as they can function differently depending on the tissue or cell type in which they act. For example, miR-221 and miR-222 have contrasting functions; they can act as tumour suppressors in erythroblast cells by targeting the oncogene, KIT (Felli *et al.*, 2005), or conversely in CLL, thyroid carcinoma and hepatocellular carcinoma, miR-221 and miR-222 can

target and de-repress several tumour suppressors including, p27, p57, phosphatase and tensin homologue (PTEN) and tissue inhibitor of metalloproteinase 3 (TIMP3) (Garofalo *et al.*, 2009), thereby acting as oncogenic molecules.

Broader dysregulation of miRNAs by altered expression of biosynthetic enzymes may be important, as demonstrated by a study which measured Dicer and Drosha levels in patients with ovarian cancer (Merritt *et al.*, 2008). Low expression of these enzymes was positively correlated with disease stage, response to treatment and outcome. Given the relatively short time since the discovery of miRNAs, research has come a long way in identifying the key players in major cancers. However, much still needs to be learned and, given the heterogeneity of cancer, this may prove challenging.

1.7.2 miRNAs and Cardiovascular Disease

There is a large body of research that supports a regulatory role for miRNAs in the function of the cardiovascular system and may implicate them in cardiovascular disease. Targeted knockdown of Dicer during late development in mice leads to dilated cardiomyopathy, decreased cardiac output and heart failure; mice die prematurely at post-natal day 4 (Chen *et al.*, 2008a). Thus, miRNAs are essential for heart development and normal cardiac function.

miRNAs have also been extensively studied in a wide variety of cardiovascular disorders and have been implicated in the pathogenesis of left ventricular hypertrophy, arrhythmias and fibrosis (for a review see Small *et al.*, (2010). Several miRNAs, including miR-1 and miR-133, are cardiac-specific and appear to contribute to the regulation of a diverse range of cardiovascular functions.

Very few studies so far have investigated what role miRNAs may have in human hypertension; this is possibly due to the complex nature of hypertension. A limited number of studies have used *in vitro* methods or animal models to evaluate their role. miR-155 has emerged as a potential candidate, which may target and repress the expression of the AngII type I receptor (AT1R) (Martin *et al.*, 2006). This repression is dependent on the presence of a sequence variant A1166C (rs5186) located in the *AT1R* gene. miRNA repression of a reporter

construct was observed with the 1166A allele but not with the 1166C allele (Sethypathy, 2007). miR-155 expression is lower in aortic samples taken from spontaneously hypertensive rats (SHR) than those from control Wistar-Kyoto rats, moreover in these samples there was decreased protein levels of the AT1R (Zheng *et al.*, 2010). Repression of AT1R by miR-155 would be expected to decrease the vasoconstrictive effects of angiotensin II, thus lowering blood pressure. Hypertension has been associated with the 1166C allele in some populations but not others; if there is a link with the 1166C and hypertension, it would be expected that patients with the 1166C allele would have reduced miR-155 repression of AT1R and, therefore, greater AngII signalling. SNPs present in the genes of other RAS components may similarly modulate the effects of miRNA regulation (Elton *et al.*, 2010).

Thus, whilst miRNAs have distinct expression patterns in cardiovascular tissue and may have major roles in heart development and disease progression (including hypertension-related cardiac pathology), their role in hypertension has not yet been incontrovertibly established.

1.8 Therapeutic Potential of miRNAs

Novel therapeutic strategies are always being sought for diseases, either to improve existing treatment (e.g. less invasive, more specific and manageable strategies), or to treat conditions that do not respond to standard treatment. Unsurprisingly, much research time and money has been devoted to identifying ways of manipulating miRNAs, either by increasing or decreasing their expression as the basis of novel therapies. Problems to be solved include the methods of delivery to target tissues, synthesising stable analogues and reducing off-target effects.

Current synthetic miRNA therapies include the use of viruses to generate premiR molecules (Kota *et al.*, 2009), or direct administration of nucleotide analogues that act as mature miRNAs (Figure 1-20). Several companies are concentrating on making these mRNA-mimics as short as possible in order to increase specificity; RXi Pharmaceuticals (Worcester, MA) are investigating the potential of 15-nucleotide-long dsRNA molecules, whereas Santaris Pharma (Hoersholm, DK) have tested molecules as short as 8 nucleotides.

Options for reducing the destructive effects of miRNA broadly fall into three areas that are summarised in Figure 1-20. They include direct inhibition of the miRNA using antisense molecules, sequestering miRNA with a miRNA-sponge (Ebert *et al.*, 2007), which has several repeats of 100% complementary miRNA sequences, or antagonising miRNA binding sites in the 3'UTR of mRNA (Figure 1-20).

Mechanisms for miRNA delivery include viral vectors or short synthetic oligonucleotides which include the miRNA sequence (Figure 1-20). While viral vectors can provide long term treatment, they also require accurate targeting to the cell nucleus (Section 1.5.2.3). Synthetic RNA molecules are easier to deliver to the cytoplasm but have very short half-lives. For this reason, several strategies that modify nucleic acid structure, helping to stabilise them have been devised (Figure 1-21). Modifications include addition of a methylene bridge between the 2'-O atom and 4'-C atom of the ribose ring. Termed Lock Nucleic Acid (LNA), this can increase the hybridisation affinity to ssRNA (Vester and Wengel, 2004). Another modification introduces a 2'-O-methyl group to the nucleotide which by reducing nuclease degradation, increases stability and also increases affinity for specific RNA molecules (Hutvagner et al., 2004). Finally, a phosphorothioate linkage can be modified; this involves the replacement of the oxygen molecule that normally links a nucleotide to its adjacent nucleotide with a sulphur molecule. Again, this increases stability and reduces plasma clearance; however, it also reduces the affinity for the target RNA. That these approaches are promising is shown by the fact that anti-miRs with both a 2'-P-methyl group and phosphorothioate link successfully target and repress mature miR-122 in the liver for up to 23 days following injection into the tail vein of mice (Krutzfeldt et al., 2005); this was the first effective use of antisense miRNA molecules in vivo.



Figure 1-20. Methods of Inhibiting miRNAs.

A - miRNA inhibition by small anti-sense molecules, which bind to the miRNA molecules, thus preventing them from acting. Delivery mechanisms include small oligonucleotide, with or without modification, or by viruses. B - miR-sponge is a reporter construct with several repeats of the antisense miRNA sequence; these bind miRNAs and inhibit them. C-miR-masking inhibitors are small molecules which competitively antagonise the miRNA binding site present on the 3'UTR.



Figure 1-21. RNA Structural Modifications.

Examples of modifications to nucleotides which stabilise small RNAs. Locked Nucleic Acid (LNA) involves adding a methylene bridge between the 2'-O atom and 4'-Carbon atom of the ribose ring. 2'O-Methyl modification introduces a 2'O-Methyl group to the nucleotide. A phosphorothioate link replaces the oxygen molecule on the nucleotide with sulphur. Taken from Garzon *et al.*, (2010).

1.8.1 Current Clinical Trials

Despite the challenges described in the previous section, there are several pharmaceutical companies who are conducting early pre-clinical trials of miRNAbased therapy. The *in vivo* delivery in small mammals could be replicated in higher mammals was first demonstrated by Elmen *et al.*, (2008), they successfully targeted the liver of African green monkey with a LNA anti-miR-122. Subsequently, several pharmaceutical companies have lodged patents for miRNA therapies and Santaris Pharma this year announced that they are commencing phase II clinical trials using a similar anti-miR-122 (SPC3249) as a treatment for Hepatitis C Virus. Several other companies (and presumably academic institutes) are actively pursuing similar therapies and, thus, the search for new treatments based on miRNAs appears highly promising.

1.8.2 miRNAs as Biomarkers

Further clinical benefits of miRNAs lie in their potential as biomarkers, providing diagnostic signatures of disease. Biomarkers aim to be non-invasive, routine identifier of disease phenotype, stage and response to treatment. DNA is routinely recovered from plasma and, more recently, techniques to isolate RNA (both coding and non-coding) from plasma have opened the door to their use as novel biomarkers (Lawrie *et al.*, 2008). miRNAs have also been successfully isolated from urine, saliva and amniotic fluid (Chen *et al.*, 2008b). The pattern of miRNAs in biological fluids is remarkably reproducible between patients (Chen *et al.*, 2008b) and as they are relatively unharmed by enzyme degradation, freeze-thawing and pH changes, they survive isolation unchanged (Mitchell *et al.*, 2008).

Recent analysis of blood samples from patients with non-small cell lung carcinoma revealed a sub-set of 24 miRNAs whose expression levels were different from those of healthy controls. miRNA levels could be measured with greater than 95% accuracy and specificity and with greater than 90% sensitivity (Keller *et al.*, 2009). Another study examined plasma from patients with colorectal cancer and identified 5 miRNAs (out of 95 miRNAs screened) were up-regulated relative to healthy controls (Ng *et al.*, 2009). The same 5 miRNAs were

up-regulated in cancerous tissue compared to adjacent non-cancerous tissue (Ng *et al.*, 2009).

These examples indicate the great potential of miRNAs to act as a diagnostic tool in disease.

1.9 miRNAs and Endocrine Disease

The only published report of miRNAs modulating aldosterone production was published in 2007 and identified an affect of miR-21 (Romero *et al.*, 2008), which is expressed in the adrenocortical cell line, H295R, and is significantly up-regulated in response to AngII stimulation. H295R cells are of cancerous origin and miR-21 has previously been implicated in oncogenesis (for review see Krichevsky and Gabriely (2009)). Experimental over-expression of miR-21 in these cells increased aldosterone production but not cortisol (Romero *et al.*, 2008). Unfortunately, the investigators failed to assess the effect of miR-21 on *CYP11B2* mRNA or other known regulators of aldosterone biosynthesis. Although this important study suggests a role for miRNAs in corticosteroid production, further work is warranted.

As discussed in section 1.7.1 miRNAs are frequently discordantly expressed in tumours, and this applies also to endocrine tumours. Of particular relevance to this thesis are two recent profiling studies, one of which profiled normal adrenal glands, benign adenomas and adrenocortical carcinoma (ACC) (Tombol *et al.*, 2009), and another which compared macronodular adrenocortical disease samples with normal control tissue (Bimpaki *et al.*, 2010). These studies used Taqman Low Density Arrays to measure the expression of 365 miRNAs, less than a third currently of the total known human miRNA species. The expression of several miRNAs was altered in the diseased tissue compared to the control tissue but neither of these studies was designed to assess miRNA regulation of the *CYP11B1* or *CYP11B2* genes directly.

Further evidence that miRNAs may be determinants of adrenal function comes from a recent study in which SF-1 positive cells (adrenal, ovary and testis cells) were selectively targeted for Dicer knockout by Cre- mediated (Huang and Yao, 2010). While embryonic development of all SF-1 organs proceeded as normal, at 18.5 days post-coitum, adrenal cortex cells became apoptotic and, 48 hours later, the primordial adrenal gland degenerated. It is interesting to note that the testes were affected in a similar manner but the ovaries were normal for the duration of the study, until post-natal day 5 (Huang and Yao, 2010). This emphasises the absolute necessity of miRNAs in late adrenal development and organ maintenance.

The review of recent studies makes clear that miRNAs are important in adrenal development and pathology, and provide preliminary evidence for miRNAmediated regulation of genes involved in corticosteroid biosynthesis. Moreover, several groups have reported the regulation of steroid receptors including MR (Sober *et al.*, 2010) and others by miRNAs (for review see Tessel (2010). miRNA involvement in the regulation of adrenocortical function is becoming established, adding another facet to an already complex system of control. A further understanding of its precise contribution will require careful dissection of its involvement at the important steps of corticosteroid synthesis and function, as described in the earlier sections of this review.

1.10 Aims

The influence of the *CYP11B1* and *CYP11B2* genes in hypertension is well established. However, much still needs to be understood about the regulation of these genes and the consequential impact on aldosterone and cortisol production. Both genes are highly polymorphic (Section 1.4.2.1). The link between these polymorphisms and enzyme activities is yet to be fully elucidated. Finally, the processes underlying the development of APA are still unclear. The novel regulator miRNA may provide some explanations will be investigated in this study.

The aims of the study were:

To investigate the 3'UTR of the *CYP11B1* and *CYP11B2* genes in order to identify putative miRNA binding sites and also the existence and location of polymorphic variation in the 3'UTR.

To establish the expression profile for miRNAs in normal adrenal glands and in aldosterone-producing adenoma samples.

To develop methods to determine the identity and effect of miRNAs acting on *CYP11B1* and *CYP11B2* mRNA.

To determine using *in vitro* systems, whether miRNAs regulate the corticosteroidogenic pathway.

2 Material and Methods

2.1 Cell Culture

Culturing of cell lines was performed in a biological class II vertical laminar flow cabinet using sterile conditions. Cells were grown in a monolayer in tissue culture flasks, with normal growth medium placed in a humidified chamber at 37° C in 5% CO₂/95% air.

HeLa cells (an immortalised cell human line derived from cervical cancer) were purchased from the European Collection of Animal Cell Cultures (Porton Down, Wiltshire, U.K.). HeLa cells were maintained in Dulbecco's modified Eagles medium supplemented with 10% (v/v) fetal calf serum (FCS), 2 mM L- glutamine and 1 IU penicillin, 100 µg/ml streptomycin (Invitrogen, Paisley, U.K.).

The H295R (strain 2) human adrenocortical tumour cell line was a kind gift from Professor William Rainey (Medical College of Georgia, GA, U.S.A.). H295R cells were maintained in Dulbecco's modified Eagles medium with F12 supplement (Invitrogen, Paisley, U.K.) containing HEPES buffer, L-Glutamine and pyridoxine HCl medium. This was supplemented with 2.5% Ultroser G (Pall Bioscience, France), 1% insulin-transferrin-selenium (ITS) (BD Biosciences, Paisley, U.K.), 1 IU penicillin, 100 µg/mL streptomycin (Invitrogen, Paisley, U.K.).

2.1.1 Cryopreservation

For long-term storage, cells were preserved by collection as described in section 2.1.3. After centrifugation, cells were resuspended in complete growth medium, supplemented with 10% (v/v) dimethyl sulphoxide (DMSO) to prevent ice crystals forming during the freezing process. Cells were aliquoted into cryopreservation vials, which were cooled at a constant -1° C/minute to -80° C using isopropanol and a Nalgene Cryo-Container and then placed into the vapour phase of liquid nitrogen for storage.

2.1.2 Revival of Cell Stocks from Liquid Nitrogen

To revive cells from cryopreservation, vials were quickly thawed in a 37° C water bath and the cell suspension immediately transferred to a universal container with 10 ml of complete growth medium. The suspension was centrifuged at 478 x g for 5 minutes to remove the DMSO. The cell pellet was resuspended in 5 ml of complete growth medium, transferred to a T-25 cell culture flask and cultured, as described in section 2.1.3.

2.1.3 Sub-Culturing Cells

Cells were sub-cultured when approximately 70-80% confluent. The medium was removed and the cells were washed with 10 ml of phosphate buffered saline (PBS), which was also discarded. To detach the cells from the flask, 5 ml of 0.025% Trypsin-EDTA solution was added to the cells and the flask placed into the incubator for approximately 5 minutes. Once the cells were successfully dispersed, 5 ml of normal growth medium was added to inactivate the Trypsin-EDTA. The cell suspension was placed in a 25 ml universal container and centrifuged at 478 x g for 5 minutes. The supernatant was discarded and the cell pellet resuspended in an appropriate volume of normal growth medium for subculturing. Typically H295R cells were sub-cultured at a ratio of 1 in 3, and HeLa cells at 1 in 8.

2.1.4 Counting Cells

Counting of cells was carried out using a Bright Line Haemocytometer (Sigma-Aldrich, Poole, U.K.). Cells were collected using Trypsin-EDTA, as described in section 2.1.3. After centrifugation, cells were resuspended in 1 ml of complete growth medium. A coverslip was placed on the haemocytometer and 20 μ l of cell suspension applied by capillary action across the chamber. Using a light microscope, the number of viable cells in each 1 mm corner square was counted and the average calculated. Each corner square on the haemocytometer, with coverslip in place, represents a total volume of 1 mm³ (equivalent to 10^{-4} cm³). Since 1 cm³ is equivalent to a volume of approximately 1 ml, the subsequent concentration of cells per ml can be determined using the following calculation:

Cell number /ml = average cell count per square x 10^4 x original volume

The calculated cell concentration (cell/ml) was then used for plating cells at the desired density.

2.1.5 Stimulation of Cells with Angiotensin II

CYP11B2 gene transcription was stimulated in H295R cells by supplementing the medium with Angiotensin II (AngII) (Sigma-Aldrich, Poole, U.K.). Normal growth medium was supplemented with AngII at a final concentration of 10 nM by adding 150 μ l of AngII (10 μ M stock) to 150 ml of growth medium and added to the cells for 1, 6 or 24 hours.

2.2 Total RNA Isolation

Specialised methods preserving the small RNA fraction were used to isolate total RNA from cells and tissue. The miRNeasy mini kit (QIAGEN, Crawley, U.K.) was used to isolate total RNA from cells and frozen tissue samples because it facilitates the purification of both large and small (<200 base pairs) RNA molecules. The RecoverAll[™] Total Nucleic Acid Isolation Kit (Applied Biosystems, Warrington, U.K.) was used to isolate total RNA from formalin-embedded paraffin-embedded tissue. RNase free plastics, reagents and handling techniques were implemented when working with RNA to minimise degradation of sample.

2.2.1 Total RNA Isolation from Cells and Frozen Tissue

For isolation of total RNA from cells, the direct lysis method of the miRNeasy Kit was followed. The cell medium was removed and the cells washed twice with PBS before adding 700 μ l of Qiazol regent directly to the cells in order to homogenise them. The Qiazol lysis reagent contains phenol and guanidine thiocyanate which efficiently lyse the cells and inhibit RNases. The cell culture vessels were placed in the -80°C freezer for at least 15 minutes to facilitate lysis then thawed at room temperature. The sample was then transferred to an RNase-free 1.5 mL tube and left at room temperature for 5 minutes.

For frozen adrenal glands, a 3 mm³ sample (approximately 40 mg of tissue) was placed into a FastPrep Lysing Matrix D Tube (MP Biomedicals, Illkirch, France) with 700 μ l of QIAzol lysis reagent. Samples were homogenised in a bench top MagNA Lyser instrument (Roche Applied Science, Indianapolis, USA) for 30 seconds at a speed of 535 rpm. The tube was briefly centrifuged to collect the

contents, which were then transferred to an RNase-free tube and left at room temperature for 5 minutes.

Following this incubation, 140 µl of chloroform was added and the tube shaken vigorously for 15 seconds, then incubated at room temperature for 3 minutes. To separate the sample into aqueous and organic phases, the samples were centrifuged at 4° C, 12,000 x g for 15 minutes. The upper, aqueous phase, containing the RNA fraction, was transferred to a clean tube and 1.5 volumes (approximately 425 µl) of 100% molecular grade ethanol added to each sample. The sample was then loaded into an RNeasy mini spin column which was centrifuged for 30 seconds at 8,000 x g to bind RNA molecules within the sample to the silica membrane. The flow-through was discarded and then the membrane washed with 700 µl of Buffer RWT. This was followed by two further washes with 500 µl of Buffer RPE, the first was centrifuged at 8,000 x g for 30 seconds and the second for 2 minutes. The spin column was then placed in a fresh collection tube and centrifuged at 13,000 x g for 1 minute to prevent carry-over of wash buffers. The column was then placed into a 1.5 ml tube, 50 µl of RNase-free water added to the centre of the membrane, left to stand for 1 minute and centrifuged at 8,000 x g for 1 minute to elute the RNA.

RNA was quantified using the Nanodrop (section 2.2.4), DNase treated (section 2.2.3) prior to qRT-PCR measurement, and stored at -80°C.

2.2.2 Total RNA Isolation from Formalin-Fixed Paraffin-Embedded Tissue

The RecoverAll[™] Total Nucleic Acid Isolation Kit (Applied Biosystems, Warrington, U.K.) was used to isolate total RNA from adrenal gland samples that had been formalin-fixed and paraffin-embedded. Four 20 µm sections of tissue were cut using a microtome by Tim Harvey (University of Glasgow, Department of Pathology) and placed in a FastPrep Lysing Matrix D Tube (MP Biomedicals, Illkirch, France) with 1 ml of xylene. The samples were placed in a benchtop MagNA Lyser instrument (Roche Applied Science, Indianapolis, USA) for 30 seconds at a speed of 535 rpm to homogenise the sample. The matrix tube was briefly centrifuged to collect the contents, which were then transferred to an
RNase-free 1.5 mL tube and placed in a 50°C heat block for 3 minutes to melt the paraffin.

The sample was separated into a tissue pellet and a xylene upper layer by centrifugation at maximum speed for 2 minutes. The xylene layer was discarded. Two washes with 100% ethanol were then carried out to remove residual traces of xylene and to aid drying of the tissue. Ethanol was carefully removed and discarded each time. The pellet was then left at room temperature for 40 minutes to air dry. 100 μ l of digestion buffer and 4 μ l of protease solution (both supplied in the kit; concentration unknown)) were added to each sample tube, gently mixed and then placed in a heat block for 15 minutes at 50°C, which was increased to 80°C for a further 15 minutes.

The sample was cooled to room temperature prior to adding 120 μ l of Isolation Additive and 275 μ l of ethanol. The solution was mixed by pipetting, then applied to a filter cartridge, placed in a collection tube, was centrifuged at 10,000 x g for 30 seconds, the flow through was discarded. One wash step with 700 μ l of Wash 1 and one with 500 μ l of Wash 2/3 was performed, centrifuging at 10,000 x g for 30 seconds each time. A final spin of 30 seconds was performed to remove residual fluid from the filter.

An on-column DNase reaction was performed by combining 6 μ l of DNase buffer, 4 μ l of DNase solution (supplied in the kit) and 50 μ l of Nuclease-free water, then adding this solution to the cartridge. The filter was incubated for 30 minutes at room temperature. The two previous wash steps were repeated, this time with a final spin for 1 minute at 10,000 x g. The filter was transferred to a new RNase-free 1.5 mL tube and 60 μ l of nuclease-free water added. The sample was incubated at room temperature for 1 minute before eluting the RNA at 10,000 x g for 1 minute. The sample quality was verified by Nanodrop (section 2.2.4) and Bioanalyser (section 2.2.5) and then stored at -80°C.

2.2.3 DNase Treatment of RNA

DNase treatment was performed to remove residual genomic DNA from RNA samples, thus ensuring qPCR measured only mRNA. DNase treatment was performed using the TURBO DNA-free^M kit (Ambion, Texas, USA). For a 50 µl RNA

sample, 2 μ l of TURBO DNase (2U/ μ l) and 5 μ l of 10X TURBO DNase Buffer was added and carefully mixed. The reaction was incubated at 37°C for 30 minutes. To inhibit the reaction, 10 μ l of DNase Inactivation reagent was added, mixed and left at room temperature for 2 minutes. The samples were then centrifuged at 10,000 x g for 1.5 minutes. The supernatant, containing RNA, was collected and stored in an RNase-free tube at -80°C.

2.2.4 Determination of Nucleic Acid Quantity

A Nanodrop[®] ND-100 Spectrophotometer with ND-1000 v3.1.0 software (Labtech International Ltd, Lewes, East Sussex, U.K.) was used to determine the quantity and quality of nucleic acids. The Nanodrop[®] was set to measure DNA or RNA as appropriate. Nuclease-free water was used to blank the Nanodrop prior to measuring samples. To load the pedestal, 2 μ l of water or sample was applied and the arm lowered. The arm automatically moves to pull the sample into a column between it and the pedestal, which creates an optimal path length for the sample to be measured. The concentration of the sample is calculated using the Beer-Lambert Law of absorption;

Concentration of DNA (μ g/ml) = (A₂₆₀ reading - A₃₂₀ reading) x 50

Concentration of RNA (μ g/ml) = (A₂₆₀ reading - A₃₂₀ reading) x 40

The ratio of absorption at 260 nm and 280 nm was used as an indicator of purity of DNA and RNA samples, with ratios of 1.8 and 2, respectively being indicative of pure samples. Additionally, the ratio of absorption at 260 nm and 230 nm was used as an indicator of RNA purity; pure RNA has a ratio of 2.0 - 2.2, (a common contaminant is phenol, which absorbs at 230 nm and is used in Total RNA extraction).

2.2.5 Agilent Bioanalyser

Total RNA samples isolated for microarray analyses were tested for degradation using an Agilent Bioanalyser 2100 and a Eukaryote Total RNA Nano Series II chip. The analysis was run at the Molecular Biology Support Unit at the University of Glasgow. Results produced included an electropherogram, with RNA peaks at 18S and 28S, and a RNA integrity number (RIN).

2.3 miRNA Microarray

To determine the miRNA expression profile of adrenal tissue (non-tumourous 'normal' and aldosterone-producing adenoma) total RNA was analysed on a miRNA microarray. This was performed by LC Sciences (Houston, Texas, U.S.A.) using 5 μ g of the total RNA previously prepared. For shipping, 0.1 volume of 3M NaOAc (pH 5.2) and 3 volumes of 100% ethanol were added, mixed and placed in an RNase-free 1.5 mL tube. The samples were packed in dry ice and ice blocks for shipping.

Samples were further quality checked prior to proceeding with the miRNA μ Paraflo[®] technology microarray, version 10.1. Briefly, the RNA was labelled with a fluorescent dye then hybridised overnight onto a microfluidic chip containing complementary probes for 723 human miRNAs and other RNA controls. The chip was imaged using a GenePiz 4000B laser scanner and digitised using Array-Pro image analysis software. Raw data matrix was then subtracted from the background matrix. Data adjustment included data filtering, Log₂ transformation, gene centring and normalisation.

2.4 Reverse Transcription

First strand complementary DNA (cDNA) was produced from RNA by reverse transcription to allow quantification of miRNA and mRNA by qRT-PCR. Three different protocols were followed to achieve this depending on the nature of the experiment and each is detailed below. For each RNA sample a no template and a no reverse transcriptase (-RT) control was included to assess the specificity of the qRT-PCR reaction to detect genomic DNA.

2.4.1 ImProm-II[™] Reverse Transcription System

The ImProm-II^M Reverse Transcription System (Promega, Madison, WI, U.S.A.) utilises random primers to reverse transcribe mRNA transcripts in the sample. Following the manufacturer's instructions, 200 ng of DNase treated RNA was combined with 1 µl of Random primers (0.5 µg/reaction) and made up to a final volume of 5 µl with RNase-free water in a 96-well plate. The samples were heated to 70°C for 5 minutes on a 96-well Dyad Disciple^M thermal cycler (MJ Research, Waltham, M.A, U.S.A.), then placed on ice for 5 minutes. The following reagents were combined in a master mix, mixed, centrifuged and then 15 µl of the master mix added to each sample.

Reagent	Volun +RT	ne (µl) -RT	Final Concentration
5X ImProm-II™ Reaction Buffer	4.0	4.0	1X
dNTP Mix	1.0	1.0	0.5 mM/dNTP
MgCl ₂	4.8	4.8	6 mM
RNasin Ribonuclease Inhibitor	0.5	0.5	
ImProm-II™ Reverse Transcriptase	1.0	0.0	
RNase-Free Water	3.7	4.7	

The samples were subjected to the conditions below on a 96-well Dyad Disciple^m thermal cycler (MJ Research, Waltham, M.A, U.S.A.) then 80 µl of nuclease-free water added to each sample, which was then stored at -20°C until required.

	Step	Time (minutes)	Temperature (°C)
1	Annealing of primers	5	25
2	Extension	60	42
3	Inactivation of RT	70	95

2.4.2 TaqMan[®] microRNA Reverse Transcription

TaqMan[®] MicroRNA Reverse Transcription Kit (Applied Biosystems, Warrington, U.K.) was used to generate cDNA of specific miRNAs or controls, to be used with Taqman[®] MicroRNA Assays. A reverse transcription master mix was prepared as follows;

Boggont	Volume (µl)	
Keagent	+RT	-RT
10X Reverse Transcription Buffer	0.750	0.750
dNTP Mix (100 mM)	0.075	0.075
RNase Inhibitor (20 U/µl)	0.094	0.094
MultiScribe™ Reverse Transcriptase (50 U/µl)	0.500	0.000
TaqMan [®] MicroRNA assay primer	1.500	1.500
RNase-Free Water	3.581	4.081

The reverse transcription master mix was added to 2 μ l of DNase treated Total RNA (5 ng/ μ l). The plate was placed on a 96-well Dyad DiscipleTM thermal cycler (MJ Research, Waltham, M.A., U.S.A.) and the samples subjected to the following conditions below. 105 μ l of RNase-free water was added to each sample and stored at -20°C.

	Step	Time (minutes)	Temperature (°C)
1	Primer Annealing	30	16
2	Extension	30	42
3	Inactivation of RT	5	85

2.4.3 miScript Reverse Transcription Kit

The miScript Reverse Transcription Kit (QIAGEN, Crawley, U.K.) comprises a unique reverse transcription mix that contains two enzymes: a poly(A) polymerase that polyadenylates mature miRNA molecules in the sample, and a traditional reverse transcriptase enzyme. The buffer contains random primers to reverse transcribe mRNA and oligo-dT primers to bind and transcribe miRNA products.

The following reaction was set up for each sample: 200 ng of DNase treated RNA was combined with 1 μ l of Reverse Transcription Mix, 4 μ l of 5X miScript Reverse Transcription Buffer and made up to 20 μ l with RNase-free water in a 96 well plate. The reaction was run on the Multi Block System Satellite 0.2 Thermo Cooler (Thermo Fisher Scientific, U.K.) set with the following below. Following the reaction, 80 μ l of RNase-free water was added to each sample and then stored at -20°C.

Step	Time (minutes)	Temperature (°C)
Reverse Transcription	120	37
Inactivation of RT Mix	5	95

2.5 Quantitative Real-Time Polymerase Chain Reaction

Quantitative real-time PCR (qRT-PCR) was performed on cDNA template to determine the relative quantities of mRNA or miRNA in a sample. The reactions were set up using the methods described below in Thermo-Fast[®] 384-well PCR plate (Thermo Fisher Scientific, U.K.) and run on an ABI PRISM 7900HT PCR system. Triplicate reactions were set up for +RT cDNA samples and a duplicate for the -RT sample. Positive controls and water blanks were also included.

2.5.1 Universal ProbeLibrary

The Universal ProbeLibrary (UPL) System (Roche Applied Science, Indianapolis, USA) comprises 165 pre-validated, real-time PCR probes that are labelled with fluorescein (FAM) at the 5' end and a dark quencher dye at the 3' end. Primers were designed using the UPL assay design centre and ordered from Eurofins MWG Operon (Ebersberg, Germany). The amplicon for each primer pair encompassed a specific probe binding site and where possible, intron spanning primers were used. Details of primers and probes can be found in (Appendix, Table 8-7-1). Reaction master mixes were set up as follows;

Reagent	Volume (µl)	Final Concentration
ABsolute [™] QPCR ROX MIX (2x)	5.0	1X
Forward Primer (10 µM)	0.4	400 nM
Reverse Primer (10 µM)	0.4	400 nM
UPL Probe (10 µM)	0.1	100 nM
Nuclease-Free Water	2.1	

Two μ l of cDNA prepared using either the ImPromII RT System or the miScript Reverse Transcription Kit (sections 2.4.1 and 2.4.3) was added to a 384-well PCR plate and the plate briefly centrifuged to collect the solution. 8 μ l of the master mix was added to the appropriate well and mixed. The plate was sealed with an optical QPCR Absolute seal (ABgene Ltd, Epsom, U.K.), centrifuged to collect the reaction and run on the ABI PRISM 7900HT cycler using SDS software with the conditions outlined below:

Step	Time	Temperature (°C)	Cycles
Enzyme Activation	15 minutes	95	1
Denaturation	15 seconds	95	40
Annealing and Extension	1 minute	60	40

2.5.2 TaqMan[®] qRT-PCR

TaqMan[®] pre-designed RT-PCR gene expression assays (Applied Biosystems, Warrington, U.K.) were used to measure relative quantities of PTK9 (Hs00702289_s1), Dicer-1 (Hs00229023_m1) and HMGA2 (Rn00583330_m1*) mRNA. A master mix was set up as detailed below, with 8 μ l of this added to 2 μ l of cDNA sample (section 2.4.1) in a 384-well optical plate. The reaction was run according to the conditions detailed in Section 2.5.1.

Reagent	Volume (µl)
ABsolute [™] QPCR ROX MIX (2x)	5
TaqMan [®] Gene expression Assay	0.5
Nuclease-Free Water	2.5

TaqMan[®] miRNA assays (Applied Biosystems, Warrington, U.K.) were used to measure relative quantities of mature miRNA, details of which are in (Appendix; Table 8-7-2). 1.5 μ l of cDNA, produced using the TaqMan[®] MicroRNA Reverse Transcription Kit (section 2.4.2), was added to a 384-well optical plate with 8.5 μ l of miRNA Taqman[®] master mix, detailed below. This reaction was run according to the cycling conditions in Section 2.5.1.

Reagent	Volume (µl)
ABsolute [™] QPCR ROX MIX (2x)	5.0
TaqMan [®] miRNA expression Assay	0.5
Nuclease-Free Water	3.0

2.5.3 miScript qRT-PCR System

Detection of mature miRNA was performed using the miScript SYBR Green PCR Kit (QIAGEN, Crawley, U.K.) and cDNA template generated using the miScript Reverse Transcription Kit (Section 2.4.3). 10X miScript Primer Assays (Appendix; Table 8-7-3) were reconstituted with 550 μ l of TE (pH 8.0) and stored at -20°C. A reaction mix was set up, as detailed below and 8 μ l of this was added to cDNA, prepared as in Section 2.4.3, in a 384 well plate.

Reagent	Volume (µl)
QuantiTect SYBR Green PCR Master Mix (2X)	10
10X miScript Universal Primer	2
10X miScript Primer Assay	2
Nuclease-Free Water	4

The reaction was subjected to the conditions detailed below on an ABI PRISM 7900HT PCR cycler. A melt curve was performed immediately after the cycling conditions to verify specificity and identity of PCR products; data was collected during the 60°C-95°C ramp phase. A single peak was observed for each primer assay, confirming a single and unique PCR product.

Step	Time	Temperature (°C)	Cycles
Enzyme Activation	15 minutes	95	1
Denature	15 seconds	95	
Annealing	30 seconds	55	40
Extension	30 seconds	70	
	15 seconds	95	
Melt curve	15 seconds	60	1
	15 seconds	95 (slow ramp)	

2.5.4 Analysis of qRT-PCR Results

The extension phase of the PCR releases the fluorescence from the respective qRT-PCR probe, which is measured by the ABI PRISM 7900HT thermo cycler. A threshold of florescence is set during the exponential phase of amplification. The cycle number at which the fluorescence signal breaches this set threshold is termed the cycle threshold (C_t).

qRT-PCR results were analysed using the relative quantification method of comparative C_t . This method assumes the efficiencies of each assays under comparison to be approximately equal. Efficiencies were calculated using serial dilutions of cDNA product and the equation:

Efficiency (%) =
$$(10^{-1/slope}) \times 100$$

The genes *GAPDH* and *B-Actin*, and small nucleolar RNA 48 (*RNU48*) were used as housekeeping genes and assays for each were run on all plates. These act as a normalising control for RNA concentration and small pipetting discrepancies. They had each been tested for consistent expression and reproducibility in appropriate samples. Housekeeping gene expression was used to calculate the ΔC_t for the given gene of interest as follows;

$$\Delta C_t = C_{t(sample)} - C_{t(Housekeeper)}$$

All experiments were compared to a reference control (e.g. Non transfected cells) and the $\Delta\Delta C_t$ calculated as follows;

$$\Delta\Delta C_{t} = \Delta C_{t(sample)} - \Delta C_{t(reference)}$$

The fold change in mRNA or miRNA was calculated as follows;

Fold Change =
$$2^{-\Delta\Delta Ct}$$

2.6 Polymerase Chain Reaction

Polymerase chain reaction (PCR) was performed on a subset of the North Glasgow Monitoring Trends and Determinants in Cardiovascular (MONICA) normotensive cohort to facilitate sequencing of the 3'UTR region of the *CYP11B1* gene.

2.6.1 PCR of CYP11B1 3'UTR

Selected DNA samples previously extracted from patients of the MONICA study were quantified using the Nanodrop (section 2.2.4) and diluted to 20 ng/ μ l. The following reaction used primers, detailed in Table 2-1, and the Expand Hi-Fidelity PCR System (Roche Applied Science, Indianapolis, USA);

Reagent	Volume per Reaction (µl)
Expand High Fidelity Buffer 10x with (MgCl ₂)	2.5
dNTP Mix	2.0
Forward Primer (10 µM)	1.0
Reverse Primer (10 µM)	1.0
Expand High-Fidelity Enzyme Mix	1.0
Nucleases-Free Water	12.5
DNA	5.0
Total	20

Table 2-1. Primers Used to Amplify CYP11B1 3'UTR.				
Name	Sequence (5'-3')	Direction	Tm (°C)	
B1 ex9 seq1F	CAGGTGGAGACACTAACCCA	Sense	59.3	
B1 ex9 R1	ATGCTCTGCCCCTGCAGCTT	Anti-Sense	61.4	

Reactions were set up in a 96-well plate and subjected to the following conditions on a 96-well Dyad DiscipleTM sample block powered by the PTC-0221 thermal cycler (MJ Research, Waltham, M.A, USA).

Time (seconds)	Temperature (°C)	Cycles
120	94	1
15	94	
30	66	10
90	72	
15	94	
30	66	20
90 + 5 seconds / cycle	72	
420	72	1

The PCR product (2,225 base pairs) was verified by agarose gel electrophoresis (section 2.6.2) prior to sequencing (section 2.7).

2.6.2 Agarose Gel Electrophoresis

Agarose gel electrophoresis was used for analysing PCR fragments or restriction endonuclease reactions. A 0.7% agarose gel was prepared by combining 0.7% (w/v) Ultrapure agarose (Invitrogen, Paisley, U.K.) with 100 ml 1x Tris/Borate/EDTA (TBE) (10 mM Tris, 10 mM boric acid, 10 mM EDTA, pH 8.3) and heating in a microwave oven for approximately 80 seconds. Ethidium bromide, 1 μ l (10 mg/ml) (Sigma-Aldrich, Poole, U.K.) was added to the liquid gel in a fume hood; this intercalates with the nucleic acids and fluoresces under UV light. The gel was poured into a gel mould with a Teflon comb and allowed to set for 30 minutes, then submerged in 1 X TBE buffer in an electrophoresis tank and the comb removed. The samples were combined with loading dye (0.02% bromophenol blue, 0.02% xylene cyanol and 2.5% glycerol) and carefully loaded into the cast wells. 5 μ l of 1 kb DNA size ladder (Promega, Madison, W.I, USA) was run to allow the resolution of molecular size. The gel was resolved at a constant voltage of 80 V for 60 minutes. DNA fragments were visualised and captured using UV light on a molecular Imager ChemiDocTM XRS+ Imaging system (Bio-Rad Laboratories, Hemel Hampstead, U.K.) and Multi-Analyst software v 1.1 (Bio-Rad).

2.7 DNA Sequencing Reaction

2.7.1 PCR Purification

The AMPure purification method (Agencourt, USA) was used to purify PCR products from PCR reagents. For PCR reactions with a final volume of 20 μ l, 36 μ l of resuspended, room temperature AMPure reagent was added to each well, the plate sealed, vortexed to mix and left to sit at room temperature for 5 minutes. The PCR products, at least 100 base pairs, bind to the magnetic beads. The plate was then placed on an SPRI (solid phase reversible immobilization) magnet for 5 minutes to draw the beads/PCR product from the liquid phase, which was subsequently cleared from the plate by inversion. The beads/PCR product was washed by two 200 μ l volumes of 70%. Following the second wash, the plate was inverted on a tissue and centrifuged at 76 x g for 30 seconds, then removed from the magnet and left to dry for 10 minutes.

The PCR product was eluted from the beads by adding 40 μ l of nuclease-free water, vortexing and placing the plate back on the magnet. This cleaned PCR product was then transferred to a new plate for use in the sequencing reaction (section 2.7.2) or stored at -20°C.

2.7.2 Sequencing Reaction

DNA sequencing was performed using 10 μ l of purified PCR product or 200 ng of plasmid DNA and the BigDye Terminator v3.1 Cycle Sequencing Kit (Applied Biosystems, Warrington, U.K.). The following components were set up in a 96-well plate:

Reagent	Volume per Reaction (µl)
ABI PRISM BigDye v3.1 Sequencing Buffer	4.0
ABI PRISM BigDye Terminator v3.1 Ready Reaction Mix	1.0
Primer (3.2 µM)	1.0
Plasmid DNA or PCR product	Variable
Nuclease-Free Water	Up to 20 µl

Details of sequencing primers can be found in Table 6-1. The sequencing reaction was subjected to the following conditions on a 96-well heating block.

-	Time (minutes)	Temperature (°C)	Cycles
-	45	96	25
	240	60	23

2.7.3 Sequencing Reaction Purification

The CleanSEQ purification method (Agencourt, USA) was used to remove reagents from the sequencing product. 10 μ l of resuspended, room temperature CleanSEQ and 62 μ l of freshly prepared 85% ethanol were each added to a 20 μ l sequencing reaction. The plate was resealed, vortexed and briefly centrifuged, then placed on a SPRI magnet for 5 minutes. The liquid phase was cleared from the wells by inversion then washed in 150 μ l of 85% ethanol for 30 seconds. The plate was inverted and centrifuged at 76 x g for 30 seconds to remove ethanol. The plate, off the magnet, was then left to air dry at room temperature for 10 minutes.

To elute the sequencing product from the beads, 40 μ l of nuclease-free water was added to each well, the plate sealed, vortexed and centrifuged then placed onto the SPRI magnet to remove beads from the purified sequencing reaction.

2.7.4 Automated Cycle Sequencing

To analyse sequencing products, 20 μ l of the purified sequencing reaction was transferred to a 96-well bar-coded plate. Sequencing was analysed using the Applied Biosystems 3730 DNA Analyzer and SeqScape v 2.1.1 software.

2.8 pEZX - 3'UTR Reporter Construct

Reporter constructs were purchased from LabOmics (Nivelles, Belgium) to determine miRNA binding to the 3'UTR. Each construct contained a pEZX-reporter backbone, which comprises a renilla reporter gene coupled to a SV40 (Simian virus 40) viral promoter, a firefly experimental gene coupled to a CMV (Cytomegalovirus) promoter with a multiple cloning site located at the 3' end of the gene, and a Kanamycin resistant gene for antibiotic selection. Constructs containing the 3'UTR of *CYP11B1* (pEZX-B1), the 3'UTR of *CYP11B2* (pEZX-B2) or a negative control empty vector (pEZX-Con) were purchased (Figure 2-1). The constructs were transformed and grown as detailed below.

2.8.1 Transformation of Competent Cells

One Shot[®] TOP10 *E.coli* competent cells (Invitrogen, Paisley, U.K.) were transformed according to the manufacturer's protocol. Briefly, 1 vial (50 μ l) of cells was thawed on ice for each reaction, and then briefly centrifuged to collect cells. 1 μ l of DNA (approximately 50 pg) was added and the tubes were gently flicked to mix. The cells were incubated on ice for 30 minutes, then heat-shocked by placing vials in a 42°C water bath for 30 seconds. The cells were cooled by placing on ice for 2 minutes. 200 μ l of room temperature SOC (super optimal broth catabolite repression) medium was added to each vial before shaking at 225 rpm, 37°C for 1 hr in an Innova[®] 44 incubator shaker (New Brunswick Scientific, St. Albans, U.K.). 200 μ l of the transformation reaction was plated onto LB (Luria Broth) Agar plates with kanamycin (50 mg/ml) and incubated overnight at 37°C.

Plates were inspected the following day. Single colonies were picked from the plate using sterile pipette tips and placed into individual 15 mL tubes with 2 ml of LB, supplemented with Kanamycin (50 mg/ml). These starter cultures were shaken overnight at 37°C at 225 rpm.



Figure 2-1. pEZX Reporter Constructs.

Outline of the pEZX vector map a) pEZX-control, b) pEZX-B1, c) pEZX-B2. Schematic generated using SerialCloner v2.0 software.

2.8.2 Small Scale DNA Purification

The QIAprep Spin Miniprep Kit (QIAGEN, Crawley, U.K.) was used to purify plasmid DNA from overnight incubations. 1 ml of the incubation was transferred into a 1.5 ml tube, which was centrifuged at 10,000 x g for 10 minutes to pellet the cells. The supernatant was discarded and the cells lysed by resuspending them in 250 µl of Buffer P1. The DNA was denatured by adding 250 µl Buffer P2 and mixing by inversion. This reaction was neutralised by adding 350 µl of Buffer N3 and immediately mixing by inversion. The samples were centrifuged for 10 minutes at 13,000 x g in a table top centrifuge. The supernatants were transferred into a QIAprep spin column, which was centrifuged at 13,000 x g for 1 minute to load DNA on to the silica membrane. The flow through was discarded. The column was washed by adding 500 µl of Buffer PB and centrifuging at 13,000 x g for 1 minute. Another wash with 750 µl of Buffer PE was performed, after which the column was centrifuged at top speed for 1 minute to ensure there was no residual buffer remaining. The column was transferred to a new 1.5 mL tube and the DNA eluted by adding 50 µl of nuclease-free water, standing at room temperature for 1 minute then centrifuging at 13,000 x g for 1 minute. Plasmid DNA was guantified by Nanodrop (Section 2.2.4) and stored at -20°C until required.

2.8.3 Large Scale DNA Purification

Large bacterial culture preparations were set up using 500 μ l of the starter culture (Section 2.8.2) added to 400 ml of LB and supplemented with kanamycin. The culture was shaken at 37°C, 225 rpm overnight.

The QIAGEN[®] Plasmid Purification Kit (Maxi) was used to purify plasmid DNA from *e. coli* cells on a large scale. Following the manufacturer's protocol, the cells were collected in an ultra centrifuge at 6,000 x g for 15 minutes at 4°C, then the cell pellet was resuspended in 10 ml of Buffer P1. To these lysed cells, 10 ml of Buffer P2 was added, mixed by shaking, then 10 ml of chilled Buffer P3 was added and mixed by inversion. The solution was centrifuged at 20,000 x g for 30 minutes at 4°C to remove cell debris. The supernatant containing the DNA was removed and applied to a QIAGEN-tip 100 and the sample allowed to run through

the resin by gravity. The QIAGEN-tip was washed twice with 10 ml of Buffer QC. The DNA was eluted with 5 ml of Buffer QF and precipitated by adding 3.5 ml of isopropanol and centrifuged at 15,000 x g for 30 minutes at 4°C. The pelleted DNA was washed with 2 ml of 70% ethanol, which was placed into multiple 1.5 ml tubes and centrifuged at maximum speed for 10 minutes. The supernatant was discarded and the pellet air dried. Finally, the DNA was eluted by resuspending and combining the DNA in nuclease-free water. The DNA was quantified (Section 2.2.4), used in transfection and stored at -20°C.

2.8.4 Restriction Endonuclease Digestion

Restriction digests were performed using enzymes shown in Table 2-2 to establish the molecular size of the plasmid and insert. Approximately 500 ng of plasmid DNA was combined with 1 μ l of enzyme (1 unit), 2 μ l of buffer, specific to each enzyme (Table 2-2), 0.2 μ l of BSA (except with *Hind*III) and nuclease-free water, up to a final volume of 20 μ l in a 96-well plate. Control reactions included those omitting DNA or enzyme. The plate was placed at 37°C for 2 hours. The product was analysed by agarose gel electrophoresis (Section 2.6.2)

	case Enzymes osed for Plasmid Digestion.			
	Enzyme	Brand	Buffer	
	HindIII	New England Biosciences	2	
	Sacl	New England Biosciences	4	
	BamHI	New England Biosciences	3	
_	Stul	Promega	В	

Table 2-2. Endonuclease Enzymes Used for Plasmid Digestion.

2.9 Transient Cell Line Transfection

2.9.1 Preparation of Small Molecules

Specific miRNA mimics (Pre-miR^M) and miRNA antagonists (Anti-miR^M) were purchased from Applied Bioscience (Warrington, U.K.) to test the effects of particular miRNAs on *CYP11B1* and *CYP11B2* mRNA *in vitro*. Scrambled control molecules for each Pre-miR and Anti-miR were also purchased. Details of the molecules used can be found in Table 5-1. Each Pre-miR^M or Anti-miR^M was resuspended to 6.25 µmoles by adding 800 µl of RNase-free water, vortexing to mix and collecting the contents by centrifugation. Molecules were stored at -20°C.

Two siRNAs designed to target human *Dicer-1* mRNA (s23755 and s23756; Figure 8-7-1) were purchased from Applied Biosystems (Warrington, U.K.) and used to determine the effect of global down regulation of miRNAs. A positive control *GAPDH*-targeted siRNA and a scrambled negative control siRNA were also purchased. The siRNA molecules were reconstituted to a final concentration of 10 μ M and stored at -20°C

2.9.2 Small Molecule Transient Transfection

Small molecules (Pre-miR^M, Anti-miR^M or siRNAs) were transfected into H295R cells in 6-well plates. Reactions were performed in triplicate. Following the transfection agent manufacturer's protocol, cells were trypsinised as in section 2.1.3 and counted as in section 2.1.4. Cells were diluted to a concentration of 2 x 10⁵ cells/ml using normal growth medium and the suspension placed in the 37°C incubator until required.

A transfection reagent master mix was prepared by adding 9 μ l of siPORT^M NeoFX transfection (Applied Bioscience, Warrington, U.K.) to 291 μ l of OptiMEM[®] I reduced serum medium (Invitrogen, Paisley, U.K.) per reaction plus a 10% excess in a 30 mL tube and kept in the dark for 10 minutes.

Small molecules were prepared for transfection by diluting them in OptiMEM[®] medium as detailed in Table 2-3.

Table 2-3. Constituents of Transfection Reaction.				
Small Molecule	Stock Concentration (µmoles)	Volume (µl)	Final concentration (nM)	Volume of OptiMEM (µl)
siRNA	10.00	9	30	291
Pre-miR™	6.25	24	50	274
Anti-miR™	6.25	24	50	274

Table 2-3. Constituents of Transfection Reaction.

The transfection agent and small molecules were combined by adding 300 μ l of the siPORTTM NeoFX/OptiMEM to each transfectant, mixing and incubating for 10 minutes at room temperature.

Following the incubation, 600 μ l of the transfection complex was then added to the cell culture plate. The cells were then removed from the incubator, resuspended and 2.4 ml of the cells added to each well to a final cell density of 4.8 x10⁵ cell/well. The plate was rocked to mix the reagents and then placed in the incubator.

After 24 hours the medium on the cells was exchanged for 3 ml of normal growth medium. After a further 24 hours the medium was removed and stored at -20°C for subsequent steroid analysis (Section 2.11) The cells were lysed for RNA (Section 2.2.1) or protein (Section 2.12) isolation.

2.9.3 Reporter Construct Transient Transfection

The pEZX-reporter construct (section 2.6) was transfected into H295R or HeLa cells in quadruplicate on a 24-well plate. The transfection was set up in a similar manner as detailed in Section 2.9.2, except that the volumes of reagents used were reduced. For a 24-well plate, 1.5 μ l of siPORT^M NeoFX and 48.5 μ l of OptiMEM[®] I were combined and incubated for 10 minutes. The pEZX-construct (100 ng/ μ l) was combined with OptiMEM[®] medium to a final concentration of 50 μ l (Table 2-4).

 Table 2-4. Constituents of Transfection Reaction for pEZX-Reporter Construct.

	Stock Concentration (ng/µl)	Volume (µl)	Final Concentration (nM)	OptiMEM Volume (µl)
pEZX construct	100	1	100	49

50 μ l of the NeoFX/OptiMEM[®] mix was added to each transfectant, mixed and then incubated for 10 minutes at room temperature. Following the incubation, 100 μ l of the transfection complex was added to each well of the cell culture plate. The cells were removed from the incubator, mixed to resuspend and 400 μ l added to each well, to a final cell density of 8x10⁴ cells/well. The plate was rocked to mix the reagents and then placed in the 37°C incubator.

The medium was changed after 24 hours. After a further 24 hours the medium was removed and discarded, and the cells were lysed with 100 μ l of 1X Passive Lysis Buffer (1X PLB) (section 2.10) and stored at -80°C, for subsequent luciferase analysis.

2.9.4 Co-transfection

The pEZX-reporter construct was co-transfected with Pre-miR^M or Anti-miR^M molecules to directly investigate the effects of miRNAs on the 3'UTR of the *CYP11B1* or *CYP11B2* genes. This was performed using HeLa cells in 24-well plates, in a similar manner to that described above. This pEZX-construct (100 ng) was added to Pre-miR^M or Anti-miR^M molecules (final concentration 50 nM) and made to a final volume of 50 µl of OptiMEM[®] medium.

2.10 Dual Reporter Luciferase Assay

Renilla and Firefly luciferase activity was measured in cells, lysed in 1X passive lysis buffer, using the Dual Luciferase Reporter Assay system (Promega, Madison, WI, USA) on a Lumat LB 9507 tube luminometer (Berthold Technologies, Harpenden, U.K.). All reagents were thawed at room temperature and then prepared by reconstituting the lyophilised luciferase assay substrate in 10 ml of luciferase assay buffer II (LAR II) and adding 400 μ l of Stop and Glo Substrate to 10 ml of Stop and Glo Buffer. The luminometer was prepared by priming line 1 with the Luciferase assay reagent and line 2 with Stop and Glow reagent. 10 μ l of lysates were added to round bottomed tubes, in duplicate, and loaded into the luminometer. The Lumat 9507 was set to automatically inject 50 μ l of LAR II reagent, pause for 2 seconds then measure luciferase for 10 seconds. This was immediately followed by an injection of 50 μ l Stop and Glo reagent, a further 2

second pause and then a luciferase measurement over 10 seconds. Firefly luciferase activity measured in relative light units (RLU) was divided by renilla luciferase activity (RLU) to normalise for transfection efficiency and the mean taken of the duplicate ratios.

2.11 Steroid Quantification

Cell media was stored at -20°C until steroids could be measured using Liquid Chromatography Tandem Mass spectrophometry (LC:MS/MS) by Miss Mary Ingram. Steroids were extracted from the media using Chem Elute cartridges (Varian, CA, USA) and eluted with dichloromethane. The sample was then evaporated to dryness under nitrogen and reconstituted in 60 μ l acetylonitrile. A 20 μ l aliquot was injected into a Polaris 5 micron, 150 mm x 2 mm C-18-A reversed phase HPLC column. Identification and quantification of steroid products was achieved by tandem mass spectrometry using a Varian 1200L mass spectrometer with a triple quadruple detector.

2.12 Determination of Protein Concentration

Total protein concentration in cells lysed in 1 X Reporter Lysis Buffer (1X RLB) was measured using a bicinchoninic acid (BCA) assay kit (PIERCE, Perbio Science, Northumberland, U.K.) following the manufacturer's protocol. A BCA working reagent (WR) was prepared by mixing BCA Reagent A with BCA Reagent B at a ratio of 50:1 (A:B). Protein standards were prepared using the albumin standard included in the kit and 1X RLB to cover a range of 25 μ g/ml - 2000 μ g/ml, 1X RLB was used as the blank for calibration. In duplicate, 25 μ l of protein standards or unknown sample was added to a 96-well flat bottom plate, to which 200 μ l of the working reagent was added. The plate was incubated at 37°C for 30 minutes then cooled prior to measuring the samples' absorbance at 560 nm, using a Wallac Victor² plate reader (Wallac, Turku, Finland). The blank absorbance reading (1X RLB) was subtracted from each sample or standard reading. A standard curve was plotted and the concentration of the samples extrapolated.

2.13 Statistical Analysis

Statistical analysis was performed using Prism 4.0 Graph Pad software. All results are expressed as mean ± SEM (standard error of the mean). *In vitro* experiments were performed in at least three technical replicates, at three independent biological times (unless otherwise stated). *In vitro* results were analysed by either an unpaired Student's t-test or one-sample t-test as stated. Analysis of results for multiple groups was performed by one-way analysis of variance (ANOVA) and Bonferroni's post-hoc test, comparing all results to a designated control group or all sets of data as required.

Confidence intervals of 95% were used, with a P-value of < 0.05 therefore being considered significant.

3 Investigating a Role for miRNA Regulation in Adrenal Cells

3.1 Introduction

miRNA molecules are novel regulators of mRNA levels and protein expression (Section 1.5 and 1.6). The RNase III, Dicer, is essential for miRNA maturation and its deletion in the SF-1 positive cells of mouse embryos prevents adrenal formation (Huang and Yao, 2010). Therefore, miRNAs are crucial to the embryonic development of the adrenal gland. However, their influence on corticosteroidogenesis has not been established. This study uses Dicer-targeted siRNAs (Section 1.5.1) and 3'UTR reporter constructs, each transfected into an adrenocortical cell line, to determine which enzymes in the corticosteroidogenic pathway are subject to miRNA-regulation.

In order to understand whether miRNAs regulate genes in the adrenal gland a suitable in vitro model is required. The most widely used cell model of adrenal steroidogenesis is the NCI-H295 cell line. It was established from an adrenocortical carcinoma mass that was surgically removed from a 48-year-old black female. Initial characterisation of this cell line concluded that it continued to synthesise and secrete most adrenal steroids (Gazdar et al., 1990) but cells had to be maintained in suspension. Subsequent manipulations of growth conditions resulted in an adherent cell line, H295R, which retains steroid producing capabilities, largely due to the use of the bovine-derived serum substitute, Ultroser G. A modified strain of this cell line was also derived using Nu-Serum in place of Ultroser G, which overcomes the limited availability which can be an issue with Ultroser G. This cell line is available from the American Type Culture Collection (ATCC) (Rainey et al., 1994). For studies in this thesis, H295R cells maintained in Ultroser G were used. Unlike the spatial separation of corticosteroidogenesis that occurs in discrete zones of the adrenal gland, the H295R cells co-express enzymes for the biosynthesis of mineralocorticoids, glucocorticoids and androgens and therefore secrete these various steroids. The cells are responsive to AnglI and K^+ and express the AT1R (Bird *et al.*, 1993). They do not express the ACTH receptor and are therefore unresponsive to ACTH stimulation (Mountjoy et al., 1994). The cells are commonly used for investigating various aspects of steroid production including transcription factor, inhibitor and mutational studies (Lehoux et al., 2001; Ye et al., 2009).

The expression of miR-21 has been assessed in H295R cells, where it was shown to increase in response to AngII stimulation and to be associated with raised aldosterone production (Romero *et al.*, 2008). However, the molecular mechanism of miR-21 action was not investigated and this remains the only report of miRNA modulation of steroid production to date. In order to test whether miRNAs exert control over the corticosteroidogenic pathway, it would be highly desirable to measure the levels of relative mRNA and steroid in order to gain the fullest possible picture of their effects.

3.2 Aims

The aim of this study was to determine whether miRNA-mediated regulation is an important factor in the regulation of steroidogenic enzyme production and of corticosteroid biosynthesis in adrenal cells. Additionally, this study examined whether miRNA regulation of the *CYP11B1* and *CYP11B2* genes occurs at their respective 3'UTRs and whether this is modulated by AngII stimulation.

3.3 Methods

3.3.1 Transfection of H295R Cells with siRNA molecules

H295R cells were transfected in 6-well plates as described in section 2.9. Six replicate wells per condition were set up on three independent occasions. The siRNA molecules used and the preparation methods are detailed in section 2.9.1. The medium was replaced on the cells after 24 hours and, after 48 hours, the medium was removed for steroid analysis by LC:MS/MS (Section 2.11). Cell lysates were prepared for protein quantification from three replicates (Section 2.12). Steroid concentrations were normalised per mg of protein and then expressed as fold-change compared to control siRNA transfected cells. The final three replicates were used for qRT-PCR analysis as described in section 2.4 and 2.5. Data were analysed using the $\Delta\Delta$ Ct method (Section 2.5.4) with β-actin as a housekeeping gene, and results expressed relative to the control siRNA transfected cells.

3.3.2 3'UTR Reporter Construct Plasmids

The pEZX-control, pEZX-B1 and pEZX-B2 3'UTR reporter construct plasmids (Figure 2-1) were prepared as described in section 2.8.3. For verification, each construct was digested with several restriction endonuclease enzymes (Section 2.8.4) and resolved by agarose gel electrophoresis (Section 2.6.2). Also, the 3'UTR insert and a small region covering the multiple cloning site of each plasmid was sequenced, as described in section 2.7.

3.3.3 Investigation of 3'UTR Activity In Vitro

H295R cells were transfected with the pEZX reporter constructs in 24-well plates as described in section 2.9.3. Cell medium was replaced at 24 hours with normal growth medium. Further, at 1, 6 and 24 hours prior to cell lysate preparation, medium was removed and replaced with normal growth medium (control, nonstimulated cells) or normal growth medium supplemented with AngII (10 nM), (Section 2.1.5). 48 Hours post-transfection, cell lysates were prepared and their luciferase activities measured DRLA (Section 2.10). As controls non-transfected cells were used as well as cells transfected with construct containing only virally-driven firefly (pGL4.13) or renilla reporter gene (pGL4.73, Promega Madison, WI, USA). These were prepared as described in section 2.8.3. and acted as transfection controls and as controls for background luminescence, and also determined whether firefly luminescence was fully quenched by the Stop and Glo reagent of the DLRA. Normalisation for transfection efficiency was achieved by dividing firefly luciferase luminescence by renilla luciferase luminescence (both expressed as RLU). The firefly to renilla ratio from AngII-stimulated cells was expressed as a percentage of that from non-stimulated time-matched control transfected cells.

3.3.4 Statistical Analysis

Statistical difference in mRNA or steroid levels in siRNA experiments was assessed by a one-way analysis of variance (ANOVA) test and by Bonferroni's post-hoc test, comparing the Dicer1 siRNA results for each with that from control siRNA-transfected cells. Reporter construct results were analysed by one-way ANOVA and Bonferroni's post-hoc tests, comparing all time-points.

For all analyses, confidence intervals of 95% were used and P < 0.05 was required for statistical significance. Data are expressed as the mean \pm standard error of the mean (SEM).

3.4 Results

3.4.1 Confirmation of siRNA Transfection

The suitability of the transfection conditions for siRNA delivery into H295R cells was assessed using a positive control siRNA targeted against *GAPDH* mRNA. Following the transfection, RNA was isolated and the levels of *GAPDH* mRNA measured by qRT-PCR. GAPDH-specific siRNA significantly reduced *GAPDH* mRNA levels to 0.46 \pm 0.03 to that of the control transfected cell (p < 0.001) (Figure 3-1).



Figure 3-1. Assessment of the GAPDH-targeted siRNA on GAPDH mRNA in H295R Cells.

H295R cells were transfected with scrambled negative control siRNA or a GAPDH-specific siRNA (final concentration 30 nM). *GAPDH* mRNA was analysed 48 hours post-transfection by qRT-PCR. Cycle threshold values were normalised to *β*-*actin* mRNA and expressed relative to control cell values. The mean of three independent biological experiments performed in triplicate, error bars represent SEM. *** p < 0.001 compared to control.

3.4.2 siRNA-mediated Dicer-knockdown in H295R cells

Following transfection of the H295R cells with Dicer1-specific siRNAs, targeting was validated by measuring *Dicer1* mRNA levels using qRT-PCR (Section 3.4.1). Dicer1 A siRNA significantly decreased *Dicer1* mRNA to 0.56 \pm 0.07 fold (p < 0.001) and Dicer1 B siRNA decreased *Dicer1* mRNA to 0.50 \pm 0.06 fold (p < 0.001), compared to control siRNA transfected cells (1.01 \pm 0.05; Figure 3-2).

This depletion of *Dicer1* mRNA had no significant effect on *StAR*, *3B-HSD11* or *HSD11B2* mRNA abundance (Figure 3-3 A,C and F). CYP11A1 mRNA abundance increased from control cell levels (1.03 \pm 0.10) following transfection of Dicer1 A siRNA (1.41 \pm 0.09; p < 0.05) and of Dicer1 B siRNA (1.42 \pm 0.10; p < 0.05; Figure 3-3B). Dicer1 A siRNA significantly increased *CYP21A1* mRNA abundance from 1.01 \pm 0.05 in control cells to 2.40 \pm 0.34 fold (p < 0.01). However, the change in *CYP21A1* mRNA with Dicer1 B siRNA (1.72 \pm 0.30) did not reach statistical significance relative to control (Figure 3-3D). The results for *CYP17A1* mRNA showed a similar pattern, with Dicer1 A siRNA significantly increasing abundance relative to control (1.73 \pm 0.22 vs 1.02 \pm 0.07; p < 0.01) but Dicer1 B siRNA (1.37 \pm 0.07) not achieving statistical significance (Figure 3-3E).

Reducing *Dicer1* mRNA led to a significant increase in the relative level of *CYP11B1* mRNA present in cells transfected with Dicer1 A siRNA from 1.00 ± 0.04 to 1.75 ± 0.24 (p < 0.05) (Figure 3-4A). There was no significant change in cells transfected with Dicer1 B siRNA (1.39 ± 0.23). In addition, there was a significant increase in *CYP11B2* mRNA abundance with Dicer1 A siRNA from 1.00 ± 0.02 to 2.12 ± 0.41 (p < 0.01). However, Dicer1 B siRNA (1.40 ± 0.09) did not result in a statistically-significant increase (Figure 3-4B).

24-Hour steroid secretion was measured in the media of transfected cells by LC:MS/MS. Figure 3-5 shows the results for steroids synthesised on the aldosterone arm of the corticosteroid pathway. Firstly, deoxycorticosterone (DOC) levels (panel A) were significantly increased relative to control cell (1.00 \pm 0.02) for both Dicer-targeting siRNAs (Dicer1 A, 1.23 \pm 0.01, p < 0.05 relative to control; Dicer1 B siRNA, 1.53 \pm 0.09, p < 0.001 relative to control). There was a significant change in corticosterone production between cells transfected with control siRNA or with Dicer1 siRNA A (1.00 \pm 0.03 and 1.10 \pm 0.06, respectively;

Figure 3-5B). In contrast, corticosterone secretion was significantly increased in cells transfected with Dicer1 B siRNA 1.32 ± 0.13 fold (p < 0.05 relative to control). A similar pattern was observed with 18-OH-corticosterone production, with no change to control levels following Dicer1 A siRNA transfection (1.00 ± 0.01 and 1.17 ± 0.06, respectively) and a slight but significant increase following Dicer1 B siRNA transfection to 1.28 ± 0.10 (p < 0.05 relative to control) (Figure 3-5C). Control aldosterone levels (1.00 ± 0.06) were similar following Dicer1 A siRNA transfection (1.06 ± 0.06) but significantly increased in Dicer1 B siRNA transfection (1.06 ± 0.06) but significantly increased in Dicer1 B siRNA transfection (1.06 ± 0.06) but significantly increased in Dicer1 B siRNA transfection (1.47 ± 0.11 fold; p < 0.01; Figure 3-5D).

Cortisol-related steroids are shown in Figure 3-6. There was no significant change in 11-deoxycortisol secretion (control cells 1.00 ± 0.01 ; Dicer1 A siRNA cells 1.14 ± 0.07 ; Dicer1 B siRNA cells 1.26 ± 0.12 ; Figure 3-6A). There was no significant difference in cortisol secretion following in Dicer1 A siRNA transfection (control cells 1.00 ± 0.01 ; Dicer1 A siRNA cells 1.07 ± 0.08) but cells transfected with Dicer1 B siRNA had significantly increased cortisol (1.33 ± 0.11 fold, p < 0.05; Figure 3-6B). Secretion of the cortisol metabolite cortisone did not change significantly in cells transfected with either Dicer siRNA molecule, relative to control (Figure 3-6C).



Figure 3-2. Assessment of the Effect of Dicer1 siRNAs on Dicer1 mRNA in H295R Cells.

H295R cells were transfected with one of two siRNAs targeted against Dicer1, designated Dicer1 A and Dicer1 B or with a scrambled negative control siRNA (final concentration 30 nM). *Dicer1* mRNA were analysed 48 hours post-transfection by qRT-PCR. Cycle threshold values were normalised to β -actin mRNA and expressed relative to the control cells. Results represent the mean of three independent biological experiments performed in triplicate; error bars represent SEM. *** p < 0.001 compared to control.



Figure 3-3. Assessment of the Effect of Dicer1 siRNAs on Steroidogenic mRNAs in H295R Cells.

H295R cells were transfected with one of two siRNAs targeted against Dicer1, designated Dicer1 A and Dicer1 B, or with a scrambled negative control siRNA (final concentration 30 nM). *StAR* mRNA (A), *CYP11A1* mRNA (B), *36HSDII* mRNA (C), *CYP21A1* mRNA (D), *CYP17A1* mRNA (E) and *HSD11B2* mRNA (F) were analysed 48 hours post-transfection by qRT-PCR. Cycle threshold values were normalised to *β*-*actin* mRNA and expressed relative to the control cells. Results represent the mean of three independent biological experiments performed in triplicate; error bars represent SEM. * p < 0.05, ** p < 0.01 compared to control.



Figure 3-4. Assessment of the Effect of Dicer1 siRNAs on *CYP11B1* and *CYP11B2* mRNA in H295R Cells.

H295R cells were transfected with one of two siRNAs targeted against Dicer1, designated Dicer1 A and Dicer1 B, or with a scrambled negative control siRNA (final concentration 30 nM). *CYP11B1* mRNA (A) and *CYP11B2* mRNA (B) were analysed 48 hours post-transfection by qRT-PCR. Cycle threshold values were normalised to *β*-actin mRNA and expressed relative to the control cells. Results represent the mean of three independent biological experiments performed in triplicate; error bars represent SEM. * p < 0.05, ** p < 0.01 compared to control.



Figure 3-5. Assessment of the Effect of Dicer1 siRNAs on Steroid Production in H295R Cells (aldosterone pathway).

H295R cells were transfected with one of two siRNAs targeted against Dicer1, designated Dicer1 A and Dicer1 B, or with a scrambled negative control siRNA (final concentration 30 nM). 24-Hour steroid secretion was measured in cell growth media by liquid chromatography with tandem mass spectrometry. Steroid production was normalised to total cell protein and expressed as fold-change relative to control cells. Production of deoxycorticosterone (A), corticosterone (B), 18-OH-corticosterone (C) and aldosterone (D) are shown. The results represent the mean of two independent biological experiments performed in replicate groups of 6; error bars represent SEM * p < 0.05, ** p < 0.01, *** p < 0.001 compared to control.





11-Deoxycortisol

А

1.75

1.50

1.25 1.00 0.75 0.50 0.25 0.00

(Fold change, pmol/ml/mg protein, relative to negative control)

Steroid Production Rate



Figure 3-6. Assessment of the Effect of Dicer1 siRNAs on Steroid Production in H295R Cells (cortisol pathway).

H295R cells were transfected with one of two siRNAs targeted against Dicer1, designated Dicer1 A and Dicer1 B, or with a scrambled negative control siRNA (final concentration 30 nM). 24-Hour steroid secretion was measured in cell growth media by liquid chromatography with tandem mass spectrometry. Steroid production was normalised to total cell protein and expressed as fold-change relative to control cells. Production of 11-deoxycortisol (A), cortisol (B) and cortisone (C) are shown. The results represent the mean of two independent biological experiments performed in replicate groups of 6; error bars represent SEM. * p < 0.05.
3.4.3 Verification of the pEZX 3'UTR Construct

pEZX-Control plasmid (Figure 2-1) comprises the pEZX backbone contains the firefly and renilla genes and is 7416 base pairs (bp) in size; this is confirmed by restriction endonuclease digestion shown in Figure 3-7. *Hind*III, *SacI* and *SpeI* all linearise the plasmid, yielding a band of that size. *Stu* I enzyme cleaves pEZX-control twice creating two DNA products; 4908 bp and 2508 bp (Figure 3-7).

The pEZX-B1 plasmid comprises the 3'UTR of the *CYP11B1* gene which is approximately 2000 bp in length. Therefore, the pEZX-B1 is 9426 bp in length (Figure 2-1), and this was confirmed by digestion using the *Hin*dIII enzyme (Figure 3-8A). Both *SacI* and *Bam*HI cleave the pEZX-B1 plasmid twice, once in the 3'UTR insert and once in the pEZX backbone. Figure 3-8A shows that *SacI* creates products approximately 7146 bp and 2280 bp in size and *Bam*HI creates products 7587 bp and 1839 bp.

The pEZX-B2 plasmid comprises the 3'UTR of the *CYP11B2* gene which is approximately 1500 bp in length. Therefore, the pEZX-B2 is 8835 bp in length (Figure 2-1), and this was confirmed by digestion using the *Hin*dIII enzyme (Figure 3-8B). *Sac*I cleaves the pEZX-B2 plasmid twice, once in the 3'UTR and once in the pEZX backbone and Figure 3-8B shows that *Sac*I creates two products approximately 7158 bp and 1677 bp in size. *Bam*HI cleaves pEZX-B2 three times, the two larger products 7600 bp and 1839 bp can be visualised in Figure 3-8B, the other DNA fragment at 276 bp was too small to be clearly visible on a 0.7% agarose gel. Finally, *Spe*I digests at either side of the 3'UTR construct creating two products, 7400 bp and 1435 bp in size (Figure 3-8B).

Additionally, the plasmids were directly sequenced (Section 2.7) and examples of the electropherograms generated for each plasmid using SeqScape software are shown in Figure 3-9. Analysis of sequencing spanning the complete insert and adjoining sections of the pEZX backbone verified the identities of the plasmids.



Figure 3-7. Restriction Endonuclease Digestion of pEZX-Control Plasmid.

pEZX-Control plasmid (1 μ g) was digested by various restriction endonuclease and resolved on 0.7% agarose gel. Promega 1 kb ladder was used for size determination; sizes indicated are in kilobase pairs.





pEZX-B1 plasmid (A) or pEZX-B2 plasmid (B) (1 μ g each) was digested by various restriction endonuclease and resolved on 0.7% agarose gel. Promega 1 kb ladder was used for size determination; sizes indicated are in kilobase pairs.



Figure 3-9. Sequence Analysis of pEZX Plasmids.

Examples of electropherogram read from sequence analysis of the pEZX-Control, pEZX-B1 and pEZX-B2 plasmids.

3.4.4 Response of 3'UTR Reporter Construct Activity to Angll Stimulation

pEZX-plasmids (Figure 2-1) were transfected into H295R cells. Firefly and renilla luciferase results (relative light units, RLU) demonstrated that all plasmids were functional and control transfections indicated that the DLRA (Dual Luciferase Reporter Assay) was an appropriate assay for measuring luciferase.

The relative levels of the firefly:renilla ratios of the pEZX-C, pEZX-B1 and pEZX-B2 reporter constructs was assessed after transfecting equal amounts (500ng) of the plasmids into H295R cells (Figure 3-10). The addition of either the 3'UTR of the *CYP11B1* or *CYP11B2* gene to the pEZX backbone caused a significant decrease in the firefly:renilla ratio (0.00048 \pm 0.00012 and 0.00050 \pm 0.00016) compared to the empty pEZX-C plasmid (0.00220 \pm 0.00053). This demonstrates the addition of the 3'UTR of the *CYP11B1* or *CYP11B2* gene is sufficient to cause negative regulation of the firefly mRNA in the H295R cells.

The response the pEZX plasmids was tested in H295R cells stimulated with AngII for 1, 6 or 24 hours (Figure 3-11). There was no significant change in the luciferase ratio for pEZX-control plasmid in the stimulated cells at any time-point (Figure 3-11A). This demonstrates that the pEZX backbone is not responsive to the corticosteroid trophin, AngII. There were no significant differences in the luciferase ratio of the pEZX-B1 plasmid after stimulation with AngII for 1 or 6 hours (105.93 \pm 5.62% and 101.39 \pm 2.88%, respectively; Figure 3-11B). When stimulated with AngII for 24 hr, the luciferase ratio increased to 192.70 \pm 28.04% of non-stimulated cells (p < 0.05; Figure 3-11B). The luciferase ratio of pEZX-B2 plasmid did not significantly differ in AngII-stimulated H295R cells at 1 or 6 hours (101.82 \pm 2.41% and 100.67 \pm 3.85%, respectively; Figure 3-11C). 24-Hour AngII stimulation increased the luciferase ratio to 220.58 \pm 43.52% (p < 0.05; Figure 3-11C).



Figure 3-10. Assessment of Relative Levels of Reporter Construct Activities.

H295R cells were transfected with the pEZX control, pEZX-B1 or pEZX-B2 then lysed after 48 hours. Firefly and renilla luciferase activities were measured using the Dual Luciferase Reporter Assay Kit and normalised for transfection efficiency using the ratio of firefly to renilla luminescence. Results represent the mean of at least three independent biological experiments performed in quadruplicate; error bars represent SEM. Statistical analysis was a 1-way ANOVA with Bonferroni's Multiple Comparison correction. * p < 0.05 relative to pEZX-C.





H295R cells were transfected with the pEZX control (A), pEZX-B1 (B) or pEZX-B2(C) then lysed after 48 hours. Prior to lysis, medium was replaced with AngII (10 nM) supplemented medium for 1, 6 or 24 hrs. Firefly and renilla luciferase activities were measured using the Dual Luciferase Reporter Assay Kit and normalised for transfection efficiency using the ratio of firefly to renilla luminescence. The results for the stimulated cells were expressed as a percentage of time matched non-stimulated control cells. Results represent the mean of at least three independent biological experiments performed in quadruplicate; error bars represent SEM. Statistical analysis was a 1-way ANOVA with Bonferroni's Multiple Comparison correction. ns; non-significant, * p < 0.05.

3.5 Discussion

The experiments presented in this chapter investigated if miRNAs mediate regulation of corticosteroid-related mRNA levels and steroid production in the adrenocortical cell line, H295R. To achieve this, miRNA levels were reduced using Dicer1-targeted siRNA and the activation of the *CYP11B1* and *CYP11B2* 3'UTR investigated in response to the classical corticotrophic peptide, AngII.

The first stage involved transient siRNA transfection into H295R cells; the transfection protocol and the positive control siRNA molecule were shown to be successful in altering *GAPDH* mRNA levels. Therefore, a method was successfully developed for small RNA transfection into the H295R cells, and was adopted for the subsequent siRNA transfections. The siRNA molecules were purchased as prevalidated siRNAs from Applied Biosciences and this study confirmed *Dicer1* mRNA silencing in H295R cells. The magnitude of *Dicer1* mRNA reduction was relatively small: 0.5 fold change relative to control cells. This is equivalent to reducing them to approximately 75% of their original levels. siRNA-silencing only affects nascent mRNAs therefore, Dicer1 enzyme synthesised prior to transfection will still be present and functional within cells, assessing the level of Dicer protein by western blot analysis would help quantify the impact of this. The reduction in Dicer1 mRNA levels is slightly less then that reported in the HeLa cell line by the manufacturer; this is likely to be caused by differences in the cell types (H295R cells are notoriously difficult to transfect) and altered concentrations of siRNA.

Despite the modest decrease in *Dicer1* mRNA, a reduction in miRNA levels would still be expected; a change of similar magnitude has already been shown to reduce levels of several miRNAs (Lee *et al.*, 2006). Yet, this protocol does not eliminate all miRNAs as some Dicer activity remains and the turnover rate of most miRNAs is still unknown (Section 1.5.2.4). To determine miRNA levels in Dicer1 knockdown cells, it would be necessary to perform miRNA microarray, qRT-PCR or northern blot analyses. Moreover, recent reports suggest that miRNAs may be formed by a Dicer1-independent pathway, and this has been demonstrated for miR-451 *in vitro* (Yang *et al.*, 2010) and in mice models (Cheloufi *et al.*, 2010;Cifuentes *et al.*, 2010). In this relatively newly-discovered pathway pre-miR-451 is cleaved by Argonaut2, the importance of this alternative

pathway remains to be evaluated. In summary, transient transfection of Dicer1targeted siRNA in H295R cells reduces maturation of novel miRNAs and thus serves as a good model to test the potential impact of miRNA-regulation on corticosteroidogenesis.

The mRNA levels of several key enzymes in the adrenal steroid production pathway were assessed cells transfected with in Dicer1-targeted siRNA. Reduction of *Dicer1* mRNA did not change the relative levels of *StAR*, *3BHSDII* or HSD11B2 mRNA, but did increase the levels of the CYP450 mRNA analysed (Sections 1.2.3 and 1.2.4). This indicates that reducing miRNAs leads to the derepression of CYP450 expression in H295R cells. The change in mRNA level was approximately 1.5 - 2 fold depending on the gene and this is typical of the finetuning effect usually exerted by miRNAs (Sevignani et al., 2006). Significant increases in the mRNA level of CYP450 mRNAs, other than CYP11A1, were only achieved with the Dicer1 A siRNA. Although an increase in these mRNA was always observed with the Dicer1 B siRNA, the magnitude of change was smaller or the error slightly larger relative to the Dicer1 A siRNA, meaning a statistical significant change was not observed. One possible explanation for this difference is that each siRNA targets different regions of Dicer1 mRNA (Appendix; Figure 8-7-1), and this may alter the mechanism and/or efficiency of silencing. This study validated two separate Dicer-targeting siRNAs for this purpose but could have also co-transfected two Dicer-targeting siRNAs.

The majority of steroids produced from H295R cells are androgens although they corticosteroid-synthesising abilities. Dicer1-targeted also retain siRNA transfected cells had increased production of DOC, B, 18-OH-B and aldosterone. Together, these steroid results and those measuring mRNA indicate that expression of the CYP21A1 and CYP11B2 enzymes may be regulated by miRNAs. Moreover, cortisol production was increased, suggesting miRNA regulation of CYP11B1 transcript abundance. The cortisol precursor, S, was not significantly increased in the knockdown cells, which is inconsistent with the role of CYP21A1 mRNA results and increased DOC production. It was not possible to assess the enzyme efficiency by comparing the relative levels of precursor and substrate (for example, the S:F ratio) as the sample size and the relative amounts of steroid produced were too small. The level of HSD11B2 mRNA was unaffected by Dicer1-targeted siRNA, which might be expected given that the conversion of cortisol to cortisone is of greater relevance to peripheral tissues, rather than in the adrenal gland.

The focus of this thesis is to investigate the role of miRNAs in adrenal corticosteroidogenesis. Therefore, the majority of work described in subsequent chapters will be concerned with the *CYP11B1* and *CYP11B2* genes. The above experiments undoubtedly support a role for miRNA regulation in these genes' expression and the generation of their relevant steroid products. To investigate whether this is mediated by direct action on the 3'UTR of CYP11B1 and/or CYP11B2 mRNAs, reporter constructs containing the 3'UTR of these genes were used. Initial experiments demonstrated that there was significant repression of the luciferase mRNA compared to a no-3'UTR control construct (as indicated by the relative firely:renilla activity ratios), this supports a role for miRNA-mediated regulation acting at the 3'UTR of both the *CYP11B1* and *CYP11B2* genes.

The construct were then further tested in H295R cells to determine whether miRNA repression could be overcome by stimulation of H295R cells with AngII, which has repeatedly been shown to increase CYP11B2 and, to a lesser extent, CYP11B1 expression (Denner et al., 1996). The presence of CYP11B1 or CYP11B2 3'UTR was sufficient to increase reporter construct activity levels in response to AnglI stimulation. This increase was only observed in cells stimulated for 24 hours and no response was observed in the control plasmid, which lacked a 3'UTR. Thus, the observed increase in activity can be attributed to the CYP11B1 or CYP11B2 3'UTRs. This is consistent with AnglI stimulating CYP11B2 or CYP11B1 protein synthesis and de-repression of the 3'UTR by miRNAs. Although the influence of AnglI on the expression levels of miRNAs cannot be determined from these experiments, a previous study using H295R cells found that miR-21 was increased following AnglI stimulation for 6, 12 and 24 hours (Romero et al., 2008) suggesting that the regulation of certain some adrenal miRNAs is sensitive to AngII. However, the expression profile of adrenal miRNAs and their response to AnglI has not been fully determined and it is, therefore, difficult to ascertain what regulatory mechanism dominates. Therefore, further experiments are required in order to determine whether the observed changes in reporter gene activity are attributable to modulation of miRNA levels or to indirect effects such as alterations in Dicer2 levels.

In conclusion, these studies have provided evidence of a role for miRNAmediated regulation of CYP450 expression, which has an impact on adrenal corticosteroidogenesis and therefore, appears to be of biological significance. The identity or contribution of individual miRNAs cannot be assessed on the basis of these experiments, but they clearly support further investigation.

4 Normal Adrenal Gland microRNA Profiling and Target-Site Prediction

4.1 Introduction

To investigate miRNA-mediated regulation of a particular mRNA, the investigator is required to have an understanding of the expression of miRNAs in the tissue or cell type that expresses the mRNA. It is then necessary to understand whether miRNA targeting of the mRNA is feasible by identifying miRNA-binding sites in its 3'UTR. The miRNA expression profile of a tissue or a cell can only be properly determined by experimental analysis, whether by qRT-PCR, miRNA microarray analysis, northern blot or deep-sequencing. However, computational strategies are available that either map experimentally validated genomic locations of miRNAs or facilitate prediction of the genetic location of novel miRNA species. Using this strategy offers the investigator the opportunity to perform a relatively quick and cost-effective analysis prior to investing in expensive 'wet-lab' techniques that may take time and require precious tissue samples.

Computational methods designed to identify genomic regions that encode primiR transcripts have been developed based on common features of miRNAs, for example: structural properties such as hairpin length, mismatched base-pair bulge size and thermodynamic stability or pri-miR sequence properties (Bentwich *et al.*, 2005;Bentwich, 2005). Frequently the algorithms used rely on evolutionary sequence conservation for identification of miRNAs; this is valid as many miRNAs are well conserved across species (Berezikov *et al.*, 2005). However, this method is limited by the accuracy of the algorithm and also by miRNAs which are not conserved, therefore it fails to detect species-specific miRNAs (Bentwich *et al.*, 2005). Validation of miRNA prediction is required and this can be achieved by various methods of direct sequencing. More recently, deep-sequencing has proved successful for identifying novel miRNAs. This involves sequencing large fractions of RNA of unknown function, with the aim of identifying novel regions which resemble a pri-miRNA or pre-miRNA (Friedlander *et al.*, 2008).

In circumstances where a specific gene target is unknown (*e.g.* in studies of defined disease phenotypes or of cell differentiation), a gene expression microarray is often performed concurrently with miRNA expression analysis. The

aim is to identify negatively correlated miRNA:mRNA combinations. This strategy is complicated as a single miRNA can regulate many genes.

In addition, in silico methods have been developed to predict mRNA targets for miRNAs. However, as will be discussed below, this is highly complex and has proven to be problematic. There are numerous miRNA databases and targetprediction algorithms but in general, the validation rate is low, possibly because miRNAs are promiscuous and believed to bind to hundreds of different targets (Lewis et al., 2003). Further, miRNA regulation does not require perfect basepairing complementarity of miRNA to mRNA and, therefore, target identification by simple BLAST (Basic Local Alignment Search Tool) base-pairing analysis cannot be used (Rehmsmeier et al., 2004). Moreover, this leads to the formation of complex RNA secondary structure. Experimental testing and validation of miRNA target sites has provided a learning dataset against which algorithms can be compared; this has improved the accuracy of the algorithms and of the target site predictions. It has also identified a region known as the 'seed site', which refers to nucleotides 2-7 of the miRNA from the 5' end (Figure 4-1). Watson-Crick base-pairing in this region is often evolutionarily conserved and leads to a higher rate of positive target predictions being generated (Lewis *et al.*, 2005). Perfect complementarity in this region is referred to as a 6mer site (Figure 4-1). Further analysis has identified that a base-pairing at nucleotide 8 further increases target recognition, as does an adenosine nucleotide at position 1 of the miRNA; instances of these are termed 7mer-m8 site and 7mer-A1 sites, respectively (Figure 4-1). miRNA:mRNA pairings which have both of these additional features are termed 8mer sites (Figure 4-1) (Lewis et al., 2003;Lewis et al., 2005). Some miRNAs have common sequences at the seed site, having evolved from a common ancestor, and are collectively known as a miRNA family (Griffiths-Jones et al., 2003). Seed site base-pairing forms one of the strongest criteria for several target prediction databases, increasing the reliability and reducing the rate of false positive predictions (Bartel, 2009). Moreover, basepairing in the seed region is sufficient to elicit miRNA-regulation as even a single nucleotide polymorphism in the seed region can disrupt miRNA function (Doench and Sharp, 2004).

Seed Sites

Alternative Base-Pairing

21 20 19 18 17 16 15 14 13 12 11 10 9 8 7 6 5 4 3 2 1 NNNNNNNNNNNNNNNNNNNN 6mer NNNNNNNNNNNNNNNNNNN Offset 6mer NNNNNNNNNNNNNNNA-5' 7mer-A1 NNNNNNNNNNNNNNNN-5' 7mer-m8 NNNNNNNNNNNN-5'

8mer

21 20 19 18 17 16 15 14 13 12 11 10 9 8 7 6 5 4 3 2 1 NNNNNNNNNNNNNNNNNNNNN 3' Supplementary Pairing NNNNNNNNNNNNNNNNNNN-5' 3' Compensatory Pairing NNNNNNNNNNNNNNNNNNN-5'

Centred Pairing

Figure 4-1. miRNA Target Recognition Base Pairing Requirements.

miRNA seed site base-pairing is shown in the left panel. Nucleotides (N) are numbered from the 5' end of the miRNA. Yellow nucleotides (N) represent Watson-Crick matches to the 3'UTR of mRNA. Red bases are the additional matches that increase target recognition and miRNA silencing and include, adenosine (A) at position 1. The right hand panel shows nucleotide matches in the 3' end that can be present with or without a canonical seed site. The bottom miRNA illustrates centered pairing.

In addition to the seed region, pairing of the mRNA target to other miRNA bases can affect its target recognition and silencing. Complementary binding at the 3' end of the miRNA (Figure 4-1) can enhance miRNA function for targets which also have a strong seed site complementarity; this is known as supplementary pairing, and refers to continuous base-pairing at nucleotides 13-16 (Grimson *et al.*, 2007). Further, enhanced base-pairing, typically from nucleotides 13-19, can make up for mismatched bases within the seed site and is known as compensatory pairing (Figure 4-1) (Grimson *et al.*, 2007;Bartel, 2009). Recently, miRNA-mediated regulation lacking both a traditional seed site and 3' complementarity has been identified. This involves alternative base-pairing of 11 or 12 continuous bases in the centre region of the miRNA (nucleotides 4-15) to a target mRNA (Figure 4-1). At high Mg^{2+} concentrations this is sufficient to cause miRNA-mediated silencing (Shin *et al.*, 2010).

Another parameter used by prediction algorithms is the degree of evolutionary sequence conservation of the miRNA species and of the target sites. A higher degree of sequence conservation is associated with a decreased number of false-positive miRNA target predictions (Lewis *et al.*, 2005;Bartel, 2009). Moreover, the conservation of most miRNA target sites identified to date is higher than would be expected by chance. Thus, they are believed to be under selective pressure and indicates a strong likelihood of biological function. However, the stringency of conservation used by bioinformatic analyses may miss miRNA-targets that are species-specific, and it has been estimated that 30% of validated miRNA:mRNA interactions are not conserved (Sethupathy *et al.*, 2006).

Finally, most target prediction algorithms do not account for RNA secondary structure of mRNA or for miRNAs which will affect the ability for base-pairing to occur efficiently; this may contribute to a poor rate of target validation. Prediction of secondary structure and the minimum free-energy may be useful for identifying interactions that are physiologically likely to occur (Robins *et al.*, 2005;Kertesz *et al.*, 2007). However, modelling RNA secondary structure is not 100% accurate and removing this feature from one of the databases does not change the specificity of target prediction (Lewis *et al.*, 2005;Bartel, 2009).

Despite the problems outlined above, target site prediction databases still provide the investigator with useful information regarding miRNA-targeting

4.2 Aims

The aims of this study were to investigate the genomic location of pri-miRNA sequences; to utilise bioinformatic prediction algorithms to identify putative miRNA binding sites in the 3'UTR of the *CYP11B1* and *CYP11B2* genes; and to establish the expression profile of miRNAs in non-tumorous (normal) adrenal glands and then to cross-reference the target and profiling data to compile a list of candidate miRNAs that may be important in the regulation of the *CYP11B1* and *CYP11B2* genes.

4.3 Materials and Methods

4.3.1 Bioinformatic Analysis

4.3.1.1 Identification of the Genomic Location of miRNAs

The genomic co-ordinates, strand location and mRNA transcript length of the human genes encoding corticosteroidogenic enzymes were identified using the Ensembl Genome Browser (release 61 - Feb. 2011). These details were used to cross-reference with known miRNA precursor sequences that are, mapped and stored in the miRBase database (release 16 - Sept. 2010).

4.3.1.2 Investigation of the Evolutionary Sequence Conservation of the CYP11B1 and CYP11B2 3'UTR

Analysis of the evolutionary sequence conservation of the 3'UTR of the *CYP11B1* and *CYP11B2* mRNA was performed using the UCSC Genome Browser Gateway (Feb. 2009 assembly). For comparison, 3 non-human primates (chimp, gorilla and rhesus macaque) and 5 other mammals (mouse, rat, rabbit, cow and dog) were used, and a graphical representation was generated.

4.3.1.3 Structural Prediction of the 3'UTR of CYP11B1 and CYP11B2

The secondary structure of the 3'UTR of *CYP11B1* and *CYP11B2* genes was predicted using the RNAFold program from the Vienna RNA Package (version 1.8.5). This was achieved by entering gene information from the UCSC Genome Browser Gateway system and exporting pictorial results and free-energy predictions from the software program.

4.3.1.4 Bioinformatic miRNA Target Site Prediction

A combinational approach using five databases was applied to identify putative miRNA binding sites in mRNAs involved with corticosteroidogenesis. The databases are listed in Table 4-1. Where required, the full-length human transcript was selected and conservation was set to low. All other parameters were left at default settings. Database predictions for each gene from the

database were recorded, pooled and duplicate results removed to create a list of putative miRNA binding sites for each gene.

4.3.2 miRNA Expression Profiling of Normal Adrenal Glands

4.3.2.1 miRNA Microarray Analysis of miRNA Expression

Total RNA was isolated from four frozen non-tumorous adrenal glands (section 2.2.1) and quantified as described in sections 2.2.4 and 2.2.5. Samples were shipped to the microarray provider (LC Sciences, Houston, Texas) and the array performed (Section 2.3). A background threshold cut-off of 500 arbitrary units was used as recommended by the array chip manufacturer; miRNAs expressed at levels greater than this threshold were deemed to be expressed in the adrenal gland.

4.3.2.2 qRT-PCR Validation of miRNA Expression

Twenty miRNAs with expression levels greater than the threshold were validated by qRT-PCR. This was performed by LC Sciences (Houston, Texas, U.S.A.) using the Applied Biosciences miRNA Taqman[®] Assay and the method described in Section 2.4.2 and 2.5.2. Two small nucleolar housekeeping RNAs, RNU48 (SNORD48) and U47 (SNORD47), were also run in the assay. RNU48 was the most stably expressed and, therefore, was used for normalisation purposes.

4.3.3 Statistical Analysis

All data are presented as mean \pm SEM. the correlation of the microarray expression data and the qRT-PCR validation was tested using Prism 4.0 Graph Pad software.

Database	Version	Algorithm	Reference
microrna.org	August 2010	miRanda - miRSVR (Support Vector Regression)	(John <i>et al.</i> , 2004;Betel <i>et al.</i> , 2010)
miR-viewer	June 2005	miRanda	http://cbio.mskcc.org/cgi-bin/mirnaviewer/mirnaviewer.pl
TargetScan	5.1, April 2009		(Lewis <i>et al.</i> , 2005;Grimson <i>et al.</i> , 2007;Friedman <i>et al.</i> , 2009)
MicroCosm Targets (formally miRBase)	5		(Griffiths-Jones et al., 2006;Griffiths-Jones et al., 2008)
TarBase	5		(Papadopoulos et al., 2009)

 Table 4-1. Bioinformatic Databases Used for Screening miRNA Target Sites.

4.4 Results

4.4.1 Genomic Prediction of pre-miRNA Coding Regions

No miRNA precursor sequences are currently mapped to the intronic regions of the nine selected corticosteroidogenic genes analysed (Table 4-2). The most proximal miRNA sequence to any of the genes was hsa-miR-146b, which is coded on the forward strand of human chromosome 10. It is approximately 300,000 bases downstream of the *CYP17A1* gene at co-ordinates 104,196,269 to 104,196,341. This miRNA is located in an intergenic region and is surrounded by several other protein-coding genes.

 Table 4-2. Genetic Location of Human Corticosteroid Genes and the Predicted miRNAs

 Sequences.

Gene	Chromosome Location	Start Co-ordinates	End Co-ordinates	Size (Bases)	Number of miRNA Precursor Sequences
CYP11B1	8	143,953,772	143,961,262	7,490	0
CYP11B2	8	143,991,975	143,999,259	7,284	0
CYP11A1	15	74,630,100	74,660,081	29,981	0
CYP17A1	10	104,590,288	104,597,290	7,002	0
CYP21A2	6	31,965,200	32,001,297	3,405	0
HSD3B2	1	119,957,673	119,965,657	7,984	0
STAR	8	38,000,226	38,008,783	8,557	0
HSD11B1	1	209,859,510	209,908,295	48,785	0
HSD11B2	16	67,465,041	67,471,456	6,415	0

Start and End co-ordinated refer to transcription start and stop sites.

4.4.2 Analysis of the 3'UTR of the CYP11B1 and CYP11B2 Genes

The predicted secondary structure of the *CYP11B1* and *CYP11B2* 3'UTRs is shown in Figure 4-2. Secondary structure can affect the accessibility of a miRNA to its target-site; if the energy required for binding is too high then binding would be unlikely. This analysis does not evaluate any miRNA binding sites but does give an estimation of the unbound secondary structure of the native 3'UTR, which appears not to have too many complex folds (Figure 4-2).

The evolutionary sequence conservation of the 3'UTR of the *CYP11B1* and *CYP11B2* 3'UTRs was assessed using the comparative genomic analysis available within the UCSC Genome browser. The regions (highlighted in blue and indicated by red dashed lines) were compared to other mammals and the results can be seen in Figure 4-3. The 3'UTR of both genes is well conserved across the higher mammals, including the chimp, gorilla and rhesus macaque, as indicated by the solid black bars, which indicate sequence similarity. The degree of similarity is substantially reduced for the mouse, rat, rabbit, cow and dog sequences (Figure 4-3).



Figure 4-2. Predicted RNA Secondary Structure of the 3'UTR of the CYP11B1 and CYP11B2 Genes.

Predicted structure of the 3'UTR of *CYP11B1* (A) and *CYP11B2* (B) from the UCSC Genome Browser Gateway and the Vienna RNA Package.



Figure 4-3. Mammalian Evolutionary Conservation of the 3'UTR of the CYP11B1 and CYP11B2 Genes.

The sequence conservation, generated by the UCSC Genome Browser for the *CYP11B1* (top) and *CYP11B2* (bottom) 3'UTR, shown by the blue bar and indicated by the dashed red lines. Conservation is represented by black bars for the species listed on the left-hand side and is defined as per default settings of the Genome Browser.

4.4.3 Identification of Putative miRNA Binding Sites

Table 4-3 lists the 3'UTR length for each of the selected corticosteroidogenic genes. Of the genes chosen, the *CYP11B1* and *CYP11B2* genes have the longest 3'UTR at 2022 base-pairs and 1428 base-pairs, respectively. This region comprises a large proportion of the total size of each mRNA transcript (Table 4-2) and the 3'UTR is relatively large in comparison to those of the other genes especially *CYP11A1* gene, whose 3'UTR comprises less than 1% of its total size (Table 4-3).

The five databases listed in Table 4-1 were searched, in turn, to identify putative miRNA binding sites in each of the selected corticosteroidogenic genes. The Tarbase database relies on validated targets and produced no miRNA target predictions for any of the genes investigated. Table 4-3 summarises the results generated from the other four databases. It lists the number of target sites predicted by each algorithm, with the number of individual miRNAs in parentheses. The collated results are shown in the final column; this excludes any duplicate predictions made by one or more algorithm.

The longest (*CYP11B1*) and shortest (*CYP17A1*) 3'UTRs match the highest and lowest number of miRNA prediction, respectively. However, the numbers of predictions do not correlate well with the size of the 3'UTR for other mRNAs. The *CYP11B2* 3'UTR has approximately the same number of predicted binding sites as *StAR* and *HSD11B1* despite being over twice as long as that of the other genes.

The extent of sequence similarity between the *CYP11B1* and *CYP11B2* genes prompted the comparison of predicted miRNA target sites. There was a large degree of similarity; in total 84 putative miRNA binding sites common to both genes are listed in Table 4-5. The identities of the 119 miRNAs predicted to bind to *CYP11B1* only are listed in Table 4-4 and the 88 miRNAs with putative binding sites to *CYP11B2* are listed in Table 4-6.

Gene	3'UTR Length (Base pairs)	Microrr	na.org	miR- vie	ewer	Target	Scan	MicroC Targe	osm ets	Unique miRNAs with Binding Sites
CYP11B1	2022	150	(142)	4	(4)	196	(156)	33	(30)	202
CY P11B 2	1428	89	(83)	10	(10)	138	(117)	6	(6)	172
CY P11 A1	213	52	(52)	1	(1)	21	(21)	43	(43)	103
CYP17A1	171	24	(24)	NA		11	(11)	23	(22)	42
CYP21A2	508	57	(56)	1 1	(10)	76	(75)	25	(22)	142
3βHSDII	414	67	(63)	NA		26	(26)	46	(44)	106
StAR	624	110	(101)	4	(4)	139	(128)	1 1	(11)	189
HSD11B1	343	164	(135)	1	(1)	50	(44)	70	(65)	168
HSD11B2	550	82	(78)	4	(4)	53	(50)	33	(31)	118

Table 4-3. Bioinformatic miRNA Target Site Predictions for Corticosteroidogenic Genes

The 3'UTR length of the genes involved in corticosteroidogenesis was identified using the UCSC Genome Browser Gateway and Ensemble Genome Browser. The number of predicted miRNA target sites from each database is listed and the number of individual miRNA predicted to bind is shown in parentheses. The final column shows the cumulative number of miRNAs target-sites, with duplicates removed. NA: Not applicable/no predictions.

Table 4-4. Identities of miRNAs with	h Putative Binding S	Sites in the CYP11B1	Gene, but not
CYP11B2.	_		

Putative miRNA binding sites in the CYP11B1 genes					
hsa-miR-105	hsa-miR-361-3p	hsa-miR-543	hsa-miR-661		
hsa-miR-149*	hsa-miR-371-5p	hsa-miR-545	hsa-miR-665		
hsa-miR-150	hsa-miR-373	hsa-miR-548d-3p	hsa-miR-744		
hsa-miR-151-5p	hsa-miR-376a	hsa-miR-548o	hsa-miR-768-5p		
hsa-miR-181a	hsa-miR-376b	hsa-miR-550	hsa-miR-876-5p		
hsa-miR-181b	hsa-miR-382	hsa-miR-551a	hsa-miR-924		
hsa-miR-181c	hsa-miR-423-5p	hsa-miR-554	hsa-miR-933		
hsa-miR-181d	hsa-miR-429	hsa-miR-561	hsa-miR-938		
hsa-miR-190b	hsa-miR-431	hsa-miR-568	hsa-miR-1197		
hsa-miR-193a-5p	hsa-miR-432	hsa-miR-583	hsa-miR-1207-5p		
hsa-miR-196a	hsa-miR-449a	hsa-miR-588	hsa-miR-1224-5p		
hsa-miR-196b	hsa-miR-449b	hsa-miR-590-3p	hsa-miR-1225-5p		
hsa-miR-200b	hsa-miR-452	hsa-miR-593	hsa-miR-1226		
hsa-miR-200c	hsa-miR-483-3p	hsa-miR-596	hsa-miR-1244		
hsa-miR-204	hsa-miR-490-3p	hsa-miR-599	hsa-miR-1248		
hsa-miR-205	hsa-miR-494	hsa-miR-608	hsa-miR-1251		
hsa-miR-210	hsa-miR-497*	hsa-miR-609	hsa-miR-1256		
hsa-miR-214	hsa-miR-503	hsa-miR-618	hsa-miR-1259		
hsa-miR-218	hsa-miR-515-3p	hsa-miR-620	hsa-miR-1269		
hsa-miR-219-1-3p	hsa-miR-516a-3p	hsa-miR-622	hsa-miR-1270		
hsa-miR-219-2-3p	hsa-miR-518d-5p	hsa-miR-626	hsa-miR-1273		
hsa-miR-221*	hsa-miR-518e*	hsa-miR-627	hsa-miR-1281		
hsa-miR-23a	hsa-miR-519a	hsa-miR-630	hsa-miR-1290		
hsa-miR-23b	hsa-miR-519b-5p	hsa-miR-636	hsa-miR-1294		
hsa-miR-297	hsa-miR-519c-3p	hsa-miR-637	hsa-miR-1300		
hsa-miR-298	hsa-miR-519c-5p	hsa-miR-643	hsa-miR-1303		
hsa-miR-299-3p	hsa-miR-519e	hsa-miR-652	hsa-miR-1304		
hsa-miR-302c	hsa-miR-526b	hsa-miR-655	hsa-miR-1321		
hsa-miR-340	hsa-miR-532-5p	hsa-miR-657	hsa-miR-1322		
hsa-miR-34c-5p	hsa-miR-539	hsa-miR-660			

For information regarding miRNA nomenclature please see Section 1.5.2.1

Table 4-	Identities of miRNAs with Putative	e Binding Sites in the CYP11B1	and CYP11B2
Genes.			

Putative miRNA bine	Putative miRNA binding sites in the CYP11B1 and the CYP11B2 genes					
hsa-miR-1	hsa-miR-378	hsa-miR-592	hsa-miR-1184			
hsa-miR-10a	hsa-miR-422a	hsa-miR-604	hsa-miR-1204			
hsa-miR-10b	hsa-miR-450b-3p	hsa-miR-613	hsa-miR-1205			
hsa-miR-138	hsa-miR-484	hsa-miR-615-5p	hsa-miR-1207-3p			
hsa-miR-140-3p	hsa-miR-485-5p	hsa-miR-624	hsa-miR-1224-3p			
hsa-miR-143	hsa-miR-486-3p	hsa-miR-625	hsa-miR-1236			
hsa-miR-146b-3p	hsa-miR-504	hsa-miR-628-5p	hsa-miR-1253			
hsa-miR-185	hsa-miR-509-3-5p	hsa-miR-632	hsa-miR-1254			
hsa-miR-188-3p	hsa-miR-509-5p	hsa-miR-638	hsa-miR-1260			
hsa-miR-198	hsa-miR-510	hsa-miR-647	hsa-miR-1265			
hsa-miR-206	hsa-miR-515-5p	hsa-miR-651	hsa-miR-1266			
hsa-miR-22	hsa-miR-516b	hsa-miR-671-5p	hsa-miR-1274a			
hsa-miR-220c	hsa-miR-520a-5p	hsa-miR-708	hsa-miR-1274b			
hsa-miR-24	hsa-miR-520f	hsa-miR-769-3p	hsa-miR-1275			
hsa-miR-28-5p	hsa-miR-548a-3p	hsa-miR-873	hsa-miR-1280			
hsa-miR-328	hsa-miR-548e	hsa-miR-874	hsa-miR-1285			
hsa-miR-339-5p	hsa-miR-548f	hsa-miR-875-3p	hsa-miR-1286			
hsa-miR-33b	hsa-miR-549	hsa-miR-877	hsa-miR-1287			
hsa-miR-34a	hsa-miR-558	hsa-miR-939	hsa-miR-1291			
hsa-miR-34c-3p	hsa-miR-571	hsa-miR-940	hsa-miR-1299			
hsa-miR-370	hsa-miR-577	hsa-miR-943	hsa-miR-1308			

Table 4-6. Identities of miRNAs with Putative Binding Sites in the *CYP11B2* Gene, but not *CYP11B1*.

Putative miRNA binding sites in the CYP11B2 genes						
hsa-miR-125a-3p	hsa-miR-362-5p	hsa-miR-606	hsa-miR-1225-3p			
hsa-miR-125a-5p	hsa-miR-367*	hsa-miR-612	hsa-miR-1228			
hsa-miR-125b	hsa-miR-378*	hsa-miR-614	hsa-miR-1229			
hsa-miR-134	hsa-miR-378b	hsa-miR-619	hsa-miR-1260b			
hsa-miR-184	hsa-miR-378c	hsa-miR-624*	hsa-miR-1914			
hsa-miR-188-5p	hsa-miR-383	hsa-miR-631	hsa-miR-1914*			
hsa-miR-18b*	hsa-miR-431*	hsa-miR-639	hsa-miR-1975			
hsa-miR-193	hsa-miR-432*	hsa-miR-650	hsa-miR-2113			
hsa-miR-193a-3p	hsa-miR-449	hsa-miR-657	hsa-miR-3065-3p			
hsa-miR-193b	hsa-miR-452*	hsa-miR-658	hsa-miR-3125			
hsa-miR-217	hsa-miR-495	hsa-miR-661	hsa-miR-3126-5p			
hsa-miR-224*	hsa-miR-500*	hsa-miR-665	hsa-miR-3127			
hsa-miR-30c-1*	hsa-miR-505*	hsa-miR-711	hsa-miR-3153			
hsa-miR-30c-2*	hsa-miR-512-3p	hsa-miR-720	hsa-miR-3162			
hsa-miR-31	hsa-miR-513a-5p	hsa-miR-744*	hsa-miR-3176			
hsa-miR-326	hsa-miR-514	hsa-miR-760	hsa-miR-3192			
hsa-miR-329	hsa-miR-523	hsa-miR-766	hsa-miR-4292			
hsa-miR-337-3p	hsa-miR-525-5p	hsa-miR-801	hsa-miR-4306			
hsa-miR-33a	hsa-miR-552	hsa-miR-888*	hsa-miR-4313			
hsa-miR-346	hsa-miR-593	hsa-miR-93*	hsa-miR-4319			
hsa-miR-34c	hsa-miR-597	hsa-miR-942	hsa-miR-4323			
hsa-miR-362-3p	hsa-miR-605	hsa-miR-1202	hsa-miR-4327			

For information regarding miRNA nomenclature please see Section 1.5.2.1

4.4.4 miRNA Expression Profiling of the Adrenal Gland

RNA extracted from four normal human adrenal glands was tested for quality prior to performing the microarray analysis. The quantity of the samples was first assessed using the Nanodrop (Section 2.2.4) then each sample was diluted to run on an Agilent Bioanalyser chip (Section 2.2.5). The electropherogram and RIN (RNA integrity number) are shown in Figure 4-4. The peaks labelled 18S and 28S represent the ribosomal RNA and are used to calculate the RIN. The first small peak is a spike-in RNA control, the second represents small RNA molecules that would encompass miRNAs. The 'Normal Adrenal 4' sample had the lowest RIN which may have proved problematic for mRNA studies, but it had a strong 'microRNA' peak. All samples were subject to further quality control testing by LC Sciences (Houston, Texas) prior to being run on the miRNA microarray. All samples successfully passed the internal quality control tests and, therefore, the miRNA microarray was performed on four normal adrenal gland samples.



Figure 4-4. Agilent Bioanalyser Electropherogram of Adrenal RNA Samples. RIN: RNA integrity Number; FU: fluorescence.

The mean microarray output intensity is shown in Figure 4-5 for each of the 728 human miRNAs on the chip (Version 10.1). The manufacturers of the microarray microfluidic chip recommended a cut off value of 500 arbitrary units (AU) to positively determine a miRNA expressed within a sample; this is represented by the red line on Figure 4-5. For future investigations miRNAs expressed below the threshold were discarded, leaving 103 miRNAs expressed in the human adrenal gland above the stated threshold. The mean expression level of these miRNAs is shown in Figure 4-6. The microarray results for individual miRNAs was relatively consistent across each of the four adrenal samples. The expression ranged from 510 ± 88 AU for miR-106a, to the highly expressed miR-26a, expressed at 31,535 \pm 2,012 AU. Some of the expressed miRNAs included both strands of the precursor-miRNA e.g. pre-miR-574 and pre-miR-768. In this case the -5p and -3p miRNA were expressed at approximately equal levels. However, for pre-miR-151 the -5p miRNA was expressed at a higher level than the -3p form, $6,052 \pm 190$ AU and 943 \pm 58 AU respectively. In contrast the expression of miR-199a-5p was (713) \pm 135 AU) the miR-199a-3p (8,405 \pm 1,282 AU). Eight members of the let-7 (a-g and i) family were highly expressed in the tissue.

To validate the miRNA microarray result, 40 miRNAs were chosen to be guantified by Tagman[®] gRT-PCR. The mean cycle threshold value for each miRNA is shown in Table 4-7. Again, the intra-sample variation was low confirming the homology of the adrenal samples tested. The qRT-PCR results were normalised to a housekeeping gene, RNU48, creating a delta Ct value (Δ Ct). The negative ΔCt was expressed to the power of 2 (representing the doubling of RNA concentration per Ct). This value is, therefore, proportional to the abundance of the miRNA in the sample. These results were highly and significantly correlated with the microarray values; $r^2 = 0.50$ and p < 0.0001 (Figure 4-7). One notable outlier was miR-638, which had a relatively high expression as quantified by microarray analysis (17,630 ± 1,681 AU), but low expression according to gRT-PCR validation (mean Ct value 29.78 \pm 0.18; 2^{- Δ Ct} equivalent: 0.004) (Figure 4-7 and Table 4-7). gRT-PCR is the gold standard for expression guantification and therefore miR-638 should be regarded as a low-expression miRNA in the adrenal gland. Other than this the qRT-PCR results confirmed the findings of the miRNA microarray experiment.



Human microRNAs

Figure 4-5. Normal Adrenal Gland miRNA Microarray Output.

The average normalised microarray expression signal of 728 human miRNAs (x-axis, in miRNA numerical order) in four normal adrenal glands. The red line represents the expression cut-off threshold of 500 Arbitrary Units (AU).



Figure 4-6. Average Microarray Signals of miRNAs Expressed in the Adrenal Gland.

The average normalised microarray expression signal of miRNAs with an expression level greater than 500 Arbitrary Units (AU). The mean of four tissues, error bars represent SEM.

miRNA Taqman® Assay	Ct value	Standard Error	miRNA Taqman® Assay	Ct value	Stan dard Error
hsa-let-7a	20.75	0.12	hsa-miR-185	31.35	0.34
hsa-let-7b	20.63	0.11	hsa-miR-195	21.42	0.12
hsa-let-7e	25.62	0.15	hsa-miR-202	25.74	0.10
hsa-miR-101	25.10	0.13	hsa-miR-214	24.08	0.13
hsa-miR-103	23.05	0.08	hsa-miR-218	25.25	0.19
hsa-miR-106a	25.76	0.09	hsa-miR-22	22.74	0.06
hsa-miR-106b	24.39	0.08	hsa-miR-23a	23.43	0.13
hsa-miR-107	27.35	0.06	hsa-miR-23b	22.58	0.11
hsa-miR-10a	24.44	0.10	hsa-miR-24	21.60	0.09
hsa-miR-10b	22.50	0.17	hsa-miR-26b	21.22	0.07
hsa-miR-125b	20.33	0.10	hsa-miR-29b	23.16	0.12
hsa-miR-126	21.38	0.14	hsa-miR-30b	21.22	0.09
hsa-miR-132	25.19	0.12	hsa-miR-30c	21.68	0.08
hsa-miR-134	25.99	0.12	hsa-miR-320	22.72	0.08
hsa-miR-143	24.45	0.24	hsa-miR-34a	23.70	0.17
hsa-miR-145	24.25	0.18	hsa-miR-379	25.38	0.07
hsa-miR-150	25.71	0.36	hsa-miR-382	25.90	0.14
hsa-miR-15a	25.57	0.12	hsa-miR-432	26.11	0.15
hsa-miR-16	20.58	0.13	hsa-miR-487b	24.62	0.14
hsa-miR-181a	24.97	0.16	hsa-miR-638	29.78	0.18

Table 4-7. qRT-PCR Expression Validation of Selected miRNAs.

Mean cycle threshold (Ct) values are from four adrenal gland samples measured by qRT-PCR in technical triplicate (three repeats per sample per assay)



Figure 4-7. Correlation of Microarray and qRT-PCR miRNA Expression.

The microarray expression levels of twenty miRNAs plotted against their qRT-PCR expression. The cycle threshold (Ct) the housekeeping reference, RNU48, was deducted from that of the miRNA (Δ Ct). The negative Δ Ct was expressed to the power of 2 (assuming 100% efficiency).

4.4.5 Adrenal-miRNAs with CYP11B1 and CYP11B2 Binding Sites

To identify the miRNAs which may regulate *CYP11B1* and *CYP11B2* mRNA in the adrenal gland, the putative miRNA-target sites and miRNA expression data sets were combined. The results of this are summarised in Figure 4-8. Half of the miRNA target sites for *CYP11B1* are shared with the target sites predicted for *CYP11B2*, whereas, the miRNAs predicted to target both genes comprise the majority of these predicted to target *CYP11B2*. In total, 24 adrenal-expressed miRNAs were predicted to bind *CYP11B1* 3'UTR and 16 to bind *CYP11B2* 3'UTR; in total, 28 miRNAs are predicted to bind to both genes. Therefore, based on the computational approaches used, 75 miRNAs expressed in the adrenal gland do not bind to the *CYP11B1* or *CYP11B2* mRNA.

Detailed information for the 28 miRNAs is shown in Table 4-8. None of these miRNAs are located in genes which have been directly implicated in corticosteroidogenesis. However, their genetic locations may prove interesting: both miR-638 and miR-214 are located in a region coding for dynamin mRNA (type 2 and 3 respectively); miR-218 and miR148* are located in the *SLIT2* gene and the *GPC1* gene, whose products are known to interact; miR-10a and miR-10b are both located in genes belonging to the homeobox transcription factor family. Finally, several miRNAs are transcribed in clusters including miR-24 and miR-23b which are in the *C9orf3* cluster.

Table 4-9 and Table 4-10 give specific information regarding the bioinformatic target site predictions, including which database(s) predicted each miRNA binding site, the number of binding sites predicted and the proposed degree of base-pairing complementarity in the seed region (Figure 4-1). These tables also contain the microarray expression values for each miRNA (Table 4-9 and Table 4-10). Only three of the miRNAs (miR-149*, miR-423-5p and miR-34a) were predicted to possess two separate target sites, for miR-149* and miR-423-5 the binding sites are relatively close, (206 and 69 bases apart, respectively). Most target site predictions were provided by only one or two of the databases but three databases (the exception being miRViewer) predict a binding site for miR-638 on the *CYP11B1* 3'UTR (Table 4-9).
Given the tendency of bioinformatics databases to rely on seed site base-pairing for miRNA-target prediction, it is unsurprising that the majority of target sites predicted have full complementarity at the seed region (Figure 4-9) and that most are 7mer predictions, indicating a strong likelihood of miRNA regulation. miR-940 is predicted to bind with the strongest seed-pairing, 8mer, to both of the genes; miR-149* is also predicted to have an 8mer site on the *CYP11B1* gene (Table 4-9). Two miRNAs, miR-125a-5p and miR-125b, are predicted to bind with a 8mer seed site to the *CYP11B2* gene (Table 4-10). These miRNAs belong to the same miRNA family and therefore share a high degree of sequence similarity with identical sequence in the seed region; for this reason their 8mer predicted site lies in exactly the same place on the 3'UTR.



Figure 4-8. Venn Diagram of Adrenal miRNAs with Putative miRNA-Target Sites in the 3'UTR of the CYP11B1 and CYP11B2 Genes

miRNA microarray analyses of non-tumorous adrenal tissue combined with bioinformatics analysis of putative miRNA binding sites in the 3'-UTR of the *CYP11B1* and *CYP11B2* genes

		Genetic	
	MIRINA Sequence	Location	Co-ordinates of MIRNA Location
hsa-miR-149°	agggagggacggggggcugugc	Intronic	2 : 241,395,418 - 241,395,506 [+]
hsa-miR-151-5p	ucgaggagcucacagucuagu	Intronic	8 : 141,742,663 - 141,742,752 [-]
hsa-miR-181a	aacauucaacgcugucggugagu	Intergenic	1 : 198,828,173 - 198,828,282 [-]
hsa-miR-214	acagcaggcacagacaggcagu	Intronic	1 : 172,107,938 - 172,108,047 [-]
hsa-miR-218	uugugcuugaucuaaccaugu	Intronic	4 : 20,529,898 - 20,530,007 [+]
hsa-miR-23a	aucacauugccagggauuucc	Intergenic	19 : 13,947,401 - 13,947,473 [-]
hsa-miR-23b	aucacauugccagggauuacc	Intronic	9 : 97,847,490 - 97,847,586 [+]
hsa-miR-382	gaaguuguucgugguggauucg	Intergenic	14 : 101,520,643 - 101,520,718 [+]
hsa-miR-423-5p	ugaggggcagagagcgagacuuu	Intronic	17 : 28,444,097 - 28,444,190 [+]
hsa-miR-432	ucuuggaguaggucauugggugg	Intergenic	14 : 101,350,820 - 101,350,913 [+]
hsa-miR-494	ucuuggaguaggucauugggugg	Intergenic	14 : 101,495,971 - 101,496,051 [+]
hsa-miR-768-5p	guuggaggaugaaaguacggagugau	Intergenic	16 : 70,349,796 - 70,349,899 [-]
hsa-miR-10a	caaauucguaucuaggggaaua	Intronic	17 : 46,657,200 - 46,657,309 [-]
hsa-miR-10b	uacccuguagaaccgaauuugug	Intronic	2 : 177,015,031 - 177,015,140 [+]
hsa-miR-140-3p	cagugguuuuacccuaugguag	Intronic	16 : 69,966,984 - 69,967,083 [+]
hsa-miR-143	ugagaugaagcacuguagcuc	Intergenic	5 : 148,808,481 - 148,808,586 [+]
hsa-miR-185	uggagagaaaggcaguuccuga	Intronic	22 : 20,020,662 - 20,020,743 [+]
hsa-miR-22	aagcugccaguugaagaacugu	Intronic	17 : 1,617,197 - 1,617,281 [-]
hsa-miR-24	uggcucaguucagcaggaacag	Intronic	9 : 97,848,303 - 97,848,370 [+]
hsa-miR-28-5p	aaggagcucacagucuauugag	Intronic	3 : 188,406,569 - 188,406,654 [+]
hsa-miR-34a	uggcagugucuuagcugguugu	Intergenic	1 : 9,211,727 - 9,211,836 [-]
hsa-miR-34c-3p	aaucacuaaccacacggccagg	Intergenic	11 : 111,384,164 - 111,384,240 [+]
hsa-miR-638	agggaucgcgggcggguggcggccu	Intronic	19 : 10,829,080 - 10,829,179 [+]
hsa-miR-940	aaggcagggcccccgcucccc	Intergenic	16 : 2,321,748 - 2,321,841 [+]
miR-125a-5p	ucccugagacccuuuaaccuguga	Intergenic	19 : 52,196,507 - 52,196,592 [+]
miR-125b	ucccugagacccuaacuuguga	Intergenic	11 : 121,970,465 - 121,970,552 [-]
miR-134	ugugacugguugaccagagggg	Intergenic	14 : 101,521,024 - 101,521,096 [+]
miR-495	aaacaaacauggugcacuucuu	Intergenic	14 : 101,500,092 - 101,500,173 [+]

Table 4-8. Genomic and Sequence Information of Adrenal-miRNAs Predicted to Bind to *CYP11B1* and *CYP11B2*.

Adrenal miRNA information obtained from miRBASE. miRNA sequence: 5'- 3' direction. Genomic co-ordinates: chromosome number: co-ordinates [strand]. The double lines separate miRNAs according to bioinformatic prediction, the top section includes miRNAs with putative binding site in *CYP11B1*, the middle section miRNAs predicted to bind to both genes; and the bottom miRNAs putative binding sites on the *CYP11B2* 3'UTR.

Database					е						
microRNA	Number of Database Hits	TargetScan	microRNA.org	miRVierwer	MicroCosm	Number of Binding Sites	Seed Site Match	Distance Between Binding Sites (nucleotides)	Micr Averag Intens	oarr je S sity (ay ignal AU)
miR-149*	1				~	2	8mer / Offset 6mer	206	608	±	89
miR-151-5p	2	~			~	1	7mer-m8		6,052	±	190
miR-181a	1	✓				1	7mer-m8		2,011	±	274
miR-214	1		~			1	Offset 6mer		6,894	±	1,106
miR-218	1	\checkmark				1	7mer-1A		642	±	197
miR-23a	2	\checkmark	~			1	7mer-m8		17,047	±	2,074
miR-23b	2	\checkmark	~			1	7mer-m8		18,033	±	1,998
miR-382	1				~	1	Offset 6mer		1,116	±	195
miR-423-5p	2	\checkmark	~			2	Offset 6mer / Offset 6mer	69	2,272	±	507
miR-432	2		~		~	1	Offset 6mer		611	±	90
miR-494	1	~				1	7mer-1A		1,576	±	369
miR-768-5p	1	~				1	7mer-1A		2,832	±	1,021
hsa-miR-10a	2	\checkmark			✓	1	7mer-1A		1,872	±	470
hsa-miR-10b	2	\checkmark			~	1	7mer-1A		9,496	±	1,152
hsa-miR-140-3p	1		~			1	7mer-m8		1,082	±	79
hsa-miR-143	1	\checkmark				1	7mer-1A		6,877	±	1,236
hsa-miR-185	1	~				1	7mer-m8		1,141	±	90
hsa-miR-22	1	\checkmark				1	7mer-m8		3,473	±	670
hsa-miR-24	2		~	~		2	Offset 6mer	930	8,938	±	810
hsa-miR-28-5p	2	\checkmark	~			1	7mer-m8		788	±	75
hsa-miR-34a	2	~	~			2	7mer-m8/Offset 6mer	1,275	1,418	±	98
hsa-miR-34c-3p	1	\checkmark				1	7mer-1A		752	±	264
hsa-miR-638	3	~	~		~	1	7mer-1A		17,630	±	1,681
hsa-miR-940	2	\checkmark	~			1	8mer		735	±	264

Table 4-9. Bioinformatic and Microarray Information of Adrenal-miRNA with Putative Binding Sites in the CYP11B1 3'UTR.

Database					е						
microRNA	Number of Database Hits	TargetScan	micro RNA. org	miRVierwer	MicroCosm	Number of Binding Sites	Seed Site Match	Distance Between Binding Sites (nucleotides)	Microar Signal Ir	ray /	Average sity (AU)
hsa-miR-125a-5p	2	\checkmark	~			1	8mer	· · ·	18,278	±	1,808
hsa-miR-125b	2	✓	~			1	8mer		22,781	±	1,558
hsa-miR-134	1	✓				1	7mer-m8		1,130	±	217
hsa-miR-495	1	~				1	7mer-m8		777	±	210
hsa-miR-10a	1	✓				1	7mer-1A		1,872	±	470
hsa-miR-10b	2	✓			\checkmark	1	7mer-1A		9,496	±	1,152
hsa-miR-140-3p	1	✓				1	7mer-m8		1,082	±	79
hsa-miR-143	1	✓				1	7mer-1A		6,877	±	1,236
hsa-miR-185	2	✓	~			1	7mer-m8		1,141	±	90
hsa-miR-22	2	~		~		1	7mer-m8		3,473	±	670
hsa-miR-24	1			✓		1	Offset 6mer		8,938	±	810
hsa-miR-28-5p	2	~	~			1	7mer-m8		788	±	75
hsa-miR-34a	1			✓		2	7mer-m8/Offset 6mer	1,275	1,418	±	98
hsa-miR-34c-3p	1	~				1	7mer-1A		752	±	264
hsa-miR-638	1	~				1	7mer-1A		17,630	±	1,681
hsa-miR-940	2	~	\checkmark			1	8mer		735	±	264

Table 4-10. Bioinformatic and Microarray Information of Adrenal-miRNA with Putative Binding Sites in the CYP11B2 3'UTR.



Figure 4-9. Seed Site Distribution of Putative miRNA Binding Sites.

Base-pairing in the seed region of miRNAs with putative binding sites in the *CYP11B1* and *CYP11B2* 3'UTR.

4.5 Discussion

The experiments presented in this chapter investigated the existence of miRNA target sites in corticosteroidogenic genes using computational methods. It also identified the array of miRNAs expressed within the adrenal gland: this data was used to construct a list of potential miRNAs which may regulate the *CYP11B1* and *CYP11B2* genes *in vivo*. The main outcome of this part of the project was the identification of 24 miRNAs expressed in the adrenal gland which may regulate *CYP11B1* transcription and 16 that may regulate *CYP11B2*.

The first stage of *in silico* analysis investigated whether any non-coding RNAs are located within the coding regions of genes involved in corticosteroidogenesis. None of the 1,048 miRNAs currently registered at the miRNA repository, MiRBase (Chobanian et al., 2003; Griffiths-Jones et al., 2006; Griffiths-Jones et al., 2008) were mapped to genomic regions encompassing the important corticosteroidogenic genes, including CYP11B1 and CYP11B2. Hence, altered transcriptional regulation of these genes or SNPs located in these genes will not directly affect the transcription or quantity of any miRNA. Therefore, miRNAs found to regulate CYP11B1 and CYP11B2 will be located in separate genomic locations and subject to their own transcriptional regulation.

In-depth analysis of the 3'UTR sequence of *CYP11B1* and *CYP11B2* revealed both to be conserved across a few, but not all, mammalian species. Conservation of the 3'UTR and, in particular the putative miRNA-target sites is proposed to be an indicator of miRNA regulation (Friedman *et al.*, 2009). miRNA sequences themselves are highly conserved, yet there are examples of mammalian-specific miRNAs which may target non-conserved sites (Sethupathy *et al.*, 2006;Friedman *et al.*, 2009). It could be argued that target site conservation is preferentially selected for or that non-conserved regions in higher mammals have evolved. The *CYP11B1* and *CYP11B2* genes have the longest 3'UTRs of all the genes analysed and this might reflect the importance of this region for regulation of mRNA stability. Studies have shown that genes involved in basic cell homeostasis, for example ribosomal genes, have shorter 3'UTRs and are resistant to miRNA-mediated repression (Stark *et al.*, 2005). Assessment of the unbound secondary mRNA structure was performed but without the addition of a miRNA,

it is difficult to interpret this. The software used to generate this structure cannot generate miRNA interactions but an alternative software package, RNAHybrid, has been developed (Kruger and Rehmsmeier, 2006). This software will be useful for assessing the impact of miRNAs on the secondary structure of the *CYP11B1* or *CYP11B2* 3'UTR and evaluating the minimum free energy of binding (Kruger and Rehmsmeier, 2006) and will be utilised in Section 5.4.9.

Another parameter which may influence miRNA regulation is the turnover rate of mRNA, which is determined by the rate of transcription vs. the rate of mRNA degradation. mRNAs with a relatively quick turnover are less likely to be subject to miRNA regulation and more likely to be regulated by other degrading factors that act on AU-rich element (AREs). However, those with longer half-lives are more likely to be regulated by miRNAs (Larsson et al., 2010). The assessment of the half-lives of CYP11B1 and CYP11B2 mRNA has been previously estimated in H295R cells. These were treated with Actinomycin D for 2 hr, then the relative abundance of the mRNA transcripts measured at hourly intervals (Lin et al., 2006). The abundance of both CYP11B1 and CYP11B2 mRNA decreased to half of the starting level by approximately 3 hr. This is on the borderline of what Larsson *et al.* (2010) describe as a slow or medium turnover rate and, based on their analysis, CYP11B1 and CYP11B2 mRNA are less likely to be regulated by miRNAs than a longer turnover rate mRNA (half-life > 1000 minutes). However, they did show that 4.4% of the slower turnover mRNAs were liable to strong repression by miRNAs, compared to 10.2% of long turnover mRNAs. There is also quite a lot of scatter within their data; 3 of the previously published data-sets used did not show a correlation of repression to turnover rate. Finally, the datasets were all generated in a non-specific cell line (HeLa), used miRNA species and defined miRNA-mediated regulation by strong repression of mRNA target (Larsson et al., 2010) which may not reflect canonical fine-tuning miRNA action in specific cell types (Sevignani *et al.*, 2006). The authors also mention that the relative abundance of mRNA could have an impact on regulation. Thus, in the CYP11B1 and CYP11B2 expression-specific cells of the adrenal cortex, this may not be a factor. The importance of mRNA turnover rate remains to be seen.

miRNA target sites were predicted by bioinformatic analysis all of the genes analysed. Of the databases employed, only Tarbase failed to predict any miRNA target sites, presumably because it relies solely on experimentally-validated miRNA targets (Papadopoulos *et al.*, 2009), and there are currently no validated miRNA, mRNA targets in this pathway (Section 1.9).

In general, there was a large degree of variation in the number of predicted sites generated by the other four databases. miR-viewer predicted the lowest number of miRNA binding sites and, for CYP17A1 and 3BHSDII mRNA, it failed to predict any sites at all. The database with the next lowest prediction count was MicroCosm Targets. The algorithms used by the TargetScan and microrna.org databases appear to have equivalent stringencies for identifying miRNA target sites, as demonstrated by their similar number of target-site predictions. Interestingly, both of these databases have user-definable parameters, e.g. the levels of miRNA sequence conservation. The results presented from these two databases in this thesis have been generated without restrictions on conservation. In fact, TargetScan identified no miRNA-target sites for evolutionarily conserved miRNAs in the CYP11B1 or CYP11B2 3'UTRs. The exact details of what governs miRNA:mRNA target interaction are still being investigated and, as such, databases are continually evolving. Therefore, it was decided no to increased the stringency of conservation required for miRNAtarget prediction, thus avoiding excessive false negative predictions, and also to use a combinational approach of several databases to improve accuracy.

The degree of variation between the databases is consistent with other reports (Bartel, 2009;Saito and Saetrom, 2010). This highlights the current lack of understanding regarding miRNA-target binding and the fact that relatively few miRNA targets have been experimentally validated. Two recent reviews on this subject highlighted the variation between the current group of bioinformatic databases and emphasised that validation is biased towards complementary seed site binding and sequence conservation (Bartel, 2009;Saito and Saetrom, 2010). However, this may itself be biased due the high weighting placed on these features by several of the commonly-used databases. Other features which are absent from the currently-available databases include alternative miRNA:mRNA complementarity (e.g. centered site base-pairing, Figure 4-1); accounting for optimal distance between multiple miRNA binding sites; or the ability to search for binding sites in the coding-region or 5' untranslated region of genes (Saito and Saetrom, 2010).

Based on computational bioinformatics it appears that all the genes within the corticosteroidogenic pathway possess miRNA binding sites and hence could be subject to miRNA-mediated regulation. As previously stated, the main focus of this thesis is the regulation of *CYP11B1* and *CYP11B2* mRNA expression, and therefore the remainder of the work presented will be concerned with these genes. Bioinformatic prediction identified 202 and 172 miRNA binding sites in the 3'UTRs of the *CYP11B1* and *CYP11B2* genes, respectively. There was a relatively high overlap of miRNA binding-site predictions; given the high degree of sequence similarity in the 3'UTRs (approximately 80%) this is probably not surprising. Moreover, it is comparable to the similarity observed in transcription factor binding sites located at the 5'UTR (Bassett *et al.*, 2000;Bassett *et al.*, 2002). However, there are numerous miRNAs with putative binding sites in only one of the genes so, potentially, miRNA-mediated regulation could be achieved by target-sites common to both genes or by specific miRNAs, thereby accounting for the differential regulation of these genes.

Two striking observations arise from the predictions: the lack of any let-7 miRNA family target-sites and the high number of miRNA* (passenger strand) predictions for the *CYP11B2* 3'UTR. The let-7 family includes the let-7c miRNA, which was the second miRNA to be identified (Reinhart *et al.*, 2000). This family is believed to have a major role in developmental timing and cell regulation (Roush and Slack, 2008) and hence would be thought unlikely to target the *CYP11B1* and *CYP11B2* genes. miRNA* are thought to be the redundant strand of the pre-miR hair-pin that generates the functioning miRNA (Section 1.5.2.1 and 1.5.2.3). However, as the function of more miRNAs is being discovered, many are being renamed to indicate from which arm they are cleaved (-3p or -5p) and, as such, do not indicate which mature arm of the pre-miRNA is likely to be functional. Therefore, the miRNA* predictions for *CYP11B2* may even represent a novel function of these * miRNAs.

To determine the miRNA expression profile of the adrenal gland, human nontumorous (termed 'normal') adrenal glands were chosen for miRNA microarray analysis. The samples were obtained from four kidney donor patients undergoing nephrectomy and their adrenal glands were retained and frozen at -80°C. RNA isolation from the tissues was successful, each sample obtaining strong RIN scores. The microarray and subsequent qRT-PCR validation indicated that the samples were of good quality and homogeneous for miRNA expression. Therefore, the results in this thesis can confidently confirm the expression of 103 human miRNAs in the adrenal gland. At the time the miRNA microarray was performed it contained probes which covered 100% of the known human miRNAs and therefore, represented the best option for assessing the relative levels of all miRNA in samples. The quality of the microarray was confirmed by qRT-PCR performed in the same samples; this correlated highly with the microarray expression results.

The samples of adrenal gland used for RNA isolation could not be cut precisely enough to include only tissue from the adrenal cortex and may also have included part of the adrenal medulla. Moreover, they would likely have contained portions of all three layers of the adrenal cortex. Therefore, the miRNA expression profile generated in this thesis is representative of the whole adrenal gland. It would be beneficial to obtain tissue from specific zones of the cortex in order to determine the zone-specific distribution, if any, of individual miRNAs. However, obtaining sufficient human, normal adrenal tissue for this has, as of yet, proved difficult.

When this investigation started, there were no existing publications describing adrenal miRNA expression. However, there have since been a small number of such studies with the main focus being on adrenal pathology e.g. adrenocortical adenoma or carcinoma. These have not studied normal adrenal function or the role of miRNAs in hypertension and have only used normal adrenal glands for purposes of comparison (Tombol et al., 2009; Soon et al., 2009; Iliopoulos et al., 2009;Bimpaki et al., 2010;Schmitz et al., 2011). In general, it is hard to compare the miRNA-expression profiling in these publications with that which is presented in this thesis. It appears that none of the publications have submitted their microarray results to a microarray repository to allow open access of individual miRNA expression results, meaning that only information regarding selected published miRNAs can be obtained. Furthermore, comparison is complicated as all of these studies used different profiling platforms to each other, and to the one used in this thesis. Only the study by Soon *et al.*, (2009) used microarray system of the same level of miRNA coverage as this study (100% of miRNAs in miRBase v10). The others used a Taqman Low Density Array[®] (Applied

Biosciences) platform which is a good alternative for assessing miRNA expression but typically contains fewer miRNA probes (365 probes) than chip-based arrays. Finally, interpreting the data is complicated due to post-assay data manipulation for example depicting expression levels by hierarchical heat-map clustering, meaning that relative quantities of miRNAs can not be compared. However, careful analysis shows that at least 33 of the 103 miRNAs identified in this thesis have been previously identified (Tombol *et al.*, 2009;Soon *et al.*, 2009;Iliopoulos et al., 2009; Bimpaki et al., 2010; Schmitz et al., 2011). This number may in fact be higher but full data-sets are not available and, for one study (Schmitz et al., 2011), the online supplementary data is not currently accessible and another presents only profiling data from tumorous tissue (Soon et al., 2009). Several miRNAs have been consistently detected both in the current study and in more than one of the other published studies: miR-100, miR-143, miR-145, miR-375, miR-424 and miR-7. Without access to the full data from the published microarray studies it is impossible to comment on how representative is the data presented in this thesis, but even the small degree of overlap described above suggests that the findings in this thesis are comparable. This current study presents the most comprehensive adrenal miRNA expression investigation in normal subjects; with the exception of Tombol et al., (2009), the other studies used either commercially-purchased, pooled adrenal samples or adjacent adrenal tissue lying next to an adrenal tumour. Tombol et al., (2009) used adrenal tissue samples obtained from patients undergoing nephrectomy for kidney tumours and, while there is no reason to suggest that the adrenal glands were also affected by tumorous tissue, the adrenal gland and kidney are linked by blood supply and miRNAs can circulate through the bloodstream (Section 1.8.2). Therefore, using the adrenal glands of kidney donor patients, as in the current study, is probably the most reliable source of normal adrenal tissue.

By combining the adrenal miRNA-expression data and the bioinformatic target site analysis, those miRNAs which may regulate *CYP11B1* and *CYP11B2* expression in the adrenal gland could be identified. This approach identified 24 miRNAs with putative binding sites on the *CYP11B1* 3'UTR and 16 on *CYP11B2* 3'UTR; 12 miRNAs were common to both. Since these data were complied, one potential *CYP11B1*-regulating miRNA, miR-768-5p, has been removed from miRBase as the genomic location from which it was transcribed has been re-annotated as

containing a small nucleolar RNA, HBII-239. Evolutionary conservation analysis supports the mapping of HBII-239, therefore this miRNA is excluded from future investigation. One adrenal-miRNA, miR-638, was predicted in three databases, the most of all the miRNAs, and may have represented the best candidate miRNA for further study if qRT-PCR quantification had not shown that it is actually relatively lowly expressed within the adrenal gland.

To identify which miRNA may be likely to regulate these genes it was interesting to note that several of the predicted miRNAs belong to the same miRNA family, e.g. miR-10a and miR-10b, which share the same target site on the 3'UTR. Moreover, other miRNAs which are not part of a miR-family also share common binding regions e.g. miR-23a, miR-23b and miR-181a which all bind to a similar region in the 3'UTR of *CYP11B1*. This may add complexity to the mechanism by which these miRNAs work; a recent study has been the first to demonstrate that more than one miRNA can regulate an mRNA (Wu *et al.*, 2010). Several adrenal miRNAs in the target lists are produced as part of a cluster, and this may represent some sort of adrenal transcriptional specificity. For example, miR-24, miR-23b and miR-27b are all found in the adrenal gland, but only miR-24 and miR-23b have binding sites in the *CYP11B1* and *CYP11B2* genes. Cluster members miR-381 and miR-134 are transcribed together but have putative binding sites on different 3'UTR (*CYP11B1* and *CYP11B2*, respectively).

A literature search found that the previously-reported functions of the miRNAs identified fall into four categories: those involved in cancer, including action on tumour suppressor genes, oncogenes or genes that alter cell cycle regulation; those with association to cardiovascular disease; those with association to neural disease; or those whose function has not yet been established. Numerous miRNAs fall into several of these categories, highlighting that each may act on several mRNA targets (Lewis *et al.*, 2003) and, in some cases, a miRNA may act in a cell-type specific manner (Section 1.7.1).

Finally, in the hope of identifying miRNAs that may be associated with hypertension, the genomic locations of the miRNAs were compared with those of published QTL (quantitative trait loci) associated with hypertension (Cowley, 2006). In total, six miRNAs are encoded in known hypertension QTL and of these, four are located in the QTL of defined causes of hypertension or identified in

non-Caucasian cohorts. Of the remaining two, the sequence of miR-10a lies within an intron of a homeobox gene on chromosome 17 at location identified in the Framingham Heart Study as being associated with hypertension (Levy *et al.*, 2009), while miR-28-5p is found in an intron of the LPP gene, at a region identified in the HERITAGE Family Study (Rice *et al.*, 2002). As previously discussed (Section 1.1.3.2), the identification of QTL for hypertension has proved difficult to replicate and to validate by functional experiments investigating the gene coded within them. However, it may be possible that miRNAs located in QTL may be the functioning regulator of pathways involved in the development of hypertension.

In conclusion, these studies have identified several putative miRNA binding sites in all of the genes involved in corticosteroid synthesis and have defined the miRNA expression profile of a normal human adrenal gland to produce a shortlist of potential miRNAs that may regulate *CYP11B1* and *CYP11B2* mRNA stability. The work presented in this chapter has also highlighted, by sequence conservation analysis and relative 3'UTR length, that the *CYP11B1* and *CYP11B2* genes are good candidates for miRNA regulation. The subset of adrenal-miRNAs which have a putative miRNA target-site on the 3'UTR are currently based on bioinformatic prediction only. Therefore, they require experimental validation which will be presented in Chapter 5 of this thesis.

5 miRNA Target-Site Validation in the 3'UTR of *CYP11B1* and *CYP11B2*.

5.1 Introduction

Putative miRNA binding sites must be experimentally investigated in order to validate bioinformatic analyses and this will be the focus of the current chapter. Validation can be achieved by several methods, most of which involve *in vitro* analysis, that may provide direct or indirect evidence of miRNA-mediated regulation. Two important considerations when designing experiments are that the measurements should be highly specific and accurate to allow for detection of the small changes typical of miRNA-mediated regulation, and that a suitable model be chosen, that best represents the *in vivo* situation.

Methods to validate miRNA target sites typically involve manipulating miRNAexpression levels, then assessing mRNA abundance or protein levels. Several examples of miRNA inhibition are given in Section 1.8 and Figure 1-20. Increasing cellular miRNA levels can be achieved by introducing exogenous miRNA mimics into cells, these can be in the form of short single-stranded RNA molecules, larger molecules which are cleaved intracellulary to form short single-stranded miRNA molecules, or plasmid constructs containing pre-miR or pri-miR sequences that are expressed under the control of a viral or cell-type specific promoter. In this chapter, short single-stranded molecules are used to manipulate miRNA abundance; these include Pre-miR[™] molecules, which increase miRNA expression, and Anti-miR[™] molecules whose complementarity to specific endogenous miRNAs causes them to be bound and competitively antagonised. The benefits of these molecules are that they do not require endogenous miRNA processing enzymes, such as Dicer, that they are specific to one miRNA (whereas those involving expression of a pre-miR will lead to exogenous expression of miRNA and miRNA* species) and that it is possible to measure their effect on miRNA expression by qRT-PCR analysis.

Two types of miRNA target validation experiment will be used in this chapter: firstly, the 3'UTR reporter plasmid construct used in Chapter 3 will be tested with specific Pre-miR^M or Anti-miR^M molecules and miRNA binding assessed by measuring the relative luciferase levels; secondly, miRNA-action will be assessed by transfecting these molecules into the human adrenocortical cell line, H295R (Section 3.1), and then assessing the levels of *CYP11B1* or *CYP11B2* mRNA by qRT-PCR. The subsequent effects on steroid production will also be assessed by measuring cortisol and aldosterone secretion in cell media.

An additional parameter that is often investigated in miRNA analysis is the effect on protein synthesis, usually measured by western blot analysis. However, there are currently no antibodies available for detecting the human *CYP11B1* and *CYP11B2* proteins specifically and effectively. Therefore, this could not be performed in this current study, although changes in steroid production can be suggestive of changes in protein abundance.

5.2 Aims

The aims of this study are to investigate the function of adrenal miRNAs predicted to have binding sites on the *CYP11B1* and/or *CYP11B2* 3'UTR with a view to validating the bioinformatic analyses. To achieve this, miRNA targeting of the 3'UTR will be assessed by reporter construct analysis, as determined by measurement of steroids and of *CYP11B1* and *CYP11B2* mRNA following transfection of H295R cells with small molecules capable of modulation miRNA levels (*i.e.* Pre-miR^m and Anti-miR^m).

5.3 Materials and Methods

5.3.1 Investigation of miRNA Binding to 3'UTR Reporter Construct

HeLa cells were co-transfected with pEZX reporter constructs (500 ng) and PremiR^m or Anti-miR^m molecules (50 nM) in 24-well plates as described in Section 2.9.4. 48 Hours post-transfection, cell lysates were prepared and their firefly and renilla luciferase activities measured using the Dual Reporter Luciferase Assay (DLRA, Sections 2.10). Control transfections and normalisation methods were identical to those described in Section 3.3.3.

5.3.2 Pre-miR[™] or Anti-miR[™] Transfection of H295R Cells

H295R cells were transfected in 6-well plates as described in Section 2.9. The Pre-miR^m or Anti-miR^m molecules used are listed Table 5-1 and were prepared as detailed in Section 2.9.1. The medium was replaced on the cells after 24 hours and, after 48 hours, this was removed for steroid analysis by LC:MS/MS (Section 2.11). Cell lysates were prepared for total RNA isolation (Section 2.2.1), ready for qRT-PCR analysis as described in Section 2.4 and 2.5. Data were analysed using the $\Delta\Delta$ Ct method (Section 2.5.4), with *GAPDH* or *RNU48* used as a housekeeping gene and results expressed relative to the negative (Pre-miR^m or Anti-miR^m) control-transfected cells.

5.3.3 Statistical Analysis

For experiments performed once in technical triplicate statistical analysis was performed using a Student's t-test.

All other experiments were performed at least in triplicate and at least two biologically-independent times. As described above, the miRNA, Pre-miR^M or Anti-miR^M molecules were normalised to a negative control transfected with a scrambled Pre-miR^M or Anti-miR as appropriate. The mean result of the biologically replicated experiments were then taken and statistical analysis was performed using a one-sample t-test, imputing either 100% or a 1-fold change as a reference level to which experimental values were compared.

For mature miRNA measurement following increasing $Pre-miR^{M}$ or Anti-miR^M concentrations, a one-way ANOVA and Bonferroni's post-hoc tests were used, comparing each concentration to the negative control cells.

For all analyses, confidence intervals of 95% were used and P < 0.05 was required for statistical significance. Data are expressed as the mean \pm standard error of the mean (SEM).

miRNA	Pre-miR [™] Product Code	Anti-miR [™] Product Code
Negative Control #1	AM17110	AM17010
Let-7c		4392431
miR-1	AM17150	
miR-10a	PM10787	AM10787
miR-10b	PM11108	AM11108
miR-125a-5p	PM12561	AM12561
miR-125b	PM10148	AM10148
miR-134	PM10341	AM10341
mir-140-3p	PM12503	AM12503
miR-143	PM10883	AM10883
miR-185	PM12486	AM12486
miR-22	PM10203	AM10203
miR-23b	PM10711	AM10711
miR-24	PM10737	AM10737
miR-34a	PM11030	AM11030
miR-382	PM10924	AM10924
miR-423-5p	PM11164	AM11164
miR-495	PM11526	AM11526
miR-638	PM11594	AM11594
miR-940	PM12798	AM12798

Table 5-1. Pre-miR[™] and Anti-miR[™] molecules.

Supplier: Applied Biosystems (Warrington, U.K.)

5.4 Results

5.4.1 Assessing Potential Repression of the pEZX Reporter Construct

To investigate the suitability of the 3'UTR reporter construct as an experimental tool, a siRNA molecule with perfect complementarity was designed to target each 3'UTR (Table 5-2). Each was co-transfected into HeLa cells with the corresponding reporter construct. The ratio of firefly luciferase to renilla luciferase was calculated and normalised to a transfected control siRNA. The *CYP11B1* 3'UTR siRNA significantly decreased the normalised ratio from the 100% control level to $16.13\% \pm 0.67$ (Figure 5-1A). Similarly, the *CYP11B2* 3'UTR siRNA significantly decreased the ratio to $52.34\% \pm 4.51$ relative to the control level of 100% (Figure 5-1B). These results offered proof of concept that miRNAs targeting the 3'UTR of either mRNA will result in significantly different luciferase activity that can be measured with a good degree of sensitivity.

 Table 5-2. siRNA Molecules Targeted to the CYP11B1 and CYP11B2 3'UTR.

siRNA Name	Sequence
CYP11B1 siRNA	UCUACAUCCUUCUCAUAUA
CYP11B2 siRNA	AUACUUAGGUAAUCAUUCC



Figure 5-1. siRNA Verification of the pEZX-B1 and pEZX-B2 Reporter Plasmids.

HeLa cells were co-transfected with (A) pEZX-B1 construct and *CYP11B1* 3'-UTR siRNA or (B) pEZX-B2 construct and *CYP11B2* 3'UTR siRNA. Firefly and renilla luciferase were measured 48 hours post-transfection using the Dual Luciferase Reporter Assay Kit. To normalise for transfection efficiency the ratio of firefly to renilla was measured and expressed as a percentage of the negative control siRNA value. Results represent the mean of three independent biological experiments, performed in quadruplicate; error bars represent standard error of the mean (SEM). ** p < 0.01, *** p < 0.001 compared to control.

5.4.2 miRNA Targeting to the 3'UTR of CYP11B1 mRNA.

miRNAs with putative binding sites in the 3'UTR of both *CYP11B1* and *CYP11B2* mRNA (Figure 4-8) were tested in turn, using the pEZX-reporter constructs to establish if the miRNA bind to the 3'UTR and cause repression. Specific single-stranded Pre-miR^M or Anti-miR^M were co-transfected into HeLa cells alongside the reporter construct to either overexpress or competitively inhibit a particular miRNA species.

Manipulating the levels of: miR-10a, miR-10b, miR-140-3p, miR-143, miR-22 and miR-34a did not change the relative levels of luciferase activity (Figure 5-2A-D: F and Figure 5-3B). Therefore, experimental data failed to validate the predicted bioinformatic target sites for these miRNAs on the *CYP11B1* 3'UTR.

However, several other miRNAs did cause changes in the relative luciferase levels. Firstly, increasing miR-185, reduced luciferase activity to $79.50 \pm 4.35\%$ (p = 0.018) although inhibiting the same miRNA with an Anti-miR^M did not increase the level of luciferase activity (Figure 5-2E). miR-24 (Figure 5-3A) and miR-638 (Figure 5-3E) a similarly consistent pattern of results with neither producing a change in luciferase activity when overexpressed despite seeing an increase in Anti-miR[™] transfected cells: miR-24 increased luciferase activity to $137.30 \pm 5.87\%$ (p = 0.023) and miR-638 increased it to 144.8 \pm 4.65% (p = 0.066). The putative miR-382 binding site on the CYP11B1 3'UTR was confirmed by changes in luciferase activity characteristic of canonical miRNA action *i.e.* Pre-miR^{\mathbb{M}} decreased luciferase activity to 58.51 ± 2.63% (p = 0.004) and Anti-miR^m transfected cells increased it to 121.90 ± 3.22% (p = 0.021) of control cells (Figure 5-3C). A similar pattern was observed for miR-940 (Figure 5-3F): Pre-miR[™] caused a large decrease in luciferase activity to 34.14 ± 2.25% (p = 0.0012). However, the Anti-miR^m only marginally increased it to 121.40 ± 8.98% and did not reach statistical significance. Manipulation of miR-423-5p caused interesting results: Anti-miR[™] cells did not cause a change in luciferase activity but Pre-miR^m increased it to 186.50 ± 8.015% (p = 0.0085). While this result does not confirm the existence of a miRNA target site it may suggest some other type of regulation (Figure 5-3D).

A summary of these results is shown in Table 5-3.



Figure 5-2. Effect of miRNAs on the Luciferase Activity of the pEZX-B1 Reporter Construct.

HeLa cells were co-transfected with the pEZX-B1 construct and Pre-miR^m or Anti-miR^m molecules for the indicated miRNAs. Firefly and renilla luciferase activity were measured 48 hours post-transfection using the Dual Luciferase Reporter Assay System. To normalise for transfection efficiency, the ratio of firefly to renilla luminescence was calculated and expressed as a percentage of the negative, scrambled control. Results represent the mean of at least three independent biological experiments, performed in quadruplicate; error bars represent standard error of the mean (SEM). * p < 0.05.



Figure 5-3. Effect of miRNAs on the Luciferase Activity of the pEZX-B1 Reporter Construct (2).

HeLa cells were co-transfected with the pEZX-B1 construct and Pre-miR^m or Anti-miR^m molecules for the indicated miRNAs. Firefly and renilla luciferase activity were measured 48 hours post-transfection using the Dual Luciferase Reporter Assay System. To normalise for transfection efficiency, the ratio of firefly to renilla luminescence was calculated and expressed as a percentage of the negative, scrambled control. Results represent the mean of at least three independent biological experiments, performed in quadruplicate; error bars represent standard error of the mean (SEM). * p < 0.05, ** <0.01 or p-value as stated.

5.4.3 miRNA Targeting to the 3'UTR of CYP11B2 mRNA.

Bioinformatic target site predictions on the *CYP11B2* mRNA were tested using the pEZX-B2 reporter construct. These experiments failed to confirm target-site predictions for miR-10a (Figure 5-4A), miR-10b (Figure 5-4B), miR-140-3p (Figure 5-4C), miR-22 (Figure 5-4F), miR-382 (Figure 5-5C) miR-423-5p (Figure 5-5D) or miR-638 (Figure 5-5E), none of which caused a significant change in luciferase activity when either overexpressed or inhibited in HeLa cells. However, in some circumstances (*e.g.* miR-22, miR-384 and miR-423-5p), large changes in luciferase activity were observed, but which failed to reach statistical significance, probably due to large standard deviations in the replicate results (Figure 5-4 and Figure 5-5).

In contrast, the results for miR-143 indicate that it does bind to the 3'UTR of *CYP11B2* (Figure 5-4D); increasing miR-143 significantly reduced luciferase activity to 79.03 \pm 4.86% (p = 0.499), while inhibiting miR-143 increased luciferase activity, 128.50 \pm 9.57% although this failed to reach statistical significance (p = 0.093). A similar pattern of results was observed for miR-185, with the Pre-miRTM significantly reducing luciferase activity (64.06 \pm 4.12%; p = 0.003), and the Anti-miRTM increasing luciferase activity (145.10 \pm 10.91%) but without reaching statistical significance (p = 0.054; Figure 5-4E).

Increasing miR-24 did not change the luciferase activity of the pEZX-B2 reporter construct in HeLa cells but inhibiting miR-24 with the Anti-miR^M molecule caused it to increase significantly to 162.80 ± 11.20% (p = 0.030, Figure 5-5A). Results also show that miR-34a overexpression reduce luciferase activity to 68.52 ± 7.20% (p = 0.485, Figure 5-5B) but there was no change in Anti-miR^M-transfected cells. A similar pattern was observed for miR-940 (Figure 5-5F) where Pre-miR^M reduced luciferase activity by a large margin (27.23 ± 5.23%; p = 0.0081) but Anti-miR^M did not achieve significance (123.00 ± 8.00; p = 0.1028).





HeLa cells were co-transfected with the pEZX-B1 construct and Pre-miR^m or Anti-miR^m molecules for the indicated miRNAs. Firefly and renilla luciferase activity were measured 48 hours post-transfection using the Dual Luciferase Reporter Assay System. To normalise for transfection efficiency, the ratio of firefly to renilla luminescence was calculated and expressed as a percentage of the negative, scrambled control. Results represent the mean of at least three independent biological experiments, performed in quadruplicate; error bars represent standard error of the mean (SEM). * p < 0.05, ** <0.01 or p-value as stated.



Figure 5-5. Effect of miRNAs on the Luciferase Activity of the pEZX-B2 Reporter Construct [2].

HeLa cells were co-transfected with the pEZX-B1 construct and Pre-miR^m or Anti-miR^m molecules for the indicated miRNAs. Firefly and renilla luciferase activity were measured 48 hours post-transfection using the Dual Luciferase Reporter Assay System. To normalise for transfection efficiency, the ratio of firefly to renilla luminescence was calculated and expressed as a percentage of the negative, scrambled control. Results represent the mean of at least three independent biological experiments, performed in quadruplicate; error bars represent standard error of the mean (SEM). * p < 0.05, ** <0.01 or p-value as stated.

5.4.4 miRNAs with Unique Target Sites on the CYP11B2 3'UTR.

In addition to testing those miRNAs which had putative miRNA binding sites in *CYP11B1* and *CYP11B2*, the four miRNAs which were predicted only to bind the *CYP11B2* 3'UTR (Figure 4-8) were also tested using the same system. The results of these experiments are shown in Figure 5-6.

Manipulating the levels of miR-134 or miR-495 did not significantly alter the luciferase activity of the pEZX-B2 reporter construct (Figure 5-6C-D). However, the existence of a miRNA binding site for miR-125a-5p and miR-125b on the *CYP11B2* 3'UTR was confirmed by these experiments; increasing the levels of miR-125a-5p in Pre-miR^M transfected cells reduced luciferase activity to a non-significant level of 63.78 ± 8.09% but decreasing miR-125a-5p in Anti-miR^M transfected cells significantly increased luciferase activity to 143.50 ± 9.02% (p = 0.040, Figure 5-6A). miR-125b Pre-miR^M transfected cells had significantly decreased luciferase activity (75.90 ± 1.28%; p = 0.033) and conversely, the Anti-miR^M transfected cells produced a significant increase in luciferase activity (148.70 ± 7.61; p = 0.023), in line with canonical miRNA action (Figure 5-6B).

A summary of the results for all miRNAs targeting the *CYP11B2* 3'UTR are shown in Table 5-4 (page 227).



Figure 5-6. Effect of *CYP11B2*-Specific Targeting miRNAs on the Luciferase Activity of the pEZX-B2 Reporter Construct.

HeLa cells were co-transfected with the pEZX-B1 construct and Pre-miR^m or Anti-miR^m molecules for the indicated miRNAs. Firefly and renilla luciferase activity were measured 48 hours post-transfection using the Dual Luciferase Reporter Assay Kit. To normalise for transfection efficiency, the ratio of firefly to renilla luminescence was calculated and expressed as a percentage of the negative, scrambled control. Results represent the mean of at least three independent biological experiments, performed in quadruplicate; error bars represent standard error of the mean (SEM). * p < 0.05.

5.4.5 Assessment of H295R Transfection with Pre-miR[™] and AntimiR[™] Molecules.

To investigate the action of miRNA on the full-length *CYP11B1* or *CYP11B2* mRNAs, the Pre-miR^M and Anti-miR^M molecules were transfected into the H295R adrenocortical cell line and their effects assessed by measuring the relative abundance of *CYP11B1* or *CYP11B2* mRNA and the levels of steroids secreted into the cell media.

First, it was necessary to assess the viability of using these small molecules and the suitability of the transfection conditions for H295R cells. A positive control experiment was performed and the levels of mature miRNA measured following transfection with three different concentrations of miR-24 Pre-miR^m or Anti-miR^m.

Figure 5-7 shows the levels of mature miR-24 measured by qRT-PCR following transfection of 1, 10 or 50 nM of the miR-24 Pre-miR^m or Anti-miR^m. A concentration-dependent increase in mature miR-24 was observed (Figure 5-7A). 1 nM increased the levels 50.24 ± 2.73-fold compared to control transfected cells but this did not reach statistical significance; 10 nM and 50 nM concentrations increased the level of mature miR-24 by 499.79 ± 67.46 (p < 0.05) and 1914.04 ± 225.22 (p < 0.001) fold, respectively. Similarly, the levels of mature miR-24 decreased in a concentration-dependent manner (1 nM: 0.705 ± 0.009; 10 nM: 0.027 ± 0.003; 50 nM: 0.023 ± 0.007, with results statistically different relative to the scrambled, negative control (Figure 5-7B).

The extent to which different miRNAs were overexpressed relative to their basal level was compared by transfecting H295R cells with Pre-miR[™] molecules for five different miRNAs and then measuring the levels of those miRNAs. Figure 5-8 shows the results for miR-10b, miR-125a-5p, miR-21, miR-24 and miR-143, all transfected at a final concentration of 50 nM. All miRNAs tested saw an increase, and there was a large range of fold-changes in the miRNA levels; the extent of this change may be influenced strongly by the basal level of the miRNA in H295R cells.

These results suggests that transfection of H295R cells with small molecules was successful and that all Pre-miR^M molecules are suitable for increasing their specific miRNAs and while Anti-miR^M molecules are capable of decreasing mature miRNA levels in a specific manner. The magnitude of change caused by Pre-miR^M transfection does vary depending on the individual miRNA species and the basal endogenous levels however, in general, there is a significant and large change with a 50 nM concentration of Pre-miR^M or Anti-miR^M.

To assess the ability of Pre-miR^m or Anti-miR^m to modulate mRNA targets, positive control experiments were set up to measure the effect of miR-1 Pre-miR^m transfection on its validated target, *PTK9* (Lim *et al.*, 2005), and of let-7c Anti-miR^m on its target, *HMGA2* (Lee and Dutta, 2007). Increasing the levels of miR-1 successfully decreased the level of *PTK9* mRNA abundance to 0.56 ± 0.05 fold (p = 0.004, Figure 5-9A), while decreasing the levels of let-7c increased *HMGA2* mRNA by 1.47 ± 0.06 fold (p = 0.014, Figure 5-9B). These results matched the expected results, as set out by the manufacturer, and confirmed that the transfection protocol is appropriate for assessing miRNA function in the H295R cell. It was therefore adopted for the subsequent experiments described in this chapter.



Figure 5-7. Mature miR-24 Levels Following Transfection with Various Concentrations of miR-24 Pre-miR[™] or Anti-miR[™].

H295R cells were transfected with (A) miR-24 Pre-miR[™] or (B) miR-24 Anti-miR[™] and equivalent scrambled negative controls at 1, 10 and 50 nM. Mature miR-24 levels were measured 48 hours post-transfection by qRT-PCR. Cycle threshold values were normalised to RNU48 mRNA and expressed relative to the control, scrambled Pre-miR[™] or Anti-miR Pre-miR[™], cells. Results represent the mean of one independent biological experiments performed in triplicate; error bars represent standard error of the mean (SEM). * p < 0.05, *** p < 0.001 compared to control.



Figure 5-8. Mature miRNAs Levels Post-Transfection in H295R Cells.

H295R cells were transfected with miR-10b, miR-125-5p, miR-21, miR-24 and miR-143 Pre-miR[™] and a scrambled negative control (final concentration 50 nM). Mature miRNA levels were measured 48 hours post-transfection by qRT-PCR. Cycle threshold values were normalised to RNU48 mRNA and expressed relative to the negative control cells. Results represent the mean of one independent biological experiments performed in triplicate; error bars represent standard error of the mean (SEM).



Figure 5-9. Verification of Pre-miR[™] and Anti-miR[™] Action Using Positive Control Target Genes.

H295R cells were transfected with (A) miR-1 Pre-miRTM, (B) let-7c Anti-miR Pre-miRTM or with a scrambled negative control (all final concentration 50 nM). (A) *PTK9* mRNA and (B) *HMGA2* mRNA levels were measured 48 hours post-transfection by qRT-PCR. Cycle threshold values were normalised to RNU48 mRNA and expressed relative to the control cells. Results represent the mean of three independent biological experiments, performed in triplicate; error bars represent standard error of the mean (SEM). * p < 0.05 and ** p < 0.01 compared to control.

5.4.6 Assessment of mRNA effect on CYP11B1 and CYP11B2 mRNA Abundance and Steroid Production.

Based on the results of the reporter construct studies described earlier in this chapter several miRNAs were chosen for further investigation in the H295R cell line, which is the best available *in vitro* model for adrenocortical tissue. The miRNA selected were miR-140-3p, miR-143, miR-185, miR-382, miR-24, miR-628 and miR-940.

miR-10b was also investigated using its specific Pre-miR^M molecule. As it did not change luciferase activity when tested on the pEZX-B1 or pEZX-B2 reporter construct, its use in H295R cells acted as an internal negative control to verify the reporter construct experiments. As expected, increasing miR-10b levels through in Pre-miR^M-transfected cells did not significantly change the levels of *CYP11B2* mRNA abundance (1.02 ± 0.12 fold, p = 0.88, Figure 5-10A) or *CYP11B1* mRNA abundance (1.10 ± 0.51, p = 0.58, Figure 5-10C); Neither aldosterone nor cortisol levels were significantly altered (Figure 5-10B and D). Therefore, both pEZX reporter constructs and H295R miRNA transfection experiments show that miR-10b does not regulate the *CYP11B1* or *CYP11B2* genes and provide evidence that these two experiment protocols provide consistent data for assessing miRNA action.

Due to some constraints (e.g. difficult cell line), many of these experiments were only performed once in triplicate; precise details can be found in the figure legends and further replication would be required to provide stronger evidence of miRNA action.

miR-140-3p Pre-miR^m transfected H295R cells had significantly reduced levels of *CYP11B2* mRNA (0.70 ± 0.03 fold of control transfected cells; p = 0.009) while Anti-miR^m transfected cells increased to *CYP11B2* mRNA 1.18 ± 0.05 fold (p = 0.095, Figure 5-11A). These mRNA changes were not supported by aldosterone levels, which were unchanged for both Pre-miR^m and Anti-miR^m cells (Figure 5-11B). The pattern of results for *CYP11B1* mRNA and cortisol production were inconsistent and, as no single result reached statistical significance, no definitive conclusion can be drawn (Figure 5-11C and D).

miR-143-transfected cells did not significantly alter *CYP11B2* or *CYP11B1* mRNA abundance, although there was a trend towards decreased *CYP11B2* levels in Pre-miR^M transfected cells (0.85 ± 0.03 fold; p = 0.055, Figure 5-12A). While cortisol levels reflected the lack of change in *CYP11B1* mRNA (Figure 5-12C and D), the aldosterone levels showed a trend towards increased production in Pre-miR^M transfected cells (1.61 ± 0.15, p = 0.077). This conflicts with the mRNA and reporter construct results, and with canonical miRNA action. Aldosterone production in Anti-miR^M transfected cells was not significantly different from control (Figure 5-12B).

H295R cells transfected with miR-185 Pre-miR^M or Anti-miR^M did not exhibit a significant change in either *CYP11B2* mRNA abundance or aldosterone production (Figure 5-13A and B). However, both the Pre-miR^M and Anti-miR^M transfected cells showed a trend towards decreased *CYP11B1* mRNA abundance (0.77 ± 0.04 fold; p = 0.055 and 0.58 ± 0.16 fold; p = 0.070, Figure 5-13C), with great variability in Anti-miR^M results. Cortisol production levels were unchanged with either molecule (Figure 5-13D). Further replicates would be required to definitively determine miR-185's effects.

The results of miR-24 Pre-miR[™] or Anti-miR[™] transfection into H295R cells support a role for miR-24 in the regulation of both CYP11B1 and CYP11B2 mRNA levels; the experiments involving miRNA were carried out three independent biological times each time in triplicate and the results are shown in Figure 5-14. Pre-miR[™] transfection reduced CYP11B2 levels to 0.58 ± 0.07 fold of control levels, which was significant (p = 0.026); moreover, this change was corroborated by a significant decrease in aldosterone production (0.78 \pm 0.02; p = 0.044). The opposite effect was observed with Anti-miR transfection: a significant increase in CYP11B2 mRNA abundance $(1.32 \pm 0.06 \text{ fold}; p = 0.035)$ and in aldosterone production $(1.22 \pm 0.03 \text{ fold}; p = 0.014)$, Figure 5-14A and B). Similar observations were made for CYP11B1 mRNA abundance and cortisol production: Pre-miR[™] transfected cells had significantly reduced CYP11B1 mRNA levels (0.638 \pm 0.06 fold; p = 0.037) and cortisol production (0.71 \pm 0.05 fold; p = 0.033). The Anti-miR[™] transfected cells had increased CYP11B1 mRNA levels $(1.38 \pm 0.05 \text{ fold}; p = 0.02)$ but no change in cortisol $(0.93 \pm 0.03 \text{ fold}; p = 0.144)$ Figure 5-14C and D).
The results for miR-382 were ambiguous; overall, the only significant result was a small increase in *CYP11B2* mRNA following Anti-miR^M transfection (1.19 ± 0.02 fold; Figure 5-15A). However, Pre-miR^M also increased *CYP11B2* abundance by a similar magnitude, although this was not statistically significant, due to a large variability in the replicates. The aldosterone results were also quite variable and, statistically, neither Pre-miR^M nor Anti-miR^M significantly changed aldosterone production in H295R cells (Figure 5-15B). There was also no significant change in *CYP11B1* mRNA or cortisol production (Figure 5-15C and D).

There was a small but significant increase in *CYP11B2* mRNA abundance following in miR-638 Pre-miR^M transfection of H295R cells (1.16 ± 0.03 fold; p = 0.047) together with a large increase in aldosterone production (1.63 ±0.21 fold) although this was not statistically significant (Figure 5-16A and B). The AntimiR^M transfected cells saw no significant changes in *CYP11B2* mRNA or aldosterone (Figure 5-16A and B). *CYP11B1* and cortisol production were also unaffected by miR-638 (Figure 5-16C and D).

Results following transfection of H295R cells with miR-940 Pre-miR^m and AntimiR^m molecules do support a regulatory effect on *CYP11B2* mRNA; levels changed to 0.79 ± 0.01 fold (p = 0.008) and 1.46 ± 0.05 fold (p = 0.009), respectively (Figure 5-17A and B). There was large variability in the aldosterone levels and no significant difference was seen in the Pre-miR^m transfected cells nor in Anti-miR^m transfected cells (1.47 ± 0.12 fold; p = 0.090, Figure 5-17A and B). The existence of a miR-940 binding site in *CYP11B1* 3'UTR was not confirmed by these transfection experiments, as there was no significant change in mRNA abundance or cortisol production following Pre-miR^m or Anti-miR^m transfection (Figure 5-17C and D).



Figure 5-10. Assessment of miR-10b Effects on *CYP11B1* and *CYP11B2* mRNA Levels and on Steroid Production.

H295R cells were transfected with miR-10b Pre-miR^M or with a scrambled negative control miRNA (Neg), at a final concentration of 50 nM. 48 Hours post-transfection, the abundance of *CYP11B2* (A) and *CYP11B1* (C) mRNA were measured by qRT-PCR. Cycle threshold values were normalised to *GAPDH* mRNA and expressed relative to the control cells. The levels of aldosterone (B) and cortisol (D) secretion were measured by LC:MS/MS. The results represent the mean of three independent biological experiments performed in triplicate; error bars represent standard error of the mean (SEM).



Figure 5-11. Assessment of miR-140-3p Effect on CYP11B1 and CYP11B2 mRNA Levels and on Steroid Production.

H295R cells were transfected with miR-140-3p Pre-miRTM, miR-140-3p Anti-miRTM or with a scrambled negative control (Neg), at a final concentration of 50 nM. 48 Hours post-transfection, the abundance of *CYP11B2* (A) and *CYP11B1* (C) mRNA were measured by qRT-PCR. Cycle threshold values were normalised to *GAPDH* mRNA and expressed relative to the control cells. The levels of aldosterone (B) and cortisol (D) secretion were measured by LC:MS/MS. Results represent the mean of one biological experiment performed in triplicate; error bars represent standard error of the mean (SEM). ** p < 0.01 compared to negative control.



Figure 5-12. Assessment of miR-143 Effect on *CYP11B1* and *CYP11B2* mRNA Levels and on Steroid Production.

H295R cells were transfected with miR-143 Pre-miR^M, miR-143 Anti-miR^M or with a scrambled negative control (Neg), at a final concentration of 50 nM. 48 Hours post-transfection, the abundance of *CYP11B2* (A) and *CYP11B1* (C) mRNA were measured by qRT-PCR. Cycle threshold values were normalised to *GAPDH* mRNA and expressed relative to the control cells. The levels of aldosterone (B) and cortisol (D) secretion were measured by LC:MS/MS. Results represent the mean of one biological experiment performed in triplicate; error bars represent standard error of the mean (SEM). Where appropriate, p-values are as stated and are relative to negative control values.



Figure 5-13. Assessment of miR-185 Effect on *CYP11B1* and *CYP11B2* mRNA Levels and on Steroid Production.

H295R cells were transfected with miR-185 Pre-miR^M, miR-185 Anti-miR^M or with a scrambled negative control (Neg), at a final concentration of 50 nM. 48 Hours post-transfection, the abundance of *CYP11B2* (A) and *CYP11B1* (C) mRNA were measured by qRT-PCR. Cycle threshold values were normalised to *GAPDH* mRNA and expressed relative to the control cells. The levels of aldosterone (B) and cortisol (D) secretion were measured by LC:MS/MS. Results represent the mean of one biological experiment performed in triplicate; error bars represent standard error of the mean (SEM). Where appropriate, p-values are as stated and are relative to negative control values.



Figure 5-14. Assessment of miR-24 Effect on *CYP11B1* and *CYP11B2* mRNA Levels and on Steroid Production.

H295R cells were transfected with miR-24 Pre-miR^M, miR-24 Anti-miR^M or with a scrambled negative control (Neg), at a final concentration of 50 nM. 48 Hours post-transfection, the abundance of *CYP11B2* (A) and *CYP11B1* (C) mRNA were measured by qRT-PCR. Cycle threshold values were normalised to *GAPDH* mRNA and expressed relative to the control cells. The levels of aldosterone (B) and cortisol (D) secretion were measured by LC:MS/MS. Results represent the mean of three biological experiments performed in triplicate; error bars represent standard error of the mean (SEM). * p < 0.05 compared to negative control.



Figure 5-15. Assessment of miR-382 Effect on *CYP11B1* and *CYP11B2* mRNA Levels and on Steroid Production.

H295R cells were transfected with miR-382 Pre-miRTM, miR-382 Anti-miRTM or with a scrambled negative control (Neg), at a final concentration of 50 nM. 48 Hours post-transfection, the abundance of *CYP11B2* (A) and *CYP11B1* (C) mRNA were measured by qRT-PCR. Cycle threshold values were normalised to *GAPDH* mRNA and expressed relative to the control cells. The levels of aldosterone (B) and cortisol (D) secretion were measured by LC:MS/MS. Results represent the mean of one biological experiment performed in triplicate; error bars represent standard error of the mean (SEM). * p < 0.05 compared to negative control.



Figure 5-16. Assessment of miR-638 Effect on *CYP11B1* and *CYP11B2* mRNA Levels and on Steroid Production.

H295R cells were transfected with miR-638 Pre-miRTM, miR-638 Anti-miRTM or with a scrambled negative control (Neg), at a final concentration of 50 nM. 48 Hours post-transfection, the abundance of *CYP11B2* (A) and *CYP11B1* (C) mRNA were measured by qRT-PCR. Cycle threshold values were normalised to *GAPDH* mRNA and expressed relative to the control cells. The levels of aldosterone (B) and cortisol (D) secretion were measured by LC:MS/MS. Results represent the mean of one biological experiment performed in triplicate; error bars represent standard error of the mean (SEM). Where appropriate p values are stated, or * represents p < 0.05 relative to negative control.



Figure 5-17. Assessment of miR-940 Effect on *CYP11B1* and *CYP11B2* mRNA Levels and on Steroid Production.

H295R cells were transfected with miR-940 Pre-miRTM, miR-940 Anti-miRTM or with a scrambled negative control (Neg), at a final concentration of 50 nM. 48 Hours post-transfection, the abundance of *CYP11B2* (A) and *CYP11B1* (C) mRNA were measured by qRT-PCR. Cycle threshold values were normalised to *GAPDH* mRNA and expressed relative to the control cells. The levels of aldosterone (B) and cortisol (D) secretion were measured by LC:MS/MS. Results represent the mean of one biological experiment performed in triplicate; error bars represent standard error of the mean (SEM). Where appropriate p values are stated, or ** represents p < 0.01 relative to negative control.

5.4.7 Effect of CYP11B2-Specific miRNAs on CYP11B2 mRNA Abundance.

The results from the pEZX reporter construct studies described earlier in this chapter supported a role for a miR-125a-5p and miR-125b binding site on the 3'UTR of CYP11B2. To test this further, the Pre-miR^M or Anti-miR^M molecules corresponding to these miRNAs were transfected into H295R cells. The results for three independent biological replicate experiments are shown in Figure 5-18.

Firstly, results confirm that the action of miR-125a-5p on *CYP11B2* is typical of that exerted by miRNAs, with increasing miRNA levels leading to decreased *CYP11B2* mRNA abundance (0.70 \pm 0.001 fold, p = 0.004) and decreasing miRNA levels leading to significant increase in *CYP11B2* mRNA abundance (1.53 \pm 0.05 fold, p = 0.009, Figure 5-18A). Similarly, *CYP11B2* mRNA levels decreased to 0.66 \pm 0.02 fold (p = 0.033) in Pre-miRTM transfected cells and increased by 1.58 \pm 0.09 fold however, this failed to attain statistical significance (p = 0.101, Figure 5-18B).

A summary of all of the miRNA results detailed in this chapter is provided in Table 5-3 and Table 5-4.



Figure 5-18. Assessment of miR-125a-5p and miR-125b Effect on CYP11B2 mRNA Levels.

H295R cells were transfected with Pre-miRTM or Anti-miRTM molecules for miR-125a-5p (A), miR-125b (B) or with a scrambled negative control (Neg), at a final concentration of 50 nM. 48 Hours post-transfection, the abundance of *CYP11B2* mRNA was measured by qRT-PCR. Cycle threshold values were normalised to *GAPDH* mRNA and expressed relative to the control cells. The results represent the mean of two biological experiment performed in triplicate; error bars represent standard error of the mean (SEM). * p < 0.05, ** p < 0.01 compared to negative control.

	Seed Site	3'U	TR	mR	NA	Cortisol	
		Pre-miR	Anti-miR	Pre-miR	Anti-miR	Pre-miR	Anti-miR
miR-10a	7mer-1A	\leftrightarrow	\leftrightarrow	-	-	-	-
miR-10b	7mer-1A	\leftrightarrow	\leftrightarrow	\leftrightarrow	\leftrightarrow	\leftrightarrow	\leftrightarrow
miR-140-3p	7mer-m8	\leftrightarrow	\leftrightarrow	\leftrightarrow	\leftrightarrow	\leftrightarrow	\leftrightarrow
miR-143	7mer-1A	\leftrightarrow	\leftrightarrow	\leftrightarrow	\leftrightarrow	\leftrightarrow	\leftrightarrow
miR-185	7mer-m8	\downarrow	\leftrightarrow	\downarrow	\leftrightarrow	\leftrightarrow	\leftrightarrow
miR-22	7mer-m8	↔	↔	-	-	-	-
miR-24	Offset 6mer	\leftrightarrow	Ť	\downarrow	1	Ļ	\leftrightarrow
miR-34a	7mer-m8	↔	\leftrightarrow	-	-	-	-
miR-382	Offset 6mer	Ļ	Ť	\leftrightarrow	\leftrightarrow	\leftrightarrow	\leftrightarrow
miR-423-5p	Offset 6mer	1	\leftrightarrow	-	-	-	-
miR-638	7mer-1A	\leftrightarrow	Ť	\leftrightarrow	\leftrightarrow	\leftrightarrow	\leftrightarrow
miR-940	8mer	\downarrow	\leftrightarrow	\leftrightarrow	\leftrightarrow	\leftrightarrow	\leftrightarrow

Table 5-3. Summary of miRNA Target-Site Validation Experiments for the CYP11B1 Gene.

Dashes (-): unmeasured parameters; unchanged arrows (↔): non-statistically significant changes.

		3'UTR		mR	RNA	Aldosterone		
MIRNA	Seed Site	Pre-miR	Anti-miR	Pre-miR	Anti-miR	Pre-miR	Anti-miR	
miR-10a	7mer-1A	\leftrightarrow	\leftrightarrow	-	-	-	-	
miR-10b	7mer-1A	\mapsto	\leftrightarrow	\leftrightarrow	\leftrightarrow	\leftrightarrow	\leftrightarrow	
miR-140-3p	7mer-m8	\leftrightarrow	\leftrightarrow	↓	\leftrightarrow	\leftrightarrow	\leftrightarrow	
miR-143	7mer-1A	↓ ↓	$\downarrow \uparrow \downarrow \downarrow \leftrightarrow$		\leftrightarrow	↔		
miR-185	7mer-m8	Ļ	↑	\leftrightarrow	\leftrightarrow	\leftrightarrow	\leftrightarrow	
miR-22	7mer-m8	1	↔	-	-	-	- ↑ ↑ -	
miR-24	Offset 6mer	\leftrightarrow	↑	Ļ	1	↓ ↓		
miR-34a	7mer-m8	↓ ↓	\leftrightarrow	-	-	-		
miR-382	-	Ļ	$\downarrow \leftrightarrow \leftrightarrow \uparrow$		\leftrightarrow	↔		
miR-423-5p	-	Ť	\leftrightarrow			-	-	
miR-638	7mer-1A	\leftrightarrow	\leftrightarrow	1	\leftrightarrow	\leftrightarrow	\leftrightarrow	
miR-940	8mer	\downarrow	\leftrightarrow	\downarrow	Ť	\leftrightarrow	\leftrightarrow	
miR-125a-5p	8mer	\leftrightarrow	1	\downarrow	1	-	-	
miR-125b	8mer	↓	↑	\downarrow	\leftrightarrow	-	-	
miR-134	7mer-m8	\leftrightarrow	\leftrightarrow	-	-	-	-	
miR- 495	7mer-m8	\leftrightarrow	\leftrightarrow	-			-	

Table 5-4. Summary of miRNA Target-Site Validation Experiments for the CYP11B2 Gene.

Dashes (-): unmeasured parameters; unchanged arrows (↔): non-statistically significant changes.

5.4.8 Investigating miRNA Clustered with miR-24.

The above results suggest a role for miR-24 in regulating the *CYP11B1* and *CYP11B2* mRNA levels and in steroid production. miR-24 is transcribed in an intronic cluster from a chromosomal location on chromosome 9, *C9orf3*, along with two other miRNA, miR-23b and miR-27b. These miRNAs are also expressed in the adrenal gland (Figure 4-6), with miR-23b predicted to bind the *CYP11B1* 3'UTR; miR-27b was not predicted to bind *CYP11B1* or *CYP11B2*. To test whether the individual components of this cluster act together to regulate the *CYP11B1* or *CYP11B2* genes, miR-23b and miR-27b were analysed using the pEZX reporter constructs in HeLa cells. The results of three independent biological experiments are shown in Figure 5-19.

Neither of these other two clustered miRNAs bind to and regulate *CYP11B1* or *CYP11B2* by canonical miRNA action. However, they do appear to be altering the luciferase activity of the reporter constructs; in miR-23b Anti-miR^m transfected cells, there was no significant change in luciferase activity (Figure 5-19A and B) but there was a significant increase in luciferase activity with the miR-23b Pre-miR^m transfected cells for both the pEZX-B2 (226.10 ± 29.29%; p = 0.049, Figure 5-19A) and pEZX-B1 reporter construct (144.00 ± 5.87%; p = 0.017, Figure 5-19B).

A similar trend was observed for miR-27b; the luciferase activity in Anti-miR^M transfected cells was not significantly different in pEZX-B2 cells and was marginally decreased in pEZX-B1 cells (83.31 ± 0.10%). This was statistically significant (p = 0.003) largely due to the small error associated with these results (Figure 5-19C and D). Furthermore, Pre-miR^M transfection increased the luciferase activity of both constructs: pEZX-B2 to 159.70 ± 6.78% (p = 0.013) and pEZX-B1 to 167.20 ± 19.22% (p = 0.177). While this result is not statistically significant, it is an increase of large magnitude (Figure 5-19C and D). The results for miR-24 are presented again in Figure 5-19E and F for comparison.





HeLa cells were co-transfected with the pEZX-B1 (A, C, E) or pEZX-B2 (B, D, F) construct and Pre-miR^M or Anti-miR^M molecules for miR-23b (A-B), miR-27b (C-D) or miR-24 (E-F). Firefly and renilla luciferase were measured 48 hours post-transfection using the Dual Luciferase Reporter Assay System To normalise for transfection efficiency, the ratio of firefly to renilla luminescence was calculated and expressed as a percentage of the negative scrambled control. Results represent the mean of at least three independent biological experiments, performed in quadruplicate; error bars represent standard error of the mean (SEM). * p < 0.05, ** p < 0.01.

5.4.9 Investigation of the miRNA Binding Sites of miR-24, miR-125a-5p and miR-125b.

In addition to other findings, this chapter has identified a regulatory effect of miR-24 on both the *CYP11B1* and *CYP11B2* 3'UTRs and of miR-125a-5p and miR-125b on the *CYP11B2*. The sequence of the 3'UTR binding site and the mature miRNA were entered into in the RNAHybrid Secondary Structure Software, which has been specifically adapted to predict secondary structure of miRNA:mRNA interactions and to give greater detail regarding these binding sites (Kruger and Rehmsmeier, 2006).

miR-24 was predicted to target the *CYP11B2* gene by only one database, miRviewer, which also predicted the same site on the *CYP11B1* gene (Table 4-9 and Table 4-10); this site is an offset 6mer site spanning bases 489 to 502 of the 3'UTR (Figure 5-20A) with a predicted minimum free energy (MFE) of -22.1 kcal/mol. A second miR-24 site was predicted by microRNAs.org at bases 1206 to 1228 of the *CYP11B1* 3'UTR (Table 4-10). This site is a 7mer-m8 site, but with a mismatch at nucleotide number 6 (Figure 5-1B); the MFE for this predicted pairing is -27.3 kcal/mol.

Two databases predicted miR-125a-5p and miR-125b to bind to *CYP11B2* 3'UTR. Each was predicted to have the highest complementarity in the seed region (8mer) and furthermore, the region of the 3'UTR to which they bind was the same for both miRNAs (bases 1,200 to 1,223). The predicted secondary structure for both miRNA:mRNA interactions are shown in Figure 5-21 with the MFE for miR-125a-5p being -24.0 kcal/mol, and -28.0 kcal/mol for miR-125b.



Figure 5-20. Predicted Secondary Structure of miR-24 Target Sites in the CYP11B1 and CYP11B2 3'UTRs.

Secondary structure and base-complementarity of miR-24 (green) binding to 3'UTR of *CYP11B1* or *CYP11B2* (red). Blue boxes indicate the 6mer seed region and extensions indicate additional target recognition feature: complementary base-pairing and the 8th nucleotide. Site 1 (A) represents the binding of miR-24 to both the *CYP11B1* and *CYP11B2* 3'UTR. Site 2 (B) only exists in the *CYP11B1* 3'UTR. Structures predicted by RNAHybrid (Kruger and Rehmsmeier, 2006)



Figure 5-21. Predicted Secondary Structure of miR-125a-5p and miR-125b Target Sites in the *CYP11B2* 3'UTR.

Secondary structure and base-complementarity of miR-125a-5p (A) and miR-125b (B) (green) binding to the *CYP11B2* 3'UTR (red). Blue boxes indicate the 6mer seed region and extensions indicate additional target recognition features: an adenosine base at nucleotide 1 on the 3'UTR and the complementary base-pairing and the 8th nucleotide. Structures predicted by RNAHybrid (Kruger and Rehmsmeier, 2006).

5.5 Discussion

The experiments presented in this chapter investigated which of the miRNA target sites in the 3'UTR of *CYP11B1* and *CYP11B2* as predicted in Chapter 4 are regulated by the adrenal miRNAs indentified in Chapter 3. Two primary methods were employed in the present chapter: 3'UTR reporter construct studies, and investigation of gain and loss of miRNA function in an adrenocortical cell line. The merits of these methods and the findings of this chapter, including the identification for the first time of miRNAs which directly regulate the *CYP11B1* and/or the *CYP11B2* genes, will be discussed below.

The first stage of miRNA target site validation utilised a 3'UTR reporter plasmid, which coupled a viral promoter-driven firefly luciferase coding region to the full length 3'UTR of the either CYP11B1 or CYP11B2 gene. This plasmid also contained a separate renilla luciferase reporter gene for the assessment of transfection efficiency. It was, therefore, not necessary to co-transfect a separate control reporter plasmid, which may have caused transfection efficiency problems, particularly as small molecules were also required to be transfected. By inserting the full length 3'UTR, rather than smaller sections that only encompass the predicted miRNA-binding region, the results of these experiments give a true reflection of miRNA:mRNA interaction and will, to a degree, account for the secondary structure present in this region (Rehmsmeier et al., 2004). The HeLa cell line was used to evaluate miRNA binding; this was chosen as it is a well established cell line that does not change dramatically over time, is easy to transfect, has been used in other miRNA target site validation experiments and does not endogenously express the CYP11B1 or CYP11B2 genes, so miRNA target site competition would not be a problem for assessing miRNA action (Lim *et al.*, 2005).

The final experimental tools used in this chapter were the Pre-miR^m and AntimiR^m molecules, which were purchased from Applied Biosystems. These are an established tool for use in miRNA investigation (Cheng *et al.*, 2005); similar types of molecule are available from other suppliers. The molecules are single stranded, so there is no risk of passenger strand expression and they are highly specific, which is important because miRNAs often only differ by one or two bases and, if specificity is not high, there may be off-target effects. Furthermore, as shown in this chapter, the actions of either the Pre-miR^M or Anti-miR^M molecules can be directly assessed by measuring the levels of mature miRNAs post-transfection. This chapter has shown both that Pre-miR^M and Anti-miR^M were highly effective at modulating the levels of specific miRNAs. The change in miRNA levels were very large and probably not representative of the change observed in physiological circumstances however, as is often the case in *in vitro* model systems, a supraphysiological change is required to investigate a pathway. Moreover, the degree of change measured for several miRNAs was vastly different yet, in all cases the increase in miRNA expression would have been sufficient to exert a measurable level of regulation. These molecules offer the best method to consistently modulate the levels of a specific miRNA in *vitro* and are suitable for measuring changes in mRNA, as shown by changes in *PTK9* or *HMGA2* mRNA abundance.

This chapter aimed to test the actions of the sixteen miRNAs predicted to have binding sites in the *CYP11B2* 3'UTR; twelve of these miRNAs also have a predicted binding site in *CYP11B1* so they were also tested for regulation that gene. During the course of this project, the bioinformatic target site predictions have been updated and now include two additional miRNAs that are predicted to bind to both genes: miR-28-5p and miR-34c-3p. Unfortunately, these miRNA were not investigated in this project and will need to be studied at a later date. Furthermore, another two miRNAs, miR-382 and miR-423-5p, are now only predicted to bind *CYP11B1* although, they have been tested against the *CYP11B2* gene. The updated list of adrenal miRNAs predicted to bind *CYP11B1* and *CYP11B2* is given in Figure 4-8.

The first phase of testing involved modulating the level of miRNAs and evaluating the response using a 3'UTR reporter construct. The results for the *CYP11B1* 3'UTR construct show that seven miRNAs either do not bind to the 3'UTR, or bind and cause effects that are not typical of the repressive miRNA action. The other five miRNAs showed promising results, suggestive of 3'UTR binding and consistent with canonical miRNA action. The influence of these miRNAs on endogenous *CYP11B1* mRNA abundance and steroid production was tested in the H295R cell line. Except for the miR-24 experiments, the other miRNA experiments were performed one biological time, with three technical

replicates. The actions of three out of the five miRNAs (miR-382, miR-638 and miR-940) were not verified in H295R cells. miR-185 and miR-24 were the only two miRNAs for which the results indicated a regulatory role. The magnitude of miR-185 action was small and will require replication to verify and the action of miR-24 will be discussed below.

A similar strategy was adopted for testing sixteen miRNAs for *CYP11B2* regulation; eight miRNAs were ruled out after reporter construct studies. Increasing miR-34a levels in HeLa cells did indicate miR-34a binding of the *CYP11B2* 3'UTR, but this miRNA was not investigated further in this project. The mRNA or steroid results for miR-185 did not verify the miRNA regulation that was observed with the reporter construct. The *CYP11B2* mRNA showed a trend towards regulation by miR-143 but aldosterone production was not consistent with this; more experimental replicates are required to understand the role of miR-143. Similarly, the results from H295R cells did not substantiate the reporter construct results indicating a role for miR-382. Given that this miRNA does not possess a putative binding site in the 3'UTR, according to the latest predictions, then the results support the current bioinformatic information.

The results from both validation experiments support a regulatory role for four miRNAs acting on the *CYP11B2* gene; three were tested comprehensively and the implications of these will be discussed below. The other one was miR-940, which consistently demonstrated regulation of *CYP11B2*, including actions at the 3'UTR and modulation of mRNA abundance. Unfortunately, only one experiment in H295R cells was performed but, with further replicates, it would be anticipated that these results would be validated and that the levels of aldosterone would be modulated accordingly.

Of the four miRNAs with predicted sites in *CYP11B2*, (miR-134 and miR-495), did not bind to the 3'UTR of *CYP11B2* and modulate luciferase activity so were not verified by this study. However, the results suggest that the other two mRNAs, miR-125a-5p and miR-125b, are genuine regulators of *CYP11B2* mRNA; both miRNAs decreased mRNA abundance when over-expressed in H295R cells. Inhibiting miR-125a-5p also increased the levels of mRNA significantly. The reporter construct studies confirmed that this regulation occurs through binding of the 3'UTR. These miRNAs are from the same family and therefore share similar sequences; they are predicted to bind to the same seed site on the *CYP11B2* 3'UTR. This target site was predicted by two databases and predicted to have the highest complementarity in the seed region, 8mer. The magnitude of change in mRNA abundance is equivalent to that previously observed and is consistent with other reported results from miRNA targets with only one seed site in a 3'UTR (Selbach *et al.*, 2008). The location of the target site is near to the Poly (A) tail of the 3'UTR; miRNAs with target sites located at the proximal or distal end of the 3'UTR are likely to more effectively target and cause repression than those targeting central regions (Grimson *et al.*, 2007). The secondary structure predicted for miR-125b has a lower MFE than that of miR-125a-5p, which may suggest that this miRNA interacts with the mRNA preferentially (Rehmsmeier *et al.*, 2004). The value of the MFE is comparable to the preferential binding sites identified for let-7 targeting of *lin-14*, a well-verified miRNA:mRNA interaction (Reinhart *et al.*, 2000;Rehmsmeier *et al.*, 2004).

These miRNAs are expressed in a wide range of tissue types (Lagos-Quintana *et al.*, 2002) and are transcribed from separate chromosomal locations: miR-125a-5p from chromosome 11, and miR-125b from chromosome 19. miR-125a is transcribed alongside two other miRNAs, let-7e and miR-99b, which are also expressed within the adrenal gland (Figure 4-6) but were not predicted to bind to the *CYP11B2* 3'UTR. However, in addition to miR-125a-5p regulating *CYP11B2*, these other miRNAs may regulate other genes, providing a novel adrenal regulatory network.

Due to their sequence similarity, miR-125a and miR-125b are frequently studied together. miR-125a-5p and miR-125b have both been shown to repress two members of the ERBB oncogene family (*ERBB2* (HER2) and *ERBB3* (HER3)). Targeting of these genes in a breast cancer cell line, by exogenous overexpression of miR-125a-5p or miR-125b, reduced cell growth and migration (Scott *et al.*, 2007). Furthermore, they have both been shown to regulate the levels of endothelin-1 (ET-1) in vascular endothelial cells by targeting preproET-1 at its 3'UTR (Lia *et al.*, 2010). Therefore, co-regulation of the *CYP11B2* gene by miR-125a-5p and miR-125b, as described in this chapter, is not unusual. The interplay between these miRNAs has not been established; it would be interesting to establish whether they act synergistically or competitively.

Increasing the levels of both of these miRNA in H295R cells or with a reporter construct may help to answer these questions.

Finally, miR-125b has been implicated in cardiac hypertrophy. It was found to be upregulated in a mouse model of cardiac hypertrophy, when compared with the hearts of control animals, although the authors failed to directly correlate increased expression of miR-125b with a target gene or pathway (van Rooij *et al.*, 2006). This miRNA has been implicated in regulation of cell proliferation and its expression is down regulated in a range of cancer tissues and cancer cell lines (Lee *et al.*, 2005;lorio *et al.*, 2005). Therefore, in adrenal cells miR-125b may have target multiplicity and control several pathways.

The only miRNA which was identified in this chapter as a regulator of both *CYP11B1* and *CYP11B2* mRNA abundance and of steroid production was miR-24. This miRNA targets one site that is common to both genes: an offset 6mer site with an additional G:U wobble pairing at base 2, which can be tolerated in miRNA seed sites (Bartel, 2009). There is a further 7mer-m8 target site unique to the *CYP11B1* 3'UTR, although this has a 7mer-m8 with a mismatch in the seed region. However, this site also contains a large degree of base-pairing complementarity to the centre region and 3' end of the miRNA, which would most likely compensate for this mismatch (Shin *et al.*, 2010).

The reporter construct studies confirmed that inhibiting the levels of miR-24 caused an increase in luciferase activity. However, the corresponding decrease was not observed when miR-24 was over expressed. One possible explanation is that high endogenous levels of miR-24 in HeLa cells may already occupy the available miR-24 targets sites and repress the construct to a maximal level (Ritchie *et al.*, 2010). Nonetheless, experiments with the Anti-miR^m molecule confirm that miR-24 regulation of the *CYP11B1* and *CYP11B2* genes occurs at the 3'UTR and investigations in H295R cells were consistent with this, showing that miR-24 acts to regulate both of these genes in a canonical miRNA manner.

miR-24 is transcribed from two genomic locations, known as miR-24-1 (intronic region on Chromosome 9) and miR-24-2 (intergenic region on Chromosome 19). These miRNAs have identical mature sequences but different primary transcripts; therefore they have similar biological function but may have

different regulation and expression patterns (Chhabra *et al.*, 2010). Each miRNA is transcribed as part of a triplet cluster that includes miR-23a and miR-27a (cluster 2) or miR-23b and miR-27b (cluster 1). The sequences of the miR-23 and miR-27 families are not identical and vary by one or two bases, meaning their mRNA targets will be different. The members of the miR-24-1 cluster were also tested for their action on *CYP11B1* and *CYP11B2* (miR-23b has a putative binding site on *CYP11B1*), but neither miRNA demonstrated canonical miRNA action. Both caused an increase in luciferase activity when over expressed in HeLa cells but the meaning of this is unclear; miR-27b is not thought to have a site on either 3'UTR and miR-23b only on *CYP11B1*, so these miRNA may be having an indirect effect that subsequently increases the transcription of luciferase.

The implications of novel regulation by miR-24 are still to be identified. It may have a role in maintaining *CYP11B1* and *CYP11B2* mRNA levels at baseline levels, or changes in the level of miR-24, for example in response to AngII, may allow for increases in *CYP11B1* and *CYP11B2* expression, thus increasing aldosterone and/or cortisol production. The members of both of the miR-24 clusters have been reported to be dysregulated in several pathologies, including several types of cancer, although there does not appear to be a consistent pattern of expression, with the miRNA clusters sometimes being increased and, at other times, reduced in cancer samples (Volinia *et al.*, 2006;Saumet *et al.*, 2009).

Previous studies have identified a role for miR-24 in the regulation of several cellular pathways (Chhabra *et al.*, 2010). The action of miR-24 appears to be cell-type specific, as it has been reported to regulate opposing cell mechanisms, such as cell proliferation and cell apoptosis (Lal *et al.*, 2008;Qian *et al.*, 2011). The majority of validated targets for miR-24 are related to regulation of the cell cycle (Chhabra *et al.*, 2010), which may explain the observed differential expression in tumour tissue. However, in this chapter miR-24 has been shown to regulate *CYP11B1* and *CYP11B2* mRNA levels in normal conditions. Whether adrenal miR-24 has a dual function and contributes to adrenal plasticity remains to be seen.

The results presented in this chapter highlight some difficulties in assessing miRNA function. Because miRNAs are promiscuous, altering the levels of only one miRNA may not represent the true physiological situation. miRNAs are

hypothesised to only modulate mRNA levels by small amounts, which may be difficult to measure unless specific and sensitive methods, such as those employed in this chapter, are used. Moreover, the results highlight the shortcomings of bioinformatic miRNA target prediction algorithms: very few of the predicted miRNAs could be validated as the target of either gene by either of the two methods used here. However, the techniques chosen and developed in this chapter are frequently used to assess miRNA targeting and they were carefully verified to ensure a good level of accuracy. They proved successful at validating some miRNA target predictions and at discounting others. Other strengths of this chapter include the use of the H295R cell line; this is a human cell line which maintains basal steroid secretion. Assessing miRNA function in these cells is the best available model for studying steroid production in the human adrenal gland. Target validation could be taken further by mutating the seed region in the 3'UTR construct and assessing whether this disrupts miRNA function. Other more sophisticated techniques include HITS-CLIP (high throughput sequencing of cross-linking immunoprecipitation), which offers the advantage of assessing the action of endogenous miRNA on endogenous mRNA targets (Chi et al., 2009). Further, modified versions of SILAC (stable isotope labelling with amino acids in cell culture), including pulsed SILAC, have been developed to monitor changes in novel protein production following miRNA transfection. This method allows for many targets of a single miRNA to be identified (Selbach et al., 2008).

In summary, these studies have investigated putative miRNA binding sites in the *CYP11B1* and *CYP11B2* 3'UTRs. This chapter has disproved some of the bioinformatic prediction and found several miRNAs to be novel regulators of *CYP11B1* mRNA and cortisol production and/or of *CYP11B2* mRNA and aldosterone production. miR-24 is the best example of a miRNA regulating the expression of these genes in an adrenal cell line. Regulation of miR-24 expression and its function under stimulated or pathophysiological conditions may prove interesting in circumstances of altered adrenal function. In conclusion, the results presented in this chapter show for the first time that the *CYP11B1* and *CYP11B2* genes are directly regulated by miRNAs.

6 Altered Adrenal-miRNA Expression and Function.

6.1 Introduction

Altered miRNA expression and/or disrupted miRNA regulation have been implicated in the aetiology of several pathologies (Section 1.7.1 and 1.7.2); investigating the influence of miRNA regulation and expression on adrenal-related disease will be the focus of the studies in this chapter.

There are numerous ways in which miRNAs may impact on disease. For example, they may be important in disease initiation and progression, they may contribute the symptoms of a disease and, additionally, aberrant miRNA expression may occur as a result of the pathology itself. A single miRNA could contribute to more than one of the deleterious effects in a pathological state. For example, miRNAs have been reported to affect cell cycle regulation and vascularisation of tissue, and also have direct effects on cell-specific genes. Finally, because normal physiological miRNA regulation is required to moderate mRNA stability by only small amounts, then any change in miRNA abundance could be associated with subtle changes in cell function and phenotype, such as increased aldosterone production and small changes in blood pressure.

Both relative miRNA expression and target site sequence can influence miRNAmediated regulation. These changes can be divided into *cis*-factors (such as polymorphisms at the genomic loci encoding miRNAs, epigenetic modifications and chromosomal alterations) or by *trans*-factors (*e.g.* polymorphisms that disrupt established miRNA target sites or create novel ones, or polymorphisms in the genes encoding proteins involved in miRNA processing and maturation) (Sethupathy and Collins, 2008;Ryan *et al.*, 2010). A specific example of a *cis*acting polymorphism and a *trans*-acting polymorphism is given in Section 1.5.2.2 and Section 1.7.2, respectively.

The initial studies in this chapter aim to explore the role of miRNAs in the aetiology of aldosterone-producing adenoma (APA), one of the common types of primary aldosteronism (PA) (Young, 2007) (Section 1.4.1.1). The cause of these spontaneous masses is unknown but they occur more frequently in patients with the -344T *CYP11B2* promoter allele (Inglis *et al.*, 2001).

In addition to the miRNA expression data for the normal adrenal gland presented in Chapter 4, a small number of studies have investigated the role of miRNAs in adrenal samples of various origins (Tombol et al., 2009;Soon et al., 2009; Iliopoulos et al., 2009; Bimpaki et al., 2010; Schmitz et al., 2011). However, only two investigated miRNA expression in adrenocortical adenoma and only a subset of these samples were classified as Conn's tissue (Schmidt et al., 2003; Soon et al., 2009). Moreover, the coverage of miRNAs and access to data presented in these studies are limited. Therefore, further studies were warranted. To this end, the miRNA expression profile of APA samples was determined by miRNA microarray, the data-set generated being comparable to the comprehensive normal adrenal gland miRNA expression profile (Chapter 4). Currently, the cause and regulation of increased CYP11B2 expression in APAs is unknown (Gomez-Sanchez and Gomez-Sanchez, 2010); the information gained from performing this microarray experiment will be useful in the identification of miRNAs involved in neoplastic adrenal growth and will also, in conjunction with earlier studies presented in this thesis, contribute to our understanding of miRNA regulation of CYP11B2 mRNA.

The latter studies in this chapter will be concerned with identifying genetic variation in the 3'UTR of the *CYP11B1* and *CYP11B2* genes and investigating how this may influence miRNA-binding. One hypothesis is that such polymorphisms may lie in miRNA binding sites and, hence, alter miRNA-mediated regulation of these genes under basal conditions, thereby leading to changes in the levels of corticosteroid production and contributing to the aetiology of essential hypertension.

6.2 Aims

The aims of this study were to identify whether miRNA-mediated regulation of corticosteroidogenic genes contribute to the development or the phenotype of adrenal pathology and to determine and investigate genetic variation in the 3'UTR of the *CYP11B1* and *CYP11B2* genes, to compare the miRNA expression in aldosterone-producing adenoma samples to that of the normal adrenal gland.

6.3 Methods

6.3.1 miRNA Expression Profiling of Aldosterone-Producing Adenomas

Total RNA was isolated from four formalin-fixed paraffin-embedded aldosteroneproducing adenoma tissue samples (section 2.2.2) and quantified as described in Sections 2.2.4 and 2.2.5. Samples were shipped to the microarray provider (LC Sciences, Houston, Texas) and the array experiment was performed (Section 2.3). A background threshold cut-off of 500 arbitrary units was used, as recommended by the array chip manufacturer. miRNAs expressed at levels greater than this threshold were deemed to be expressed in the adrenal gland.

A cross-array normalisation was performed between normal adrenal and APA microarray experiments. The relative signal intensities of normal and APA samples were compared.

6.3.1.1 Statistical Analysis of Expression Data

The correlation of the microarray expression data and miRNA expression differences (Student's t-test) were tested using Prism 4.0 Graph Pad software. For all analyses, confidence intervals of 95% were used and p < 0.05 was required for statistical significance. Data are expressed as the mean \pm SEM.

6.3.2 Genotyping of the 3'UTR of CYP11B1

6.3.2.1 Subjects

Twenty-six normotensive patients participating in the North Glasgow Monitoring Trends and Determinants in Cardiovascular (MONICA IV) were selected for analysis (Fraser *et al.*, 1999). These subjects had previously been sequenced for two common genetic variants: the *CYP11B2* promoter -344 C/T SNP and the intron 2 conversion. The characteristics of these patients have previously been published (Barr *et al.*, 2007).

6.3.2.2 Sequence Analysis

The full-length *CYP11B1* 3'UTR (2014 base-pairs) was first amplified by PCR (Section 2.6.1), verified by gel electrophoresis (Section 2.6.2) and the products were then sequenced according to methods described in Section 2.7 using the primers in Table 6-1.

Primer Name	Sequence (5'-3')	Direction	
B1 ex9 seq1F	CAGGTGGAGACACTAACCCA	Sense	
B1 ex9 seq3F	AGTCTCACATGTCCCTGTTC	Sense	
B1 ex9 seq5F	AAGGACTCAGACGAGTTTTA	Sense	
B1 ex9 seq7F	AAGAAAACGCCATAGACTGG	Sense	
B1 ex9 R1	ATGCTCTGCCCCTGCAGCTT	Anti-sense	

Table 6-1. Primers Used to Sequence the *CYP11B1* 3'UTR.

6.3.2.3 Statistical Analysis of Genetic Sequences

Sequencing was visualised using the SeqScape v. 2.1.1 software. Data taken from Ensembl release 66 (Feb 2012) using genotype data from the 1000 genome CEU population; low coverage panel was analysed using Haploview v. 4.2 Software (Barrett *et al.*, 2005). Genotype distribution was calculated by using the Hardy Weinberg equilibrium and LD calculated using D'.

6.4 Results

6.4.1 miRNA Expression Profiling of Aldosterone-Producing Adenoma Tissue.

The results of the microarray experiment are shown in Figure 6-1; the red line represents the expression cut-off of 500 AU, 62 miRNAs were expressed in APA tissue above the stated threshold. The mean expression level of these miRNAs is shown in Figure 6-2. The signal intensity for individual miRNAs was consistent between the samples.

To compare the relative expression levels of miRNAs in aldosterone-producing adenoma samples to those in the normal adrenal gland, the two miRNA microarrays were normalised using a global normalisation approach which allows for direct comparison of the two data-sets. This adjusted the signal intensities for the normal adrenal gland miRNA expression results so that the signal intensity of 67 miRNAs was greater than 500 AU following normalisation. Of these 51 were also expressed in the APA samples and 16 were not. Eleven miRNAs were only expressed in APA samples (Figure 6-3).

The signals from the normal and APA samples were highly correlated (p < 0.0001, r^2 0.825; purple dots, Figure 6-4). This correlation was maintained when only samples expressed above 500 AU were analysed (p < 0.0001, r^2 0.8703; red diamonds, Figure 6-4). 27 highly-expressed miRNAs are differentially expressed by greater than 2-fold (Figure 6-4). Table 6-2 lists the mean expression levels of all the miRNAs which differed significantly between samples types, including those which are classified as being expressed in only one tissue, as determined by microarray analysis.



Human microRNAs

Figure 6-1. Aldosterone-Producing Adenoma miRNA Microarray Output.

The average normalised microarray expression signal of 728 human miRNAs (x-axis, in miRNA numerical order) in four aldosterone-producing adenoma samples. Red line represents the expression cut-off threshold of 500 Arbitrary Units (AU).



Human miRNA

Figure 6-2. Average Microarray Signals of miRNAs Expressed in Aldosterone-Producing Adenoma.

The average normalised microarray expression signal of miRNAs with an expression level greater than 500 Arbitrary Units (AU). The mean of four tissues samples; error bars represent SEM.



Figure 6-3. Number of miRNAs Expressed in Normal and APA Samples.

The number of miRNAs expressed in each sample type: normal (non-tumorous adrenal gland) and APA (aldosterone-producing adenoma) following miRNA microarray normalisation.



Figure 6-4. Correlation of miRNA Expression in Normal and APA Adrenal Tissue.

miRNA expression levels (purple: < 500 AU; red: >500AU) from normal adrenal gland (x-axis) and APA samples (y-axis) on log scales. Blue line indicates change of greater than 2-fold.

Table 6-2. Normalised Adrenal and APA Microarray Signals.

let-7a16,002 \pm 1,18923,866 \pm 895< 0.01	miRNA	Norma	l Ad	renal	ŀ	٩PA		p-value
let-7b10,072 \pm 85519,235 \pm 2,202< 0.01	let-7a	16,002	±	1,189	23,866	±	895	< 0.01
let-7c13,232 \pm 1,04720,486 \pm 1,593< 0.01let-7d11,534 \pm 75614,091 \pm 480< 0.05	let-7b	10,072	±	855	19,235	±	2,202	< 0.01
let-7d11,534 \pm 75614,091 \pm 480< 0.05let-7f13,709 \pm 55718,592 \pm 407< 0.01	let-7c	13,232	±	1,047	20,486	±	1,593	< 0.01
let-7f13,709 \pm 55718,592 \pm 407< 0.01miR-1031,109 \pm 1381,813 \pm 217< 0.05	let-7d	11,534	±	756	14,091	±	480	< 0.05
miR-1031,109 \pm 1381,813 \pm 217< 0.05miR-107958 \pm 1201,527 \pm 176< 0.05	let-7f	13,709	±	557	18,592	±	407	< 0.01
miR-107958 \pm 1201,527 \pm 176< 0.05miR-10b2,835 \pm 3741,438 \pm 107< 0.05	miR-103	1,109	±	138	1,813	±	217	< 0.05
miR-10b $2,835$ \pm 374 $1,438$ \pm 107 < 0.05 miR-125a-5p $6,919$ \pm 715 $4,475$ \pm 466 < 0.05 miR-126 $5,331$ \pm 511 $2,029$ \pm 373 < 0.01 miR-140-3p 320 \pm 25 567 \pm 22 < 0.01 miR-15a 868 \pm 212 143 \pm 35 < 0.01 miR-15a 868 \pm 122 143 \pm 35 < 0.01 miR-195 $5,193$ \pm 568 $2,267$ \pm 712 < 0.05 miR-202* $7,125$ \pm 735 577 \pm 250 < 0.001 miR-21 $7,871$ \pm $1,345$ $2,284$ \pm 943 < 0.05 miR-23a $7,247$ \pm 265 $4,436$ \pm 317 < 0.05 miR-24 $2,625$ \pm 225 $1,691$ \pm 219 < 0.05 miR-27a $2,040$ \pm 138 495 \pm 76 < 0.001 miR-29a $5,160$ \pm 553 $1,269$ \pm 195 < 0.01 miR-30a 914 \pm 82 253 \pm 58 < 0.05 miR-30b $1,759$ \pm 153 $1,666$ \pm 152 < 0.05 miR-30b $1,759$ \pm 153 $1,666$ \pm 34 < 0.01 miR-30b $1,759$ <td< td=""><td>miR-107</td><td>958</td><td>±</td><td>120</td><td>1,527</td><td>±</td><td>176</td><td>< 0.05</td></td<>	miR-107	958	±	120	1,527	±	176	< 0.05
miR-125a-5p $6,919$ \pm 715 $4,475$ \pm 466 < 0.05 miR-126 $5,331$ \pm 511 $2,029$ \pm 373 < 0.01 miR-140-3p 320 \pm 25 567 \pm 22 < 0.01 miR-148a $1,680$ \pm 278 169 \pm 33 < 0.01 miR-15a 868 \pm 122 143 \pm 35 < 0.01 miR-195 $5,193$ \pm 568 $2,267$ \pm 712 < 0.05 miR-202* $7,125$ \pm 735 577 \pm 250 < 0.001 miR-21 $7,871$ \pm $1,345$ $2,284$ \pm 943 < 0.05 miR-23a $7,247$ \pm 265 $4,436$ \pm 317 < 0.05 miR-24 $2,625$ \pm 225 $1,691$ \pm 219 < 0.05 miR-27a $2,040$ \pm 138 495 \pm 76 < 0.001 miR-29b $5,160$ \pm 553 $1,269$ \pm 195 < 0.01 miR-29c $3,407$ \pm 308 141 \pm 37 < 0.001 miR-30b $1,759$ \pm 153 $1,066$ \pm 152 < 0.05 miR-30b $1,759$ \pm 153 $1,666$ \pm 34 < 0.01 miR-320 $2,416$ \pm 347 $4,507$ \pm 543 < 0.01 miR-326 460 <t< td=""><td>miR-10b</td><td>2,835</td><td>±</td><td>374</td><td>1,438</td><td>±</td><td>107</td><td>< 0.05</td></t<>	miR-10b	2,835	±	374	1,438	±	107	< 0.05
miR-1265,331 \pm 5112,029 \pm 373 $<$ 0.01miR-140-3p320 \pm 25567 \pm 22 $<$ 0.01miR-148a1,680 \pm 278169 \pm 33 $<$ 0.01miR-15a868 \pm 122143 \pm 35 $<$ 0.01miR-1955,193 \pm 5682,267 \pm 712 $<$ 0.05miR-202*7,125 \pm 735577 \pm 250 $<$ 0.001miR-217,871 \pm 1,3452,284 \pm 943 $<$ 0.05miR-23a7,247 \pm 2654,436 \pm 317 $<$ 0.01miR-242,625 \pm 2251,691 \pm 219 $<$ 0.05miR-27a2,040 \pm 138495 \pm 76 $<$ 0.001miR-27b2,522 \pm 147792 \pm 78 $<$ 0.001miR-29a5,160 \pm 5531,269 \pm 195 $<$ 0.01miR-30a914 \pm 82253 \pm 58 $<$ 0.05miR-30b1,759 \pm 1531,066 \pm 152 $<$ 0.05miR-30c1,668 \pm 1501,113 \pm 109 $<$ 0.05miR-3202,416 \pm 3474,507 \pm 543 $<$ 0.01miR-325821 \pm 179136 \pm 34 $<$ 0.01miR-3261,317 \pm 139 <td>miR-125a-5p</td> <td>6,919</td> <td>±</td> <td>715</td> <td>4,475</td> <td>±</td> <td>466</td> <td>< 0.05</td>	miR-125a-5p	6,919	±	715	4,475	±	466	< 0.05
miR-140-3p 320 ± 25 567 ± 22 <0.01 miR-148a $1,680 \pm 278$ 169 ± 33 <0.01 miR-15a 868 ± 122 143 ± 35 <0.01 miR-195 $5,193 \pm 568$ $2,267 \pm 712$ <0.05 miR-202* $7,125 \pm 735$ 577 ± 250 <0.001 miR-21 $7,871 \pm 1,345$ $2,284 \pm 943$ <0.05 miR-23a $7,247 \pm 265$ $4,436 \pm 317$ <0.05 miR-23b $7,622 \pm 177$ $4,290 \pm 317$ <0.05 miR-24 $2,625 \pm 225$ $1,691 \pm 219$ <0.05 miR-27a $2,040 \pm 138$ 495 ± 76 <0.001 miR-29a $5,160 \pm 553$ $1,269 \pm 195$ <0.001 miR-29c $3,407 \pm 308$ 141 ± 37 <0.001 miR-30a 914 ± 82 253 ± 58 <0.05 miR-30b $1,759 \pm 153$ $1,066 \pm 152$ <0.05 miR-30b $1,564 \pm 337$ <0.05 miR-30b $1,564 \pm 0.05$	miR-126	5,331	±	511	2,029	±	373	< 0.01
miR-148a1,680 \pm 278169 \pm 33< 0.01miR-15a868 \pm 122143 \pm 35< 0.01	miR-140-3p	320	±	25	567	±	22	< 0.01
miR-15a868 \pm 122143 \pm 35 < 0.01 miR-1955,193 \pm 5682,267 \pm 712 < 0.05 miR-202*7,125 \pm 735577 \pm 250 < 0.001 miR-217,871 \pm 1,3452,284 \pm 943 < 0.05 miR-23a7,247 \pm 2654,436 \pm 317 < 0.01 miR-23b7,622 \pm 1774,290 \pm 317 < 0.01 miR-242,625 \pm 2251,691 \pm 219 < 0.05 miR-27a2,040 \pm 138495 \pm 76 < 0.001 miR-27b2,522 \pm 147792 \pm 78 < 0.001 miR-29a5,160 \pm 5531,269 \pm 195 < 0.01 miR-29b561 \pm 9169 \pm 20 < 0.001 miR-30a914 \pm 82253 \pm 58 < 0.05 miR-30b1,759 \pm 1531,066 \pm 152 < 0.05 miR-30c1,668 \pm 1501,113 \pm 109 < 0.05 miR-3202,416 \pm 3474,507 \pm 543 < 0.01 miR-325821 \pm 179136 \pm 34 < 0.01 miR-34c-3p209 \pm 641,493 \pm 235 < 0.05 miR-36164 \pm 33 <td< td=""><td>miR-148a</td><td>1,680</td><td>±</td><td>278</td><td>169</td><td>±</td><td>33</td><td>< 0.01</td></td<>	miR-148a	1,680	±	278	169	±	33	< 0.01
miR-195 $5,193$ \pm 568 $2,267$ \pm 712 < 0.05 miR-202* $7,125$ \pm 735 577 \pm 250 < 0.001 miR-21 $7,871$ \pm $1,345$ $2,284$ \pm 943 < 0.05 miR-23a $7,247$ \pm 265 $4,436$ \pm 317 < 0.01 miR-23b $7,622$ \pm 177 $4,290$ \pm 317 < 0.01 miR-24 $2,625$ \pm 225 $1,691$ \pm 219 < 0.05 miR-27a $2,040$ \pm 138 495 \pm 76 < 0.001 miR-27b $2,522$ \pm 147 792 \pm 78 < 0.001 miR-29a $5,160$ \pm 553 $1,269$ \pm 195 < 0.01 miR-30a 914 \pm 82 253 \pm 58 < 0.05 miR-30b $1,759$ \pm 153 $1,066$ \pm 152 < 0.05 miR-30c $1,668$ \pm 150 $1,113$ \pm 109 < 0.05 miR-320 $2,416$ \pm 347 $4,507$ \pm 543 < 0.01 miR-335 821 \pm 179 136 \pm 34 < 0.01 miR-34c-3p 209 \pm 64 $1,493$ \pm 235 < 0.05 miR-361-5p $1,347$ \pm 139 $1,944$ \pm 88 < 0.01 miR-342 197	miR-15a	868	±	122	143	±	35	< 0.01
miR-202*7,125±735577±250< 0.001miR-217,871±1,3452,284±943< 0.05	miR-195	5,193	±	568	2,267	±	712	< 0.05
miR-21 $7,871 \pm$ $1,345$ $2,284 \pm$ 943 < 0.05 miR-23a $7,247 \pm$ 265 $4,436 \pm$ 317 < 0.05 miR-23b $7,622 \pm$ 177 $4,290 \pm$ 317 < 0.01 miR-24 $2,625 \pm$ 225 $1,691 \pm$ 219 < 0.05 miR-27a $2,040 \pm$ 138 $495 \pm$ 76 < 0.001 miR-27b $2,522 \pm$ 147 $792 \pm$ 78 < 0.001 miR-29a $5,160 \pm$ 553 $1,269 \pm$ 195 < 0.01 miR-29b $561 \pm$ 91 $69 \pm$ 20 < 0.001 miR-29c $3,407 \pm$ 308 $141 \pm$ 37 < 0.001 miR-30a $914 \pm$ 82 $253 \pm$ 58 < 0.05 miR-30b $1,759 \pm$ 153 $1,066 \pm$ 152 < 0.05 miR-30c $1,668 \pm$ 150 $1,113 \pm$ 109 < 0.05 miR-320 $2,416 \pm$ 347 $4,507 \pm$ 543 < 0.01 miR-325 $821 \pm$ 179 $136 \pm$ 34 < 0.01 miR-365 $460 \pm$ 67 $115 \pm$ 23 < 0.01 miR-361-5p $1,347 \pm$ 139 $1,944 \pm$ 88 < 0.01 miR-361 $64 \pm$ 33 $4,226 \pm$ $2,809 <$ < 0.05 miR-361 $64 \pm$ 33 $4,226 \pm$ $2,809 <$ < 0.05 miR-574-5p $244 \pm$ 86 $1,919 \pm$ $470 <$ < 0.05 miR-768-3p $771 \pm$ 280	miR-202*	7,125	±	735	577	±	250	< 0.001
miR-23a $7,247 \pm$ 265 $4,436 \pm$ 317 < 0.05 miR-23b $7,622 \pm$ 177 $4,290 \pm$ 317 < 0.01 miR-24 $2,625 \pm$ 225 $1,691 \pm$ 219 < 0.05 miR-27a $2,040 \pm$ 138 $495 \pm$ 76 < 0.001 miR-27b $2,522 \pm$ 147 $792 \pm$ 78 < 0.001 miR-29a $5,160 \pm$ 553 $1,269 \pm$ 195 < 0.01 miR-29b $561 \pm$ 91 $69 \pm$ 20 < 0.001 miR-29c $3,407 \pm$ 308 $141 \pm$ 37 < 0.001 miR-30a $914 \pm$ 82 $253 \pm$ 58 < 0.05 miR-30b $1,759 \pm$ 153 $1,066 \pm$ 152 < 0.05 miR-30c $1,668 \pm$ 150 $1,113 \pm$ 109 < 0.05 miR-320 $2,416 \pm$ 347 $4,507 \pm$ 543 < 0.01 miR-335 $821 \pm$ 179 $136 \pm$ 34 < 0.01 miR-365 $460 \pm$ 67 $115 \pm$ 23 < 0.05 miR-361 - 5p $1,347 \pm$ 139 $1,944 \pm$ 88 < 0.01 miR-424 $824 \pm$ 163 $74 \pm$ 23 < 0.01 miR-451 $64 \pm$ 33 $4,226 \pm$ $2,809$ < 0.05 miR-574-5p $244 \pm$ 86 $1,919 \pm$ 470 < 0.05 miR-768-5p $857 \pm$ 353 $4,863 \pm$ $1,099$ < 0.05 miR-768-5p $857 \pm$ 353 <	miR-21	7,871	±	1,345	2,284	±	943	< 0.05
miR-23b $7,622$ \pm 177 $4,290$ \pm 317 < 0.01 miR-24 $2,625$ \pm 225 $1,691$ \pm 219 < 0.05 miR-27a $2,040$ \pm 138 495 \pm 76 < 0.001 miR-27b $2,522$ \pm 147 792 \pm 78 < 0.001 miR-29a $5,160$ \pm 553 $1,269$ \pm 195 < 0.01 miR-29b 561 \pm 91 69 \pm 20 < 0.001 miR-29c $3,407$ \pm 308 141 \pm 37 < 0.001 miR-30a 914 \pm 82 253 \pm 58 < 0.05 miR-30b $1,759$ \pm 153 $1,066$ \pm 152 < 0.05 miR-30c $1,668$ \pm 150 $1,113$ \pm 109 < 0.05 miR-320 $2,416$ \pm 347 $4,507$ \pm 543 < 0.01 miR-335 821 \pm 179 136 \pm 34 < 0.01 miR-34c-3p 209 \pm 64 $1,493$ \pm 235 < 0.05 miR-361 546 67 115 \pm 23 < 0.01 miR-321 197 \pm 29 693 \pm 131 < 0.05 miR-321 197 \pm 280 $3,699$ \pm 753 < 0.05 miR-574-5p 244 \pm 86 $1,9$	miR-23a	7,247	±	265	4,436	±	317	< 0.05
miR-24 $2,625 \pm$ 225 $1,691 \pm$ $219 < < 0.05$ miR-27a $2,040 \pm$ 138 $495 \pm$ $76 < < 0.001$ miR-27b $2,522 \pm$ 147 $792 \pm$ $78 < < 0.001$ miR-29a $5,160 \pm$ 553 $1,269 \pm$ $195 < < 0.01$ miR-29b $561 \pm$ $91 $ $69 \pm$ $20 < < 0.001$ miR-29c $3,407 \pm$ $308 $ $141 \pm$ $37 < < 0.001$ miR-30a $914 \pm$ 82 $253 \pm$ $58 < < 0.05$ miR-30b $1,759 \pm$ $153 $ $1,066 \pm$ $152 < < 0.05$ miR-30c $1,668 \pm$ $150 $ $1,113 \pm$ $109 < < 0.05$ miR-30c $1,668 \pm$ $150 $ $1,113 \pm$ $109 < < 0.05$ miR-30c $1,668 \pm$ $150 $ $1,113 \pm$ $235 < < 0.05$ miR-30b $1,759 \pm$ $64 $ $1,493 \pm$ $235 < < 0.05$ miR-320 $2,416 \pm$ $347 $ $4,507 \pm$ $543 < < 0.01$ miR-335 $821 \pm$ $179 $ $136 \pm$ $34 < < 0.01$ miR-365 $460 \pm$ $67 $ $115 \pm$ $23 < < 0.05$ miR-361 -5p $1,347 \pm$ $139 $ $1,944 \pm$ $88 < < 0.01$ miR-365 $460 \pm$ $67 $ $115 \pm$ $23 < < 0.05$ miR-451 $64 \pm$ $33 $ $4,226 \pm$ $2,809 < < 0.05$ miR-574-5p $244 \pm$ $86 $ $1,919 \pm$ $470 < < 0.05$ miR-768-5p $857 \pm$ $353 $ $4,863 \pm$ $1,099 < < 0.05$ miR-768-5p $857 \pm$ $353 $ $4,863 \pm$ $1,099 < < 0.05$ <td>miR-23b</td> <td>7,622</td> <td>±</td> <td>177</td> <td>4,290</td> <td>±</td> <td>317</td> <td>< 0.01</td>	miR-23b	7,622	±	177	4,290	±	317	< 0.01
miR-27a2,040 \pm 138495 \pm 76< 0.001miR-27b2,522 \pm 147792 \pm 78< 0.001	miR-24	2,625	±	225	1,691	±	219	< 0.05
miR-27b2,522 \pm 147792 \pm 78< 0.001miR-29a5,160 \pm 5531,269 \pm 195< 0.01	miR-27a	2,040	±	138	495	±	76	< 0.001
miR-29a $5,160 \pm 553$ $1,269 \pm 195 < 0.01$ miR-29b 561 ± 91 $69 \pm 20 < 0.001$ miR-29c $3,407 \pm 308$ $141 \pm 37 < 0.001$ miR-30a 914 ± 82 $253 \pm 58 < 0.05$ miR-30b $1,759 \pm 153$ $1,066 \pm 152 < 0.05$ miR-30c $1,668 \pm 150$ $1,113 \pm 109 < 0.05$ miR-320 $2,416 \pm 347$ $4,507 \pm 543 < 0.01$ miR-335 821 ± 179 $136 \pm 34 < 0.01$ miR-361-5p $1,347 \pm 139$ $1,944 \pm 88 < 0.01$ miR-365 460 ± 67 $115 \pm 23 < 0.05$ miR-424 824 ± 163 $74 \pm 23 < 0.01$ miR-574-5p 244 ± 86 $1,919 \pm 470 < 0.05$ miR-768-3p 771 ± 280 $3,699 \pm 753 < 0.05$ miR-768-5p 857 ± 353 $4,863 \pm 1,099 < 0.05$ miR-768-5p 857 ± 353 $4,863 \pm 1,099 < 0.05$ miR-92a $1,421 \pm 82$ $2,649 \pm 171 < 0.001$ miR-92b 428 ± 32 $633 \pm 54 < 0.05$ miR-92b 428 ± 32 $633 \pm 54 < 0.05$	miR-27b	2,522	±	147	792	±	78	< 0.001
miR-29b 561 ± 91 $69 \pm 20 < 0.001$ miR-29c $3,407 \pm 308$ $141 \pm 37 < 0.001$ miR-30a 914 ± 82 $253 \pm 58 < 0.05$ miR-30b $1,759 \pm 153$ $1,066 \pm 152 < 0.05$ miR-30c $1,668 \pm 150$ $1,113 \pm 109 < 0.05$ miR-320 $2,416 \pm 347$ $4,507 \pm 543 < 0.01$ miR-32b $2,416 \pm 179$ $136 \pm 34 < 0.01$ miR-32c $2,416 \pm 179$ $136 \pm 34 < 0.01$ miR-34c-3p 209 ± 64 $1,493 \pm 235 < 0.05$ miR-361-5p $1,347 \pm 139$ $1,944 \pm 88 < 0.01$ miR-365 460 ± 67 $115 \pm 23 < 0.01$ miR-424 824 ± 163 $74 \pm 23 < 0.01$ miR-451 64 ± 333 $4,226 \pm 2,809 < 0.05$ miR-574-5p 244 ± 86 $1,919 \pm 470 < 0.05$ miR-768-3p 771 ± 280 $3,699 \pm 753 < 0.05$ miR-768-5p 857 ± 353 $4,863 \pm 1,099 < 0.05$ miR-92a $1,421 \pm 82$ $2,649 \pm 171 < 0.001$ miR-92b 428 ± 32 $633 \pm 54 < 0.05$	miR-29a	5,160	±	553	1,269	±	195	< 0.01
miR-29c $3,407 \pm$ 308 $141 \pm$ $37 < < 0.001$ miR-30a914 \pm 82 $253 \pm$ $58 < < 0.05$ miR-30b $1,759 \pm$ 153 $1,066 \pm$ $152 < < 0.05$ miR-30c $1,668 \pm$ 150 $1,113 \pm$ $109 < < 0.05$ miR-320 $2,416 \pm$ 347 $4,507 \pm$ $543 < < 0.01$ miR-335 $821 \pm$ 179 $136 \pm$ $34 < < 0.01$ miR-34c-3p $209 \pm$ 64 $1,493 \pm$ $235 < < 0.05$ miR-361-5p $1,347 \pm$ 139 $1,944 \pm$ $88 < < 0.01$ miR-365 $460 \pm$ 67 $115 \pm$ $23 < < 0.01$ miR-424 $824 \pm$ 163 $74 \pm$ $23 < < 0.01$ miR-451 $64 \pm$ 33 $4,226 \pm$ $2,809 < < 0.05$ miR-574-5p $244 \pm$ 86 $1,919 \pm$ $470 < < 0.05$ miR-768-3p $771 \pm$ 280 $3,699 \pm$ $753 < < 0.05$ miR-768-5p $857 \pm$ 353 $4,863 \pm$ $1,099 < < 0.05$ miR-92a $1,421 \pm$ 82 $2,649 \pm$ $171 < < 0.001$ miR-92b $428 \pm$ 32 $633 \pm$ $54 < < 0.05$	miR-29b	561	±	91	69	±	20	< 0.001
miR-30a914 \pm 82253 \pm 58< 0.05miR-30b1,759 \pm 1531,066 \pm 152< 0.05	miR-29c	3,407	±	308	141	±	37	< 0.001
miR-30b $1,759$ \pm 153 $1,066$ \pm 152 < 0.05 miR-30c $1,668$ \pm 150 $1,113$ \pm 109 < 0.05 miR-320 $2,416$ \pm 347 $4,507$ \pm 543 < 0.01 miR-335 821 \pm 179 136 \pm 34 < 0.01 miR-34c-3p 209 \pm 64 $1,493$ \pm 235 < 0.05 miR-361-5p $1,347$ \pm 139 $1,944$ \pm 88 < 0.01 miR-365 460 \pm 67 115 \pm 23 < 0.01 miR-424 824 \pm 163 74 \pm 23 < 0.01 miR-451 64 \pm 33 $4,226$ \pm $2,809$ < 0.05 miR-574-5p 244 \pm 86 $1,919$ \pm 470 < 0.05 miR-768-3p 771 \pm 280 $3,699$ \pm 753 < 0.05 miR-768-5p 857 \pm 353 $4,863$ \pm $1,099$ < 0.05 miR-92a $1,421$ \pm 82 $2,649$ \pm 171 < 0.001 miR-92b 428 \pm 32 633 \pm 54 < 0.05	miR-30a	914	±	82	253	±	58	< 0.05
miR-30c $1,668 \pm 150$ $1,113 \pm 109 < 0.05$ miR-320 $2,416 \pm 347$ $4,507 \pm 543 < 0.01$ miR-335 821 ± 179 $136 \pm 34 < 0.01$ miR-34c-3p 209 ± 64 $1,493 \pm 235 < 0.05$ miR-361-5p $1,347 \pm 139$ $1,944 \pm 88 < 0.01$ miR-365 460 ± 67 $115 \pm 23 < 0.01$ miR-424 824 ± 163 $74 \pm 23 < 0.01$ miR-451 64 ± 33 $4,226 \pm 2,809 < 0.05$ miR-574-5p 244 ± 86 $1,919 \pm 470 < 0.05$ miR-768-3p 771 ± 280 $3,699 \pm 753 < 0.05$ miR-768-5p 857 ± 353 $4,863 \pm 1,099 < 0.05$ miR-92a $1,421 \pm 82$ $2,649 \pm 171 < 0.001$ miR-92b 428 ± 32 $633 \pm 54 < 0.05$ miR-99b 671 ± 87 $1,285 \pm 166 < 0.05$	miR-30b	1,759	±	153	1,066	±	152	< 0.05
miR-320 $2,416 \pm$ 347 $4,507 \pm$ $543 < 0.01$ miR-335 $821 \pm$ 179 $136 \pm$ $34 < 0.01$ miR-34c-3p $209 \pm$ 64 $1,493 \pm$ $235 < 0.05$ miR-361-5p $1,347 \pm$ 139 $1,944 \pm$ $88 < 0.01$ miR-365 $460 \pm$ 67 $115 \pm$ $23 < 0.01$ miR-424 $824 \pm$ 163 $74 \pm$ $23 < 0.01$ miR-432 $197 \pm$ 29 $693 \pm$ $131 < 0.05$ miR-574-5p $244 \pm$ 86 $1,919 \pm$ $470 < 0.05$ miR-768-3p $771 \pm$ 280 $3,699 \pm$ $753 < 0.05$ miR-768-5p $857 \pm$ 353 $4,863 \pm$ $1,099 < 0.05$ miR-92a $1,421 \pm$ 82 $2,649 \pm$ $171 < 0.001$ miR-92b $428 \pm$ 32 $633 \pm$ $54 < 0.05$	miR-30c	1,668	±	150	1,113	±	109	< 0.05
miR-335 821 ± 179 $136 \pm 34 < 0.01$ miR-34c-3p 209 ± 64 $1,493 \pm 235 < 0.05$ miR-361-5p $1,347 \pm 139$ $1,944 \pm 88 < 0.01$ miR-365 460 ± 67 $115 \pm 23 < 0.01$ miR-424 824 ± 163 $74 \pm 23 < 0.01$ miR-432 197 ± 29 $693 \pm 131 < 0.05$ miR-574-5p 244 ± 86 $1,919 \pm 470 < 0.05$ miR-768-3p 771 ± 280 $3,699 \pm 753 < 0.05$ miR-768-5p 857 ± 353 $4,863 \pm 1,099 < 0.05$ miR-92a $1,421 \pm 82$ $2,649 \pm 171 < 0.001$ miR-92b 428 ± 32 $633 \pm 54 < 0.05$ miR-99b 671 ± 87 $1285 \pm 166 < 0.05$	miR-320	2,416	±	347	4,507	±	543	< 0.01
miR-34c-3p 209 ± 64 $1,493 \pm 235 < 0.05$ miR-361-5p $1,347 \pm 139$ $1,944 \pm 88 < 0.01$ miR-365 460 ± 67 $115 \pm 23 < 0.01$ miR-424 824 ± 163 $74 \pm 23 < 0.01$ miR-432 197 ± 29 $693 \pm 131 < 0.05$ miR-451 64 ± 33 $4,226 \pm 2,809 < 0.05$ miR-768-3p 771 ± 280 $3,699 \pm 753 < 0.05$ miR-768-5p 857 ± 353 $4,863 \pm 1,099 < 0.05$ miR-92a $1,421 \pm 82$ $2,649 \pm 171 < 0.001$ miR-92b 428 ± 32 $633 \pm 54 < 0.05$	miR-335	821	±	179	136	±	34	< 0.01
miR-361-5p $1,347 \pm$ 139 $1,944 \pm$ 88 < 0.01miR-365 $460 \pm$ 67 $115 \pm$ 23 < 0.01	miR-34c-3p	209	±	64	1,493	±	235	< 0.05
miR-365 460 ± 67 $115 \pm 23 < 0.01$ miR-424 824 ± 163 $74 \pm 23 < 0.01$ miR-432 197 ± 29 $693 \pm 131 < 0.05$ miR-451 64 ± 33 $4,226 \pm 2,809 < 0.05$ miR-574-5p 244 ± 86 $1,919 \pm 470 < 0.05$ miR-768-3p 771 ± 280 $3,699 \pm 753 < 0.05$ miR-768-5p 857 ± 353 $4,863 \pm 1,099 < 0.05$ miR-92a $1,421 \pm 82$ $2,649 \pm 171 < 0.001$ miR-92b 428 ± 32 $633 \pm 54 < 0.05$ miR-92b 671 ± 87 $1285 \pm 166 < 0.05$, miR-361-5p	1,347	±	139	1,944	±	88	< 0.01
miR-424 824 ± 163 74 ± 23 < 0.01 miR-432 197 ± 29 693 ± 131 < 0.05 miR-451 64 ± 33 $4,226 \pm 2,809$ < 0.05 miR-574-5p 244 ± 86 $1,919 \pm 470$ < 0.05 miR-768-3p 771 ± 280 $3,699 \pm 753$ < 0.05 miR-768-5p 857 ± 353 $4,863 \pm 1,099$ < 0.05 miR-92a $1,421 \pm 82$ $2,649 \pm 171$ < 0.001 miR-92b 428 ± 32 633 ± 54 < 0.05 miR-99b 671 ± 87 $1,285 \pm 166$ < 0.05	, miR-365	460	±	67	, 115	±	23	< 0.01
miR-432197 \pm 29693 \pm 131< 0.05miR-45164 \pm 334,226 \pm 2,809< 0.05miR-574-5p244 \pm 861,919 \pm 470< 0.05miR-768-3p771 \pm 2803,699 \pm 753< 0.05miR-768-5p857 \pm 3534,863 \pm 1,099< 0.05miR-92a1,421 \pm 822,649 \pm 171< 0.001miR-92b428 \pm 32633 \pm 54< 0.05miR-99b671 \pm 871,285 \pm 166< 0.05	miR-424	824	±	163	74	±	23	< 0.01
miR-451 64 ± 33 $4,226 \pm 2,809 < 0.05$ miR-574-5p 244 ± 86 $1,919 \pm 470 < 0.05$ miR-768-3p 771 ± 280 $3,699 \pm 753 < 0.05$ miR-768-5p 857 ± 353 $4,863 \pm 1,099 < 0.05$ miR-92a $1,421 \pm 82$ $2,649 \pm 171 < 0.001$ miR-92b 428 ± 32 $633 \pm 54 < 0.05$ miR-99b 671 ± 87 $1,285 \pm 166 < 0.05$	miR-432	197	±	29	693	±	131	< 0.05
miR-574-5p $244 \pm$ 86 $1,919 \pm$ $470 < 0.05$ miR-768-3p $771 \pm$ 280 $3,699 \pm$ $753 < 0.05$ miR-768-5p $857 \pm$ 353 $4,863 \pm$ $1,099 < 0.05$ miR-92a $1,421 \pm$ 82 $2,649 \pm$ $171 < 0.001$ miR-92b $428 \pm$ 32 $633 \pm$ $54 < 0.05$ miR-99b $671 \pm$ 87 $1,285 \pm$ $166 < 0.05$	miR-451	64	±	33	4.226	±	2,809	< 0.05
miR-768-3p771 \pm 2803,699 \pm 753< 0.05miR-768-5p857 \pm 3534,863 \pm 1,099< 0.05miR-92a1,421 \pm 822,649 \pm 171< 0.001miR-92b428 \pm 32633 \pm 54< 0.05miR-99b671 \pm 871,285 \pm 166< 0.05	miR-574-5p	244	±	86	1.919	±	470	< 0.05
miR-768-5p857 \pm 3534,863 \pm 1,099< 0.05miR-92a1,421 \pm 822,649 \pm 171< 0.001miR-92b428 \pm 32633 \pm 54< 0.05miR-99b671 \pm 871,285 \pm 166< 0.05	miR-768-3p	771	+	280	3.699	+	753	< 0.05
miR-92a1,421 \pm 822,649 \pm 171 < 0.001miR-92b428 \pm 32633 \pm 54 < 0.05miR-99b671 \pm 871.285 \pm 166 < 0.05	miR-768-5p	857	- +	353	4,863	+	1.099	< 0.05
miR-92b428 \pm 32633 \pm 54< 0.05miR-99b671 \pm 871.285 \pm 166< 0.05	miR-92a	1.471	+	87	2.649	+	171	< 0.001
miR-99b $671 + 87 1.285 + 166 < 0.05$	miR-92b	478	∸ +	32	633	∸ +	54	< 0.05
	miR-99h	671	+	87	1 285	+	166	< 0.05

6.4.2 APA miRNAs Targeting the CYP11B1 and CYP11B2 Genes.

The role of APA-expressed miRNAs in the regulation of the *CYP11B1* and *CYP11B2* genes was investigated by comparing the miRNA array with the putative target sites identified in these genes by bioinformatic target site (see Chapter 4). APA miRNAs with target sites within these genes are shown in Figure 6-5; 18 miRNAs were predicted to target the 3'UTR of *CYP11B1* and 11 to bind *CYP11B2*. Of these, 8 miRNAs had target sites common to both genes.

Merging the miRNA target site lists for the normal adrenal and APA samples identified that all miRNAs predicted to target the *CYP11B1* or *CYP11B2* 3'UTR are expressed in the normal adrenal gland (Figure 6-6), although some are expressed at significantly different levels between sample groups (Table 6-2). However, 7 miRNAs are only detectable in normal adrenal gland and not in APA samples. Of these, 2 target *CYP11B1*, 1 targets *CYP11B2* and 4 target both genes (Figure 6-6).

The relative, normalised expression levels of the miRNAs expressed in both tissue types at significantly different levels are depicted in Figure 6-7. Two of the *CYP11B1*-targeting miRNAs, miR-23a and miR-23b, are expressed at significantly lower levels in APA tissue than normal adrenal gland, whereas the abundance of miR-432 and miR-768-5p is significantly lower in APA tissue than normal adrenal gland (Figure 6-7 and Table 6-2). Only one miRNA which exclusively targets the *CYP11B2* 3'UTR, miR-125a-5p, is present in different levels being significantly higher in normal adrenal gland than APA tissue (Figure 6-7 and Table 6-2).

Finally, 4 miRNAs predicted to target both *CYP11B1* and *CYP11B2* are differentially expressed in these tissues (Figure 6-7 and Table 6-2). The levels of miR-10b and miR-24 are significantly higher in normal tissue than in APA, whereas miR-140-3p and miR-34c-3p are expressed at higher levels in APA samples relative to normal adrenal tissue (Figure 6-7 and Table 6-2).

It should be reiterated the removal of miR-768-5p from the miRNA repository and therefore does not represent a likely regulator of adrenal physiology.



Figure 6-5. Venn Diagram of APA Adrenal miRNAs Predicted to Bind to the CYP11B1 and CYP11B2 Genes

Results of miRNA microarray analyses of aldosterone-producing adenoma adrenal tissue combined with data from bioinformatics searches for putative miRNA binding sites in the 3'-UTR of the *CYP11B1* and *CYP11B2* genes.



Figure 6-6. Putative miRNA Binding Sites in *CYP11B1* and *CYP11B2* and miRNA Expression in Adrenal Tissue.




Results show the relative expression levels of miRNAs in normal adrenal (Red) or APA (blue) tissue. Results are presented according to whether predicted miRNA binding occurs in *CYP11B1* and/or *CYP11B2*. * p < 0.05, ** p < 0.01.

6.4.3 Genetic Variation at the 3'UTR of CYP11B1 and CYP11B2 Genes

Approximately 2,000 base-pairs of the 3'UTR of *CYP11B1* were successfully amplified (Figure 6-8) and sequenced (Figure 6-9) in 26 normotensive patients, which were stratified based on their genotype at the -344 C/T (rs1799998) and intron 2 conversion loci: TT/ConCon (n = 11); CC/WtWt (n = 10); TT/WtWt (n = 5) (Table 6-4). This small sub-set of subjects was chosen to allow the identification of novel SNPs across this region and to ensure all three major haplotypes in the region were represented. Twelve polymorphisms were identified across the region; all were single-base changes (Table 6-3) and the alleles corresponding to each genotype group are shown in Table 6-4. All twelve of the SNPs identified are recorded in the Single Nucleotide Polymorphism Database (dbSNP).

Non-random selection of subjects in this study prevents analysis of allele and haplotypes frequencies or calculation of LD. The pattern of SNPs certainly supports high LD, as is observed across the whole *CYP11B1/B2* locus (Davies *et al.*, 2009). To explore this further, population data from the 1000 genome project was evaluated for each of the SNPs; unfortunately for two SNPs (rs5301 and rs72552270) no population data was available (Table 6-5). Analysis indicates that for the majority of SNPs the two alleles are present at approximately equal frequency in a Caucasian population, but for three SNPs (rs61752812, rs1752809, rs61752805) the minor allele is only rarely observed in a Caucasian population; therefore were excluded from linkage disequilibrium analysis. The LD plot demonstrated a high degree of Linkage Disequilibrium between seven 3'UTR SNPs (Figure 6-10). Comparable population data was not available for *CYP11B1* promoter SNPs (rs4471016 and rs4313136) thus, the level of linkage disequilibrium across the gene could not be determined.

The 3'UTR of *CYP11B2* has previously been sequenced in the same cohort of patients by other members of the group during a full gene SNP discovery investigation (Barr *et al.*, 2007). This analysis identified four genetic variants within the 3'UTR, the details of the variants - 3 single-base changes and a 3-base insertion (rs113094040) - are described in Table 6-6. The base changes

attributed to the polymorphisms and their association with the -344 C/T and intron 2 polymorphisms are shown in Table 6-7. One of these polymorphisms appears to be a novel finding; no previous report of it was found in dbSNP, and as such does not have a reference sequence (rs) number. Again, for statistical analysis genotype from genotype data taken from a small subset (low coverage panel) of subjects in the 1000 genome project (Table 6-8); this could only be achieved for two SNPs (rs28491316 and rs3802230) and *CYP11B2* -344C/T promoter SNP (rs1799998). All three of these variants are in high LD (Figure 6-12).



Figure 6-8. CYP11B1 3'UTR PCR Amplification in MONICA Patients.

PCR product from a random selection of MONICA patients was resolved on a 0.7% agarose gel. Promega 1 kb ladder was used for size determination; sizes indicated are in kilobase pairs.



Figure 6-9. Genotype Analysis of CYP11B1 3'UTR

An example of the electropherogram reads from one subject of each haplotype group (indicated on the left hand-side; -344(promoter SNP)/intron conversion status). This example shows the C/T SNP (re58961462) and the A/G SNP (rs57991037).

SNP	Postion	Location	Alleles
rs4736312	143953937	7926	T/G
rs1134096	143954223	7640	G/T
rs1134095	143954290	7573	C/T
rs61752812	143954372	7491	G/A
rs61752809	143954501	7362	A/G
rs7003319	143954747	7116	A/G
rs5017238	143954769	7094	C/T
rs61752805	143954866	6997	G/C
rs12543598	143955318	6545	A/C
rs5299	143955471	6392	A/G
rs5301	143955669	6590	A/G
rs72552270	143955891	6812	G/A

Table 6-3. Characteristics of Polymorphisms Genotyped in the CYP11B1 3'UTR.

Table 6-4. Polymorphisms Identified in the *CYP11B1* 3'UTR (grouped according to -344 and IC genotype).

		Position										
	6392	6545	6590	6812	6997	7094	7116	7362	7491	7573	7640	7926
TT/ConCon	А	А	А	G	G	С	А	А	G	С	G	Т
CC/WtWt	G	С	G	А	G/C [#]	Т	G	А	G	Т	Т	G
TT/WtWt	G	С	G/A*	А	G	Т	G	G/A*	G/A*	Т	Т	G

#: 2 subjects were heterozygous (G/C) at position 6997, 8 subjects were homozygous (G/G)

*: 3 subjects were heterozygous (G/A) at positions 6590, 7362 and 7491. One subject was homozygous (G/G) and one homozygous (A).

SNP	Postion		MAF	Obs. Het	HWE (p)
rs4736312	143953937	3'UTR	0.48	0.50	1.00
rs1134096	143954223	3'UTR	0.49	0.48	0.95
rs1134095	143954290	3'UTR	0.49	0.48	0.95
rs61752812	143954372	3'UTR	0.06	0.12	1.00
rs61752809	143954501	3'UTR	0.06	0.12	1.00
rs7003319	143954747	3'UTR	0.49	0.48	0.95
rs5017238	143954769	3'UTR	0.49	0.48	0.95
rs61752805	143954866	3'UTR	0.04	0.08	1.00
rs12543598	143955318	3'UTR	0.48	0.45	0.57
rs5299	143955471	3'UTR	0.49	0.48	0.95
rs5301	143955669	3'UTR	No popula	tion data a	vailable
rs72552270	143955891	3'UTR	No popula	tion data a	vailable
rs4471016	143963087	Promoter	No popula	tion data a	vailable
rs4313136	143963117	Promoter	No popula	tion data a	vailable

Table 6-5. Statistical Analysis of CYP11B1 Polymorphisms.

Data taken from Ensembl release 66 (Feb 2012) using genotype data from the 1000 genome CEU population, low coverage panel.



Figure 6-10. LD Plot and Haplotypes of Polymorphisms in the CYP11B1 3'UTR.

The LD plot (generated with the D' method) of *CYP11B1* polymorphisms, generated using genotype data from the 1000 genome CEU population, low coverage panel.



Figure 6-11. Genotype Analysis of CYP11B2 3'UTR

An example of the electropherogram read from one subject of each genotype group (indicated on the left hand-side). This example shows the TCC insertion (rs57991037).

Table 6-6. Characteristics of the Polymorphisms Genotyped in the CYP11B2 3'UTR.

SNP	Postion	Location	Alleles
rs113094040	143993104	6756	-/TCC
rs28491316	143992661	7199	G/A
rs3802230	143992864	6996	G/T
Unassigned	143992411	6543	A/G

Table 6-7. Polymorphisms Identified in the *CYP11B2* 3'UTR (grouped according to -344 and IC genotype).

	Position				
	6543	6756	6996	7199	
TT/ConCon	А		G	G	
CC/WtWt	G	тсс	Т	А	
TT/WtWt	G	тсс	Т	G	

Table 6-8. Statistical Analysis of CYP11B2 Polymorphisms.

SNP	Postion		MAF	Obs. Het	HWE (p)
rs1799998	143999600	Promoter	0.43	0.50	1.00
rs28491316	143992661	3'UTR	0.45	0.47	0.8
rs3802230	143992864	3'UTR	0.50	0.50	1.00
Unassigned	143992411	3'UTR	No popula	tion data av	/ailable
rs113094040	143963117	3'UTR	No popula	tion data av	vailable

Data taken from Ensembl release 66 (Feb 2012) using genotype data from the 1000 genome CEU population, low coverage panel.



Figure 6-12. LD Plot and Haplotypes of Polymorphisms in the CYP11B2 3'UTR.

The LD plot (generated with the D' method) of *CYP11B2* polymorphisms, generated using genotype data from the 1000 genome CEU population, low coverage panel.

6.4.4 Impact of Polymorphisms on miRNA Binding Sites

The putative binding sites of miRNAs expressed in the normal adrenal gland (Figure 4-8) were mapped to the 3'UTR of the *CYP11B1* or *CYP11B2* gene to determine whether genetic variation in this region could disrupt a miRNA binding site (Figure 6-13 and Figure 6-14).

Three *CYP11B1* 3'UTR SNPs lie in putative miRNA binding sites: rs5299 and miR-140-3p: rs57250946 and miR-22; rs61752812 and miR-494 (shown in red in Figure 6-13). According to TargetScan or microrna.org predictions, none of the SNPs are located in the seed site of the miRNA binding site, or in other complementary base-matches between the 3'UTR and the other regions of the miRNA. Furthermore, 13 miRNA target-sites lie within 50 base-pairs of a SNP (indicated in yellow in Figure 6-13) whereas, the binding sites for the other 10 miRNAs do not lie in close proximity to a 3'UTR SNP.

Of the four polymorphisms in the *CYP11B2* 3'UTR (Figure 6-14), only one SNP is located in a miRNA binding site: rs380223 which lies in the common miR-10a/miR-10b region; again this SNP does not lie in the seed region. Another 9 miRNA binding sites lie within 50 bases of a SNP (yellow in Figure 6-14) and 6 binding sites were not in close proximity to a polymorphism.



Figure 6-13. Schematic Diagram of SNPs and miRNA Binding Sites in the CYP11B1 3'UTR.

Location of single nucleotide polymorphisms (SNPs) on the 3'UTR of the *CYP11B1* gene. Adrenal-expressed miRNA binding sites are colour-coded. Red: SNP located in binding region; Yellow: SNP with 50 base-pairs (bp) of binding site; Purple: binding site not in close proximity to SNP. Diagram not to scale.



Figure 6-14. Schematic Diagram of SNPs and miRNA Binding Sites in the CYP11B2 3'UTR.

Location of single nucleotide polymorphisms (SNPs) on the 3'UTR of the *CYP11B2* gene. Adrenal-expressed miRNA binding sites are colour-coded. Red: SNP located in binding region; Yellow: SNP with 50 base-pairs (bp) of binding site; Purple: binding site not in close proximity to SNP. Diagram not to scale.

6.5 Discussion

The experiments presented in this chapter investigated the role of miRNA in adrenal pathologies and the influence of SNPs in the 3'UTR of the *CYP11B1* and *CYP11B2* genes on miRNA binding.

This study successfully determined the miRNA expression profile of APA samples, identifying several miRNAs not expressed in normal tissue and many more with differential expression levels. However, the observed differences in miRNA expression shown and discussed here need to be validated by qRT-PCR analysis in order to confirm the accuracy of the array data and to determine more precisely the magnitude of these changes. This work will be conducted in the future, but time constraints prevent their presentation in this thesis.

The APA tissue samples were fixed in formalin and embedded in paraffin blocks (FFPE), which is normally problematic as it leads to degradation of longer RNA species. However, due to their small size, miRNAs can successfully be isolated and measured in archived preserved tissue (Xi *et al.*, 2007;Li *et al.*, 2007). As shown in this study, utilising archived tissue increases the range of viable tissues available for miRNA studies. Four samples were used in this study to establish the APA miRNA expression profile and this number is comparable to those used in the two previous adrenal miRNA investigations (Soon *et al.*, 2009;Schmitz *et al.*, 2011). Given the appropriate sample number for human adrenal miRNA profiling and the low intrasample variation in expression pattern, the results appear to confirm the homogeneity of the selected samples and the accuracy of expression measurements.

Despite the miRNA microarrays for the two sample types (APA and normal adrenal) being performed at different times, care was taken to permit data-sets to be compared accurately; arrays were performed using the same method, at the same supplier and using the same microfluidic chips with identical miRNA coverage. Post-array analysis utilised internal spike-in controls to cross-normalise the array data output, allowing signals to be compared between chips. There was a good number of APA-expressed miRNAs with target sites in the *CYP11B1* and/or *CYP11B2* genes and this was similar to those previously observed

for normal adrenal tissue (Chapter 4). Furthermore, there was overlap of miRNA species with several being expressed in both sample types. This adds confidence to the miRNA expression profile data generated this part of the study and indicates that the miRNA coverage and tissues types were appropriate and of suitable stringency.

Seven miRNAs which were predicted to have binding sites in *CYP11B1* and *CYP11B2* genes were only expressed in normal adrenal gland. Additionally, nine miRNAs targeting both *CYP11B1* and *CYP11B2* were expressed at significantly different levels in the two tissue types. This may suggest that these miRNAs are important in regulatory processes surrounding adrenal pathology. Moreover, several of the miRNAs identified have also been implicated in tumourogenesis: miR-24 represses the translation of p16 (Cyclin-dependent kinase inhibitor 2A) (Lal *et al.*, 2008) and miR-10b is strongly implicated in breast cancer (Ma *et al.*, 2007). This may indicate that, in addition to regulating adrenal-specific genes, these miRNAs can change cell cycle regulation or cell morphology, leading to neoplastic growths.

Two of the miRNAs which target CYP11B1 and CYP11B2, miR-23 and miR-24, are transcribed in two separate clusters (Section 6.5). All cluster members (including miR-27a and miR-27b which are not predicted to bind CYP11B1 or CYP11B2) are expressed at lower levels in APA samples. The miRNAs from the miR-24-2 cluster are located on a intergenic region, whereas the genomic locus which includes the miR-24-1 cluster, C9orf3, is believed to encode a novel metalloprotease called aminopeptidase-O (AP-O) (Diaz-Perales et al., 2005). AP-O may be a novel dysregulated gene in APA. An attempt was made to measure the relative mRNA levels of AP-O in normal and APA samples but the RNA quality of the samples was not suitable for mRNA gRT-PCR analysis. The role of AP-O in adrenal physiology has not been studied, but AP-O is proposed to be important in vascular cell biology (Xi et al., 2007) and another aminopeptidase (AP-A) is integral to the RAAS, cleaving angiotensin II into angiotensin III; it also plays a critical role in blood pressure regulation (Ahmad and Ward, 1990; Mitsui et al., 2003). Further investigation of the AP-O protein and the miRNAs transcribed from its eight introns is required in order to establish their role in adrenal function.

This study also analysed 26 subjects who had been selected from a larger normotensive cardiovascular population study on the basis of their genotype at the -344 C/T and IC loci. These polymorphisms have been shown to be linked to increased aldosterone production, inefficient 11B-hydroxylation and hypertension (Davies *et al.*, 2009). They are also in tight LD, the frequencies of their haplotypes in a normotensive Caucasian population are: CC/WtWt (45%), TT/ConCon (38%) and T/Wt (16%) (Davies *et al.*, 1999).

Previous sequencing analysis identified 4 polymorphisms in the *CYP11B2* 3'UTR and this current investigation identified 12 further SNPs in the *CYP11B1* 3'UTR. Population data from the 1000 genome project found that seven *CYP11B1* 3'UTR SNPs are in high LD and additionally, two *CYP11B2* 3'UTR and the -344 promoter are in high LD. This agrees with previous assessment of LD across the locus (Barr *et al.*, 2007). Therefore, the causative mechanism governing the association of the -344C/T SNP with hypertension (Sookoian *et al.*, 2007) or the *CYP11B1* promoter SNPs with inefficient 11B-hydroxylation (Barr *et al.*, 2006) could lie with variation in the 3'UTR.

Currently, the molecular mechanism to explain the relationship between *CYP11B1/CYP11B2* genotype and blood pressure phenotype remains elusive. A study of genetic variants and gene expression in brain cortical samples showed that genes with a high degree of variation are more likely to be regulated by miRNAs than less variable genes (Zhang and Su, 2008). Therefore, regulation of the *CYP11B1* and *CYP11B2* genes by miRNA offers a novel method of gene regulation, which may control basal mRNA expression levels but may also be influential in cases of altered gene transcription and steroid production. As the subjects in this study were drawn solely from a normal Caucasian normotensive population (Fraser *et al.*, 1999), the possibility exist that, in a hypertensive population, the frequencies of these haplotypes may vary, or that there may even be additional novel SNPs. Additionally, this study only evaluated 26 subjects which would not detect very rare SNPs *i.e.* those present in less than 0.01% of the population.

This study confirmed that the highly polymorphic nature of this locus extends to the 3' regulatory region, but none of the 3'UTR SNPs were located in the seed region of a putative miRNA target site. The seed site only accounts for 8 bases of a miRNA and complementary binding in other regions of the miRNA have been shown to be important (Grimson et al., 2007; Shin et al., 2010) but, according to my bioinformatic analysis, no SNP is located in a complementary mRNA:miRNA base-pairing region either. Yet, SNPs may still influence miRNA action in other ways; mismatches and bulges are frequently tolerated by miRNAs, as are G:U wobble. Therefore, it is possible that a SNP in a miRNA-binding region may alter the binding capacity and/or structure of miRNA: target interactions. Thus, while the influence of SNPs may extend beyond simple base-pairing, this would need to be tested in vitro. Furthermore, the collective influence of SNPs with a haplotype may be more important than individual SNPs, and be more representative of actual physiology. One way of testing this would be to mutate the 3'UTR of the pEZX reporter construct (Chapter 3), in order to recreate the various haplotypes then test the relative responses in vitro. Sequence analysis of the pEZX-B1 and pEZX-B2 plasmid constructs (Section 3.4.3) indicates that they contain the alleles identified in the CC/WtWt subjects (Table 6-4 and Table 6-7). Furthermore, experiments presented in this thesis have tested miRNA action in the H295R cell line; the 3'UTR of CYP11B1 or CYP11B2 not been sequenced in DNA isolated from these cells. They have however, been sequenced for the common polymorphism and contain the CC allele at -344 and the 'wild type' sequence at intron 2. Based on information in this chapter the sequences at the 3'UTR could be inferred, but direct sequencing would need to be carried out to confirm this.

Bioinformatic analysis in Chapter 4 confirmed that no pri-miRNA or pre-miRNA sequences are located in the genomic region encompassing these genes: ruling out the possibility of an intronic SNP that causes altered miRNA-expression or function. Investigating the other possible *cis*-factors are beyond the scope of this project but, should a *CYP11B1* or *CYP11B2*-regulating miRNA(s) be identified, then exploration of the epigenetic regulation of that miRNA would clearly prove to be of interest (Lehmann *et al.*, 2008;Ryan *et al.*, 2010).

Finally, polymorphic variations need not be located in a miRNA-binding site to influence miRNA-action; there is evidence that 3'UTR SNPs located adjacent to miRNA binding sites have a profound influence on miRNA-mediated regulation. The C829T polymorphism in the 3'UTR of the human *DHFR* (Dihydrofolate reductase) gene is located 14 base-pairs downstream of a verified miR-24 target

site. Cells expressing the T allele (at position 829) DHFR exhibit no repression when miR-24 levels are increased, in contrast with the observed effect in wildtype (C allele at position 829) DHFR cells (Mishra *et al.*, 2007). It is possible that SNPs in the *CYP11B1* and *CYP11B2* genes are acting in a similar manner. Such SNPs could influence miRNA regulation by altering the secondary structure of the 3'UTR, which may in turn affect the miRNA target-site accessibility or prevent the association of the large protein complexes that make up RISC (RNA-induced silencing complex) and which are crucial to miRNA action (Kertesz *et al.*, 2007;Mishra *et al.*, 2007;Jinek and Doudna, 2009).

In summary, miRNA expression levels were shown to be altered in APA samples relative to normal controls, and this could be indicative of their role in the pathogenesis of adrenal disease and in hypertension. Additionally, some miRNA binding sites in the *CYP11B1* and *CYP11B2* genes are located in regions which possess a common genetic variant, while others are located proximal to variants and may also play a role in miRNA regulation. Further *in vitro* studies would be helpful in understanding the influence of both individual SNPs and the full haplotype on miRNA regulation.

7 General Discussion

Essential hypertension (EH) is a common condition and a risk factor for cardiovascular and cerebrovascular disease. The aetiology of EH is unknown but both lifestyle and genetic factors are known to contribute. The genes involved in the adrenal corticosteroid biosynthesis pathway are a major candidate system; understanding the regulation of these genes should increase our current understanding of EH.

There are multiple factors which control the regulation of aldosterone and cortisol production in the adrenal cortex, including those which alter the expression of the CYP11B2 and CYP11B1 enzymes that respectively catalyse the final steps in their production. The roles these enzymes (and their genes) play in the regulation of blood pressure, and their contribution to the development of hypertension, were outlined in <u>Chapter 1</u>. The aim of this project was to investigate the existence of a novel *CYP11B1* and *CYP11B2* regulating mechanism involving miRNAs.

This study is the first to examine, directly, the regulation of *CYP11B1* and *CYP11B2* by miRNAs, and this was accomplished using both direct and indirect methods performed *in vitro*, *in silico* or *ex vivo*. In <u>Chapter 3</u>, the levels of mRNA encoding Dicer mRNA (an important enzyme in the process of miRNA maturation) was reduced in H295R cells. Knockdown of miRNA levels by this method increased the abundance of several key corticosteroidogenic mRNAs (including *CYP11B1* and *CYP11B2*) and of their corresponding steroid products. Cloning of the *CYP11B1* or *CYP11B2* 3'UTR into a reporter construct indicated that these regions were capable of, and sufficient to, exert a negative regulatory effect on the expression of a reporter gene. It was further demonstrated that this regulation is responsive to AngII stimulation. Taken together, these studies supported the initial hypothesis of miRNA-mediated regulation of adrenal corticosteroidogenesis.

The studies in <u>Chapter 4</u> utilised bioinformatic prediction algorithms to identify miRNAs likely to target the 3'UTR of *CYP11B1* and *CYP11B2* mRNAs. Based on miRNA target site prediction and analyses of the 3'UTR sequences (which included relative length, predicted sequence conservation and RNA secondary structure), *in silico* methods concluded that it was possible that miRNAs can target *CYP11B1* and/or *CYP11B2* in this way. Prediction of biological processes by bioinformatic algorithms is generally problematic; the imperfect nature of miRNA binding to the target site complicates it further. The currently available databases are built on our best understanding of miRNA biology, but the novelty and rapid development of the field entails continual modification of their algorithms. Even in the three-year duration of this study, predicted target sites have changed; interestingly my validation studies (Chapter 5) support the more recent predictions suggesting that these updates have, on the whole, tended towards improvements. There will always be an inherent error associated with bioinformatic predictions and this study has identified several false-positive predictions; one would imagine that the frequency of these will be reduced as algorithms are refined. However, this study has not investigated the presence of false-negative target site predictions. Recent findings have identified miRNAmediated regulation of mRNAs that do not possess strict base-pairing complementarity in the seed region (Grimson et al., 2007; Shin et al., 2010). This would, at least, suggest the possibility of false-negative predictions given that many of the databases are designed to favour seed site sequence matches. To exclude this possibility, the binding site properties of all adrenal miRNAs would need to be tested in vitro with CYP11B1 and CYP11B2 3'UTR requiring a lot of time and expense. Therefore, the rationale in this study was to use five databases in order to broaden the prediction and include algorithms of varying stringencies. As these algorithms continue to develop the list of potential targeting miRNAs will need to be updated and tested, as appropriate.

miRNA expression analysis was performed in <u>Chapter 4</u> for normal human adrenal gland samples and in <u>Chapter 6</u> for aldosterone-producing adenoma samples. miRNA microarray and qRT-PCR experiments both identified the presence of specific miRNAs within these tissues; in total, 103 miRNAs were detected above a pre-determined threshold in the normal adrenal gland, compared with 67 in the APA samples. A microarray signal intensity of 500 arbitrary units (AU) was chosen, based on advice from the microarray chip manufacturer that presumably takes into account the scanner's lower limit of detection and the efficiency of hybridisation. Of course several miRNAs expressed marginally below this borderline may still be of importance in adrenal physiology. One obvious reason for this is that a miRNA may only be expressed in a certain adrenal cell type, meaning its concentration is diluted within the heterogeneous adrenal cell populations analysed here. Zonal micro-dissection and subsequent expression analysis (presumable by qRT-PCR, as the quantity of isolated RNA would be low) would help to determine if this is the case. Analysis of the microarray data requires a cut-off value to be selected and enforced in order to avoid both experimental noise and error; the threshold of 500 AU used here enabled the identification of a manageable number of adrenal miRNAs to be carried forward for subsequent investigation. It was also logical to focus initial studies on miRNAs expressed at higher levels given their likelihood of producing more pronounced biological response.

The studies described in <u>Chapter 5</u> involved the *in vitro* overexpression or competitive inhibition of miRNAs in order to assess the validity of miRNA binding sites predicted in the previous chapters. The effect of altered miRNA levels was assessed by reporter construct assays and by direct measurement of mRNA and steroids. These studies were useful for screening the predicted miRNA target sites; several of these were shown to be false positives, while a small number were found to be ambiguous and one showed an action opposite to that generally accepted to be exerted by miRNAs; others were successfully validated. The miRNA that gave the most striking and consistent results for targeting both *CYP11B1* and *CYP11B2* was miR-24. Analysis of adrenal miRNAs predicted only to target the *CYP11B2* 3'UTR confirmed miR-125a-5p and miR-125b as novel regulators, although effects on steroid secretion are still to be assessed.

Expression levels of miR-24 (and of a number of other miRNAs) were shown to be dysregulated in APA samples relative to normal adrenals. This raises the possibility of a role for miR-24 in adrenal pathology and this requires further investigation, possibly utilising *in vivo* studies, for example altering the levels of miR-24 in rats and measuring their steroid production, level of *CYP11B1* and *CYP11B2* expression and cardiovascular parameters.

The experimental work described in this thesis has identified the importance of miRNA-mediated regulation to the adrenal gland. There are several interesting further experiments which now suggest themselves. Firstly, examining the spatial expression of individual miRNAs in adrenal sections would be of interest; there are a number of methods through which this can be achieved, such as by *in situ* hybridisation using either fluorescent probes (Silahtaroglu *et al.*, 2007) or

labelled LNA probes (Nelson *et al.*, 2006). These have been successfully used to identify miRNA localisation in formalin-fixed, paraffin-embedded preserved tissue samples (Nelson *et al.*, 2006) and this is especially pertinent given the difficulty in obtaining good-quality fresh or frozen human adrenal gland sections. Assessment of adrenal miRNA localisation would determine whether a particular miRNA is universally expressed, is confined to a specific zone or exhibits differential expression levels across cell types. The zone-specific expression of *CYP11B1* and *CYP11B2* means that information gained from these experiments could clarify miRNAs' relative impact on these genes.

Secondly, to establish whether miRNAs identified in the present study contribute to other adrenal pathologies. The information gained by doing so may help to further narrow down those miRNAs that target CYP11B2. To investigate miRNA involvement in essential hypertension (EH), it would be desirable to examine miRNA expression in the adrenal glands of EH subjects (specifically those with a high ARR), but practical and ethical consideration make this impossible. Where human studies are impractical, it is usual to turn to animal models, but much is still not known about the similarity and conservation of miRNA-mediated mechanisms across species. While, superficially, it may seem attractive to examine the impact of miR-24 knockout on steroid secretion and blood pressure in a mouse model, many differences exist that could render its relevance to negligible. These include important physiological human difference (corticosterone is the major rodent glucocorticoid), before one even considered whether miRNA target sites or expression patterns are maintained across species. Although this may prove to be the case, careful and extensive studies would have to be conducted before the suitability of such animal models was confirmed.

However, human studies may not be reliant on prohibitively invasive methods in order to analyse miRNA action recent publications have successfully demonstrated that miRNAs levels can be measured in blood (Section 1.8.2). That is miRNAs expressed in one organ can be released into the circulation and travel to act in a different organ, therefore miRNAs may act like a novel endocrine signal. In time, develop into a novel diagnostic tool or permit the generation of miRNA expression profiles indicative of the EH patient. This study has only investigated the role of individual miRNAs in gene regulation but, in normal physiology, it is likely that multiple miRNAs act to regulate the abundance of mRNAs and that the regulation of the miRNAs themselves is crucial to their effect. Furthermore, post-transcriptional regulation by miRNAs may only constitute to a small proportion of the total regulatory influences exerted over the *CYP11B1* and *CYP11B2* genes. Other known or potential factors that contribute to their regulation include epigenetic modulation (*e.g.* methylation or histone acetylation), transcription factors and post-translational mechanisms (e.g. phosphorylation). In addition to the regulation of *CYP11B1* and *CYP11B2*, a greater knowledge miRNA-mediated regulation throughout adrenal gland may contribute to a better understanding of the regulatory pathways that control adrenal development (Huang and Yao, 2010), adrenal cell differentiation and medullary catecholamine synthesis.

In the more distant future, the regulatory pathways identified in this thesis may prove useful as novel diagnostic tools (possibly through stratification of patient groups) and therapies. Currently, there are no specific aldosterone synthase inhibitors on the market; results presented in this thesis show potential for the development and use of miRNA-based therapies to treat disorders of aldosterone production, particularly hypertension. As previously mentioned there are already early-stage clinical trials of miRNA-targeted treatments in progress (Section 1.8.1), and, as the field continues to mature, the range of conditions for which miRNAs are used may become many and diverse.

In summary, the main findings of this thesis include: the identification of the a general mechanism of miRNA-mediated regulation in an adrenal cell line; the determination of miRNA expression profiles for normal adrenal gland and APA samples; the identification of predicted miRNA binding sites in the 3'UTR of *CYP11B1* and *CYP11B2*, and the development of *in vitro* systems to test and validate several of these. The most prominent regulatory mechanism identified in the course of this project involved the direct and canonical regulation of *CYP11B1* and *CYP11B2* mRNA abundance and steroid secretion by miR-24. Additionally, levels of miR-24 were found to be lower in APA samples than in the normal adrenal gland; this signals a possible role for miR-24 in adrenal pathophysiology.

In conclusion, I have, for the first time; demonstrated the direct involvement of miRNAs in the regulation of *CYP11B1* and *CYP11B2* expression. The discovery of this novel regulatory mechanism raises the possibility, in the longer term, of miRNA-based therapies for the treatment of aldosterone excess and essential hypertension. Such therapies permit the specific targeting and fine control of this system, leading to better management of hypertension and, therefore, reducing the incidence of cardiovascular disease.

8 Appendices

Gene	Forward Primer	Reverse Primer	UPL Probe #
B-actin	CCAACCGCGAGAAGATGA	CCAGAGGCGTACAGGGATAG	64
GAPDH	GCTCTCTGCTCCTCCTGTTC	ACGACCAAATCCGTTGACTC	60
StAR	TACGTGGCTACTCAGCATCG	ACCTGGTTGATGATGCTCTTG	83
CYP11A1	AGGAGGGGTGGACACGAC	TTGCGTGCCATCTCATACA	59
CYP21A1	GAGGGCACAGTCATCATTCC	GCTCCAGGAAGCGATCAG	14
CYP17A1	CTATGCTCATCCCCCACAG	TTGTCCACAGCAAACTCACC	67
3BHSDII	AGGCCTTCAGACCAGAATTG	CCTCAAGTACAGTCAGCTTGG	50
CYP11B1	ACTAGGGCCCATTTTCAGGT	GGCAGCATCACACACC	68
CYP11B2	GCACCTGCACCTGGAGATG	CACACACCATGCGTGGTCC	57
HSD11BI	CAATGGAAGCATTGTTGTTCG	GGCAGCAACCATTGGATAAG	20
HSD11BII	GGGTCAAGGTCAGCATCATC	CACTGACCCACGTTTCTCAC	71

Table 8-7-1. Universal Probe Library qRT-PCR Assays.

The UPL Probe # refers to the specific fluorescent probe supplied in the Universal Probe Library Human Set (Roche Applied Sciences) which is unique to the amplicon.

Assay	miRNA Sequence	Product Code
miR-24	UGGCUCAGUUCAGCAGGAACAG	000402
miR-21	UAGCUUAUCAGACUGAUGUUGA	000397
miR-125a-5p	UCCCUGAGACCCUUUAACCUGUGA	002198
miR-10b	UACCCUGUAGAACCGAAUUUGUG	002218
miR-143	UGAGAUGAAGCACUGUAGCUC	002249
RNU48	GATGACCCCAGGTAACTCTGAG TGTGTCGCTGATGCCATCACCG CAGCGCTCTGACC	001006

Table 8-7-2. TaqMan[®] miRNA Assays

|--|

Assay	miRNA Sequence	Product Code
miR-24	UGGCUCAGUUCAGCAGGAACAG	MS00006552
SNORD48		MS00007511



Human *Dicer-1* mRNA structure; exons shown in blue boxes. s23755:siRNA A; s23756:siRNA B

9 References

Adrogue, H.J. and N.E.Madias. 2007. Mechanisms of disease: Sodium and potassium in the pathogenesis of hypertension. *New England Journal of Medicine* 356:1966-1978.

Ahmad, S. and P.E.Ward. 1990. Role of Aminopeptidase Activity in the Regulation of the Pressor Activity of Circulating Angiotensins. *Journal of Pharmacology and Experimental Therapeutics* 252:643-650.

Alvarez-Madrazo, S., S.Padmanabhan, B.M.Mayosi, H.Watkins, P.Avery, A.M.Wallace, R.Fraser, E.Davies, B.Keavney, and J.M.Connell. 2009. Familial and Phenotypic Associations of the Aldosterone Renin Ratio. *Journal of Clinical Endocrinology & Metabolism* 94:4324-4333.

Alzamora, R., L.Michea, and E.T.Marusic. 2000. Role of 11 beta-hydroxysteroid dehydrogenase in nongenomic aldosterone effects in human arteries. *Hypertension* 35:1099-1104.

Ambros, V., B.Bartel, D.P.Bartel, C.B.Burge, J.C.Carrington, X.M.Chen, G.Dreyfuss, S.R.Eddy, S.Griffiths-Jones, M.Marshall, M.Matzke, G.Ruvkun, and T.Tuschl. 2003. A uniform system for microRNA annotation. *Rna-A Publication of the Rna Society* 9:277-279.

Arakane, F., T.Sugawara, H.Nishino, Z.M.Liu, J.A.Holt, D.Pain, D.M.Stocco, W.L.Miller, and J.F.Strauss. 1996. Steroidogenic acute regulatory protein (StAR) retains activity in the absence of its mitochondrial import sequence: Implications for the mechanism of StAR action. *Proceedings of the National Academy of Sciences of the United States of America* 93:13731-13736.

Arriza, J.L., C.Weinberger, G.Cerelli, T.M.Glaser, B.L.Handelin, D.E.Housman, and R.M.Evans. 1987. Cloning of Human Mineralocorticoid Receptor Complementary-Dna - Structural and Functional Kinship with the Glucocorticoid Receptor. *Science* 237:268-275.

Asher, C., H.Wald, B.C.Rossier, and H.Garty. 1996. Aldosterone-induced increase in the abundance of Na+ channel subunits. *American Journal of Physiology-Cell Physiology* 40:C605-C611.

Bagga, S., J.Bracht, S.Hunter, K.Massirer, J.Holtz, R.Eachus, and A.E.Pasquinelli. 2005. Regulation by let-7 and lin-4 miRNAs results in target mRNA degradation. *Cell* 122:553-563.

Ballard, P.L., J.D.Baxter, S.J.Higgins, G.G.Rousseau, and G.M.Tomkins. 1974. General Presence of Glucocorticoid Receptors in Mammalian-Tissues. *Endocrinology* 94:998-1002.

Barr, M., S.M.Mackenzie, E.C.Friel, C.D.Holloway, D.M.Wilkinson, N.J.R.Brain, M.C.Ingram, R.Fraser, M.Brown, N.J.Samani, M.Caulfield, P.B.Munroe, M.Farrall, J.Webster, D.Clayton, A.F.Dominiczak, J.M.C.Connell, and E.Davies. 2007. Polymorphic variation in the 11 beta-hydroxylase gene associates with reduced 11-hydroxylase efficiency. *Hypertension* 49:113-119. Barr, M., S.M.Mackenzie, D.M.Wilkinson, C.D.Holloway, E.C.Friel, S.Miller, T.MacDonald, R.Fraser, J.M.C.Connell, and E.Davies. 2006. Functional effects of genetic variants in the 11 beta-hydroxylase (CYP11B1) gene. *Clinical Endocrinology* 65:816-825.

Barrett, J.C., B.Fry, J.Maller, and M.J.Daly. 2005. Haploview: analysis and visualization of LD and haplotype maps. *Bioinformatics* 21:263-265.

Bartel, D.P. 2009. MicroRNAs: Target Recognition and Regulatory Functions. *Cell* 136:215-233.

Bashkirov, V.I., H.Scherthan, J.A.Solinger, J.M.Buerstedde, and W.D.Heyer. 1997. A mouse cytoplasmic exoribonuclease (mXRN1p) with preference for G4 tetraplex substrates. *Journal of Cell Biology* 136:761-773.

Bassett, M.H., P.C.White, and W.E.Rainey. 2004. The regulation of aldosterone synthase expression. *Molecular and Cellular Endocrinology* 217:67-74.

Bassett, M.H., Y.Zhang, C.Clyne, P.C.White, and W.E.Rainey. 2002. Differential regulation of aldosterone synthase and 11 beta-hydroxylase transcription by steroidogenic factor-1. *Journal of Molecular Endocrinology* 28:125-135.

Bassett, M.H., Y.Zhang, P.C.White, and W.E.Rainey. 2000. Regulation of human CYP11B2 and CYP11B1: Comparing the role of the common CRE/Ad1 element. *Endocrine Research* 26:941-951.

Beguin, P., G.Crambert, S.Guennoun, H.Garty, J.D.Horisberger, and K.Geering. 2001. CHIF, a member of the FXYD protein family, is a regulator of Na,K-ATPase distinct from the gamma-subunit. *Embo Journal* 20:3993-4002.

Behm-Ansmant, I., J.Rehwinkel, T.Doerks, A.Stark, P.Bork, and E.Izaurralde. 2006. MRNA degradation by miRNAs and GW182 requires both CCR4 : NOT deadenylase and DCP1 : DCP2 decapping complexes. *Genes & Development* 20:1885-1898.

Bentwich, I. 2005. Prediction and validation of microRNAs and their targets. *Febs Letters* 579:5904-5910.

Bentwich, I., A.Avniel, Y.Karov, R.Aharonov, S.Gilad, O.Barad, A.Barzilai, P.Einat, U.Einav, E.Meiri, E.Sharon, Y.Spector, and Z.Bentwich. 2005. Identification of hundreds of conserved and nonconserved human microRNAs. *Nature Genetics* 37:766-770.

Berezikov, E., W.J.Chung, J.Willis, E.Cuppen, and E.C.Lai. 2007. Mammalian mirtron genes. *Molecular Cell* 28:328-336.

Berezikov, E., V.Guryev, J.van de Belt, E.Wienholds, R.H.A.Plasterk, and E.Cuppen. 2005. Phylogenetic shadowing and computational identification of human microRNA genes. *Cell* 120:21-24.

Bernstein, E., S.Y.Kim, M.A.Carmell, E.P.Murchison, H.Alcorn, M.Z.Li, A.A.Mills, S.J.Elledge, K.V.Anderson, and G.J.Hannon. 2003. Dicer is essential for mouse development. *Nature Genetics* 35:215-217.

Berube, D., V.L.The, Y.Lachance, R.Gagne, and F.Labrie. 1989. The Gene Encoding Human 3-Beta-Hydroxysteroid Dehydrogenase Isomerase Mapped to the P13-Band of Chromosome-1. *Cytogenetics and Cell Genetics* 51:962.

Betel, D., A.Koppal, P.Agius, C.Sander, and C.Leslie. 2010. Comprehensive modeling of microRNA targets predicts functional non-conserved and non-canonical sites. *Genome Biology* 11.

Bhalla, V., R.Soundararajan, A.C.Pao, H.Y.Li, and D.Pearce. 2006. Disinhibitory pathways for control of sodium transport: regulation of ENaC by SGK1 and GILZ. *American Journal of Physiology-Renal Physiology* 291:F714-F721.

Bhattacharyya, S.N., R.Habermacher, U.Martine, E.I.Closs, and W.Filipowicz. 2006. Relief of microRNA-mediated translational repression in human cells subjected to stress. *Cell* 125:1111-1124.

Biglieri, E.G., M.A.Herron, and N.Brust. 1966. 17-Hydroxylation Deficiency in Man. *Journal of Clinical Investigation* 45:1946-&.

Bimpaki, E.I., D.Iliopoulos, A.Moraitis, and C.A.Stratakis. 2010. MicroRNA signature in massive macronodular adrenocortical disease and implications for adrenocortical tumourigenesis. *Clinical Endocrinology* 72:744-751.

Binder, A. 2007. A review of the genetics of essential hypertension. *Current Opinion in Cardiology* 22:176-184.

Bird, I.M., N.A.Hanley, R.A.Word, J.M.Mathis, J.L.Mccarthy, J.I.Mason, and W.E.Rainey. 1993. Human Nci-H295 Adrenocortical Carcinoma-Cells - A Model for Angiotensin-Ii-Responsive Aldosterone Secretion. *Endocrinology* 133:1555-1561.

Biron, P., J.G.Mongeau, and D.Bertrand. 1976. Familial Aggregation of Blood-Pressure in 558 Adopted-Children. *Canadian Medical Association Journal* 115:773-774.

Blow, M.J., R.J.Grocock, S.van Dongen, A.J.Enright, E.Dicks, P.A.Futreal, R.Wooster, and M.R.Stratton. 2006. RNA editing of human microRNAs. *Genome Biology* 7.

Blumenfeld, J.D., J.E.Sealey, Y.Schlussel, E.D.Vaughan, T.A.Sos, S.A.Atlas, F.B.Muller, R.Acevedo, S.Ulick, and J.H.Laragh. 1994. Diagnosis and Treatment of Primary Hyperaldosteronism. *Annals of Internal Medicine* 121:877-885.

Bohmer, S., C.Carapito, B.Wilzewski, E.Leize, A.Van Dorsselaer, and R.Bernhardt. 2006. Analysis of aldosterone-induced differential receptorindependent protein patterns using 2D-electrophoresis and mass spectrometry. *Biological Chemistry* 387:917-929.

Borchert, G.M., W.Lanier, and B.L.Davidson. 2006. RNA polymerase III transcribes human microRNAs. *Nature Structural & Molecular Biology* 13:1097-1101.

Brand, E., N.Chatelain, P.Mulatero, I.Fery, K.Curnow, X.Jeunemaitre, P.Corvol, L.Pascoe, and F.Soubrier. 1998. Structural analysis and evaluation of the aldosterone synthase gene in hypertension. *Hypertension* 32:198-204.

Brennan, F.E. and P.J.Fuller. 1999. Acute regulation by corticosteroids of channel-inducing factor gene messenger ribonucleic acid in the distal colon. *Endocrinology* 140:1213-1218.

Brilla, C.G. and K.T.Weber. 1992. Mineralocorticoid Excess, Dietary-Sodium, and Myocardial Fibrosis. *Journal of Laboratory and Clinical Medicine* 120:893-901.

Bureik, M., A.Zollner, N.Schuster, M.Montenarh, and R.Bernhardt. 2005. Phosphorylation of bovine Adrenodoxin by protein kinase CK2 affects the interaction with its redox partner cytochrome p450(scc) (CYP11A1). *Biochemistry* 44:3821-3830.

Burton, P.R., D.G.Clayton, L.R.Cardon, N.Craddock, P.Deloukas, A.Duncanson, D.P.Kwiatkowski, M.I.McCarthy, W.H.Ouwehand, N.J.Samani, J.A.Todd, P.Donnelly, J.C.Barrett, D.Davison, D.Easton, D.Evans, H.T.Leung, J.L.Marchini, A.P.Morris, C.C.A.Spencer, M.D.Tobin, A.P.Attwood, J.P.Boorman, B.Cant, U.Everson, J.M.Hussey, J.D.Jolley, A.S.Knight, K.Koch, E.Meech, S.Nutland, C.V.Prowse, H.E.Stevens, N.C.Taylor, G.R.Walters, N.M.Walker, N.A.Watkins, T.Winzer, R.W.Jones, W.L.McArdle, S.M.Ring, D.P.Strachan, M.Pembrey, G.Breen, D.St Clair, S.Caesar, K.Gordon-Smith, L.Jones, C.Fraser, E.K.Green, D.Grozeva, M.L.Hamshere, P.A.Holmans, I.R.Jones, G.Kirov, V.Moskvina, I.Nikolov, M.C.O'Donovan, M.J.Owen, D.A.Collier, A.Elkin, A.Farmer, R.Williamson, P.McGuffin, A.H.Young, I.N.Ferrier, S.G.Ball, A.J.Balmforth, J.H.Barrett, D.T.Bishop, M.M.Iles, A.Magbool, N.Yuldasheva, A.S.Hall, P.S.Braund, R.J.Dixon, M.Mangino, S.Stevens, J.R.Thompson, F.Bredin, M. Tremelling, M. Parkes, H. Drummond, C. W. Lees, E. R. Nimmo, J. Satsangi, S.A.Fisher, A.Forbes, C.M.Lewis, C.M.Onnie, N.J.Prescott, J.Sanderson, C.G.Mathew, J.Barbour, M.K.Mohiuddin, C.E.Todhunter, J.C.Mansfield, T.Ahmad, F.R.Cummings, D.P.Jewell, J.Webster, M.J.Brown, G.M.Lathrop, J.Connell, A.Dominiczak, C.A.B.Marcano, B.Burke, R.Dobson, J.Gungadoo, K.L.Lee, P.B.Munroe, S.J.Newhouse, A.Onipinla, C.Wallace, M.Z.Xue, M.Caulfield, M.Farrall, A.Barton, I.N.Bruce, H.Donovan, S.Eyre, P.D.Gilbert, S.L.Hider, A.M.Hinks, S.L.John, C.Potter, A.J.Silman, D.P.M.Symmons, W.Thomson, J.Worthington, D.B.Dunger, B.Widmer, T.M.Frayling, R.M.Freathy, H.Lango, J.R.B.Perry, B.M.Shields, M.N.Weedon, A.T.Hattersley, K.S.Elliott, C.J.Groves, C.M.Lindgren, N.W.Rayner, N.J.Timpson, E.Zeggini, M.Newport, G.Sirugo, E.Lyons, F.Vannberg, M.A.Brown, J.A.Franklyn, J.M.Heward, M.J.Simmonds, A.V.S.Hill, L.A.Bradbury, C.Farrar, J.J.Pointon, P.Wordsmith, S.C.L.Gough, S.Seal, M.R.Stratton, N.Rahman, M.Ban, A.Goris, S.J.Sawcer, A.Compston, D.Conway, M.Jallow, S.J.Bumpstead, A.Chaney, K.Downes, M.J.R.Ghori, R.Gwilliam, M.Inouye, A.Keniry, E.King, R.McGinnis, S.Potter, R.Ravindrarajah, P.Whittaker, D.Withers, D.Easton, J.Pereira-Gale, I.B.Hallgrimsdottir, B.N.Howie, Z.Su, Y.Y.Teo, D.Vukcevic, D.Bentley, M.Caulfield, C.G.Mathew, and J.Worthington. 2007. Genome-wide association study of 14,000 cases of seven common diseases and 3,000 shared controls. Nature 447:661-678.

Butterworth, M.B., R.S.Edinger, J.P.Johnson, and R.A.Frizzell. 2005. Acute ENaC stimulation by cAMP in a kidney cell line is mediated by exocytic insertion from a recycling channel pool. *Journal of General Physiology* 125:81-101.

Calin, G.A., C.D.Dumitru, M.Shimizu, R.Bichi, S.Zupo, E.Noch, H.Aldler, S.Rattan, M.Keating, K.Rai, L.Rassenti, T.Kipps, M.Negrini, F.Bullrich, and

C.M.Croce. 2002. Frequent deletions and down-regulation of micro-RNA genes miR15 and miR16 at 13q14 in chronic lymphocytic leukemia. *Proceedings of the National Academy of Sciences of the United States of America* 99:15524-15529.

Calin, G.A., M.Ferracin, A.Cimmino, G.Di Leva, M.Shimizu, S.E.Wojcik, M.V.Iorio, R.Visone, N.I.Sever, M.Fabbri, R.Iuliano, T.Palumbo, F.Pichiorri, C.Roldo, R.Garzon, C.Sevignani, L.Rassenti, H.Alder, S.Volinia, C.G.Liu, T.J.Kipps, M.Negrini, and C.M.Croce. 2005. A MicroRNA signature associated with prognosis and progression in chronic lymphocytic leukemia. *New England Journal of Medicine* 353:1793-1801.

Cammarota, M., L.R.M.Bevilaqua, P.R.Dunkley, and J.A.P.Rostas. 2001. Angiotensin II promotes the phosphorylation of cyclic AMP-responsive element binding protein (CREB) at Ser133 through an ERK1/2-dependent mechanism. *Journal of Neurochemistry* 79:1122-1128.

Caron, K.M., S.C.Soo, W.C.Wetsel, D.M.Stocco, B.J.Clark, and K.L.Parker. 1997. Targeted disruption of the mouse gene encoding steroidogenic acute regulatory protein provides insights into congenital lipoid adrenal hyperplasia. *Proceedings* of the National Academy of Sciences of the United States of America 94:11540-11545.

Carretero, O.A. and S.Oparil. 2000. Essential hypertension Part I: Definition and etiology. *Circulation* 101:329-335.

Cat, A.N.D., V.Griol-Charhbili, L.Loufrani, C.Labat, L.Benjamin, N.Farman, P.Lacolley, D.Henrion, and F.Jaisser. 2010. The endothelial mineralocorticoid receptor regulates vasoconstrictor tone and blood pressure. *Faseb Journal* 24:2454-2463.

Caulfield, M., P.Munroe, J.Pembroke, N.Samani, A.Dominiczak, M.Brown, N.Benjamin, J.Webster, P.Ratcliffe, S.O'Shea, J.Papp, E.Taylor, R.Dobson, J.Knight, S.Newhouse, J.Hooper, W.Lee, N.Brain, D.Clayton, G.M.Lathrop, M.Farrall, and J.Connell. 2003. Genome-wide mapping of human loci for essential hypertension. *Lancet* 361:2118-2123.

Chatterjee, S. and H.Grosshans. 2009. Active turnover modulates mature microRNA activity in Caenorhabditis elegans. *Nature* 461:546-U120.

Cheloufi, S., C.O.Dos Santos, M.M.W.Chong, and G.J.Hannon. 2010. A Dicerindependent miRNA biogenesis pathway that requires Ago catalysis. *Nature* 465:584-U76.

Chen, J.F., E.P.Murchison, R.Tang, T.E.Callis, M.Tatsuguchi, Z.Deng, M.Rojas, S.M.Hammond, M.D.Schneider, C.H.Selzman, G.Meissner, C.Patterson, G.J.Hannon, and D.Z.Wang. 2008a. Targeted deletion of Dicer in the heart leads to dilated cardiomyopathy and heart failure. *Proceedings of the National Academy of Sciences of the United States of America* 105:2111-2116.

Chen, S.Y., A.Bhargava, L.Mastroberardino, O.C.Meijer, J.Wang, P.Buse, G.L.Firestone, F.Verrey, and D.Pearce. 1999. Epithelial sodium channel regulated by aldosterone-induced protein sgk. *Proceedings of the National Academy of Sciences of the United States of America* 96:2514-2519.

Chen, X., Y.Ba, L.J.Ma, X.Cai, Y.Yin, K.H.Wang, J.G.Guo, Y.J.Zhang, J.N.Chen, X.Guo, Q.B.Li, X.Y.Li, W.J.Wang, Y.Zhang, J.Wang, X.Y.Jiang, Y.Xiang, C.Xu, P.P.Zheng, J.B.Zhang, R.Q.Li, H.J.Zhang, X.B.Shang, T.Gong, G.Ning, J.Wang, K.Zen, J.F.Zhang, and C.Y.Zhang. 2008b. Characterization of microRNAs in serum: a novel class of biomarkers for diagnosis of cancer and other diseases. *Cell Research* 18:997-1006.

Cheng, A.M., M.W.Byrom, J.Shelton, and L.P.Ford. 2005. Antisense inhibition of human miRNAs and indications for an involvement of miRNA in cell growth and apoptosis. *Nucleic Acids Research* 33:1290-1297.

Chhabra, R., R.Dubey, and N.Saini. 2010. Cooperative and individualistic functions of the microRNAs in the miR-23a similar to 27a similar to 24-2 cluster and its implication in human diseases. *Molecular Cancer* 9.

Chi, S.W., J.B.Zang, A.Mele, and R.B.Darnell. 2009. Argonaute HITS-CLIP decodes microRNA-mRNA interaction maps. *Nature* 460:479-486.

Chobanian, A.V., G.L.Bakris, H.R.Black, W.C.Cushman, L.A.Green, J.L.Izzo, D.W.Jones, B.J.Materson, S.Oparil, J.T.Wright, and E.J.Roccella. 2003. Seventh Report of the Joint National Committee on Prevention, Detection, Evaluation, and Treatment of High Blood Pressure. *Hypertension* 42:1206-1252.

Christ, M., K.Douwes, C.Eisen, G.Bechtner, K.Theisen, and M.Wehling. 1995. Rapid Effects of Aldosterone on Sodium-Transport in Vascular Smooth-Muscle Cells. *Hypertension* 25:117-123.

Chu, C.Y. and T.M.Rana. 2006. Translation repression in human cells by microRNA-induced gene silencing requires RCK/p54. *Plos Biology* 4:1122-1136.

Cifuentes, D., H.L.Xue, D.W.Taylor, H.Patnode, Y.Mishima, S.Cheloufi, E.B.Ma, S.Mane, G.J.Hannon, N.D.Lawson, S.A.Wolfe, and A.J.Giraldez. 2010. A Novel miRNA Processing Pathway Independent of Dicer Requires Argonaute2 Catalytic Activity. *Science* 328:1694-1698.

Clyne, C.D., Y.Zhang, L.Slutsker, J.M.Mathis, P.C.White, and W.E.Rainey. 1997. Angiotensin II and potassium regulate human CYP11B2 transcription through common cis-elements. *Molecular Endocrinology* 11:638-649.

Condon, J.C., V.Pezzi, B.M.Drummond, S.Yin, and W.E.Rainey. 2002. Calmodulin-dependent kinase I regulates adrenal cell expression of aldosterone synthase. *Endocrinology* 143:3651-3657.

Conn, J.W. 1955. Painting Background .1. Primary Aldosteronism, A New Clinical Syndrome .2. *Journal of Laboratory and Clinical Medicine* 45:3-17.

Connell, J.M.C., C.J.Kenyon, M.Ingram, C.Holloway, A.Jamieson, M.Panarelli, G.Inglis, and R.Fraser. 1996. Corticosteroids in essential hypertension: Multiple candidate loci and phenotypic variation. *Clinical and Experimental Pharmacology and Physiology* 23:369-374.

Couzin, J. 2002. Breakthrough of the year. Small RNAs make big splash. *Science* 298:2296-2297.

Cowley, A.W. 2006. The genetic dissection of essential hypertension. *Nature Reviews Genetics* 7:829-840.

Curnow, K.M., M.T.Tusieluna, L.Pascoe, R.Natarajan, J.L.Gu, J.L.Nadler, and P.C.White. 1991. The Product of the Cyp11B2 Gene Is Required for Aldosterone Biosynthesis in the Human Adrenal-Cortex. *Molecular Endocrinology* 5:1513-1522.

Czirjak, G. and P.Enyedi. 2002. Formation of functional heterodimers between the TASK-1 and TASK-3 two-pore domain potassium channel subunits. *Journal of Biological Chemistry* 277:5426-5432.

Czirjak, G., T.Fischer, A.Spat, F.Lesage, and P.Enyedi. 2000. TASK (TWIK-related acid-sensitive K+ channel) is expressed in glomerulosa cells of rat adrenal cortex and inhibited by angiotensin II. *Molecular Endocrinology* 14:863-874.

Davies, E., C.D.Holloway, M.C.Ingram, G.C.Inglis, E.C.Friel, C.Morrison, N.H.Anderson, R.Fraser, and J.M.C.Connell. 1999. Aldosterone excretion rate and blood pressure in essential hypertension are related to polymorphic differences in the aldosterone synthase gene CYP11B2. *Hypertension* 33:703-707.

Davies, E., S.M.Mackenzie, E.M.Freel, S.Alvarez-Madrazo, R.Fraser, and J.M.Connell. 2009. Altered corticosteroid biosynthesis in essential hypertension: A digenic phenomenon. *Mol. Cell Endocrinol*. 300:185-191.

Davis, B.N., A.C.Hilyard, G.Lagna, and A.Hata. 2008. SMAD proteins control DROSHA-mediated microRNA maturation. *Nature* 454:56-U2.

de Simone, G., A.P.Tommaselli, R.Rossi, R.Valentino, R.Lauria, F.Scopacasa, and G.Lombardi. 1985. Partial Deficiency of Adrenal 11-Hydroxylase - A Possible Cause of Primary Hypertension. *Hypertension* 7:204-210.

Debonneville, C., S.Y.Flores, E.Kamynina, P.J.Plant, C.Tauxe, M.A.Thomas, C.Munster, A.Chraibi, J.H.Pratt, J.D.Horisberger, D.Pearce, J.Loffing, and O.Staub. 2001. Phosphorylation of Nedd4-2 by Sgk1 regulates epithelial Na+ channel cell surface expression. *Embo Journal* 20:7052-7059.

Delles, C., M.W.McBride, D.Graham, S.Padmanabhan, and A.F.Dominiczak. 2010. Genetics of hypertension: From experimental animals to humans. *Biochimica et Biophysica Acta-Molecular Basis of Disease* 1802:1299-1308.

Denner, K., J.Doehmer, and R.Bernhardt. 1995. Cloning of Cyp11B1 and Cyp11B2 from Normal Human Adrenal and Their Functional Expression in Cos-7 and V79 Chinese-Hamster Cells. *Endocrine Research* 21:443-448.

Denner, K., W.E.Rainey, V.Pezzi, I.M.Bird, R.Bernhardt, and J.M.Mathis. 1996. Differential regulation of 11 beta-hydroxylase and aldosterone synthase in human adrenocortical H295R cells. *Molecular and Cellular Endocrinology* 121:87-91.

Diaz-Perales, A., V.Quesada, L.M.Sanchez, A.P.Ugalde, M.F.Suarez, A.Fueyo, and C.Lopez-Otin. 2005. Identification of human aminopeptidase O, a novel metalloprotease with structural similarity to aminopeptidase B and leukotriene A(4) hydrolase. *Journal of Biological Chemistry* 280:14310-14317.

Dickson, M.E. and C.D.Sigmund. 2006. Genetic basis of hypertension - Revisiting angiotensinogen. *Hypertension* 48:14-20.

Diederichs, S. and D.A.Haber. 2007. Dual role for argonautes in MicroRNA processing and Posttranscriptional regulation of MicroRNA expression. *Cell* 131:1097-1108.

Doench, J.G. and P.A.Sharp. 2004. Specificity of microRNA target selection in translational repression. *Genes & Development* 18:504-511.

Doolan, C.M., G.C.O'Sullivan, and B.J.Harvey. 1998. Rapid effects of corticosteroids on cytosolic protein kinase C and intracellular calcium concentration in human distal colon. *Molecular and Cellular Endocrinology* 138:71-79.

Drouin, J., Y.L.Sun, M.Chamberland, Y.Gauthier, A.Delean, M.Nemer, and T.J.Schmidt. 1993. Novel Glucocorticoid Receptor Complex with Dna Element of the Hormone-Repressed Pomc Gene. *Embo Journal* 12:145-156.

Drumm, K., T.R.Kress, B.Gassner, A.W.Krug, and M.Gekle. 2006. Aldosterone stimulates activity and surface expression of NHE3 in human primary renal proximal tubule epithelial cells (RPTEC). *Cellular Physiology and Biochemistry* 17:21-28.

Duan, R.H., C.H.Pak, and P.Jin. 2007. Single nucleotide polymorphism associated with mature miR-125a alters the processing of pri-miRNA. *Human Molecular Genetics* 16:1124-1131.

Dudley, D.T., R.L.Panek, T.C.Major, G.H.Lu, R.F.Bruns, B.A.Klinkefus, J.C.Hodges, and R.E.Weishaar. 1990. Subclasses of Angiotensin-Ii Binding-Sites and Their Functional-Significance. *Molecular Pharmacology* 38:370-377.

Duursma, A.M., M.Kedde, M.Schrier, C.Le Sage, and R.Agami. 2008. miR-148 targets human DNMT3b protein coding region. *Rna-A Publication of the Rna Society* 14:872-877.

Ebert, M.S., J.R.Neilson, and P.A.Sharp. 2007. MicroRNA sponges: competitive inhibitors of small RNAs in mammalian cells. *Nature Methods* 4:721-726.

Edelman, I.S., G.A.Porter, and R.Bogoroch. 1963. On Mechanism of Action of Aldosterone on Sodium Transport - Role of Protein Synthesis. *Proceedings of the National Academy of Sciences of the United States of America* 50:1169-&.

Edwards, C.R.W., D.Burt, M.A.Mcintyre, E.R.Dekloet, P.M.Stewart, L.Brett, W.S.Sutanto, and C.Monder. 1988. Localization of 11-Beta-Hydroxysteroid Dehydrogenase Tissue Specific Protector of the Mineralocorticoid Receptor. *Lancet* 2:986-989.

Ehret, G.B. 2010. Genome-Wide Association Studies: Contribution of Genomics to Understanding Blood Pressure and Essential Hypertension. *Current Hypertension Reports* 12:17-25.

Elmen, J., M.Lindow, S.Schutz, M.Lawrence, A.Petri, S.Obad, M.Lindholm, M.Hedtjarn, H.F.Hansen, U.Berger, S.Gullans, P.Kearney, P.Sarnow,

E.M.Straarup, and S.Kauppinen. 2008. LNA-mediated microRNA silencing in nonhuman primates. *Nature* 452:896-U10.

Elton, T.S., S.E.Sansom, and M.M.Martin. 2010. Cardiovascular Disease, Single Nucleotide Polymorphisms; and the Renin Angiotensin System: Is There a MicroRNA Connection? *Int. J. Hypertens.* 2010.

Eulalio, A., E.Huntzinger, and E.Izaurralde. 2008. GW182 interaction with Argonaute is essential for miRNA-mediated translational repression and mRNA decay. *Nature Structural & Molecular Biology* 15:346-353.

Farazi, T.A., S.A.Juranek, and T.Tuschl. 2008. The growing catalog of small RNAs and their association with distinct Argonaute/Piwi family members. *Development* 135:1201-1214.

Farazi, T.A., J.I.Spitzer, P.Morozov, and T.Tuschl. 2011. miRNAs in human cancer. *Journal of Pathology* 223:102-115.

Feinleib, M., R.J.Garrison, R.Fabsitz, J.C.Christian, Z.Hrubec, N.O.Borhani, W.B.Kannel, R.Rosenman, J.T.Schwartz, and J.O.Wagner. 1977. Nhlbi Twin Study of Cardiovascular-Disease Risk-Factors - Methodology and Summary of Results. *American Journal of Epidemiology* 106:284-295.

Felli, N., L.Fontana, E.Pelosi, R.Botta, D.Bonci, F.Facchiano, F.Liuzzi, V.Lulli, O.Morsilli, S.Santoro, M.Valtieri, G.A.Calin, C.G.Liu, A.Sorrentino, C.M.Croce, and C.Peschle. 2005. MicroRNAs 221 and 222 inhibit normal erythropoiesis and erythroleukemic cell growth via kit receptor down-modulation. *Proceedings of the National Academy of Sciences of the United States of America* 102:18081-18086.

Fire, A., S.Q.Xu, M.K.Montgomery, S.A.Kostas, S.E.Driver, and C.C.Mello. 1998. Potent and specific genetic interference by double-stranded RNA in Caenorhabditis elegans. *Nature* 391:806-811.

Forman, J.J., A.Legesse-Miller, and H.A.Coller. 2008. A search for conserved sequences in coding regions reveals that the let-7 microRNA targets Dicer within its coding sequence. *Proceedings of the National Academy of Sciences of the United States of America* 105:14879-14884.

Fraser, R., M.C.Ingram, N.H.Anderson, C.Morrison, E.Davies, and J.M.C.Connell. 1999. Cortisol effects on body mass, blood pressure, and cholesterol in the general population. *Hypertension* 33:1364-1368.

Friedlander, M.R., W.Chen, C.Adamidi, J.Maaskola, R.Einspanier, S.Knespel, and N.Rajewsky. 2008. Discovering microRNAs from deep sequencing data using miRDeep. *Nature Biotechnology* 26:407-415.

Friedman, R.C., K.K.H.Farh, C.B.Burge, and D.P.Bartel. 2009. Most mammalian mRNAs are conserved targets of microRNAs. *Genome Research* 19:92-105.

Fukuda, T., K.Yamagata, S.Fujiyama, T.Matsumoto, I.Koshida, K.Yoshimura, M.Mihara, M.Naitou, H.Endoh, T.Nakamura, C.Akimoto, Y.Yamamoto, T.Katagiri, C.Foulds, S.Takezawa, H.Kitagawa, K.Takeyama, B.W.O'Malley, and S.Kato.

2007. DEAD-box RNA helicase subunits of the Drosha complex are required for processing of rRNA and a subset of microRNAs. *Nature Cell Biology* 9:604-U221.

Funder, J.W. 1997. Glucocorticoid and mineralocorticoid receptors: Biology and clinical relevance. *Annual Review of Medicine* 48:231-240.

Funder, J.W. 2005. The nongenomic actions of aldosterone. *Endocrine Reviews* 26:313-321.

Funder, J.W., R.M.Carey, C.Fardella, C.E.Gomez-Sanchez, F.Mantero, M.Stowasser, W.F.Young, and V.M.Montori. 2008. Case detection, diagnosis, and treatment of patients with primary aldosteronism: An endocrine society clinical practice guideline. *Journal of Clinical Endocrinology & Metabolism* 93:3266-3281.

Funder, J.W., P.T.Pearce, R.Smith, and A.I.Smith. 1988. Mineralocorticoid Action - Target Tissue-Specificity Is Enzyme, Not Receptor, Mediated. *Science* 242:583-585.

Garofalo, M., G.Di Leva, G.Romano, G.Nuovo, S.S.Suh, A.Ngankeu, C.Taccioli, F.Pichiorri, H.Alder, P.Secchiero, P.Gasparini, A.Gonelli, S.Costinean, M.Acunzo, G.Condorelli, and C.M.Croce. 2009. miR-221&222 Regulate TRAIL Resistance and Enhance Tumorigenicity through PTEN and TIMP3 Downregulation. *Cancer Cell* 16:498-509.

Garzon, R., G.Marcucci, and C.M.Croce. 2010. Targeting microRNAs in cancer: rationale, strategies and challenges. *Nature Reviews Drug Discovery* 9:775-789.

Gazdar, A.F., H.K.Oie, C.H.Shackleton, T.R.Chen, T.J.Triche, C.E.Myers, G.P.Chrousos, M.F.Brennan, C.A.Stein, and R.V.Larocca. 1990. Establishment and Characterization of A Human Adrenocortical Carcinoma Cell-Line That Expresses Multiple Pathways of Steroid-Biosynthesis. *Cancer Research* 50:5488-5496.

Gekle, M., N.Golenhofen, H.Oberleithner, and S.Silbernagl. 1996. Rapid activation of Na+/H+ exchange by aldosterone in renal epithelial cells requires Ca2+ and stimulation of a plasma membrane proton conductance. *Proceedings of the National Academy of Sciences of the United States of America* 93:10500-10504.

Ghildiyal, M. and P.D.Zamore. 2009. Small silencing RNAs: an expanding universe. *Nature Reviews Genetics* 10:94-108.

Gomez-Sanchez, C.E. and E.P.Gomez-Sanchez. 2010. Aldosterone-Producing Adenomas Mining for Genes. *Hypertension* 55:1306-1307.

Gregory, R.I., T.P.Chendrimada, N.Cooch, and R.Shiekhattar. 2005. Human RISC couples microRNA biogenesis and posttranscriptional gene silencing. *Cell* 123:631-640.

Gregory, R.I., K.P.Yan, G.Amuthan, T.Chendrimada, B.Doratotaj, N.Cooch, and R.Shiekhattar. 2004. The Microprocessor complex mediates the genesis of microRNAs. *Nature* 432:235-240.

Griffiths-Jones, S. 2004. The microRNA Registry. *Nucleic Acids Research* 32:D109-D111.

Griffiths-Jones, S., A.Bateman, M.Marshall, A.Khanna, and S.R.Eddy. 2003. Rfam: an RNA family database. *Nucleic Acids Research* 31:439-441.

Griffiths-Jones, S., R.J.Grocock, S.van Dongen, A.Bateman, and A.J.Enright. 2006. miRBase: microRNA sequences, targets and gene nomenclature. *Nucleic Acids Research* 34:D140-D144.

Griffiths-Jones, S., H.K.Saini, S.van Dongen, and A.J.Enright. 2008. miRBase: tools for microRNA genomics. *Nucleic Acids Research* 36:D154-D158.

Grimson, A., K.K.H.Farh, W.K.Johnston, P.Garrett-Engele, L.P.Lim, and D.P.Bartel. 2007. MicroRNA targeting specificity in mammals: Determinants beyond seed pairing. *Molecular Cell* 27:91-105.

Grishok, A., A.E.Pasquinelli, D.Conte, N.Li, S.Parrish, I.Ha, D.L.Baillie, A.Fire, G.Ruvkun, and C.C.Mello. 2001. Genes and mechanisms related to RNA interference regulate expression of the small temporal RNAs that control C-elegans developmental timing. *Cell* 106:23-34.

Grossmann, C. and M.Gekle. 2009. New aspects of rapid aldosterone signaling. *Molecular and Cellular Endocrinology* 308:53-62.

Guo, H.L., N.T.Ingolia, J.S.Weissman, and D.P.Bartel. 2010. Mammalian microRNAs predominantly act to decrease target mRNA levels. *Nature* 466:835-U66.

Gwynne, J.T. and J.F.Strauss. 1982. The Role of Lipoproteins in Steroidogenesis and Cholesterol-Metabolism in Steroidogenic Glands. *Endocrine Reviews* 3:299-329.

Haase, A.D., L.Jaskiewicz, H.D.Zhang, S.Laine, R.Sack, A.Gatignol, and W.Filipowicz. 2005. TRBP, a regulator of cellular PKR and HIV-1 virus expression, interacts with Dicer and functions in RNA silencing. *Embo Reports* 6:961-967.

Hajnoczky, G., G.Csordas, A.Bago, A.T.Chiu, and A.Spat. 1992. Angiotensin-Ii Exerts Its Effect on Aldosterone Production and Potassium Permeability Through Receptor Subtype-At1 in Rat Adrenal Glomerulosa Cells. *Biochemical Pharmacology* 43:1009-1012.

Haseroth, K., D.Gerdes, S.Berger, M.Feuring, A.Gunther, C.Herbst, M.Christ, and M.Wehling. 1999. Rapid nongenomic effects of aldosterone in mineralocorticoidreceptor-knockout mice. *Biochemical and Biophysical Research Communications* 266:257-261.

Hautanen, A., L.Lankinen, M.Kupari, O.A.Janne, H.Adlercreutz, H.Nikkila, and P.C.White. 1998. Associations between aldosterone synthase gene polymorphism and the adrenocortical function in males. *Journal of Internal Medicine* 244:11-18.

Hayashita, Y., H.Osada, Y.Tatematsu, H.Yamada, K.Yanagisawa, S.Tomida, Y.Yatabe, K.Kawahara, Y.Sekido, and T.Takahashi. 2005. A polycistronic
microRNA cluster, miR-17-92, is overexpressed in human lung cancers and enhances cell proliferation. *Cancer Research* 65:9628-9632.

Heitzmann, D., R.Derand, S.Jungbauer, S.Bandulik, C.Sterner, F.Schweda, A.El Wakil, E.Lalli, N.Guy, R.Mengual, M.Reichold, I.Tegtmeier, S.Bendahhou, C.E.Gomez-Sanchez, M.I.Aller, W.Wisden, A.Weber, F.Lesage, R.Warth, and J.Barhanin. 2008. Invalidation of TASK1 potassium channels disrupts adrenal gland zonation and mineralocorticoid homeostasis. *Embo Journal* 27:179-187.

Himathongkam, T., R.G.Dluhy, and G.H.Williams. 1975. Potassium Aldosterone-Renin Interrelationships. *Journal of Clinical Endocrinology & Metabolism* 41:153-159.

Hiramatsu, K., T.Yamada, Y.Yukimura, I.Komiya, K.Ichikawa, M.Ishihara, H.Nagata, and T.Izumiyama. 1981. A Screening-Test to Identify Aldosterone-Producing Adenoma by Measuring Plasma-Renin Activity - Results in Hypertensive Patients. *Archives of Internal Medicine* 141:1589-1593.

Honda, M., W.Nowaczynski, G.P.Guthrie, F.H.Messerli, G.Tolis, O.Kuchel, and J.Genest. 1977. Response of Several Adrenal Steroids to Acth Stimulation in Essential Hypertension. *Journal of Clinical Endocrinology & Metabolism* 44:264-272.

Horisberger, J.D. 1998. Amiloride-sensitive Na channels. *Current Opinion in Cell Biology* 10:443-449.

Horisberger, J.D. and J.Diezi. 1984. Inhibition of Aldosterone-Induced Antinatriuresis and Kaliuresis by Actinomycin-D. *American Journal of Physiology* 246:F201-F204.

Huang, C.C.J. and H.H.C.Yao. 2010. Inactivation of Dicer1 in Steroidogenic factor 1-positive cells reveals tissue-specific requirement for Dicer1 in adrenal, testis, and ovary. *Bmc Developmental Biology* 10.

Hutvagner, G., J.McLachlan, A.E.Pasquinelli, E.Balint, T.Tuschl, and P.D.Zamore. 2001. A cellular function for the RNA-interference enzyme Dicer in the maturation of the let-7 small temporal RNA. *Science* 293:834-838.

Hutvagner, G., M.J.Simard, C.C.Mello, and P.D.Zamore. 2004. Sequence-specific inhibition of small RNA function. *Plos Biology* 2:465-475.

Iliopoulos, D., E.I.Bimpaki, M.Nesterova, and C.A.Stratakis. 2009. MicroRNA Signature of Primary Pigmented Nodular Adrenocortical Disease: Clinical Correlations and Regulation of Wnt Signaling. *Cancer Research* 69:3278-3282.

Imrie, H., M.Freel, B.M.Mayosi, E.Davies, R.Fraser, M.Ingram, H.J.Cordell, M.Farrall, P.J.Avery, H.Watkins, B.Keavney, and J.M.C.Connell. 2006. Association between aldosterone production and variation in the 11 betahydroxylase (CYP11B1) gene. *Journal of Clinical Endocrinology & Metabolism* 91:5051-5056.

Inglis, G.C., P.F.Plouin, E.C.Friel, E.Davies, R.Fraser, and J.M.Connell. 2001. Polymorphic differences from normal in the aldosterone synthase gene (CYP11B2) in patients with primary hyperaldosteronism and adrenal tumour (Conn's syndrome). *Clinical Endocrinology* 54:725-730.

Iorio, M.V., M.Ferracin, C.G.Liu, A.Veronese, R.Spizzo, S.Sabbioni, E.Magri,
M.Pedriali, M.Fabbri, M.Campiglio, S.Menard, J.P.Palazzo, A.Rosenberg,
P.Musiani, S.Volinia, I.Nenci, G.A.Calin, P.Querzoli, M.Negrini, and C.M.Croce.
2005. MicroRNA gene expression deregulation in human breast cancer. *Cancer Research* 65:7065-7070.

Jeska, Y.W.A., A.So, L.Kelemen, N.Sukor, C.Willys, B.Bulmer, R.D.Gordon, D.Duffy, and M.Stowasser. 2008. Examination of chromosome 7p22 candidate genes RBaK, PMS2 and GNA12 in familial hyperaldosteronism type II. *Clinical and Experimental Pharmacology and Physiology* 35:380-385.

Jeunemaitre, X. 2008. Genetics of the human renin angiotensin system. *Journal* of Molecular Medicine-Jmm 86:637-641.

Jinek, M. and J.A.Doudna. 2009. A three-dimensional view of the molecular machinery of RNA interference. *Nature* 457:405-412.

John, B., A.J.Enright, A.Aravin, T.Tuschl, C.Sander, and D.S.Marks. 2004. Human MicroRNA targets. *Plos Biology* 2:1862-1879.

Kahvejian, A., Y.V.Svitkin, R.Sukarieh, M.N.M'Boutchou, and N.Sonenberg. 2005. Mammalian poly(A)-binding protein is a eukaryotic translation initiation factor, which acts via multiple mechanisms. *Genes & Development* 19:104-113.

Katoh, T., Y.Sakaguchi, K.Miyauchi, T.Suzuki, S.Kashiwabara, T.Baba, and T.Suzuki. 2009. Selective stabilization of mammalian microRNAs by 3 ' adenylation mediated by the cytoplasmic poly(A) polymerase GLD-2. *Genes & Development* 23:433-438.

Kawahara, Y., M.Megraw, E.Kreider, H.Iizasa, L.Valente, A.G.Hatzigeorgiou, and K.Nishikura. 2008. Frequency and fate of microRNA editing in human brain. *Nucleic Acids Research* 36:5270-5280.

Kawamoto, T., Y.Mitsuuchi, K.Toda, Y.Yokoyama, K.Miyahara, S.Miura, T.Ohnishi, Y.Ichikawa, K.Nakao, H.Imura, S.Ulick, and Y.Shizuta. 1992. Role of Steroid 11-Beta-Hydroxylase and Steroid 18-Hydroxylase in the Biosynthesis of Glucocorticoids and Mineralocorticoids in Humans. *Proceedings of the National Academy of Sciences of the United States of America* 89:1458-1462.

Keller, A., P.Leidinger, A.Borries, A.Wendschlag, F.Wucherpfennig, M.Scheffler, H.Huwer, H.P.Lenhof, and E.Meese. 2009. miRNAs in lung cancer - Studying complex fingerprints in patient's blood cells by microarray experiments. *Bmc Cancer* 9.

Kennon, B., M.C.Ingram, E.C.Friel, N.H.Anderson, S.M.Mackenzie, E.Davies, L.Shakerdi, A.M.Wallace, R.Fraser, and J.M.C.Connell. 2004. Aldosterone synthase gene variation and adrenocortical response to sodium status, angiotensin II and ACTH in normal male subjects. *Clinical Endocrinology* 61:174-181. Kertesz, M., N.Iovino, U.Unnerstall, U.Gaul, and E.Segal. 2007. The role of site accessibility in microRNA target recognition. *Nature Genetics* 39:1278-1284.

Khvorova, A., A.Reynolds, and S.D.Jayasena. 2003. Functional siRNAs and rniRNAs exhibit strand bias. *Cell* 115:209-216.

Kim, V.N., J.Han, and M.C.Siomi. 2009. Biogenesis of small RNAs in animals. *Nature Reviews Molecular Cell Biology* 10:126-139.

Kinch, L.N. and N.V.Grishin. 2009. The human Ago2 MC region does not contain an eIF4E-like mRNA cap binding motif. *Biology Direct* 4.

Kiriakidou, M., G.S.Tan, S.Lamprinaki, M.Planell-Saguer, P.T.Nelson, and Z.Mourelatos. 2007. An mRNA m(7)G cap binding-like motif within human Ago2 represses translation. *Cell* 129:1141-1151.

Kirita, S., T.Hashimoto, M.Kitajima, S.I.Honda, K.I.Morohashi, and T.Omura. 1990. Structural-Analysis of Multiple Bovine P-450(11-Beta) Genes and Their Promoter Activities. *Journal of Biochemistry* 108:1030-1041.

Komiya, I., T.Yamada, M.Takara, T.Asawa, M.Shimabukuro, T.Nishimori, and N.Takasu. 2000. Lys(173)Arg and 344T/C variants of CYP11B2 in Japanese patients with low-renin hypertension. *Hypertension* 35:699-703.

Koppel, H., M.Christ, B.A.Yard, P.C.Bar, F.J.van der Woude, and M.Wehling. 2003. Nongenomic effects of aldosterone on human renal cells. *Journal of Clinical Endocrinology & Metabolism* 88:1297-1302.

Kota, J., R.R.Chivukula, K.A.O'Donnell, E.A.Wentzel, C.L.Montgomery, H.W.Hwang, T.C.Chang, P.Vivekanandan, M.Torbenson, K.R.Clark, J.R.Mendell, and J.T.Mendell. 2009. Therapeutic microRNA Delivery Suppresses Tumorigenesis in a Murine Liver Cancer Model. *Cell* 137:1005-1017.

Kotelevtsev, Y., R.W.Brown, S.Fleming, C.Kenyon, C.R.W.Edwards, J.R.Seckl, and A.J.Mullins. 1999. Hypertension in mice lacking 11 beta-hydroxysteroid dehydrogenase type 2. *Journal of Clinical Investigation* 103:683-689.

Krichevsky, A.M. and G.Gabriely. 2009. miR-21: a small multi-faceted RNA. *Journal of Cellular and Molecular Medicine* 13:x-53.

Krone, N., V.Dhir, H.E.Ivison, and W.Arlt. 2007. Congenital adrenal hyperplasia and P450 oxidoreductase deficiency. *Clinical Endocrinology* 66:162-172.

Krueger, R.J. and N.R.Ormejohnson. 1983. Acute Adrenocorticotropic Hormone Stimulation of Adrenal Corticosteroidogenesis - Discovery of A Rapidly Induced Protein. *Journal of Biological Chemistry* 258:159-167.

Kruger, J. and M.Rehmsmeier. 2006. RNAhybrid: microRNA target prediction easy, fast and flexible. *Nucleic Acids Research* 34:W451-W454.

Krutzfeldt, J., N.Rajewsky, R.Braich, K.G.Rajeev, T.Tuschl, M.Manoharan, and M.Stoffel. 2005. Silencing of microRNAs in vivo with 'antagomirs'. *Nature* 438:685-689.

Kshirsagar, A.V., M.Carpenter, H.Bang, S.B.Wyatt, and R.E.Colindres. 2006. Blood pressure usually considered normal is associated with an elevated risk of cardiovascular disease. *American Journal of Medicine* 119:133-141.

Kulkarni, M., S.Ozgur, and G.Stoecklin. 2010. On track with P-bodies. *Biochemical Society Transactions* 38:242-251.

Lachance, Y., V.Luuthe, H.Verreault, M.Dumont, E.Rheaume, G.Leblanc, and F.Labrie. 1991. Structure of the Human Type-Ii 3-Beta-Hydroxysteroid Dehydrogenase Delta-5-Delta-4 Isomerase (3-Beta-Hsd) Gene - Adrenal and Gonadal Specificity. *Dna and Cell Biology* 10:701-711.

Lagos-Quintana, M., R.Rauhut, W.Lendeckel, and T.Tuschl. 2001. Identification of novel genes coding for small expressed RNAs. *Science* 294:853-858.

Lagos-Quintana, M., R.Rauhut, A.Yalcin, J.Meyer, W.Lendeckel, and T.Tuschl. 2002. Identification of tissue-specific microRNAs from mouse. *Current Biology* 12:735-739.

Lal, A., H.H.Kim, K.Abdelmohsen, Y.Kuwano, R.Pullmann, S.Srikantan, R.Subrahmanyam, J.L.Martindale, X.L.Yang, F.Ahmed, F.Navarro, D.Dykxhoorn, J.Lieberman, and M.Gorospe. 2008. p16(INK4a) Translation Suppressed by miR-24. *Plos One* 3.

Larsson, E., C.Sander, and D.Marks. 2010. mRNA turnover rate limits siRNA and microRNA efficacy. *Molecular Systems Biology* 6.

Lawes, C.M.M., S.Vander Hoorn, and A.Rodgers. 2008. Global burden of blood-pressure-related disease, 2001. *Lancet* 371:1513-1518.

Lawrie, C.H., S.Gal, H.M.Dunlop, B.Pushkaran, A.P.Liggins, K.Pulford, A.H.Banham, F.Pezzella, J.Boultwood, J.S.Wainscoat, C.S.R.Hatton, and A.L.Harris. 2008. Detection of elevated levels of tumour-associated microRNAs in serum of patients with diffuse large B-cell lymphoma. *British Journal of Haematology* 141:672-675.

Lee, I., S.S.Ajay, J.I.Yook, H.S.Kim, S.H.Hong, N.H.Kim, S.M.Dhanasekaran, A.M.Chinnaiyan, and B.D.Athey. 2009. New class of microRNA targets containing simultaneous 5 '-UTR and 3 '-UTR interaction sites. *Genome Research* 19:1175-1183.

Lee, R.C., R.L.Feinbaum, and V.Ambros. 1993. The C-Elegans Heterochronic Gene Lin-4 Encodes Small Rnas with Antisense Complementarity to Lin-14. *Cell* 75:843-854.

Lee, Y., I.Hur, S.Y.Park, Y.K.Kim, M.R.Suh, and V.N.Kim. 2006. The role of PACT in the RNA silencing pathway. *Embo Journal* 25:522-532.

Lee, Y., M.Kim, J.J.Han, K.H.Yeom, S.Lee, S.H.Baek, and V.N.Kim. 2004a. MicroRNA genes are transcribed by RNA polymerase II. *Embo Journal* 23:4051-4060.

Lee, Y.S. and A.Dutta. 2007. The tumor suppressor microRNA let-7 represses the HMGA2 oncogene. *Genes & Development* 21:1025-1030.

Lee, Y.S., H.K.Kim, S.M.Chung, K.S.Kim, and A.Dutta. 2005. Depletion of human micro-RNA miR-125b reveals that it is critical for the proliferation of differentiated cells but not for the downregulation of putative targets during differentiation. *Journal of Biological Chemistry* 280:16635-16641.

Lee, Y.S., K.Nakahara, J.W.Pham, K.Kim, Z.Y.He, E.J.Sontheimer, and R.W.Carthew. 2004b. Distinct roles for Drosophila Dicer-1 and Dicer-2 in the siRNA/miRNA silencing pathways. *Cell* 117:69-81.

Lehmann, U., B.Hasemeier, M.Christgen, M.Muller, D.Romermann, F.Langer, and H.Kreipe. 2008. Epigenetic inactivation of microRNA gene hsa-mir-9-1 in human breast cancer. *Journal of Pathology* 214:17-24.

Lehoux, J.G., G.Dupuis, and A.Lefebvre. 2001. Control of CYP11B2 gene expression through differential regulation of its promoter by atypical and conventional protein kinase C isoforms. *Journal of Biological Chemistry* 276:8021-8028.

Levy, D., G.B.Ehret, K.Rice, G.C.Verwoert, L.J.Launer, A.Dehghan, N.L.Glazer, A.C.Morrison, A.D.Johnson, T.Aspelund, Y.Aulchenko, T.Lumley, A.Kottgen, R.S.Vasan, F.Rivadeneira, G.Eiriksdottir, X.Q.Guo, D.E.Arking, G.F.Mitchell, F.U.S.Mattace-Raso, A.V.Smith, K.Taylor, R.B.Scharpf, S.J.Hwang, E.J.G.Sijbrands, J.Bis, T.B.Harris, S.K.Ganesh, C.J.O'Donnell, A.Hofman, J.I.Rotter, J.Coresh, E.J.Benjamin, A.G.Uitterlinden, G.Heiss, C.S.Fox, J.C.M.Witteman, E.Boerwinkle, T.J.Wang, V.Gudnason, M.G.Larson, A.Chakravarti, B.M.Psaty, and C.M.van Duijn. 2009. Genome-wide association study of blood pressure and hypertension. *Nature Genetics* 41:677-687.

Lewis, B.P., C.B.Burge, and D.P.Bartel. 2005. Conserved seed pairing, often flanked by adenosines, indicates that thousands of human genes are microRNA targets. *Cell* 120:15-20.

Lewis, B.P., I.H.Shih, M.W.Jones-Rhoades, D.P.Bartel, and C.B.Burge. 2003. Prediction of mammalian microRNA targets. *Cell* 115:787-798.

Li, J.H., P.Smyth, R.Flavin, S.Cahill, K.Denning, S.Aherne, S.M.Guenther, J.J.O'Leary, and O.Sheils. 2007. Comparison of miRNA expression patterns using total RNA extracted from matched samples of formalin-fixed paraffin-embedded (FFPE) cells and snap frozen cells. *Bmc Biotechnology* 7.

Li, J.J., Z.Y.Yang, B.Yu, J.Liu, and X.M.Chen. 2005. Methylation protects miRNAs and siRNAs from a 3 '-end uridylation activity in Arabildopsis. *Current Biology* 15:1501-1507.

Li, S.Q., S.L.Lian, J.J.Moser, M.L.Fritzler, M.J.Fritzler, M.Satoh, and E.K.L.Chan. 2008. Identification of GW182 and its novel isoform TNGW1 as translational repressors in Ago2-mediated silencing. *Journal of Cell Science* 121:4134-4144.

Lia, D., P.Y.Yang, Q.H.Xiong, X.H.Song, X.Q.Yang, L.Liu, W.J.Yuan, and Y.C.Rui. 2010. MicroRNA-125a/b-5p inhibits endothelin-1 expression in vascular endothelial cells. *Journal of Hypertension* 28:1646-1654.

Lieberman, S. and Y.Y.Lin. 2001. Reflections on sterol sidechain cleavage process catalyzed by cytochrome P450(scc). *Journal of Steroid Biochemistry and Molecular Biology* 78:1-14.

Lifton, R.P., R.G.Dluhy, M.Powers, G.M.Rich, S.Cook, S.Ulick, and J.M.Lalouel. 1992. A Chimeric 11-Beta-Hydroxylase Aldosterone Synthase Gene Causes Glucocorticoid-Remediable Aldosteronism and Human Hypertension. *Nature* 355:262-265.

Lifton, R.P., A.G.Gharavi, and D.S.Geller. 2001. Molecular mechanisms of human hypertension. *Cell* 104:545-556.

Lim, L.P., N.C.Lau, P.Garrett-Engele, A.Grimson, J.M.Schelter, J.Castle, D.P.Bartel, P.S.Linsley, and J.M.Johnson. 2005. Microarray analysis shows that some microRNAs downregulate large numbers of target mRNAs. *Nature* 433:769-773.

Lim, P.O., T.M.MacDonald, C.Holloway, E.Friel, N.H.Anderson, E.Dow, R.T.Jung, E.Davies, R.Fraser, and J.M.C.Connell. 2002. Variation at the aldosterone synthase (CYP11B2) locus contributes to hypertension in subjects with a raised aldosterone-to-renin ratio. *Journal of Clinical Endocrinology & Metabolism* 87:4398-4402.

Lin, D., T.Sugawara, J.F.Strauss, B.J.Clark, D.M.Stocco, P.Saenger, A.Rogol, and W.L.Miller. 1995. Role of Steroidogenic Acute Regulatory Protein in Adrenal and Gonadal Steroidogenesis. *Science* 267:1828-1831.

Lin, T.C.E., S.C.Chien, P.C.Hsu, and L.A.Li. 2006. Mechanistic study of polychlorinated biphenyl 126-induced CYP11B1 and CYP11B2 up-regulation. *Endocrinology* 147:1536-1544.

Liu, J.D., M.A.Carmell, F.V.Rivas, C.G.Marsden, J.M.Thomson, J.J.Song, S.M.Hammond, L.Joshua-Tor, and G.J.Hannon. 2004. Argonaute2 is the catalytic engine of mammalian RNAi. *Science* 305:1437-1441.

Longini, I.M., M.W.Higgins, P.C.Hinton, P.P.Moll, and J.B.Keller. 1984. Environmental and Genetic Sources of Familial Aggregation of Blood-Pressure in Tecumseh, Michigan. *American Journal of Epidemiology* 120:131-144.

Luciano, D.J., H.Mirsky, N.J.Vendetti, and S.Maas. 2004. RNA editing of a miRNA precursor. *Rna-A Publication of the Rna Society* 10:1174-1177.

Lund, E., S.Guttinger, A.Calado, J.E.Dahlberg, and U.Kutay. 2004. Nuclear export of microRNA precursors. *Science* 303:95-98.

Ma, E., I.J.Macrae, J.F.Kirsch, and J.A.Doudna. 2008. Autoinhibition of human dicer by its internal helicase domain. *Journal of Molecular Biology* 380:237-243.

Ma, L., J.Teruya-Feldstein, and R.A.Weinberg. 2007. Tumour invasion and metastasis initiated by microRNA 10b in breast cancer. *Nature* 449:682-6U2.

Macgregor, G.A., F.E.Best, J.M.Cam, N.D.Markandu, D.M.Elder, G.A.Sagnella, and M.Squires. 1982. Double-Blind Randomized Crossover Trial of Moderate Sodium Restriction in Essential-Hypertension. *Lancet* 1:351-355.

Mangelsdorf, D.J., C.Thummel, M.Beato, P.Herrlich, G.Schutz, K.Umesono, B.Blumberg, P.Kastner, M.Mark, P.Chambon, and R.M.Evans. 1995. The Nuclear Receptor Superfamily - the 2Nd Decade. *Cell* 83:835-839.

Maroney, P.A., Y.Yu, J.Fisher, and T.W.Nilsen. 2006. Evidence that microRNAs are associated with translating messenger RNAs in human cells. *Nature Structural & Molecular Biology* 13:1102-1107.

Martin, M.M., E.J.Lee, J.A.Buckenberger, T.D.Schmittgen, and T.S.Elton. 2006. MicroRNA-155 regulates human angiotensin II type 1 receptor expression in fibroblasts. *Journal of Biological Chemistry* 281:18277-18284.

Matranga, C., Y.Tomari, C.Shin, D.P.Bartel, and P.D.Zamore. 2005. Passengerstrand cleavage facilitates assembly of siRNA into Ago2-containing RNAi enzyme complexes. *Cell* 123:607-620.

Meneton, P., X.Jeunemaitre, H.E.De Wardener, and G.A.Macgregor. 2005. Links between dietary salt intake, renal salt handling, blood pressure, and cardiovascular diseases. *Physiological Reviews* 85:679-715.

Merritt, W.M., Y.G.Lin, L.Y.Han, A.A.Kamat, W.A.Spannuth, R.Schmandt, D.Urbauer, L.A.Pennacchio, J.Cheng, A.M.Nick, M.T.Deavers, A.Mourad-Zeidan, H.Wang, P.Mueller, M.E.Lenburg, J.W.Gray, S.Mok, M.J.Birrer, G.Lopez-Berestein, R.L.Coleman, M.Bar-Eli, and A.K.Sood. 2008. Dicer, Drosha, and Outcomes in Patients with Ovarian Cancer. *New England Journal of Medicine* 359:2641-2650.

Mick, V.E., O.A.Itani, R.W.Loftus, R.F.Husted, T.J.Schmidt, and C.P.Thomas. 2001. The alpha-subunit of the epithelial sodium channel is an aldosterone-induced transcript in mammalian collecting ducts, and this transcriptional response is mediated via distinct cis-elements in the 5 '-flanking region of the gene. *Molecular Endocrinology* 15:575-588.

Miller, W.L. 2007. StAR search - What we know about how the steroidogenic acute regulatory protein mediates mitochondrial cholesterol import. *Molecular Endocrinology* 21:589-601.

Milliez, P., X.Girerd, P.F.Plouin, J.Blacher, M.E.Safar, and J.J.Mourad. 2005. Evidence for an increased rate of cardiovascular events in patients with primary aldosteronism. *Journal of the American College of Cardiology* 45:1243-1248.

Mishra, P.J., R.Humeniuk, P.J.Mishra, G.S.A.Longo-Sorbello, D.Banerjee, and J.R.Bertino. 2007. A miR-24 microRNA binding-site polymorphism in dihydrofolate reductase gene leads to methotrexate resistance. *Proceedings of the National Academy of Sciences of the United States of America* 104:13513-13518.

Mitchell, P.S., R.K.Parkin, E.M.Kroh, B.R.Fritz, S.K.Wyman, E.L.Pogosova-Agadjanyan, A.Peterson, J.Noteboom, K.C.O'Briant, A.Allen, D.W.Lin, N.Urban, C.W.Drescher, B.S.Knudsen, D.L.Stirewalt, R.Gentleman, R.L.Vessella, P.S.Nelson, D.B.Martin, and M.Tewari. 2008. Circulating microRNAs as stable blood-based markers for cancer detection. *Proceedings of the National Academy* of Sciences of the United States of America 105:10513-10518. Mitsui, T., S.Nomura, M.Okada, Y.Ohno, H.Kobayashi, Y.Nakashima, Y.Murata, M.Takeuchi, N.Kuno, T.Nagasaka, J.Y.Wang, M.D.Cooper, and S.Mizutani. 2003. Hypertension and angiotensin II hypersensitivity in aminopeptidase A-deficient mice. *Molecular Medicine* 9:57-62.

Montori, V.M. and W.F.Young. 2002. Use of plasma aldosterone concentrationto-plasma renin activity ratio as a screening test for primary aldosteronism - A systematic review of the literature. *Endocrinology and Metabolism Clinics of North America* 31:619-+.

Mornet, E., J.Dupont, A.Vitek, and P.C.White. 1989. Characterization of 2 Genes Encoding Human Steroid 11-Beta-Hydroxylase (P-45011-Beta). *Journal of Biological Chemistry* 264:20961-20967.

Morrison, N., S.B.Harrap, J.L.Arriza, E.Boyd, and J.M.Connor. 1990. Regional Chromosomal Assignment of the Human Mineralocorticoid Receptor Gene to 4Q31.1. *Human Genetics* 85:130-132.

Mountjoy, K.G., I.M.Bird, W.E.Rainey, and R.D.Cone. 1994. Acth Induces Up-Regulation of Acth Receptor Messenger-Rna in Mouse and Human Adrenocortical Cell-Lines. *Molecular and Cellular Endocrinology* 99:R17-R20.

Mourelatos, Z., J.Dostie, S.Paushkin, A.Sharma, B.Charroux, L.Abel, J.Rappsilber, M.Mann, and G.Dreyfuss. 2002. miRNPs: a novel class of ribonucleoproteins containing numerous microRNAs. *Genes & Development* 16:720-728.

Mune, T., F.M.Rogerson, H.Nikkila, A.K.Agarwal, and P.C.White. 1995. Human Hypertension Caused by Mutations in the Kidney Isozyme of 11-Beta-Hydroxysteroid Dehydrogenase. *Nature Genetics* 10:394-399.

Napoli, C., C.Lemieux, and R.Jorgensen. 1990. Introduction of A Chimeric Chalcone Synthase Gene Into Petunia Results in Reversible Co-Suppression of Homologous Genes in Trans. *Plant Cell* 2:279-289.

Nelson, D.R., L.Koymans, T.Kamataki, J.J.Stegeman, R.Feyereisen, D.J.Waxman, M.R.Waterman, O.Gotoh, M.J.Coon, R.W.Estabrook, I.C.Gunsalus, and D.W.Nebert. 1996. P450 superfamily: Update on new sequences, gene mapping, accession numbers and nomenclature. *Pharmacogenetics* 6:1-42.

Nelson, P.T., D.A.Baldwin, W.P.Kloosterman, S.Kauppinen, R.H.Plasterk, and Z.Mourelatos. 2006. RAKE and LNA-ISH reveal microRNA expression and localization in archival human brain. *RNA*. 12:187-191.

Newton-Cheh, C., T.Johnson, V.Gateva, M.D.Tobin, M.Bochud, L.Coin, S.S.Najjar, J.H.Zhao, S.C.Heath, S.Eyheramendy, K.Papadakis, B.F.Voight, L.J.Scott, F.Zhang, M.Farrall, T.Tanaka, C.Wallace, J.C.Chambers, K.T.Khaw, P.Nilsson, P.van der Harst, S.Polidoro, D.E.Grobbee, N.C.Onland-Moret, M.L.Bots, L.V.Wain, K.S.Elliott, A.Teumer, J.Luan, G.Lucas, J.Kuusisto, P.R.Burton, D.Hadley, W.L.McArdle, M.Brown, A.Dominiczak, S.J.Newhouse, N.J.Samani, J.Webster, E.Zeggini, J.S.Beckmann, S.Bergmann, N.Lim, K.Song, P.Vollenweider, G.Waeber, D.M.Waterworth, X.Yuan, L.Groop, M.Orho-Melander, A.Allione, A.Di Gregorio, S.Guarrera, S.Panico, F.Ricceri, V.Romanazzi, C.Sacerdote, P.Vineis, I.Barroso, M.S.Sandhu, R.N.Luben, G.J.Crawford, P.Jousilahti, M.Perola, M.Boehnke, L.L.Bonnycastle, F.S.Collins, A.U.Jackson, K.L.Mohlke, H.M.Stringham, T.T.Valle, C.J.Willer, R.N.Bergman, M.A.Morken, A.Doring, C.Gieger, T.Illig, T.Meitinger, E.Org, A.Pfeufer, H.E.Wichmann, S.Kathiresan, J.Marrugat, C.J.O'Donnell, S.M.Schwartz, D.S.Siscovick, I.Subirana, N.B.Freimer, A.L.Hartikainen, M.I.McCarthy, P.F.O'Reilly, L.Peltonen, A.Pouta, P.E.de Jong, H.Snieder, W.H.van Gilst, R.Clarke, A.Goel, A.Hamsten, J.F.Peden, U.Seedorf, A.C.Syvanen, G.Tognoni, E.G.Lakatta, S.Sanna, P.Scheet, D.Schlessinger, A.Scuteri, M.Dorr, F.Ernst, S.B.Felix, G.Homuth, R.Lorbeer, T.Reffelmann, R.Rettig, U.Volker, P.Galan, I.G.Gut, S.Hercberg, G.M.Lathrop, D.Zelenika, P.Deloukas, N.Soranzo, F.M.Williams, G.Zhai, V.Salomaa, M.Laakso, R.Elosua, N.G.Forouhi, H.Volzke, C.S.Uiterwaal, Y.T.van der Schouw, M.E.Numans, G.Matullo, G.Navis, G.Berglund, S.A.Bingham, J.S.Kooner, J.M.Connell, S.Bandinelli, L.Ferrucci, H.Watkins, T.D.Spector, J.Tuomilehto, D.Altshuler, D.P.Strachan, M.Laan, P.Meneton, N.J.Wareham, M.Uda, M.R.Jarvelin, V.Mooser, O.Melander, R.J.F.Loos, P.Elliott, G.R.Abecasis, M.Caulfield, and P.B.Munroe. 2009. Genomewide association study identifies eight loci associated with blood pressure. Nature Genetics 41:666-676.

Ng, E.K.O., W.W.S.Chong, H.Jin, E.K.Y.Lam, V.Y.Shin, J.Yu, T.C.W.Poon, S.S.M.Ng, and J.J.Y.Sung. 2009. Differential expression of microRNAs in plasma of patients with colorectal cancer: a potential marker for colorectal cancer screening. *Gut* 58:1375-1381.

Nogueira, E.F., W.B.Bollag, and W.E.Rainey. 2009. Angiotensin II regulation of adrenocortical gene transcription. *Molecular and Cellular Endocrinology* 302:230-236.

Nogueira, E.F. and W.E.Rainey. 2010. Regulation of Aldosterone Synthase by Activator Transcription Factor/cAMP Response Element-Binding Protein Family Members. *Endocrinology* 151:1060-1070.

Nottrott, S., M.J.Simard, and J.D.Richter. 2006. Human let-7a miRNA blocks protein production on actively translating polyribosomes. *Nature Structural & Molecular Biology* 13:1108-1114.

Odermatt, A. and A.G.Atanasov. 2009. Mineralocorticoid receptors: Emerging complexity and functional diversity. *Steroids* 74:163-171.

Ogishima, T., H.Suzuki, J.Hata, F.Mitani, and Y.Ishimura. 1992. Zone-Specific Expression of Aldosterone Synthase Cytochrome-P-450 and Cytochrome-P-45011-Beta in Rat Adrenal-Cortex - Histochemical Basis for the Functional Zonation. *Endocrinology* 130:2971-2977.

Okamura, K., J.W.Hagen, H.Duan, D.M.Tyler, and E.C.Lai. 2007. The mirtron pathway generates microRNA-class regulatory RNAs in Drosophila. *Cell* 130:89-100.

Olsen, P.H. and V.Ambros. 1999. The lin-4 regulatory RNA controls developmental timing in Caenorhabditis elegans by blocking LIN-14 protein synthesis after the initiation of translation. *Developmental Biology* 216:671-680.

Orom, U.A., F.C.Nielsen, and A.H.Lund. 2008. MicroRNA-10a binds the 5 ' UTR of ribosomal protein mRNAs and enhances their translation. *Molecular Cell* 30:460-471.

Packer, A.N., Y.Xing, S.Q.Harper, L.Jones, and B.L.Davidson. 2008. The Bifunctional microRNA miR-9/miR-9*Regulates REST and CoREST and Is Downregulated in Huntington's Disease. *Journal of Neuroscience* 28:14341-14346.

Paillard, F., D.Chansel, E.Brand, A.Benetos, F.Thomas, S.Czekalski, R.Ardaillou, and F.Soubrier. 1999. Genotype-phenotype relationships for the reninangiotensin-aldosterone system in a normal population. *Hypertension* 34:423-429.

Papadopoulos, G.L., M.Reczko, V.A.Simossis, P.Sethupathy, and A.G.Hatzigeorgiou. 2009. The database of experimentally supported targets: a functional update of TarBase. *Nucleic Acids Research* 37:D155-D158.

Pasquinelli, A.E., B.J.Reinhart, F.Slack, M.Q.Martindale, M.I.Kuroda, B.Maller, D.C.Hayward, E.E.Ball, B.Degnan, P.Muller, J.Spring, A.Srinivasan, M.Fishman, J.Finnerty, J.Corbo, M.Levine, P.Leahy, E.Davidson, and G.Ruvkun. 2000. Conservation of the sequence and temporal expression of let-7 heterochronic regulatory RNA. *Nature* 408:86-89.

Payne, A.H. and D.B.Hales. 2004. Overview of steroidogenic enzymes in the pathway from cholesterol to active steroid hormones. *Endocrine Reviews* 25:947-970.

Pelletier, J. and N.Sonenberg. 1988. Internal Initiation of Translation of Eukaryotic Messenger-Rna Directed by A Sequence Derived from Poliovirus Rna. *Nature* 334:320-325.

Penning, T.M. 1997. Molecular endocrinology of hydroxysteroid dehydrogenases. *Endocrine Reviews* 18:281-305.

Petersen, C.P., M.E.Bordeleau, J.Pelletier, and P.A.Sharp. 2006. Short RNAs repress translation after initiation in mammalian cells. *Molecular Cell* 21:533-542.

Pezzi, V., C.D.Clyne, S.Ando, J.M.Mathis, and W.E.Rainey. 1997. Ca2+-regulated expression of aldosterone synthase is mediated by calmodulin and calmodulin-dependent protein kinases. *Endocrinology* 138:835-838.

Phillips, M.C., W.J.Johnson, and G.H.Rothblat. 1987. Mechanisms and Consequences of Cellular Cholesterol Exchange and Transfer. *Biochimica et Biophysica Acta* 906:223-276.

Pillai, R.S., S.N.Bhattacharyya, C.G.Artus, T.Zoller, N.Cougot, E.Basyuk, E.Bertrand, and W.Filipowicz. 2005. Inhibition of translational initiation by Let-7 microRNA in human cells. *Science* 309:1573-1576.

Pitt, B., F.Zannad, W.J.Remme, R.Cody, A.Castaigne, A.Perez, J.Palensky, and J.Wittes. 1999. The effect of spironolactone on morbidity and mortality in patients with severe heart failure. *New England Journal of Medicine* 341:709-717.

Pojoga, L., S.Gautier, H.Blanc, T.T.Guyene, O.Poirier, F.Cambien, and A.Benetos. 1998. Genetic determination of plasma aldosterone levels in essential hypertension. *American Journal of Hypertension* 11:856-860.

Pralong, W.F., L.Hunyady, P.Varnai, C.B.Wollheim, and A.Spat. 1992. Pyridine-Nucleotide Redox State Parallels Production of Aldosterone in Potassium-Stimulated Adrenal Glomerulosa Cells. *Proceedings of the National Academy of Sciences of the United States of America* 89:132-136.

Prasad, A., S.Narayanan, S.Husain, F.Padder, M.Waclawiw, N.Epstein, and A.A.Quyyumi. 2000. Insertion-deletion polymorphism of the ACE gene modulates reversibility of endothelial dysfunction with ACE inhibition. *Circulation* 102:35-41.

Qian, L., L.W.Van Laake, Y.Huang, S.Y.Liu, M.F.Wendland, and D.Srivastava. 2011. miR-24 inhibits apoptosis and represses Bim in mouse cardiomyocytes. *Journal of Experimental Medicine* 208:549-560.

Rainey, W.E., I.M.Bird, and J.I.Mason. 1994. The Nci-H295 Cell-Line - A Pluripotent Model for Human Adrenocortical Studies. *Molecular and Cellular Endocrinology* 100:45-50.

Raveche, E.S., E.Salerno, B.J.Scaglione, V.Manohar, F.Abbasi, Y.C.Lin, T.Fredrickson, P.Landgraf, S.Ramachandra, K.Huppi, J.R.Toro, V.E.Zenger, R.A.Metcalf, and G.E.Marti. 2007. Abnormal microRNA-16 locus with synteny to human 13q14 linked to CLL in NZB mice. *Blood* 109:5079-5086.

Rehmsmeier, M., P.Steffen, M.Hochsmann, and R.Giegerich. 2004. Fast and effective prediction of microRNA/target duplexes. *Rna-A Publication of the Rna Society* 10:1507-1517.

Reinhart, B.J., F.J.Slack, M.Basson, A.E.Pasquinelli, J.C.Bettinger, A.E.Rougvie, H.R.Horvitz, and G.Ruvkun. 2000. The 21-nucleotide let-7 RNA regulates developmental timing in Caenorhabditis elegans. *Nature* 403:901-906.

Rheaume, E., J.Simard, Y.Morel, F.Mebarki, M.Zachmann, M.G.Forest, M.I.New, and F.Labrie. 1992. Congenital Adrenal-Hyperplasia Due to Point Mutations in the Type-Ii 3-Beta-Hydroxysteroid Dehydrogenase Gene. *Nature Genetics* 1:239-245.

Rice, T., T.Rankinen, Y.C.Chagnon, M.A.Province, L.Perusse, A.S.Leon, J.S.Skinner, J.H.Wilmore, C.Bouchard, and D.C.Rao. 2002. Genomewide linkage scan of resting blood pressure - HERITAGE family study. *Hypertension* 39:1037-1043.

Rice, T., G.P.Vogler, L.Perusse, C.Bouchard, and D.C.Rao. 1989. Cardiovascular Risk-Factors in A French-Canadian Population - Resolution of Genetic and Familial Environmental-Effects on Blood-Pressure Using Twins, Adoptees, and Extensive Information on Environmental Correlates. *Genetic Epidemiology* 6:571-588.

Ritchie, W., S.Flamant, and J.E.J.Rasko. 2010. mimiRNA: a microRNA expression profiler and classification resource designed to identify functional correlations between microRNAs and their targets. *Bioinformatics* 26:223-227.

Robins, H., Y.Li, and R.W.Padgett. 2005. Incorporating structure to predict microRNA targets. *Proceedings of the National Academy of Sciences of the United States of America* 102:4006-4009.

Romero, D.G., M.W.Plonczynski, C.A.Carvajal, E.P.Gomez-Sanchez, and C.E.Gomez-Sanchez. 2008. Microribonucleic acid-21 increases aldosterone secretion and proliferation in H295R human adrenocortical cells. *Endocrinology* 149:2477-2483.

Romero, D.G., S.Rilli, M.W.Plonczynski, L.L.Yanes, M.Y.Zhou, E.P.Gomez-Sanchez, and C.E.Gomez-Sanchez. 2007. Adrenal transcription regulatory genes modulated by angiotensin II and their role in steroidogenesis. *Physiological Genomics* 30:26-34.

Roush, S. and F.J.Slack. 2008. The let-7 family of microRNAs. *Trends in Cell Biology* 18:505-516.

Ruby, J.G., C.H.Jan, and D.P.Bartel. 2007. Intronic microRNA precursors that bypass Drosha processing. *Nature* 448:83-86.

Ryan, B.M., A.I.Robles, and C.C.Harris. 2010. Genetic variation in microRNA networks: the implications for cancer research. *Nature Reviews Cancer* 10:389-402.

Saito, T. and P.Saetrom. 2010. MicroRNAs - targeting and target prediction. *New Biotechnology* 27:243-249.

Salzman, D.W., J.Shubert-Coleman, and H.Furneaux. 2007. P68 RNA helicase unwinds the human let-7 microRNA precursor duplex and is required for let-7directed silencing of gene expression. *Journal of Biological Chemistry* 282:32773-32779.

Sapolsky, R.M., L.M.Romero, and A.U.Munck. 2000. How do glucocorticoids influence stress responses? Integrating permissive, suppressive, stimulatory, and preparative actions. *Endocrine Reviews* 21:55-89.

Saumet, A., G.Vetter, M.Bouttier, E.Portales-Casamar, W.W.Wasserman, T.Maurin, B.Mari, P.Barbry, L.Vallar, E.Friederich, K.Arar, B.Cassinat, C.Chomienne, and C.H.Lecellier. 2009. Transcriptional repression of microRNA genes by PML-RARA increases expression of key cancer proteins in acute promyelocytic leukemia. *Blood* 113:412-421.

Scadden, A.D.J. 2005. The RISC subunit Tudor-SN binds to hyper-edited doublestranded RNA and promotes its cleavage. *Nature Structural & Molecular Biology* 12:489-496.

Schmidt, B.M.W., S.Oehmer, C.Delles, R.Bratke, M.P.Schneider, A.Klingbeil, E.H.Fleischmann, and R.E.Schmieder. 2003. Rapid nongenomic effects of aldosterone on human forearm vasculature. *Hypertension* 42:156-160.

Schmitz, K.J., J.Helwig, S.Bertram, S.Y.Sheu, A.C.Suttorp, J.Seggewiss, E.Willscher, M.K.Walz, K.Worm, and K.W.Schmid. 2011. Differential expression of microRNA-675, microRNA-139-3p and microRNA-335 in benign and malignant adrenocortical tumours. J. Clin. Pathol.

Schwarz, D.S., G.Hutvagner, T.Du, Z.S.Xu, N.Aronin, and P.D.Zamore. 2003. Asymmetry in the assembly of the RNAi enzyme complex. *Cell* 115:199-208.

Scott, G.K., A.Goga, D.Bhaumik, C.E.Berger, C.S.Sullivan, and C.C.Benz. 2007. Coordinate suppression of ERBB2 and ERBB3 by enforced expression of micro-RNA miR-125a or miR-125b. *Journal of Biological Chemistry* 282:1479-1486.

Selbach, M., B.Schwanhausser, N.Thierfelder, Z.Fang, R.Khanin, and N.Rajewsky. 2008. Widespread changes in protein synthesis induced by microRNAs. *Nature* 455:58-63.

Sethi, A.A., B.G.Nordestgaard, and A.Tybjaerg-Hansen. 2003. Angiotensinogen gene polymorphism, plasma angiotensinogen, and risk of hypertension and ischemic heart disease - A meta-analysis. *Arteriosclerosis Thrombosis and Vascular Biology* 23:1269-1275.

Sethupathy, P. and F.S.Collins. 2008. MicroRNA target site polymorphisms and human disease. *Trends in Genetics* 24:489-497.

Sethupathy, P., M.Megraw, and A.G.Hatzigeorgiou. 2006. A guide through present computational approaches for the identification of mammalian microRNA targets. *Nature Methods* 3:881-886.

Sevignani, C., G.A.Calin, L.D.Siracusa, and C.M.Croce. 2006. Mammalian microRNAs: a small world for fine-tuning gene expression. *Mammalian Genome* 17:189-202.

Shanmugam, S., P.Corvol, and J.M.Gasc. 1996. Angiotensin II type 2 receptor mRNA expression in the developing cardiopulmonary system of the rat. *Hypertension* 28:91-97.

Sharma, K.K., A.Lindqvist, X.J.Zhou, R.J.Auchus, T.M.Penning, and S.Andersson. 2006. Deoxycorticosterone inactivation by AKR1C3 in human mineralocorticoid target tissues. *Molecular and Cellular Endocrinology* 248:79-86.

Sheth, U. and R.Parker. 2003. Decapping and decay of messenger RNA occur in cytoplasmic processing bodies. *Science* 300:805-808.

Shin, C., J.W.Nam, K.K.H.Farh, H.R.Chiang, A.Shkumatava, and D.P.Bartel. 2010. Expanding the MicroRNA Targeting Code: Functional Sites with Centered Pairing. *Molecular Cell* 38:789-802.

Shinzawa, K., S.Ishibashi, M.Murakoshi, K.Watanabe, S.Kominami, A.Kawahara, and S.Takemori. 1988. Relationship Between Zonal Distribution of Microsomal Cytochrome P-450S (P-45017-Alpha,Lyase and P-450C21) and Steroidogenic Activities in Guinea-Pig Adrenal-Cortex. *Journal of Endocrinology* 119:191-&.

Silahtaroglu, A.N., D.Nolting, L.Dyrskjot, E.Berezikov, M.Moller, N.Tommerup, and S.Kauppinen. 2007. Detection of microRNAs in frozen tissue sections by fluorescence in situ hybridization using locked nucleic acid probes and tyramide signal amplification. *Nature Protocols* 2:2520-2528.

Small, E.M., J.R.O'Rourke, V.Moresi, L.B.Sutherland, J.McAnally, R.D.Gerard, J.A.Richardson, and E.N.Olson. 2010. Regulation of PI3-kinase/Akt signaling by

muscle-enriched microRNA-486. *Proceedings of the National Academy of Sciences of the United States of America* 107:4218-4223.

Sober, S., M.Laan, and T.Annilo. 2010. MicroRNAs miR-124 and miR-135a are potential regulators of the mineralocorticoid receptor gene (NR3C2) expression. *Biochemical and Biophysical Research Communications* 391:727-732.

Sookoian, S., T.F.Gianotti, C.D.Gonzalez, and C.J.Pirola. 2007. Association of the C-344T aldosterone synthase gene variant with essential hypertension: a meta-analysis. *Journal of Hypertension* 25:5-13.

Soon, P.S.H., L.J.Tacon, A.J.Gill, C.P.Bambach, M.S.Sywak, P.R.Campbell, M.W.Yeh, S.G.Wong, R.J.Clifton-Bligh, B.G.Robinson, and S.B.Sidhu. 2009. miR-195 and miR-483-5p Identified as Predictors of Poor Prognosis in Adrenocortical Cancer. *Clinical Cancer Research* 15:7684-7692.

Spach, C. and D.H.Streeten. 1964. Retardation of Sodium Exchange in Dog Erythrocytes by Physiological Concentrations of Aldosterone in Vitro. *Journal of Clinical Investigation* 43:217-&.

Stamler, J., G.Rose, R.Stamler, P.Elliott, A.Dyer, and M.Marmot. 1989. Intersalt Study Findings - Public-Health and Medical-Care Implications. *Hypertension* 14:570-577.

Stark, A., J.Brennecke, N.Bushati, R.B.Russell, and S.M.Cohen. 2005. Animal microRNAs confer robustness to gene expression and have a significant impact on 3 ' UTR evolution. *Cell* 123:1133-1146.

Stewart, P.M., Z.S.Krozowski, A.Gupta, D.V.Milford, A.J.Howie, M.C.Sheppard, and C.B.Whorwood. 1996. Hypertension in the syndrome of apparent mineralocorticoid excess due to mutation of the 11 beta-hydroxysteroid dehydrogenase type 2 gene. *Lancet* 347:88-91.

Stocco, D.M. and B.J.Clark. 1996. Regulation of the acute production of steroids in steroidogenic cells. *Endocrine Reviews* 17:221-244.

Stow, L.R., M.L.Gumz, I.J.Lynch, M.M.Greenlee, A.Rudin, B.D.Cain, and C.S.Wingo. 2009. Aldosterone Modulates Steroid Receptor Binding to the Endothelin-1 Gene (edn1). *Journal of Biological Chemistry* 284:30087-30096.

Strazzullo, P., L.D'Elia, N.B.Kandala, and F.P.Cappuccio. 2009. Salt intake, stroke, and cardiovascular disease: meta-analysis of prospective studies. *British Medical Journal* 339.

Ten, S., M.New, and N.Maclaren. 2001. Addison's disease 2001. Journal of Clinical Endocrinology & Metabolism 86:2909-2922.

Tessel, M.A., N.L.Krett, and S.T.Rosen. 2010. Steroid receptor and microRNA regulation in cancer. *Current Opinion in Oncology* 22:592-597.

Thomas, W. and B.J.Harvey. 2010. Mechanisms Underlying Rapid Aldosterone Effects in the Kidney. *Annu. Rev. Physiol*.

Timmermans, P.B.M.W., P.C.Wong, A.T.Chiu, W.F.Herblin, P.Benfield, D.J.Carini, R.J.Lee, R.R.Wexler, J.A.M.Saye, and R.D.Smith. 1993. Angiotensin-Ii

Receptors and Angiotensin-Ii Receptor Antagonists. *Pharmacological Reviews* 45:205-251.

Todd-Turla, K.M., J.Schnermann, G.Fejestoth, A.Narayfejestoth, A.Smart, P.D.Killen, and J.P.Briggs. 1993. Distribution of Mineralocorticoid and Glucocorticoid Receptor Messenger-Rna Along the Nephron. *American Journal of Physiology* 264:F781-F791.

Tombol, Z., P.M.Szabo, V.Molnar, Z.Wiener, G.Tolgyesi, J.Horanyi, P.Riesz, P.Reismann, A.Patocs, I.Liko, R.C.Gaillard, A.Falus, K.Racz, and P.Igaz. 2009. Integrative molecular bioinformatics study of human adrenocortical tumors: microRNA, tissue-specific target prediction, and pathway analysis. *Endocrine-Related Cancer* 16:895-906.

Tsujita, Y., N.Iwai, T.Katsuya, J.Higaki, T.Ogihara, S.Tamaki, M.Kinoshita, T.Mannami, J.Ogata, and S.Baba. 2001. Lack of association between genetic polymorphism of CYP11B2 and hypertension in Japanese: The Suita Study. *Hypertension Research* 24:105-109.

Ulick, S. 1976. Diagnosis and Nomenclature of Disorders of Terminal Portion of Aldosterone Biosynthetic-Pathway. *Journal of Clinical Endocrinology & Metabolism* 43:92-96.

Ulick, S., L.S.Levine, P.Gunczler, G.Zanconato, L.C.Ramirez, W.Rauh, A.Rosler, H.L.Bradlow, and M.I.New. 1979. Syndrome of Apparent Mineralocorticoid Excess Associated with Defects in the Peripheral Metabolism of Cortisol. *Journal of Clinical Endocrinology & Metabolism* 49:757-766.

van Rooij, E., L.B.Sutherland, N.Liu, A.H.Williams, J.McAnally, R.D.Gerard, J.A.Richardson, and E.N.Olson. 2006. A signature pattern of stress-responsive microRNAs that can evoke cardiac hypertrophy and heart failure. *Proceedings of the National Academy of Sciences of the United States of America* 103:18255-18260.

Vester, B. and J.Wengel. 2004. LNA (Locked nucleic acid): High-affinity targeting of complementary RNA and DNA. *Biochemistry* 43:13233-13241.

Viengchareun, S., D.Le Menuet, L.Martinerie, M.Munier, L.Pascual-Le Tallec, and M.Lombes. 2007. The mineralocorticoid receptor: insights into its molecular and (patho)physiological biology. *Nucl. Recept. Signal.* 5:e012.

Vinson, G.P. and J.P.Coghlan. 2010. Expanding view of aldosterone action, with an emphasis on rapid action. *Clinical and Experimental Pharmacology and Physiology* 37:410-416.

Volinia, S., G.A.Calin, C.G.Liu, S.Ambs, A.Cimmino, F.Petrocca, R.Visone, M.Iorio, C.Roldo, M.Ferracin, R.L.Prueitt, N.Yanaihara, G.Lanza, A.Scarpa, A.Vecchione, M.Negrini, C.C.Harris, and C.M.Croce. 2006. A microRNA expression signature of human solid tumors defines cancer gene targets. *Proceedings of the National Academy of Sciences of the United States of America* 103:2257-2261.

Walker, B.R. 2007. Glucocorticoids and cardiovascular disease. *European Journal* of Endocrinology 157:545-559.

Wang, X.L., M.Bassett, Y.Zhang, S.Yin, C.Clyne, P.C.White, and W.E.Rainey. 2000. Transcriptional regulation of human 11 beta-hydroxylase (hCYP11B1). *Endocrinology* 141:3587-3594.

Ward, R. 1990. Hypertension: Pathophysiology, Diagnosis and Management. Raven Press, New York. 81-100 pp.

Wehling, M., J.Kasmayr, and K.Theisen. 1989. Fast Effects of Aldosterone on Electrolytes in Human-Lymphocytes Are Mediated by the Sodium-Proton-Exchanger of the Cell-Membrane. *Biochemical and Biophysical Research Communications* 164:961-967.

Whelton, P.K., J.He, J.A.Cutler, F.L.Brancati, L.J.Appel, D.Follmann, and M.J.Klag. 1997. Effects of oral potassium on blood pressure - Meta-analysis of randomized controlled clinical trials. *Jama-Journal of the American Medical Association* 277:1624-1632.

White, P.C. 2004. Aldosterone synthase deficiency and related disorders. *Molecular and Cellular Endocrinology* 217:81-87.

White, P.C., M.I.New, and B.Dupont. 1986. Structure of Human Steroid 21-Hydroxylase Genes. *Proceedings of the National Academy of Sciences of the United States of America* 83:5111-5115.

White, P.C. and L.Slutsker. 1995. Haplotype Analysis of Cyp11B2. *Endocrine Research* 21:437-442.

White, P.C. and P.W.Speiser. 2000. Congenital adrenal hyperplasia due to 21hydroxylase deficiency. *Endocrine Reviews* 21:245-291.

Wightman, B., I.Ha, and G.Ruvkun. 1993. Posttranscriptional Regulation of the Heterochronic Gene Lin-14 by Lin-4 Mediates Temporal Pattern-Formation in C-Elegans. *Cell* 75:855-862.

Wolf, S.C., M.Schultze, T.Risler, T.Rieg, F.Lang, K.Schulze-Osthoff, and B.R.Brehm. 2006. Stimulation of serum- and glucocorticoid-regulated kinase-1 gene expression by endothelin-1. *Biochemical Pharmacology* 71:1175-1183.

Wong, S., F.E.Brennan, M.J.Young, P.J.Fuller, and T.J.Cole. 2007. A direct effect of aldosterone on endothelin-1 gene expression in vivo. *Endocrinology* 148:1511-1517.

Wu, L.G., J.H.Fan, and J.G.Belasco. 2006. MicroRNAs direct rapid deadenylation of mRNA. *Proceedings of the National Academy of Sciences of the United States of America* 103:4034-4039.

Wu, S., S.Huang, J.Ding, Y.Zhao, L.Liang, T.Liu, R.Zhan, and X.He. 2010. Multiple microRNAs modulate p21Cip1/Waf1 expression by directly targeting its 3' untranslated region. *Oncogene* 29:2302-2308.

Wulff, P., V.Vallon, D.Y.Huang, H.Volkl, F.Yu, K.Richter, M.Jansen, M.Schlunz, K.Klingel, J.Loffing, G.Kauselmann, M.R.Bosl, F.Lang, and D.Kuhl. 2002. Impaired renal Na+ retention in the sgk1-knockout mouse. *Journal of Clinical Investigation* 110:1263-1268. Xi, Y., G.Nakajima, E.Gavin, C.G.Morris, K.Kudo, K.Hayashi, and J.Ju. 2007. Systematic analysis of microRNA expression of RNA extracted from fresh frozen and formalin-fixed paraffin-embedded samples. *RNA*. 13:1668-1674.

Yamashiro, T., H.Kuge, J.Zhang, and K.Honke. 2010. Calcineurin mediates the angiotensin II-induced aldosterone synthesis in the adrenal glands by upregulation of transcription of the CYP11B2 gene. *Journal of Biochemistry* 148:115-123.

Yang, J.S., T.Maurin, N.Robine, K.D.Rasmussen, K.L.Jeffrey, R.Chandwani, E.P.Papapetrou, M.Sadelain, D.O'Carroll, and E.C.Lai. 2010. Conserved vertebrate mir-451 provides a platform for Dicer-independent, Ago2-mediated microRNA biogenesis. *Proceedings of the National Academy of Sciences of the United States of America* 107:15163-15168.

Yang, W.D., T.P.Chendrimada, Q.D.Wang, M.Higuchi, P.H.Seeburg, R.Shiekhattar, and K.Nishikura. 2006. Modulation of microRNA processing and expression through RNA editing by ADAR deaminases. *Nature Structural & Molecular Biology* 13:13-21.

Yaworsky, D.C., B.Y.Baker, H.S.Bose, K.B.Best, L.B.Jensen, J.D.Bell, M.A.Baldwin, and W.L.Miller. 2005. pH-dependent interactions of the carboxylterminal helix of steroidogenic acute regulatory protein with synthetic membranes. *Journal of Biological Chemistry* 280:2045-2054.

Ye, P., Y.Nakamura, E.Lalli, and W.E.Rainey. 2009. Differential Effects of High and Low Steroidogenic Factor-1 Expression on CYP11B2 Expression and Aldosterone Production in Adrenocortical Cells. *Endocrinology* 150:1303-1309.

Yi, R., Y.Qin, I.G.Macara, and B.R.Cullen. 2003. Exportin-5 mediates the nuclear export of pre-microRNAs and short hairpin RNAs. *Genes & Development* 17:3011-3016.

Young, M., G.Head, and J.Funder. 1995. Determinants of Cardiac Fibrosis in Experimental Hypermineralocorticoid States. *American Journal of Physiology-Endocrinology and Metabolism* 32:E657-E662.

Young, W.F. 2007. Primary aldosteronism: renaissance of a syndrome. *Clinical Endocrinology* 66:607-618.

Yu, B., Z.Y.Yang, J.J.Li, S.Minakhina, M.C.Yang, R.W.Padgett, R.Steward, and X.M.Chen. 2005. Methylation as a crucial step in plant microRNA biogenesis. *Science* 307:932-935.

Zeng, Y. and B.R.Cullen. 2004. Structural requirements for pre-microRNA binding and nuclear export by Exportin 5. *Nucleic Acids Research* 32:4776-4785.

Zhang, L.H., H.Rodriguez, S.Ohno, and W.L.Miller. 1995. Serine Phosphorylation of Human P450C17 Increases 17,20-Lyase Activity - Implications for Adrenarche and the Polycystic-Ovary-Syndrome. *Proceedings of the National Academy of Sciences of the United States of America* 92:10619-10623.

Zhang, R. and B.Su. 2008. MicroRNA regulation and the variability of human cortical gene expression. *Nucleic Acids Research* 36:4621-4628.

Zheng, L.A., C.C.Xu, W.D.Chen, W.L.Shen, C.C.Ruan, L.M.Zhu, D.L.Zhu, and P.J.Gao. 2010. MicroRNA-155 regulates angiotensin II type 1 receptor expression and phenotypic differentiation in vascular adventitial fibroblasts. *Biochemical and Biophysical Research Communications* 400:483-488.